Cyclic Interference Alignment in Multi-User Communication Networks

Von der Fakultät für Elektrotechnik und Informationstechnik der Rheinisch-Westfälischen Technischen Hochschule Aachen zur Erlangung des akademischen Grades eines Doktors der Ingenieurwissenschaften genehmigte Dissertation

vorgelegt von

Diplom-Ingenieur Henning Maier

aus Altenberge

Berichter: Universitätsprofessor Dr. rer. nat. Rudolf Mathar Universitätsprofessor Dr.-Ing. Aydin Sezgin

Tag der mündlichen Prüfung: 24. Februar 2015

Diese Dissertation ist auf den Internetseiten der Hochschulbibliothek online verfügbar.

Berichte aus der Informationstechnik

Henning Maier

Cyclic Interference Alignment in Multi-User Communication Networks

Shaker Verlag Aachen 2015

Bibliographic information published by the Deutsche Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available in the Internet at http://dnb.d-nb.de.

Zugl.: D 82 (Diss. RWTH Aachen University, 2015)

Copyright Shaker Verlag 2015 All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior permission of the publishers.

Printed in Germany.

ISBN 978-3-8440-3723-4 ISSN 1610-9406

Shaker Verlag GmbH • P.O. BOX 101818 • D-52018 Aachen Phone: 0049/2407/9596-0 • Telefax: 0049/2407/9596-9 Internet: www.shaker.de • e-mail: info@shaker.de

Acknowledgements

First of all, I would like to thank my supervisor Rudolf Mathar for providing me the opportunity to pursue fascinating research at the *Institute for Theoretical Information Technology* (TI) at RWTH Aachen University, Germany. Moreover, I am especially grateful to both Rudolf Mathar and Johannes Schmitz for their contributing parts in igniting and developing the idea of *Cyclic Interference Alignment* as it is treated in this thesis. Thanks to all my current and former colleagues at the institute, in particular Michael Reyer and Meik Dörpinghaus for helpful hints surrounding the challenging path to this thesis. I thank Xiang Xu for having had a very pleasant time in our shared office. Furthermore, I would like to thank Cornelia Hanf for her kind help in many organizational matters and our network administrators Detlef Maus and Ansgar Johannes Pflipsen for keeping the local network and computer systems running smoothly.

My cordial gratitude goes to Anas Chaaban and Aydin Sezgin at Ruhr-Universität Bochum (RUB) for their excellent support and the prosperous cooperation on 3-way channels and for their contributions as co-authors of several publications that are part of this thesis.

I would like to thank the *Deutsche Forschungsgemeinschaft* (DFG), the organizers of the DFG-program *Communications in Interference Limited Networks* (COIN), and all the participating scientists who provided an annual forum for presenting and discussing recent results.

Thanks to the organizers and participants of the European (Winter) School of Information Theory (ESIT) in Barcelona, Spain, held during March 2011, and in Ohrid, Macedonia, held during April 2013, for providing a great opportunity to discuss and build up scientific networks with many resourceful engineers and scientists within the communications and information theory community.

I am very thankful to Kolja Siebert, Seyed Omid Taghizadeh Motlagh, Halil Alper Tokel, Milan Zivkovic and Johannes Schmitz for supporting me during the rehearsals of my presentation for the doctor-exam.

Als letztes möchte ich ganz besonders herzlich meiner Familie sowie natürlich all meinen einzigartigen Freunden danken, die mir immer mit Liebe, Freundschaft, und Unterstützung in allen Lebenslagen zur Seite standen.

Contents

1	Introduction				
	1.1	1 Capacity Approximation			
	1.2	2 The Fundamental Principle of Interference Alignment			
	1.3	Motiva	ation	3	
	1.4	Contri	butions	4	
	1.5	Organ	ization	6	
	1.6	Notati	on	7	
2	Basi	ic Mult	i-User Communication Models	9	
	2.1	Comm	non Parameters of the Considered Channel Models	9	
	2.2	Gauss	ian MIMO Channel Model (GMCM)	9	
		2.2.1	GMCM: Definition	10	
		2.2.2	Degrees-of-Freedom: A Capacity Approximation for the GMCM	10	
		2.2.3	Relativity of Alignment: Feasibility Conditions	11	
	2.3	Linear	Deterministic Channel Model (LDCM)	11	
		2.3.1	LDCM: Definition	12	
		2.3.2	Capacity Approximation	12	
	2.4	Cyclic	Polynomial Channel Model (CPCM)	14	
		2.4.1	CPCM: Definition	15	
		2.4.2	Degrees-of-Freedom: Capacity of the CPCM	16	
		2.4.3	Relativity of Alignment: Separability Conditions	17	
		2.4.4	CPCM versus LDCM: An Algebraic Extension	18	
3	Mul	ti-User	Single-Hop Unidirectional Communications	21	
	3.1	Eleme	ntary Unidirectional Multi-User Channels	22	
		3.1.1	Cyclic Polynomial Multiple-Access Channel	22	
		3.1.2	Orthogonal Multiple-Access (MA)	23	
		3.1.3	Cyclic Polynomial Broadcast Channel	24	
		3.1.4	Cyclic Polynomial 2-User Interference Channel	25	
		3.1.5	Linear Encoding and Decoding (LEaD)	28	
	3.2 A General Upper Bound for $K_{\text{Tx}} \times K_{\text{Rx}}$ - User X-Networks			30	
	3.3	Cyclic	IA on the Cyclic Polynomial 2 - User X - Channel	31	
		3.3.1	Perfect Cyclic Interference Alignment	31	
		3.3.2	Complementary Reciprocal Symmetry of Cyclic IA	35	
		3.3.3	X-Channel with Generalized Message Lengths	36	
		3.3.4	Achievability of Perfect Cyclic IA with Linear Coding	37	
	3.4	Cyclic	IA on the Cyclic Polynomial <i>K</i> -User Interference Channel	43	

		3.4.1	Perfect Cyclic Interference Alignment	43
		3.4.2	Common Eigenvectors in Separate Subspaces	47
		3.4.3	On the Feasibility Conditions of the K-User Interference Channel	47
	3.5	Infeasi	bility of Perfect Cyclic IA in Multi-User X-Networks	48
		3.5.1	Cyclic Polynomial 3-User X-Network	49
		3.5.2	Infeasibility Problem of Perfect Cyclic Interference Alignment .	50
		3.5.3	Common Eigenvectors in Separate Subspaces - Revisited	52
	3.6	3-User	X-Networks with Minimal Backhaul	52
		3.6.1	Cyclic IA with Minimal Feedforward Backhaul Networks	53
		3.6.2	Cyclic IAC over Minimal Receiver Backhaul Networks	55
		3.6.3	Cyclic IN over Minimal Transmitter Backhaul Networks	56
		3.6.4	Combined Cyclic IAC and IN over Minimal Tx/Rx-BHNs	56
		3.6.5	Examples for the Given Cyclic IA, IAC and IN Schemes	57
	3.7	Summa	ary	58
4	Mul	ti-User	Two-Hop and Two-Way Relay-Networks	61
	4.1	Cyclic	Interference Neutralization	61
		4.1.1	Cyclic Polynomial $2 \times 2 \times 2$ Relay-Interference Channel	62
		4.1.2	Cut-Set Upper Bounds	64
		4.1.3	Achievability	64
		4.1.4	Aligned IN on MIMO $2 \times 2 \times 2$ Relay-Interference Channels	68
	4.2	Two-W	Vay Cyclic Interference Neutralization	73
		4.2.1	Cyclic Polynomial Two-Way $2\times 2\times 2$ Relay-Interference Channel	74
		4.2.2	Cut-Set Upper Bounds	75
		4.2.3	Achievability	76
		4.2.4	Discussion	80
	4.3	Cascad	led Two-Way Relaying	80
		4.3.1	Cyclic Polynomial Cascaded Two-Way Relay Networks	81
		4.3.2	Upper Bounds	83
		4.3.3	Achievability	84
		4.3.4	User-Relay Duality	87
	4.4	Summa	ary	88
5	Mul	ti-Way	Communications with Three Users	89
	5.1	Cyclic	Interference Alignment for 3-Way and Y- Channels $\ldots \ldots \ldots$	90
		5.1.1	Cyclic Polynomial 3-User Y-Channel	90
		5.1.2	Cyclic Polynomial 3-Way Channel	92
		5.1.3	Upper Bounds	93
		5.1.4	Achievable Scheme for the Y-Channel	93
		5.1.5	Δ -Y Transformation	98
		5.1.6	Achievable Scheme for the 3-Way Channel	99
		5.1.7	Discussion	100
	5.2	Capaci	ity Region of the Reciprocal Linear Deterministic 3-Way Channel	101
		5.2.1	System Model: Linear Deterministic 3-Way Channel	101
		5.2.2	Outer Bounds of Capacity Region: 3-Way Channel	102
		5.2.3	System Model: Linear Deterministic Y-Channel	104

	5.3	5.2.4 Outer Bounds of Capacity Region: Y-Channel 10 5.2.5 Δ -Y Transformation	5569445678		
	5.4	Symmetric Degrees of Freedom of the MIMO $M_{\rm Tx} \times M_{\rm Rx}$ 3-Way Channel125.4.1 MIMO $M_{\rm Tx} \times M_{\rm Rx}$ 3-Way Channel5.4.2 Upper Bounds5.4.3 Achievability5.4.4 Discussion			
	5.5	Extending the CPCM by Multi-Antenna Constraints	1		
	5.6	Summary	2		
6	Expl	ring Practical Applications for Cyclic Interference Alignment 13.	5		
-	6.1	On Practical Implementations of IA	5		
	6.2	Cyclic IA by Propagation Delay	6		
		6.2.1 Deep-Space Communications	7		
		6.2.2 Underwater Communications	7		
		6.2.3 Node Placement in Euclidean Space	8		
		$6.2.4 2-\text{User } X-\text{Channel} \dots \dots \dots \dots \dots \dots \dots 13$	8		
	6.3	Summary	1		
7	Con	lusions 14	3		
	7.1	Outlook	4		
۸		. 14	-		
Аμ	A B	Constraints of the 3-User X-Network with Minimal Backhaul \ldots 14 Example for Cyclic Interference Neutralization on the $2 \times 2 \times 2$ Relay-	5		
		Interference Channel	8		
	С	Genie-Aided Upper Bounds of the $D3C(n_1, n_2, n_3)$ with $n_3 \ge n_2 \ge n_1$ 14	9		
Lis	t of a	cronyms 15	1		
l is	tof	igures 15	3		
		10	9		
Lis	List of Theorems 15				
Bil	Bibliography 1				
Cu	Curriculum Vitae 1				

1 Introduction

In the year 1948, Claude Shannon established a mathematical theory in his seminal work [1] for the elementary description of information transmission over noisy communication systems. He provided a conceptual probabilistic model for a point-to-point channel between a single transmitter and a single receiver impaired by noise. A main part of his contribution was deriving and proving the maximal data-rate for reliable communication on a Gaussian point-to-point channel under fixed power and bandwidth constraints. But actually achieving the maximal data-rate in an implemented communication system has been exceedingly challenging over several decades. Since then, Shannon's work inspired many scientists and engineers to design elaborate channel codes that counteract transmission errors in order to achieve the maximal data-rate. Nowadays, it is indeed possible to communicate with data rates pretty close to the *Shannon-capacity* with reasonable effort [2].

However, in contemporary wireless communication systems there are multiple users sharing the same communication medium. In contrast to omnipresent noise, the resulting multi-user interference is not a naturally occurring phenomenon - it is rather a self-induced impairment caused by multiple simultaneous transmissions. The ongoing trend of increasing wireless communication traffic leads to a tight bottleneck of the maximally achievable data rates per user.

So far, most conventional signalling strategies (cf. [3], [4], [5]) basically avoid interference by signal orthogonalization, by treating weak interference as noise or by cancelling strong interference. In case of orthogonalization, the available resources (e.g., time, bandwidth, space) are elaborately partitioned among the users so that each user obtains a limited share of an ordinary interference-free point-to-point channel to convey their dedicated signals to the intended receivers, so that non-dedicated users are masked out. For instance, if the data rate is demanded to be equal among each user, it will scale reciprocally. In other words, each one of K > 1 users only achieves the $\frac{1}{K}$ -th ratio of the sum-rate of a point-to-point channel. Whereas treating interference as noise is indeed optimal for the case of very weak interference power as shown in [6], [7]and [8], the approach clearly results in very low data rates for moderate and strong interference power, especially for a high number of users. Similarly, cancelling interference is most effective for very strong interference power only. The most relevant cases in between, i. e., for multi-user channels with moderate interference power or even generic channel conditions, the sum-capacity and the capacity region of the individual rates per user remains unsolved and highly challenging so far.

1.1 Capacity Approximation

Even for the most elementary 2-user interference channel, the exact capacity characterization is not fully known and proven yet. However, it turned out that it is worthwhile to approximate the channel capacity of multi-user channels with several different techniques (cf. [5]). Within the scope of this theses, especially the degrees-of-freedom (DoF) measure for capacity approximation at high signal-to-noise-ratio (SNR), is of particular interest.

Furthermore, to ease an accurate capacity approximation, simplified multi-user channel models are considered by focussing on the challenging effects of multiple signal interactions and by de-emphasizing the effects of noise. The *linear deterministic chan*nel model (LDCM) [9] provides a promising model that yet accurately approximates these effects in terms of the highly relevant Gaussian channel model. As a result, this approach supported and inspired several seminal works studying the capacity of various elementary channels in terms of the highly relevant Gaussian channel model. In some cases the capacity is even characterized within a limited number of bits (e. g., the 2-user interference channel [4] and the relay channel [9]). These derivations also utilize genie-based bounding techniques as developed in [10] and [4]. But when considering interference channels with more than two users, the optimal communication schemes and their capacity are far less understood. Note that approximating the capacity for channels approximated the LDCM becomes increasingly challenging for multiple users as well.

1.2 The Fundamental Principle of Interference Alignment

Multi-user interference is conventionally considered as a harmful and hence unwanted impairment that must be circumvented. A novel approach is to permit interfering signals by designing them cooperatively such that the aggregate interference signals are contained to a finite subspace at each user. This approach is called *interference alignment (IA)* and introduced in the seminal publications [3] and [11].

The key idea of IA is to optimally arrange the signal ling-dimensions among different users such that the dimensions of interfering signals overlay, while the dedicated signals remain separate and hence decodable at the intended receivers. In the optimal case, the dimensions occupied by the interfering signals fully overlap, as if each user only experiences one virtual interferer in the system. Especially for a multi-user interference channel as depicted in Figure 1.1 for three users, the data rate per user approximately scales by $\frac{1}{2}$ w.r.t. the capacity of interference-free point-to-point links for each dedicated user pair. It is remarkable that this scaling is independent of the total number of users K. This is a substantial gain, when compared to an orthogonalized communication scheme. By applying IA, the aforementioned bottleneck can already be overcome for $K \geq 3$ communicating user pairs.

This theoretical idea of IA has opened the door to efficient communication schemes [12] with multiple users, multiple relays, bidirectional communications, etc. Nonetheless, there are still many challenging limitations in both theory and practice to cope



Figure 1.1: The fundamental principle of interference alignment: In a 3-user interference channel, 3 transmitters Tx_i communicate with 3 receivers Rx_i in pairwise distinct links. The interference of the 2 non-dedicated users is aligned to an interference subspace (red, dotted lines) at each receiver, while the dedicated signal (blue, solid lines) remains distinct.

with, e.g., the high demands on computational complexity and extremely accurate knowledge of the channel gains between all users.

A fundamental demand to enable IA - the *relativity of alignment* [12] - is that the channel itself must support a sufficient structure or variability so that an alignment scheme is feasible. Interestingly, IA is applied in a diverse number of substantially different schemes. It is enabled in multiple-input multiple-output (MIMO) channels by symbol-extensions in time or frequency [3], by rational independent dimensions [13], by signal-scale [14], on pairs of complementary matrices in ergodic channels [15], by propagation delay [3], [16], by (asymmetric complex) phase [17], on lattices [18], etc.

Clearly, IA has already drawn a lot of attention from the research areas of information theory, signal processing and even in distributed data-storage systems [19] and it still remains in the spotlight. For a more extensive overview and references on IA, we point the interested reader to the comprehensive survey [12].

1.3 Motivation

In this dissertation, our main focus is to introduce and develop a novel model for multi-user communication channels - the cyclic polynomial channel model (CPCM). Inspired by the convenient algebraic framework of cyclic codes, our model enlightens a theoretical approach of IA using a polynomial ring. The CPCM is closely related to the widely accepted LDCM [9] and also motivated by the elementary example of IA by propagation delay given in [3]. A key difference to the LDCM is the use of a cyclic shift instead of a linear shift to model the impact of multiple interfering signals. From a mathematical perspective, such cyclic shifts uncover several valuable properties, that are hidden within the LDCM. It is a particular drawback of the LDCM that, for multi-user scenarios, the exponentially increasing number of parameters is hard to handle. In the literature, the authors mostly concentrate on multi-user channels

with symmetrically parameterized channel gains, e. g., [20], [21], [22], or on arbitrary channel gains with only two user-pairs, e. g., [23], [24]. The proposed CPCM, however, is particularly tailored to tackle problems with asymmetric channel gains and multiple-users rather than directly focussing on the capacity-approximation within a limited number of bits.

1.4 Contributions

The CPCM supports the investigation of the following aspects:

In our initial research idea, the proposed CPCM was mainly dedicated to generalize the concept of *IA by propagation delay* for arbitrary channels, since unrolled cyclic shifts can be interpreted as propagation delays over the channel.

Moreover, the *capacity-achieving schemes* for the CPCM are inherently provided for *asymmetric channel gains* in multi-user systems. Deriving optimal schemes for the related LDCM is very challenging, since tracking the dependencies on arbitrary channel gains is immensely cumbersome if many users are involved.

While the conventional approach is to compute the achievable rates for a fixed channel, we turn the problem around: We fix a set of demanded rates and formulate *separability conditions* that must be satisfied by the channel to ensure decodability. These conditions are also useful explain particular singularities in the related LDCM.

Note that there is still a highly active and conflicting discussion about using *algebraic structure* versus *random codes* [25] in network and multi-user information theory. In this light, considering a model with cyclic shifts is quite amenable for an algebraic investigation.

In this thesis, we cover several different multi-user networks with the following properties:

Elementary unidirectional networks: The multiple-access channel, the broadcast channel, the interference channel and the X-channel. (Chapter 3)

Some users in the networks are equipped with backhaul links for constrained cooperation to provide a specific set of *cognitive messages*. (Chapter 3)

Relaying: Relays are auxiliary devices to enhance the channel by forwarding a function of the received signals and to enable some efficient schemes. (Chapter 4)

Multi-way communications: If users act as transceivers, they can simultaneously transmit and receive over a bidirectional channel. (Chapters 4 and 5)

In order to achieve the maximal DoF for these networks, our proposed communication schemes are mainly based on: *Orthogonal multiple-access*: We show that the conventional othogonalization of signals is already optimal for some elementary channels. This is coherent with known results and intended to introduce the readers to the proposed model. (Chapter 3)

Linear coding and interference cancellation: However, these orthogonal schemes are already restrictive for interference channels. We derive linear coding strategies to resolve interference by cancellation. (Chapter 3)

Cyclic interference alignment: We develop cyclic IA schemes coherent to the fundamental IA principle described above. The provided capacity results are suitable for known results (e.g., for the X-channel and the K-user interference channel) of conventional channel models, in case they are already available. (Chapter 3)

Cyclic interference neutralization (IN): Especially in systems with multiple relays, there are multiple instances of the same interference from different relays. These signals can be aligned into the same subspace to neutralize each other over the air. (Chapter 4)

Network-coded cyclic signal alignment (SA): In bidirectional communications with relays, dedicated bidirectional signals are aligned so that a simple network code [26] is used at the receivers. (Chapters 4 and 5)

In the course of this dissertation, we do not only discuss DoF-achieving schemes for various channels, we also reveal several interesting phenomena and elaborate corresponding insights for multi-user systems:

- (A) We observe different *duality relationships*: (Chapters 3, 4 and 5)
 - (1) Cyclic IA is dual to cyclic IN in 3-user X-networks with a backhaul network of cognitive messages at the transmitters or receivers. (Chapter 3)
 - (2) We observe a user-relay duality, for a cascaded chain of two-way relay networks, i.e., the roles of users and relays are swapped. It is linked to a unicast-multicast¹ duality in the uplink and its reciprocal downlink transmission. (Chapter 4)
 - (3) We observe a Δ-Y relationship between 3-way channels and Y-channels, which is motivated by the well-known Δ-Y transformation in electrical circuit theory. (Chapter 5)
- (B) We observe a *reciprocal alignment property* for a channel with two transmitters and two receivers: It basically states that aligning signals from the two transmitters at a primary receiver provides a particular interference pattern at the secondary receiver, while, vice versa, aligning two signals at the secondary receiver in an analogous way, provides the reciprocal interference pattern of the previous alignment at the primary transmitter. This property is exploited in the capacity-achieving scheme of the 2-user X- channel. (Chapter 3)

¹In a *unicast* transmission, a single message is only dedicated for one receiver only. In a *multicast* transmission, a single messages is intended for multiple receivers at the same time.

(C) Moreover, we highlight several common properties that appear in Gaussian channels, MIMO channels and linear deterministic channels, likewise. Especially, constant MIMO channels exhibit a limitation caused by common eigenvectors when applying a perfect IA scheme. In the CPCM, an analogous property is also present for cyclic IA, but it only limits the feasibility of the communication system to a certain subset of channel matrices. (Chapters 3, 4 and 5)

Albeit, some of these phenomena, i.e., parts of (A)(1) and (C), were at least partially observed within the conventional channel models already, these insights extend the yet available knowledge and also serve as an additional confirmation for the validity of our proposed model.

Parts of this dissertation are published in [27], [28], [29], [30], [31], [32], [33] and [34].

1.5 Organization

In Chapter 2, we first recapitulate two basic models that are also treated in this thesis, before introducing the CPCM in detail. Then, optimal communication schemes for several elementary unidirectional communication channels are discussed in Chapter 3 using the CPCM. Furthermore, the 2-user X-channel, the 3-user X-channel, and the K-user interference channel are investigated for the application of cyclic IA. In Chapter 4, we extend the given model with relays and we apply cyclic IN. In a subsequent step, also two-way relaying and corresponding network-coded signal alignment schemes are included. In Chapter 5, we focus on 3-way communications that involve an elaborate combination of the methods discussed in the previous chapters. The 3-way channel is discussed in terms of the CPCM and in terms of the LDCM and Gaussian MIMO channel model (GMCM) for comparison. In Chapter 6, we briefly explore practical issues of IA in general and discuss IA by propagation delay in particular. The conclusion of this dissertation and directions for future works are provided in Chapter 7.

1.6 Notation

Symbol	Description
a, A	scalar/random variables (lower/upper case letters)
$\boldsymbol{a}, \boldsymbol{A}$	row vectors/matrices (boldface lower/upper case letters)
\mathcal{A}	sets (calligraphic upper case letters)
$0_{a imes b}$	$a \times b$ zero matrix
$oldsymbol{I}_{a imes b}$	$a \times b$ identity matrix
$1_{a imes b}$	$a \times b$ one matrix
span(A)	linear column-span of \boldsymbol{A}
$rank(oldsymbol{A})$	rank of A
$dim(oldsymbol{A})$	dimension of \boldsymbol{A}
$det(oldsymbol{A})$	determinant of \boldsymbol{A}
A^{T}	transposed matrix of \boldsymbol{A}
A	Moore-Penrose pseudo-inverse of a non-square \boldsymbol{A}
A^{-1}	inverse matrix of \boldsymbol{A}
$diag(a_1, \ldots, a_n)$	a square diagonal matrix with diagonal entries $a_1,, a_n$
$[a]^+$	$\max(0, a)$ positive part of $a \in \mathbb{R}$
$\binom{n}{k}$	binomial coefficient - n choose k
H(X)	entropy of a random variable X
I(X;Y)	mutual information between X and Y
$\mathcal{CN}(oldsymbol{\mu}, oldsymbol{\Sigma})$	complex normal distribution with mean vector μ
	and covariance matrix Σ
x	an indeterminate, to address offsets in a polynomial
p(x)	a polynomial $\sum_{i=0}^{n-1} p^{[i]} x^i$ with coefficients $p^{[i]}$
Tx	transmitter (source)
Rx	receiver (destination)
Т	transceiver (source/destination)
R	relay

Figure 1.2: Notation.

2 Basic Multi-User Communication Models

In this chapter, we will briefly recapitulate the preliminary properties of two widely established wireless channel models used in the literature on multi-user information theory:

The Gaussian MIMO Channel Model (GMCM), and

the Linear Deterministic Channel Model (LDCM).

Then, we introduce our novel channel model using polynomial rings:

The Cyclic Polynomial Channel Model (CPCM)¹.

We will elaborate several common properties between the CPCM and the two other channel models within the course of this thesis.

2.1 Common Parameters of the Considered Channel Models

The following common parameters and sets will be consistently used for all considered channel models unless it is stated otherwise. We consider multi-user interference channels with $K_{\text{Tx}} \in \mathbb{N}$ transmitters and $K_{\text{Rx}} \in \mathbb{N}$ receivers. Transmitters are denoted by Tx_i with index $i \in \mathcal{K}_{\text{Tx}} = \{1, ..., K_{\text{Tx}}\}$. Receivers are denoted by Rx_j for $j \in \mathcal{K}_{\text{Rx}} = \{1, ..., K_{\text{Rx}}\}$, respectively. A dedicated message from Tx_i to Rx_j is denoted by \boldsymbol{w}_{ji} and has rate R_{ji} for $i \in \mathcal{K}_{\text{Tx}}$, $j \in \mathcal{K}_{\text{Rx}}$.

In case that transmitters and receivers are combined into a single device, it is called a transceiver T_i . We will further assume that transceivers are capable of perfect full-duplex operation. The set of transceiver-indices is denoted by $\mathcal{K}_T = \{1, ..., K_T\}$ for K_T transceivers T_i . A relay R_i , with index $i \in \mathcal{K}_R$ is an auxiliary device that can both transmit and receive (and hence forward) messages. But in contrast to a transceiver, a relay does not initiate own messages. The set of relay-indices is denoted by $\mathcal{K}_R = \{1, ..., K_R\}$ for K_R relays.

2.2 Gaussian MIMO Channel Model (GMCM)

First, we briefly consider the *GMCM* describing multi-antenna systems with multiple users as considered in [35] for instance. We will discuss some results concerning *timevarying* MIMO channels in Section 4.1.4, and also results of the *time-constant* MIMO

¹Parts of this work have been published in [27].

3-way channel in Sections 5.3 and 5.4. Note that the distinction between time-varying and time-constant MIMO channels is essential in the design of feasible IA schemes for multi-user MIMO interference channels.

2.2.1 GMCM: Definition

Let a transmitter Tx_i be equipped with an arbitrary number of $M_{\operatorname{Tx}_i} \in \mathbb{N}$ transmit antennas, and a receiver Rx_j with M_{Rx_i} receive antennas. The signal transmitted at time instance n from Tx_i is a vector $\boldsymbol{x}_i(n) \in \mathbb{C}^{M_{\operatorname{Tx}_i} \times 1}$ satisfying a power constraint P. The channel matrix for the MIMO channel from Tx_i to Rx_j is denoted by $\boldsymbol{H}_{ji} \in \mathbb{C}^{M_{\operatorname{Rx}_j} \times M_{\operatorname{Tx}_i}}$. These random channel matrices are generated i.i.d. from a continuous probability distribution. We assume full and global channel state information (CSI), i. e., all channel matrices are perfectly known at each user. Depending on the discussed communication problems in the later chapters, we will specify whether the channel coefficients are time-varying or constant throughout the whole communication. The received signal at Rx_j is a vector $\boldsymbol{y}_j(n) \in \mathbb{C}^{M_{\operatorname{Rx}_j} \times 1}$. $\boldsymbol{y}_j(n)$ is a superposition of the transmitted signals from each Tx_i , weighted by \boldsymbol{H}_{ji} , respectively, and i.i.d. complex additive white Gaussian noise (AWGN) $\boldsymbol{z}_j \sim \mathcal{CN}(\boldsymbol{0}_{M_{\operatorname{Rx}_i} \times 1}, \boldsymbol{I}_{M_{\operatorname{Rx}_i} \times M_{\operatorname{Rx}_i}})$:

$$\boldsymbol{y}_{j}(n) = \sum_{i \in \mathcal{K}_{T}} \boldsymbol{H}_{ji} \boldsymbol{x}_{i}(n) + \boldsymbol{z}_{j}(n).$$
(2.1)

2.2.2 Degrees-of-Freedom: A Capacity Approximation for the GMCM

It has been a very challenging task to characterize the maximal achievable rates, i.e., the exact sum-capacity and the capacity region, for most of such Gaussian MIMO multi-user channels. In order to yet obtain an sufficiently accurate insight into these central problems, a widely used approach is to describe the scaling behaviour of the channel capacity within the high SNR regime. Such an approximation of the sum-capacity $C_{\Sigma}(P)$ is represented by:

$$C_{\Sigma}(P) = \operatorname{DoF} \cdot \log(P) + o(P). \tag{2.2}$$

The DoF-measure² is defined in [3] and [37] as the *pre-log factor* of the achieved sumcapacity $C_{\Sigma}(P)$:

$$DoF = \lim_{P \to \infty} \frac{C_{\Sigma}(P)}{\log(P)}.$$
(2.3)

The DoF become accurate at high SNR, i.e., for power $P \to \infty$ as the approximation error vanishes with $o(P) \to 0$.

We will particularly apply the GMCM in Subsection 4.1.4, and in Sections 5.3 and 5.4.

²The DoF are also known as multiplexing gain (cf. [35], [36]).

2.2.3 Relativity of Alignment: Feasibility Conditions

The *relativity of interference alignment* as discussed in [12] is an elementary property that must be satisfied for any IA scheme known up to now. This property basically demands that each receiver requires an independent and sufficiently different view of the channel to enable IA. For the IA schemes considered in [3], [37] and [38], this property is satisfied by either considering time-varying channel coefficients in the single antenna case, or by employing multiple antennas at each receiver, so that the provided schemes do not degenerate and invalidate the use of IA. Similarly, for the ergodic IA scheme in [15], this property is provided by time-varying channel coefficients generated by an ergodic process.

Enabling IA is also described by satisfying a set of *feasibility conditions* for multiuser MIMO channels with constant coefficients as discussed in [39] and [40]. These feasibility conditions demand that there is an interference free-space for a certain set of decodable dedicated signals while interference is completely eliminated after zeroforcing.

In order to ensure that each dedicated signal is linearly decodable at its desired receiver, we demand that the following properties for the proposed communication schemes hold:

Linear interference-free: All signals that are dedicated for the receiver are allocated to a subspace that is linear independent of the interfering signals which are dedicated for other receivers.

Linear decodability: Multiple dedicated signals for the same receiver must be received linearly independent within the dedicated signal subspace.

2.3 Linear Deterministic Channel Model (LDCM)

In order to approximate the capacity of Gaussian multi-user networks with more emphasis on the relative strength of the channel gains, a wireless channel can be characterized in the high-SNR regime by the *linear deterministic channel model* (LDCM). The model was introduced by Avestimehr, Diggavi and Tse³ in [9] and [41] and in the dissertation of Avestimehr [42]. This model is a special class of the general deterministic channel model introduced by El-Gamal and Costa in [43].

One of the main goals of the LDCM is to simplify the analysis of the capacity of Gaussian relay and interference channels with multiple users. Its underlying concept is to eclipse the impact of noise and to highlight the influence of essential properties of wireless signal interaction. In particular the authors of [9] focus on the effects caused by *broadcasting, interference* and *signal-scale* in the shared medium.

Furthermore, the LDCM is a useful tool for deriving the generalized degrees-offreedom (GDoF) as considered in [4], which is a more accurate measure of the approximate capacity than the DoF. For highly accurate capacity characterizations whose capacity approximation error is within a constant gap, a refined linear deterministic channel model using a lower triangular Toeplitz matrix is proposed in [44]. However,

 $^{^3\}mathrm{The}$ LDCM is also called the ADT-model.

the GDoF-metric is not in the scope of this thesis and hence its discussion is omitted. The concept of the constant-gap capacity approximation is only briefly recapitulated.

2.3.1 LDCM: Definition

In the LDCM of [41, Section II], the transmitted signals of each user are described by a sequence of $q \in \mathbb{N}$ bits. Each bit is indexed from the least significant bit level 1 to the most significant bit level q in a $1 \times q$ input bit vector $\boldsymbol{x} = (x(q), ..., x(1))$. The q bit levels transmitted from Tx_i to Rx_j are linearly shifted down by $q - n_{ji}$ levels, so that a number of $n_{ji} \in \mathbb{N}$ bits remains above the noise threshold. This is a so-called *signalscale* representation (*signal-scale*-property), since the q bit levels are scaled down at the receivers. These *bit-pipes* emulate the attenuation of the channel gains from a transmitter to a receiver.

The parameter q for the maximal level is determined by the strongest channel gain, i.e., $q = \max_{i \in \mathcal{K}_{Tx}, j \in \mathcal{K}_{Rx}}(n_{ji})$. For the linear shifting operation, a $q \times q$ shift matrix with ones on the lower side-diagonal and zeros otherwise is used:

$$\boldsymbol{S} = \begin{pmatrix} \boldsymbol{0}_{1\times q} & \boldsymbol{0} \\ \boldsymbol{I}_{q-1\times q-1} & \boldsymbol{0}_{q-1\times 1} \end{pmatrix}.$$
(2.4)

S is a *nil-potent* matrix, since its q-th power and higher powers yield $S^q = \mathbf{0}_{q \times q}$. The linear shift matrix is not invertible and its determinant is $\det(S) = 0$. By applying the shift matrix $q - n_{ji}$ times on x provides the following $1 \times q$ output bit vector y_j :

$$\boldsymbol{y}_j^{\mathsf{T}} = \boldsymbol{S}^{q-n_{ji}} \boldsymbol{x}_i^{\mathsf{T}}$$

= $(0, ..., 0, x_i(q), ..., x_i(q-n_{ji})).$

We obtain a linear down-shifted version of input \boldsymbol{x}_i with $n_{ji} \leq q$ potentially non-zero bit levels remaining in \boldsymbol{y}_i . Bit levels that are shifted below the lowest index of \boldsymbol{y}_i are truncated and lost. This truncation represents the influence of the noise threshold on the received signal. By extending this operation to multiple transmitters and receivers, the received signal yields a superposition of all transmitted signals, which are each shifted by $q - n_{ji}$:

$$\boldsymbol{y}_{j}^{\mathsf{T}} = \sum_{i \in \mathcal{K}_{\mathrm{Tx}}} \boldsymbol{S}^{q-n_{ji}} \boldsymbol{x}_{i}^{\mathsf{T}}, \qquad (2.5)$$

for $j \in \mathcal{K}_{\mathrm{Rx}}$, respectively. The signals of the transmitter are broadcast to each receiver (*broadcast*-property) and the receivers obtain superimposed bits of multiple transmitters (*interference*-property). An exemplary 2-user multiple-access channel is depicted in Figure 2.1 in terms of the LDCM. In order to compute the capacity by means of the LDCM, upper bounds on the maximal number of conveyable bits must be derived. Furthermore, achievable coding schemes must be designed to verify these upper bounds.

2.3.2 Capacity Approximation

The strength of the channel gains in the LDCM and the complex Gaussian channel model are related by $n_{ji} = \lceil \log_2(P) \rceil^+$. For a point-to-point channel from Tx₁ to Rx₁,



Figure 2.1: A linear deterministic multiple-access channel with two transmitters, one receiver, channel gains $n_{11} = 3$, $n_{12} = 4$, and a number of q = 4 bit levels is depicted. Most significant bits are the topmost and least significant bits are the lowermost bit levels.

the mutual information of the linear deterministic point to point channel is:

=

$$C_{\text{LDCM,P2P}} = I(X_1; Y_1) = H(\boldsymbol{S}^{q-n_{11}} \boldsymbol{x}_1^{\mathsf{T}}) = H(\boldsymbol{y}_1^{\mathsf{T}})$$
(2.6)

$$= \sum_{i=1}^{n_{11}} H(y_1(i) \mid y_1(i-1), ..., y_1(1))$$
(2.7)

$$= n_{11}$$
. (2.8)

The entropy in the last line is maximized by the Bernoulli distribution with independent components with parameter $\frac{1}{2}$. This is a capacity approximation within a margin of at most one bit w.r.t. the complex Gaussian channel point-to-point capacity (cf. [1]) for high SNR:

$$\log_2(1+P) \le n_{11} = \left[\log_2(P)\right]^+ \le \log_2(2+2P) \le \log_2(1+P) + 1.$$
(2.9)

A well-known result as, e. g., given in [45, Section 15.3.6], is that the capacity region of the Gaussian multiple-access channel is characterized by the following upper bounds on the rates R_{11} and R_{12} :

$$R_{11} \le I(X_1; Y_1 \mid X_2) = \log_2(1 + P_1),$$
 (2.10)

$$R_{12} \le I(X_2; Y_1 \mid X_1) = \log_2(1 + P_2), \tag{2.11}$$

$$R_{11} + R_{12} \le I(X_1, X_2; Y_1) = \log_2(1 + P_1 + P_2), \tag{2.12}$$

for some input distribution $f_1(x_1)f_2(x_2)$ satisfying $\mathsf{E}(X_1^2) \leq P_1$ and $\mathsf{E}(X_1^2) \leq P_2$ for zero mean unit-variance real-valued Gaussian noise.

In case of the LDCM [41], the analogous representation of the capacity region, i.e., the closure of all achievable rate vectors, for the multiple-access channel is:

$$R_{11} \le I(X_1; Y_1 \mid X_2) = n_{11}, \tag{2.13}$$

$$R_{12} \le I(X_2; Y_1 \mid X_1) = n_{12}, \tag{2.14}$$

$$R_{11} + R_{12} \le I(X_1, X_2; Y_1) = \max(n_{11}, n_{12}), \tag{2.15}$$

by computing the mutual information in terms of the LDCM, with $i \neq j \in \{1, 2\}$:

$$\begin{split} I(X_i; Y_1 \mid X_j) &= H(Y_1 \mid X_j) - H(Y_1 \mid X_1, X_2) \\ &= H(Y_1 \mid X_j) \\ &= H(S^{q-n_{11}} \boldsymbol{x}_1^{\mathsf{T}} + S^{q-n_{12}} \boldsymbol{x}_2^{\mathsf{T}} \mid S^{q-n_{1j}} \boldsymbol{x}_j^{\mathsf{T}}) \\ &= H(S^{q-n_{1i}} \boldsymbol{x}_i^{\mathsf{T}}) \\ &= n_{1i}, \\ I(X_1, X_2; Y_1) &= H(Y_1) - H(Y_1 \mid X_1, X_2) = H(Y_1) \\ &= H(S^{q-n_{11}} \boldsymbol{x}_1^{\mathsf{T}} + S^{q-n_{12}} \boldsymbol{x}_2^{\mathsf{T}}). \\ &= \max(n_{11}, n_{12}). \end{split}$$

This capacity region can be achieved using orthogonal multiple-access schemes. Furthermore, the capacity region of the LDCM is actually within one bit of its Gaussian counterpart (cf. in [9] and [41]).

$$\begin{aligned} \max(\log_2(1+P_1), \log_2(1+P_2)) &\leq \log_2(1+P_1+P_2) \\ &\leq \max(n_{11}, n_{22}) \\ &= \max(\lceil \log_2(P_1) \rceil^+, \lceil \log_2(P_2) \rceil^+) \\ &\leq \max(\log_2(2+2P_1), \log_2(2+2P_2)) \\ &\leq \max(\log_2(1+P_1), \log_2(1+P_2)) + 1. \end{aligned}$$

The LDCM and its extensions have already gained a lot of interest by many researchers working in the realm of multi-user information theory. There is an enormous amount of works dedicated to the characterization the capacity of various multi-user channels based on the LDCM. Among these, we would like to point to other elementary examples in [9] and to some works on *signal-scale interference alignment* [20], [21], [22], [46], *interference neutralization* [47], [48], and *user-cooperation* [23].

Albeit the LDCM considerably simplifies the derivation of capacity-achieving schemes for symmetric channel gains and for small networks, it becomes increasingly challenging to compute the capacity of multi-user networks with non-symmetric channel gains. Furthermore, the capacity expressions derived for the LDCM exhibit some exceptional singularities, especially for fully-symmetric channels (cf. [21] and [22], for instance). In the following, we will introduce our main channel model of this dissertation that deals with the aforementioned challenging problems.

2.4 Cyclic Polynomial Channel Model (CPCM)

In contrast to the LDCM, which is essentially based on a linear shift of bit-vectors, the proposed CPCM is based on a polynomial ring with cyclic shifts within a finite number of dimensions. Our mathematical framework is inspired by algebraic cyclic codes as in [49] and by the related LDCM of [9] as discussed in the previous section. A particular benefit of the CPCM concerns the algebraically convenient description by polynomials. While conventional algebraic codes mainly focus on the detection and correction of errors caused by noise (cf. [50]) the CPCM is dedicated to characterize the effects of multi-user interference, even for arbitrary cyclic shifts.



Figure 2.2: In a cyclic polynomial channel with K_{Tx} transmitters Tx_i and K_{Rx} receivers Rx_j , messages are conveyed by polynomials $v_i(x)$ over the channel by the transfer matrix D. The polynomials $r_j(x)$ comprise the received messages.

We would like to remark, that a closely related channel model for multi-user line-ofsight (LoS) channels was introduced in [51]. Therein, similar effects of cyclic shifting are discussed as well. But in contrast to the approaches pursued in this thesis, the considered scheme in [51] is derived from a graph-theoretic perspective. Furthermore, another related model is investigated in [52] for an orthogonal frequency division multiplexing (OFDM) communication system in the time-frequency domain. The approach in [52] presents a channel decomposition with circulant matrices describing cyclic shifts.

2.4.1 CPCM: Definition

The CPCM provides a conceptual description of a wireless channel with $K_{\text{Tx}} + K_{\text{Rx}}$ users, i. e., K_{Tx} transmitters and K_{Rx} receivers, as depicted in Figure 2.2. We denote a source message to be conveyed from Tx_i to Rx_j , for $i \in \mathcal{K}_{\text{Tx}}$ and $j \in \mathcal{K}_{\text{Rx}}$, by \boldsymbol{w}_{ji} . Such a message \boldsymbol{w}_{ji} comprises $m_{ji} \in \mathbb{N}_0$ subordinate messages (*submessages*) and is denoted by:

$$\boldsymbol{w}_{ji} = (W_{ji}^{[0]}, ..., W_{ji}^{[m_{ji}-1]}), \ W_{ji}^{[l]} \in \mathbb{F},$$
(2.16)

for $l = 1, ..., m_{ji} - 1$. The total number of submessages m_{ji} may also be interpreted as rate demand between Tx_i and Rx_j . The communication channel has $n \in \mathbb{N}$ dimensions. The assignment of messages to these dimensions is represented by polynomials in the indeterminate x of degree n - 1. As for the well-known cyclic codes, we consider a commutative polynomial ring $\mathbb{F}(x)$ modulo $x^n - 1$ with coefficients over a field \mathbb{F} (e.g., [53, Chapter 7 §2, p. 189]). The coefficients denote the carried information.

Encoding Scheme

All dedicated messages to be transmitted by Tx_i are encoded into the coefficients of the polynomial $v_i(x)$ by the encoding function e_i :

$$e_i: (\boldsymbol{w}_{1i}, ..., \boldsymbol{w}_{K_{\mathrm{Rx}}i}) \mapsto v_i(x).$$

The set of dedicated submessages is encoded in the coefficients of $v_i(x)$ with a linear code. A single position within the *n* dimensions⁴ is addressed by an offset $x^0, x^1, ..., x^{n-1}$, from 0 (zero-offset) to n-1 (maximal offset).

⁴The terms 'offsets' and 'dimensions' are synonymously used in this thesis.

Channel Transfer Matrix and Received Signal

An individual cyclic shift between Tx_i to Rx_j is denoted by the multiplication with a monomial $d_{ji} \in \mathcal{D}$ for $\mathcal{D} = \{x^k | k \in \mathbb{N}\} \cup \{0\}$. A zero-valued entry $d_{ji} = 0$ means that there is no link between Tx_i to Rx_j . The channel transfer matrix is assumed to be static and fully known to all users during the whole transmission, and it is defined by:

$$\boldsymbol{D} = (d_{ji})_{1 \le j \le K_{\mathrm{Rx}}, 1 \le i \le K_{\mathrm{Tx}}}.$$
(2.17)

The (cyclic polynomial) transfer function for K_{Tx} transmitters and K_{Rx} receivers is compactly described by a matrix multiplication of the vector of input polynomials \boldsymbol{v} with the channel matrix \boldsymbol{D} , such that the vector of output polynomials \boldsymbol{r} yields:

$$\boldsymbol{r}^{\mathsf{T}} \equiv \boldsymbol{D}\boldsymbol{v}^{\mathsf{T}} \mod(\boldsymbol{x}^n - 1). \tag{2.18}$$

with vectors $\boldsymbol{v} = (v_1(x), ..., v_{K_{Tx}}(x))$ and $\boldsymbol{r} = (r_1(x), ..., r_{K_{Rx}}(x))$. The modulo operation is taken element-wise. The received signal at Rx_j is a superposition of transmitted polynomials:

$$r_j(x) = \sum_{i \in \mathcal{K}_{\mathrm{Tx}}} d_{ji} v_i(x) \mod (x^n - 1).$$
(2.19)

Decoding Scheme

The received coefficients of the polynomials $r_j(x)$ are linearly decoded to obtain an estimate of the dedicated messages \widehat{w}_{ji} . The decoding function for $r_j(x)$ is:

$$f_j: r_j(x) \mapsto (\widehat{\boldsymbol{w}}_{j1}, ..., \widehat{\boldsymbol{w}}_{jK_{\text{Rx}}}).$$

The dedicated messages are correctly decoded if $\hat{w}_{ji} = w_{ji}$ holds for all dedicated transmitter-receiver pairs. If a dedicated message w_{ji} cannot be linearly resolved, then there exists no error-free decoding scheme.

Example: Point-to-Point Channel

Consider the elementary point-to-point channel $(K_{\text{Rx}} = K_{\text{Tx}} = 1)$ with $n \ge 1$ dimensions. Let only a single submessage W_{11} be transmitted at offset $x^{p_{11}}$ and cyclically shifted over the channel by $d_{11} = x^{\delta_{11}}$. For $v_1(x) = W_{11}x^{p_{11}}$ the polynomial received at Rx_1 is:

$$r_1(x) = d_{11}W_{11}x^{p_{11}} \equiv W_{11}x^{\delta_{11}+p_{11}} \mod (x^n - 1).$$

The dedicated message W_{11} is decoded at offset $x^{\delta_{11}+p_{11}} \mod (x^n-1)$.

2.4.2 Degrees-of-Freedom: Capacity of the CPCM

In order to evaluate the performance of a communication scheme over the given channel, we consider an analogous DoF-measure as defined in Section 2.2.2.

Assuming i. i. d. zero mean unit variance Gaussian noise at the receivers and an average power constraint P per message within each dimension, a single interferencefree link between Tx_i and Rx_j has a capacity of $\log(P)$ bits per dimension at high SNR. Recall from (2.3) that the DoF-measure is defined by:

$$\mathrm{DoF} = \lim_{P \to \infty} \frac{C_{\Sigma}(P)}{\log(P)}.$$

In the presented cyclic polynomial channel model, we define the achieved DoF by the total number of submessages M received interference-free per n dimensions:

$$DoF = \lim_{P \to \infty} \frac{\frac{M}{n} \log(P)}{\log(P)} = \frac{M}{n}.$$
(2.20)

The messaging matrix⁵ M denotes the number of demanded submessages dedicated to be conveyed from Tx_i to Rx_j per period of n dimensions:

$$\boldsymbol{M} = (m_{ji})_{1 \le j \le K_{\text{Rx}}, 1 \le i \le K_{\text{Tx}}}, \ m_{ji} \in \mathbb{N}_0.$$

$$(2.21)$$

The total number of messages in the system is computed by:

$$M = \sum_{j \in \mathcal{K}_{\mathrm{Rx}}} \sum_{i \in \mathcal{K}_{\mathrm{Tx}}} m_{ji}.$$
 (2.22)

From the number of dedicated and interfering submessages at each receiver, *upper* bounds on the DoF can be derived (e.g., Section 3.2). The **main task** is to design optimal encoding and decoding schemes for a particular setup of users and rate demands within the CPCM such that these upper bounds are achieved.

Recall the previous example: It is obvious that a messages \boldsymbol{w}_{11} with $m_{11} = L, L \in \mathbb{N}$, submessages can be conveyed within at least $n \geq L$ dimensions in the point-to-point channel. Clearly, if these L submessages are each allocated to distinct dimensions, the single and merely shifted received signal is linearly decodable. If we choose n = L, then $\frac{M}{n} = \frac{L}{L} = 1$ DoF is always achievable. This result is in accordance with the DoF of the Gaussian point-to-point channel.

2.4.3 Relativity of Alignment: Separability Conditions

In terms of the CPCM, communication channels with multi-user interference are fully described by their channel matrices and their messaging matrices. Since we assume that the channel matrix is fully known to each user, one can always trace back all transmitter offsets that are incident to a particular receiver offset.

The proposed *separability conditions*⁶ are slightly stricter than the feasibility conditions, which were briefly discussed in Section 2.2.3. In particular, we will further impose that the signals are interference-free and decodable *per each offset* and not only linear decodability. Hence, if one of the transmitter offsets contains a dedicated signal for a particular receiver offset, the incident transmitter offsets may not carry any desired or interfering signals and must remain silent. Otherwise, the dedicated signal would be subject to interference on that offset. In several cases these conditions are

⁵This matrix is basically a *unicast traffic matrix* as defined in [54], but without a normalization to one.

⁶The terminology is also used similarly in the discussion of real interference alignment in [13].

rather strict and effectively narrow the total number of channels down to the number of feasible channels.

The formulation of these separability conditions provides a useful tool to classify elementary properties of the channel matrix. We will elaborate the formal description of the separability conditions for the CPCM (cf. Sections 3.1, 4.1.1 and 4.2.3 in particular). As a result, we can determine the maximal number of achievable DoF and the feasibility of the proposed communication scheme based on the given channel matrix and the underlying message demands only. We will also discuss an interesting link between the separability conditions of cyclic IA on the CPCM and the feasibility conditions of MIMO IA on the GMCM in Section 3.4.3.

2.4.4 CPCM versus LDCM: An Algebraic Extension

There are various other ways to mathematically describe the cyclic shifts of the CPCM. Another option than using a polynomial ring to describe the behaviour of cyclic shifts, is to use vectors and powers of a particular circulant⁷ $n \times n$ matrix:

$$\boldsymbol{T} = \begin{pmatrix} \boldsymbol{0}_{1 \times n-1} & 1\\ \boldsymbol{I}_{n-1 \times n-1} & \boldsymbol{0}_{n-1 \times 1} \end{pmatrix}.$$
(2.23)

Clearly, this matrix is almost identical to the linear shift matrix in (2.4) for the LDCM, except the top-right element. However, in contrast to the linear shift matrix S, the circulant shift matrix T is not nil-potent, so that no integer exponent will produce a zero-matrix $\mathbf{0}_{n\times n}$. The cyclic shift circulant matrix is invertible and its determinant is $\det(T) = 1$. T is also orthogonal, since $T^{\mathsf{T}} = T^{-1}$ holds.

Furthermore, we can describe the input signal from a transmitter Tx_i , $i \in \mathcal{K}_{\operatorname{Tx}}$, as an $1 \times n$ vector $\boldsymbol{v}_i = (v_i(0), \dots, v_i(n-1))$. The output signal is an $1 \times n$ vector $\boldsymbol{r}_j = (r_j(1), \dots, r_j(n-1))$ at Rx_j , which is a superposition of the input signals cyclically shifted by δ_{ji} positions:

$$\boldsymbol{r}_{j}^{\mathsf{T}} = \sum_{i=1}^{K_{\mathsf{Tx}}} \boldsymbol{T}^{\delta_{ji}} \boldsymbol{v}_{i}^{\mathsf{T}}, \qquad (2.24)$$

for each $j \in \mathcal{K}_{Rx}$. Since the offsets $x^0, ..., x^{n-1}$ may be interpreted as basis-vectors of an *n* dimensional vector space, we can also apply several useful operations from linear algebra like rank(·), dim(·), etc., to this representation.

Each channel, based on the CPCM, has a specific counterpart in the LDCM (and vice versa). We replace the shift matrices $\mathbf{S} \leftrightarrow \mathbf{T}$ and hence the gains $\delta_{ji} \pmod{n} \leftrightarrow n - n_{ji}$:

$$\boldsymbol{r}_{j}^{\mathsf{T}} = \sum_{i=1}^{K_{\mathrm{Tx}}} \boldsymbol{T}^{\delta_{ji}} \boldsymbol{v}_{i}^{\mathsf{T}}, \qquad (2.25)$$

$$\boldsymbol{y}_{j}^{\mathsf{T}} = \sum_{i=1}^{K_{\mathsf{Tx}}} \boldsymbol{S}^{n-n_{ji}} \boldsymbol{x}_{i}^{\mathsf{T}}.$$
 (2.26)

The LDCM basically provides a particular subset of bit-pipes of the LDCM. Thus, by using the cyclic shift operation, we highlight several algebraic properties that are concealed within the LDCM. Only if all channel parameters of the K-user interference

⁷A circulant $n \times n$ matrix $T_n = (t_{j,i})_{0 \le i,j \le n-1}$ has a certain symmetry described by $t_{j,i} = t_{i-j \pmod{n}}$. Note that this is also a Frobenius companion matrix of the polynomial $\sum_{k=0}^{n-1} x^k$.

channel and the 2-user X-channel are fully-symmetric (i. e., $\delta_{ji} = 0$ for the CPCM, and $n_{ji} = q$ for the LDCM), then the channel matrices are raised by a power of zero. Each subchannel would degenerate to an $n \times n$ identity matrix $I_{n \times n}$:

$$S^0 = T^0 = I_{n \times n}. \tag{2.27}$$

As a result, both the CPCM and the LDCM exactly coincide in that case. This fullysymmetric parameterization is an interesting exception mentioned in the characterization of the linear deterministic and symmetric K-user interference channel in [22, Theorem 3.1 & Remark 1], and also for the linear deterministic 2-user X-channel in [21]. Singularities are observed in the sum-rate for fully-symmetric channel gains. For those cases, the authors of [22], [21] show that the achievable rates do not exceed the rates of an orthogonal multiple-access scheme.

Reconsidering this property from the viewpoint of the CPCM, these exceptional symmetries violate the separability conditions of the K-user interference channel (cf. Theorems 3.6 and 3.8) and also of the 2-user X-channel (cf. Theorem 3.9) as we will see in the next chapter. Hence, the CPCM does not only show an important analogy to the LDCM for this case, it also contributes to the analysis of the LDCM by providing a set of potential symmetry-candidates where the LDCM might display this exceptional behaviour. This insight also applies to more general channel symmetries. Furthermore, the proposed communication schemes derived from the CPCM might also indicate a first step showing how to align interference in the LDCM under general channel symmetries.

3 Multi-User Single-Hop Unidirectional Communications

In this chapter, we first consider the DoF of some elementary and well-known channels:

The K-user multiple-access channel,

the K-user broadcast channel, and

the 2-user interference channel.

By applying the CPCM to these elementary channels, we introduce our basic methodology and validate known results on the DoF from a different perspective. With this approach, we also pursue a similar strategy as the authors of [35], [9], [13], [55], and [56], who also introduce new approximation measures or novel channel models for interference networks. In [35], the DoF-analysis of multi-user MIMO channels is introduced. In [9], the authors establish the LDCM. The idea of *real interference alignment* is presented in [13], and the *discrete superposition model* is introduced in [55]. A related finite-field channel model is introduced in [56]. In our next step, we discuss the DoF of three unidirectional channels based on the CPCM:

A general **upper bound**¹ on the DoF of a $K_{Tx} \times K_{Rx}$ -user X-network,

the 2-user X- channel¹,

the K-user interference channel¹, and

the 3-user X-network².

Unidirectional Unicast Communications

In a unidirectional unicast communication system with multiple users, a message is dedicated to be conveyed from a single transmitter Tx_i to a single dedicated receiver Rx_i only. No message is dedicated for a group of multiple receivers as in the case of multicast communications. But nonetheless, a transmitter may transmit multiple different messages that are each dedicated for different receivers, and a receiver may also receive different dedicated messages from multiple transmitters.

¹Parts of this work have been published in [27].

²Parts of this work have been published in [32].

3.1 Elementary Unidirectional Multi-User Channels

First, we apply the CPCM to well-known multi-user channels as the *multiple-access* channel, the broadcast channel and the 2-user interference channel and derive optimal communication schemes to achieve the maximal number of DoF within n dimensions, respectively.

3.1.1 Cyclic Polynomial Multiple-Access Channel



Figure 3.1: The cyclic polynomial multiple-access channel.

Our first example of a unicast communication system with multiple users is a multiple-access channel (MAC). In a K-user MAC, there are $K_{\text{Tx}} = K$ transmitters and only $K_{\text{Rx}} = 1$ receiver. An illustration of the MAC is depicted in Figure 3.1. In this case the messaging matrix degenerates to a $1 \times K$ row vector:

$$M = (m_{11}, ..., m_{1K}). \tag{3.1}$$

Each Tx_i transmits a message \boldsymbol{w}_{1i} that is dedicated for the single receiver Rx_1 . The m_{1i} submessages from Tx_i to Rx_1 are denoted by \boldsymbol{w}_{1i} and the corresponding offsets for the allocation in $v_i(x)$ are denoted by the parameters $p_i^{[l]}$, $l = 0, ..., m_{1i} - 1$, in the vector \boldsymbol{p}_{1i} :

$$\boldsymbol{w}_{1i} = (W_{1i}^{[0]}, \dots, W_{1i}^{[m_{1i}-1]}), \tag{3.2}$$

$$\boldsymbol{p}_{1i} = (p_{1i}^{[0]}, \dots, p_{1i}^{[m_{1i}-1]}). \tag{3.3}$$

The polynomials $v_i(x)$ allocate the m_{1i} submessages by:

$$v_i(x) = \sum_{t=0}^{m_{1i}-1} W_{1i}^{[t]} x^{p_{1i}[t]}.$$
(3.4)

Here, the vector of transmitted polynomials is $\mathbf{v} = (v_1(x), ..., v_K(x))$ and the vector of received polynomials degenerates to a single polynomial $\mathbf{r} = r_1(x)$. From (2.19), we obtain:

$$r_1(x) \equiv \sum_{i \in \mathcal{K}} d_{1i} v_i(x) \mod (x^n - 1).$$
(3.5)

The submessages dedicated for Rx_1 must be linearly decodable. Hence, we demand that each submessage is effectively received in a separate dimension after the linear decoding procedure is performed. Thus, to keep all dedicated messages segregated

	x^0	x^1	x^2	x^3	x^4
$v_1(x)$	$W_{11}^{[0]}$	0	0	$W_{11}^{[1]}$	0
$v_2(x)$	$W_{12}^{[0]}$	0	0	0	0
$v_3(x)$	0	$W_{13}^{[1]}$	0	0	$W_{13}^{[0]}$
$r_1(x)$	$W_{13}^{[1]}$	$W_{11}^{[0]}$	$W_{12}^{[0]}$	$W_{13}^{[0]}$	$W_{11}^{[1]}$

Figure 3.2: This is a valid solution for orthogonal MA in a 3-user MAC for n = 5 dimensions, with $m_{11} = 2$, $m_{12} = 1$, and $m_{13} = 2$ submessages, and with cyclic 'right-shifts' $d_{11} = x$, $d_{12} = x^2$, and $d_{13} = x^4$. All separability conditions are fulfilled and Rx₁ can decode all five submessages achieving 1 DoF in total. The parameters are: $p_{11}^{[0]} = 0$, $p_{11}^{[1]} = 3$, $p_{11}^{[0]} = 0$, $p_{13}^{[0]} = 4$, $p_{13}^{[1]} = 1$.

at Rx_1 , we demand that the following *multiple-access* conditions pairwise hold for all distinct $i \neq k \in \mathcal{K}_{Tx}$:

$$d_{1i}x^{p_{1i}^{[l]}} \not\equiv d_{1k}x^{p_{1k}^{[l]}} \mod(x^n - 1), \tag{3.6}$$

for all $l \in \{0, ..., m_{1i} - 1\}$, $t \in \{0, ..., m_{1k} - 1\}$. Furthermore, the encoder must be linear. We demand that different submessages from the same Tx_i are also pairwise distinct:

$$d_{1i}x^{p_{1i}^{[l]}} \not\equiv d_{1k}x^{p_{1i}^{[t]}} \mod(x^n - 1), \tag{3.7}$$

for all $l, t \in \{0, ..., m_{1i} - 1\}, l \neq t$.

3.1.2 Orthogonal Multiple-Access (MA)

An orthogonal multiple-access (MA) scheme is a very basic allocation procedure for the offset-parameters p_{1i} , for all $i \in \mathcal{K}_{\text{Tx}}$. Each transmitted submessage is allocated at a distinct offsets at Rx₁. We will show that orthogonal MA achieves the following sum-DoF for $M = \sum_{i=1}^{K} m_{1i}$ submessages over $n = \sum_{k=1}^{K} m_{1k}$ dimensions:

$$\text{DoF} \le \frac{\sum_{i=1}^{K} m_{1i}}{\sum_{k=1}^{K} m_{1k}} = 1.$$
(3.8)

An exemplary encoding that uses an orthogonal MA scheme is depicted in the table of Figure 3.2. An orthogonal MAC scheme is already sufficient to achieve the upper bound on the DoF of the MAC.

Lemma 3.1. The upper bound of 1 DoF in a MAC is achieved by orthogonal MA for $n = \sum_{i=1}^{K} m_{1i}$ dimensions.

Proof:

(a) Necessity of $n \ge \sum_{i=1}^{K} m_{1i}$ dimensions:

Since all submessage must be linearly decodable at receiver Rx_1 , the number of dimensions n is lower bounded by the total number of all submessages $n \geq \sum_{i=1}^{K} m_{1i}$. Otherwise dedicated messages would overlap at Rx_1 and can not be resolved linearly.

(b) Sufficiency of orthogonal MA to achieve 1 DoF:

We may assume any channel matrix D with arbitrary $d_{ij} \in D$. W.l.o.g., we first fix all parameters of p_{11} , such that all the $p_{11}^{[\cdot]}$ are pairwise distinct within the offsets 0, ..., n-1. Now, a number of m_{11} offsets at Rx₁ is already used. Then, we iteratively fix the other offsets p_{1i} for i = 2, ..., K, so that they do not overlap with the already fixed offsets at receiver Rx₁. All submessages are orthogonal to each other within ndimensions. This is clearly feasible since all Tx_i can accommodate m_{1i} submessages at any offset within $n = \sum_{i=1}^{K} m_{1i}$ dimensions. Altogether, Rx₁ can decode all messages and the MA scheme achieves the upper bound of 1 DoF in total.

3.1.3 Cyclic Polynomial Broadcast Channel



Figure 3.3: The cyclic polynomial broadcast channel.

An inverted example of the MAC is the cyclic polynomial broadcast channel (BC). The K-user BC has one transmitter Tx_1 and $K_{Tx} = K$ receivers Rx_j , with $j \in \mathcal{K}_{Rx}$. The BC is depicted in Figure 3.3. Transmitter Tx_1 intends to convey each message w_{j1} to its dedicated receiver Rx_j . The messaging matrix degenerates to a $K \times 1$ column vector:

$$\boldsymbol{M} = (m_{11}, m_{21}, \dots, m_{K1})^{\mathsf{T}}.$$
(3.9)

Analogously to the MAC, we define the vectors for submessages and offset parameters:

$$\boldsymbol{w}_{j1} = (W_{j1}^{[0]}, ..., W_{j1}^{[m_{j1}-1]}), \tag{3.10}$$

$$\boldsymbol{p}_{j1} = (p_{j1}^{[0]}, \dots, p_{j1}^{[m_{j1}-1]}). \tag{3.11}$$

The transmitted signal from Tx_1 is:

$$v_1(x) = \sum_{j \in \mathcal{K}_{\text{Tx}}} \sum_{k=0}^{m_{j1}-1} W_{j1}^{[k]} x^{p_{j1}^{[k]}}.$$
(3.12)

The received signal at Rx_j is:

$$r_j(x) = d_{j1}v_1(x). \tag{3.13}$$

Messages, that are dedicated for receiver Rx_j but not for Rx_k , must be linear independent at Tx_1 . Again we demand that the submessages are allocated to distinct

dimensions. More precisely, the *intra-user interference conditions* for different receivers $j \neq k \in \mathcal{K}_{Rx}$ are expressed by:

$$x^{p_{j_1}^{[l]}} \not\equiv x^{p_{k_1}^{[l]}} \mod (x^n - 1), \tag{3.14}$$

for $l \in \{0, ..., m_{j1} - 1\}$, $t \in \{0, ..., m_{k1} - 1\}$. The following lemma is basically analogous to Theorem 3.1.

Lemma 3.2. The upper bound of 1 DoF in a BC is achieved by an orthogonal MA scheme for $n = \sum_{i=1}^{K} m_{1i}$ dimensions.

Proof:

(a) Necessity of $n \ge \sum_{j=1}^{K} m_{j1}$ dimensions:

Since we demand a linear encoder, each submessage must already be linearly resolvable at the transmitter. Thus, the proof is basically analogous to Theorem 3.1 (a) for swapped indices.

(b) Sufficiency to achieve 1 DoF by orthogonal MA:

We assume a channel matrix D with arbitrary $d_{j1} \in \mathcal{D}$. W.l.o.g., we may fix p_{11} first, such that m_{11} offsets are used at Tx_1 . Then, we iteratively fix the offsets p_{j1} , for j = 2, ..., K, so that they do not overlap and remain orthogonal to the already fixed offsets at Tx_1 .

The received signals $r_j(x)$ are merely cyclically shifted versions of the transmitted signal $v_1(x)$. In the received signals $r_j(x)$, the dimensions with non-dedicated messages are discarded and each dedicated submessage \boldsymbol{w}_{j1} is decoded at receiver Rx_j . Thus, the intra-user interference conditions are satisfied by construction. Note that even all non-dedicated messages are decodable at each receiver. Altogether, the MA scheme achieves $\frac{M}{n} = \frac{\sum_{i=1}^{K} m_{j1}}{\sum_{i=1}^{K} m_{j1}} = 1$ DoF on the given BC.

3.1.4 Cyclic Polynomial 2-User Interference Channel

Yet another elementary unicast communication problem is described by the 2-user interference channel (2IFC). The corresponding Gaussian 2IFC is a fundamentally important object in the research area of multi-user information theory and has already been thoroughly discussed over several decades. However, note that the exact capacity characterization of the Gaussian 2IFC is still not known yet. Particularly important works on the 2IFC are [57], [4], [6], [7], [8], [58], [14], and [36], discussing interference cancellation, genie-aided upper-bounds, treating interference as noise, etc. The DoF are known to be upper bounded by 1 as shown in [36]. The capacity region of the linear deterministic 2IFC is provided in [59].

The cyclic polynomial 2IFC comprises $K_{\text{Tx}} = 2$ transmitters and $K_{\text{Rx}} = 2$ receivers, and thus we have $\mathcal{K}_{\text{Rx}} = \mathcal{K}_{\text{Tx}} = \mathcal{K}$. Tx₁ desires to convey a message \boldsymbol{w}_{11} to Rx₁ with m_{11} submessages and concurrently, Tx₂ desires to convey \boldsymbol{w}_{22} to Rx₂ with m_{22} submessages. This corresponds to a 2 × 2 diagonal messaging matrix:

$$M = diag(m_{11}, m_{22}).$$
 (3.15)



Figure 3.4: Messaging in the cyclic polynomial interference channel.

We have $M = m_{11} + m_{22}$ submessages in total. The channel matrix for the 2IFC is defined as:

$$\boldsymbol{D} = \begin{pmatrix} d_{11} & d_{12} \\ d_{21} & d_{22} \end{pmatrix}. \tag{3.16}$$

The signals of each transmitter interfere at the undesired receivers over the crosschannels d_{12} and d_{21} . An illustration of the 2IFC is depicted in Figure 3.4. We denote the vector of submessages and of parameters for signals from Tx_i to Rx_i , with $i \in \mathcal{K}$, by:

$$\boldsymbol{w}_{ii} = (W_{ii}^{[0]}, \dots, W_{ii}^{[m_{ii}-1]}), \qquad (3.17)$$

$$\boldsymbol{p}_{ii} = (p_{ii}^{[0]}, \dots, p_{ii}^{[m_{ii}-1]}). \tag{3.18}$$

The transmitted signals from Tx_1 and Tx_2 are:

$$v_1(x) = \sum_{k=0}^{m_{11}-1} W_{11}^{[k]} x^{p_{11}^{[k]}}, \tag{3.19}$$

$$v_2(x) = \sum_{k=0}^{m_{22}-1} W_{22}^{[k]} x^{p_{22}^{[k]}}.$$
(3.20)

The polynomials received at Rx_1 and Rx_2 yield:

$$r_1(x) \equiv d_{11}v_1(x) + d_{12}v_2(x) \mod(x^n - 1),$$

$$r_2(x) \equiv d_{21}v_1(x) + d_{22}v_2(x) \mod(x^n - 1).$$

The 2IFC introduces another elementary type of interference: *inter-user inter-ference*. We demand that the dedicated messages are received separately from the undesired interfering messages. In the 2IFC, there are only two *inter-user interference* conditions to consider:

$$d_{11}x^{p_{11}^{[l]}} \not\equiv d_{12}x^{p_{22}^{[t]}} \mod(x^n - 1), \tag{3.21}$$

$$d_{21}x^{p_{11}^{[l]}} \neq d_{22}x^{p_{22}^{[l]}} \mod(x^n - 1), \tag{3.22}$$

for all $l \in \{0, 1, ..., m_{11} - 1\}$ and all $t \in \{0, 1, ..., m_{22} - 1\}$.

Lemma 3.3. If $det(D) \equiv 0 \mod(x^n - 1)$ holds, orthogonal MA is sufficient to achieve the upper bound of 1 DoF in the 2IFC.

Proof:

(a) Necessity of $n \ge m_{11} + m_{22}$ dimensions:

At Rx_1 , a number of m_{11} dedicated submessages must be linearly decodable and at
least m_{22} dimensions are occupied by interference. And vice versa at Rx₂, m_{22} dedicated submessages must be decodable and at least m_{11} dimensions are occupied by interference. Hence, at both receivers the necessary number of dimensions is lower bounded by $n = m_{11} + m_{22}$.

(b) Sufficiency of orthogonal MA to achieve 1 DoF: The given condition on the determinant implies:

$$det(D) \equiv 0 \mod(x^n - 1)$$

$$\Leftrightarrow d_{11}d_{22} - d_{12}d_{21} \equiv 0 \mod(x^n - 1)$$

$$\Leftrightarrow d_{11}d_{12}^{-1} \equiv d_{21}d_{22}^{-1} \mod(x^n - 1).$$
(3.23)

Due to this property, the separability conditions (3.21) and (3.22) coincide. We fix both parameters p_{11} and p_{22} such that orthogonality of their received offsets at Rx₁, and hence (3.21), is fulfilled. (3.22) will be fulfilled accordingly. As a result, all signals are received in orthogonal dimensions and both receivers can decode their dedicated signals (and even their interfering signals) within $n = m_{11} + m_{22}$ dimensions.

Clearly, if the converse statement $\det(D) \neq 0 \mod(x^n - 1)$ holds, a scheme purely based on orthogonal MA is not sufficient to achieve 1 DoF in the 2IFC exactly. A subset of signals will interfere and prohibit decodability of all signals. Thus $n > m_{11} + m_{22}$ dimensions are necessary if only orthogonal MA is applied.

There are two approaches to tackle this problem. The first approach is accomplished as follows. An orthogonal MA scheme that uses sufficiently long messages and a wellchosen guard interval is capable to asymptotically achieve 1 DoF. Let the cyclic shifts that are imposed by the channel be parameterized by δ_{ij} , i.e., we have $d_{ij} = x^{\delta_{ij}}$.

Corollary 3.4. An orthogonal multiple-access scheme **asymptotically** achieves 1 DoF in the 2-user interference channel for $m_{11} + m_{22} \gg \delta_{11} + \delta_{12} + \delta_{21} + \delta_{22}$.

Proof:

Let $n = m_{11} + m_{22} + 2\Delta_0$ with the guard interval $\Delta_0 = \delta_{12} + \delta_{22} + \delta_{11} + \delta_{21} \ge 0$. We first consider an allocation for Rx_1 and fix the parameters in p_{11} such that a continuous frame of m_{11} dimensions is allocated for w_{11} at Rx_1 within the dimensions $0, ..., m_{11}-1$. Then, we leave a guard interval of Δ_0 dimensions at Rx_1 and fix the parameters in p_{22} such that a continuous frame of m_{22} dimensions is allocated for w_{22} at Rx_1 within the dimensions $m_{11} + \Delta_0, ..., m_{11} + \Delta_0 + m_{22} - 1$. The remaining Δ_0 dimensions within $m_{11} + \Delta_0 + m_{22}, ..., m_{11} + \Delta_0 + m_{22} + \Delta_0 - 1$ are unused at Rx_1 . Clearly, the signals at Rx_1 do not interfere within $n = m_{11} + \Delta_0 + m_{22} + \Delta_0$ dimensions for $\Delta_0 \ge 0$.

At Rx_2 , this allocation also yields two frames of m_{11} and m_{22} dimensions, respectively. The frame of m_{11} dimensions occupies the dimensions:

$$0 - \delta_{11} + \delta_{21}, \dots, m_{11} - \delta_{11} + \delta_{21}.$$

The frame of m_{22} dimensions occupies the dimensions

$$m_{11} + \Delta_0 - \delta_{12} + \delta_{22}, \dots, m_{11} + \Delta_0 + m_{22} - \delta_{12} + \delta_{22}.$$

At Rx_2 , the gap of unused frames on the 'right-hand side' of m_{11} occupies a number of:

$$\Delta_1 = -\delta_{12} + \delta_{22} + m_{11} + \Delta_0 - (-\delta_{11} + \delta_{21} + m_{11})$$

= $2(\delta_{11} + \delta_{22}) \ge 0$ (3.24)

dimensions. The second gap of unused frames on the 'right-hand side' of m_{22} has:

$$\Delta_2 = -\delta_{11} + \delta_{21} + n - (-\delta_{12} + \delta_{22} + m_{11} + \Delta_0 + m_{22})$$

= $2(\delta_{12} + \delta_{21}) \ge 0$ (3.25)

dimensions. As Δ_0 , Δ_1 and Δ_2 are all non-negative, the signals do neither interfere at Rx₂. Altogether, for $m_{11} + m_{22} \gg 2\Delta_0 = \Delta_1 + \Delta_2$, the MA scheme asymptotically achieves the value one:

$$\mathrm{DoF} \leq \frac{M}{n} = \frac{m_{11} + m_{22}}{m_{11} + m_{22} + 2\Delta_0} \to 1. \tag{3.26}$$

We will see in the following, that 1 DoF is also exactly achievable for the case $det(D) \neq 0 \mod(x^n - 1)$ by applying a *linear encoding and decoding (LEaD)* scheme with interference cancellation.

3.1.5 Linear Encoding and Decoding (LEaD)

In our previous considerations, the overlap of multiple signals is avoided beforehand by applying an orthogonal MA scheme so that each submessage is allocated to an individual dimension, respectively. This is independent of whether it is a dedicated or an interfering signal. Basically, an MA scheme treats undesired signals in the same manner as desired signals and then discards the interference afterwards. A LEaD scheme, however, provides the opportunity to decode overlapping interference from the received signal by using known side-information about interfering signals. In terms of the CPCM, the basic idea is to repeat the interfering signals within the yet unused and interference-free signal space of the undesired receiver. These repeated signals provide full side-information about the interference in order to cancel the overlapping interference from the dedicated signals. There are related LEaD schemes used for the LDCM (e.g., as in [59] and [21]). In the following, we define some auxiliary sets to describe the collection of dedicated offsets, of interfering offsets, etc.

The set of all offsets is defined by:

$$\mathcal{N} = \{0, ..., n-1\}.$$
(3.27)

The set of dedicated offsets from Tx_i allocated at Rx_j , for $i, j \in \mathcal{K}$, is defined by:

$$\mathcal{S}_{ji} = \{k \in \mathbb{N} \mid d_{ji} x^{p_{ji}^{[l]}} \equiv x^k \mod (x^n - 1), l \in \{0, ..., m_{ji} - 1\}\}.$$
(3.28)

The set of interfering offsets from Tx_j received at Rx_j , for $i, j, t \in \mathcal{K}$, is defined by:

$$\mathcal{I}_{ji} = \{k \in \mathbb{N} \mid d_{ji} x^{p_{ti}^{[l]}} \equiv x^k \mod (x^n - 1), l \in \{0, ..., m_{ti} - 1\}, t \neq j\}.$$
(3.29)

If an offset contains neither dedicated nor interfering signals, it is unused and empty. For the case of the 2IFC, the set of unused offsets at Rx_i is:

$$\mathcal{U}_i = \mathcal{N} \smallsetminus (S_{ii} \cup \mathcal{I}_{ij}), \tag{3.30}$$

with $i \neq j \in \mathcal{K}$. Altogether, the particular sets of interest for the 2IFC are \mathcal{N} , \mathcal{S}_{11} , \mathcal{S}_{22} , \mathcal{I}_{12} , \mathcal{I}_{21} , \mathcal{U}_1 and \mathcal{U}_2 .

Theorem 3.5. A combined orthogonal multiple-access and LEaD scheme achieves the upper bound of 1 DoF of the 2IFC.

Proof:

(a) Necessity of $n \ge m_{11} + m_{22}$ dimensions:

The proof of (a) is already provided in Lemma 3.3 (a). Moreover, note that:

$$n \le m_{11} + m_{22} = |\mathcal{S}_{11}| + |\mathcal{I}_{12}| = |\mathcal{S}_{22}| + |\mathcal{I}_{21}| = |\mathcal{N}|. \tag{3.31}$$

(b) Sufficiency of orthogonal MA and LEaD to achieve 1 DoF:

For $\det(D) \equiv 0 \mod (x^n - 1)$, achievability by orthogonal MA has already been proven in Lemma 3.3 (b). Hence, it suffices to consider the complementary case with $\det(D) \not\equiv 0 \mod (x^n - 1)$.

W.l.o.g. we fix vectors p_{11} and p_{22} so that the inter-user interference condition in (3.21) holds, i.e., we apply an orthogonal offset allocation at Rx₁. Then, there is no overlapping interference at Rx₁ and we obtain the following properties:

$$S_{11} \cap \mathcal{I}_{12} = \emptyset, \ S_{11} \cup \mathcal{I}_{12} = \mathcal{N}, \ \mathcal{U}_1 = \emptyset.$$
 (3.32)

Some offsets will violate (3.22) and overlap at Rx_2 :

$$S_{22} \cap \mathcal{I}_{21} \neq \emptyset, \ S_{22} \cup \mathcal{I}_{21} \subset \mathcal{N}, \ \mathcal{U}_2 \neq \emptyset.$$
 (3.33)

But, there are also unused offsets at Tx_2 in the non-empty set \mathcal{U}_2 :

$$\mathcal{U}_2 = \mathcal{N} \setminus (\mathcal{S}_{22} \cup \mathcal{I}_{21}). \tag{3.34}$$

The number of unused offsets $|\mathcal{U}_2|$ and the number of overlapping offsets $|\mathcal{S}_{22} \cap \mathcal{I}_{21}|$ at Rx_2 is equal due to the inclusion-exclusion principle of united sets and due to (3.31):

$$\begin{aligned} |\mathcal{U}_2| &= |\mathcal{N}| - |\mathcal{S}_{22} \cup \mathcal{I}_{21}| = |\mathcal{N}| - (|\mathcal{S}_{22}| + |\mathcal{I}_{21}| - |\mathcal{S}_{22} \cap \mathcal{I}_{21}|) \\ &= |\mathcal{N}| - (|\mathcal{N}| - |\mathcal{S}_{22} \cap \mathcal{I}_{21}|) = |\mathcal{S}_{22} \cap \mathcal{I}_{21}|. \end{aligned}$$
(3.35)

In order to enable Rx_2 to cancel the overlapping interference from Tx_1 in $S_{22} \cap \mathcal{I}_{21}$, Tx_1 repeats the overlapping part of its interfering signals within the offsets of \mathcal{U}_2 . Now, Rx_2 cancels the known interference from Tx_1 and hence decodes its dedicated signal w_{22} . Rx_1 cancels the repeated signals from Tx_1 using the decodable signals received in \mathcal{S}_{11} . This scheme corresponds to a simple linear decoding procedure. Both receivers can decode their dedicated and even the interfering signals. Altogether, the combined orthogonal MA and LEaD scheme achieves 1 DoF for a D with arbitrary $d_{ii} \in \mathcal{D}$.

Up to this point, we have considered three elementary channels and introduced the corresponding separability conditions. We provided achievable schemes based on MA and LEaD. In the following, we focus on more extensive multi-user channels and we will also introduce cyclic IA in particular.

3.2 A General Upper Bound for $K_{Tx} \times K_{Rx}$ - User *X*-Networks

Before continuing with further unidirectional networks, we provide a more general upper bound on the DoF and combine the proof of the optimality for the theorems given in this chapter. Note that the general upper bound is basically analogous to [37, Theorem 1], albeit applied to the CPCM.

We denote the collection of dedicated messages received at \mathbf{Rx}_j by the *j*-th row vector of M:

$$\boldsymbol{\mu}_{j} = (m_{j1}, \dots, m_{jK_{\mathrm{Tx}}}), \tag{3.36}$$

and we obtain $M = (\mu_1^{\mathsf{T}}, ..., \mu_{K_{\mathsf{Rx}}}^{\mathsf{T}})^{\mathsf{T}}$. The total number of dedicated messages to be conveyed from all transmitters to Rx_j corresponds to summing up the entries in μ_j :

$$\|\boldsymbol{\mu}_j\|_1 = \sum_{i=1}^{K_{\mathrm{Tx}}} m_{ji}.$$

The collection of messages transmitted from Tx_i is given by the *i*-th column vector of M:

$$\boldsymbol{\nu}_i^{\mathsf{T}} = (m_{1i}, \dots, m_{K_{\mathrm{Rx}}i}),$$

so that $M = (\nu_1, ..., \nu_{K_{Tx}})$. The total number of messages transmitted from Tx_i is the sum of elements in v_i , respectively:

$$\|\boldsymbol{\nu}_i^\mathsf{T}\|_1 = \sum_{j=1}^{K_{\mathrm{Rx}}} m_{ji}.$$

In general, the *multiple-access interference conditions* demand separability of all $\|\boldsymbol{\mu}_{j}\|_{1}$ desired messages, so that these messages are received interference-free at Rx_j. Furthermore, the *intra-user interference conditions* demand separability of all $\|\boldsymbol{\nu}_{i}^{\mathsf{T}}\|_{1}$ messages transmitted from Tx_i. For a pair (Rx_j, Tx_i), there must be at least:

$$\|\mu_{j}\|_{1} + \|\nu_{i}^{\mathsf{T}}\|_{1} - m_{ji}$$

dimensions to ensure the two given separability conditions hold. Since entry m_{ji} appears twice when adding the 1-norm of the corresponding row vectors and column vectors, it must be subtracted once.

To further ensure that the *inter-user interference* conditions hold, the maximum of the sum $\|\boldsymbol{\mu}_j\|_1 + \|\boldsymbol{\nu}_i^{\mathsf{T}}\|_1 - m_{ji}$ over all $i \in \mathcal{K}_{\mathsf{Tx}}, j \in \mathcal{K}_{\mathsf{Rx}}$ provides the minimal feasible *n*. Altogether, the minimal necessary number of *n* dimensions is lower bounded by:

$$n \geq \max_{j \in \mathcal{K}_{\text{Rx}, i \in \mathcal{K}_{\text{Tx}}}} \left(\|\boldsymbol{\mu}_{j}\|_{1} + \|\boldsymbol{\nu}_{i}^{\text{T}}\|_{1} - m_{ji} \right) \\ = \max_{j \in \mathcal{K}_{\text{Rx}, i \in \mathcal{K}_{\text{Tx}}}} \left(\sum_{k=1}^{K_{\text{Tx}}} m_{jk} + \sum_{l=1}^{K_{\text{Rx}}} m_{li} - m_{ji} \right),$$
(3.37)

with $j \in \mathcal{K}_{\text{Rx}}$ and $i \in \mathcal{K}_{\text{Tx}}$ for a fixed messaging matrix M. Thus, from the definition in (2.20), the DoF of a $K_{\text{Tx}} \times K_{\text{Rx}}$ -user X-channel are upper bounded by:

$$\text{DoF} \leq \frac{\sum_{i=1}^{K_{\text{Rx}}} \sum_{i=1}^{K_{\text{Rx}}} m_{ji}}{\max_{j \in \mathcal{K}_{\text{Rx}}, i \in \mathcal{K}_{\text{Tx}}} \left(\sum_{k=1}^{K_{\text{Rx}}} m_{jk} + \sum_{l=1}^{K_{\text{Rx}}} m_{li} - m_{ji} \right)}.$$
 (3.38)



Figure 3.5: The polynomial 2-user X-channel with messages W_{11} , W_{12} , W_{21} and W_{22} between Tx₁, Tx₂, Rx₁ and Rx₂ and the cyclic shifts d_{11} , d_{12} , d_{21} and d_{22} of the transfer matrix **D**.

3.3 Cyclic IA on the Cyclic Polynomial 2 - User *X*- Channel

A 2-user X-channel is physically equivalent to a 2IFC. However, it has a number of four independent messages w_{11} , w_{21} , w_{12} and w_{22} instead of only two independent messages w_{11} and w_{22} for two separate dedicated user-pairs. The sets of indices are again $\mathcal{K}_{Tx} = \mathcal{K}_{Rx} = \mathcal{K} = \{1, 2\}$. The 2-user X-channel as depicted in Figure 3.5 does not only combine the properties of a 2-user multiple-access, a 2-user broadcast and a 2-user interference channel, it also serves as the most basic example for the application of cyclic IA. The X-channel has also been considered in [11] and in [38] for the initial development of the IA principle for time-varying MIMO X-channels.

In terms of the CPCM, the task is to convey and decode four dedicated messages w_{11} , w_{12} , w_{21} and w_{22} interference-free within *n* dimensions. The scheme is optimal if *n* is minimal and still feasible. The channel transfer matrix of the X-channel is the same as in (3.16), namely:

$$\boldsymbol{D} = \left(\begin{array}{cc} d_{11} & d_{12} \\ d_{21} & d_{22} \end{array} \right).$$

For an introductory discussion of cyclic IA, we begin with a simplified messaging matrix $M = \mathbf{1}_{2\times 2}$ with only a single submessage per user-pair. The general case with arbitrary message lengths will be treated afterwards in Sections 3.3.3 and 3.3.4. To simplify notation, and since $m_{ji} = 1$ for all $j, i \in \mathcal{K}$ holds, the superscripts [0] are omitted, i. e., we use $W_{ij}^{[0]} = W_{ji} = w_{ji}$ and $p_{ij}^{[0]} = p_{ji}$ here.

3.3.1 Perfect Cyclic Interference Alignment

We call interfering submessages to be *aligned* if they are received within the same dimension at an undesired receiver. In order to align the submessages W_{22} and W_{21} of $v_1(x)$ and $v_2(x)$, they must overlap in the same dimension at Rx_1 , but they must remain distinct in different dimensions at Rx_2 . To align the submessages W_{11} and W_{12} of $v_1(x)$ and $v_2(x)$, they must overlap at Rx_2 , but remain distinct at Rx_1 , accordingly. An example of such an alignment is shown in the table of Figure 3.6. The following

	x^0	x^1	x^2
$v_1(x)$	W ₁₁	0	W_{21}
$v_2(x)$	0	W_{12}	W_{22}
$r_1(x)$	$W_{21} + W_{22}$	W_{11}	W_{12}
$r_2(x)$	$W_{11} + W_{12}$	W_{22}	W_{21}

Figure 3.6: Cyclic IA is applied on a 2-user X- channel with n = 3 dimensions. The following cyclic 'right-shifts' $d_{11} = d_{12} = x^1$, $d_{21} = x^3$ and $d_{22} = x^2$ are assumed. The parameters are $p_{11} = 0$, $p_{12} = 1$, $p_{21} = 2$, $p_{22} = 2$ and n = 3 and $\frac{4}{3}$ DoF are achieved. Aligned interference is highlighted in red.

polynomials with the offset parameters $p_{11}, p_{12}, p_{21}, p_{22} \in \mathbb{N}_0$ are used for transmission:

$$v_1(x) = W_{11}x^{p_{11}} + W_{21}x^{p_{21}}, (3.39)$$

$$v_2(x) = W_{12}x^{p_{12}} + W_{22}x^{p_{22}}.$$
(3.40)

Then by (2.18), the received polynomials at Rx_1 and Rx_2 yield:

$$r_{1}(x) \equiv d_{11}W_{11}x^{p_{11}} + d_{12}W_{12}x^{p_{12}} + d_{11}W_{21}x^{p_{21}} + d_{12}W_{22}x^{p_{22}} \mod(x^{n} - 1),$$

$$r_{2}(x) \equiv d_{21}W_{21}x^{p_{21}} + d_{22}W_{22}x^{p_{22}} + d_{23}W_{23}x^{p_{23}} + d_{23}W$$

The messages dedicated for Rx_j must be linearly decodable. Thus, we demand that the *multiple-access interference conditions* hold with the indices $j \in \mathcal{K}$, and $i \neq l \in \mathcal{K}$:

$$d_{ji}x^{p_{ji}} \neq d_{jl}x^{p_{jl}} \mod(x^n - 1).$$
(3.43)

The two messages from Tx_i dedicated for different receivers Rx_j and Rx_k must also be linear decodable. Accordingly, we demand that the *intra-user interference conditions* with $j \neq k \in \mathcal{K}$, $i \in \mathcal{K}$, hold:

$$x^{p_{ji}} \not\equiv x^{p_{ki}} \mod(x^n - 1).$$
 (3.44)

Moreover, the interfering messages must also be received separately from the dedicated messages at each Rx_j . Hence, the *inter-user interference conditions* must hold with $j \neq k \in \mathcal{K}$, $i \neq l \in \mathcal{K}$:

$$d_{ji}x^{p_{ji}} \neq d_{jl}x^{p_{kl}} \mod(x^n - 1).$$
(3.45)

These negated congruences (3.43), (3.44) and (3.45) are the *separability conditions* of the cyclic polynomial 2-user X-channel. The separability conditions do not preclude to align intra- and inter-user interference to a single dimension at each Rx_j , which is expressed by:

$$d_{ji}x^{p_{ki}} \equiv d_{jl}x^{p_{kl}} \mod(x^n - 1), \tag{3.46}$$

with $j \neq k \in \mathcal{K}$ and $i \neq l \in \mathcal{K}$. Such an alignment is called *perfect* because the two different interference signals perfectly overlap in exactly the same dimensions per receiver.

We further remark that (3.46) substituted into (3.45) yields (3.44):

$$d_{ji}x^{p_{ji}} \not\equiv d_{jl}x^{p_{kl}} \mod(x^n - 1)$$

$$\Rightarrow d_{ji}x^{p_{ji}} \not\equiv d_{ji}x^{p_{ki}} \mod(x^n - 1)$$

$$\Rightarrow x^{p_{ji}} \not\equiv x^{p_{ki}} \mod(x^n - 1),$$

with the indices $j \neq k \in \mathcal{K}$ and $i \neq l \in \mathcal{K}$. Hence, we can neglect the condition (3.45) if both (3.44) and (3.46) hold.

From the upper bound (3.38) and $M = \mathbf{1}_{2\times 2}$, the 2 user X-channel is upper bounded by $\frac{4}{3}$ DoF.

Theorem 3.6. A perfect cyclic IA scheme for the cyclic polynomial X-channel satisfying the separability conditions of the X-channel exists, if and only if the condition $\det(\mathbf{D}) \neq 0 \mod(x^n - 1)$ and n = 3 hold. Then, cyclic IA achieves the upper bound of $\frac{4}{3}$ DoF.

Proof:

(a) Necessity of $det(D) \not\equiv 0 \mod(x^n - 1), n \in \mathbb{N}$:

By assuming the contraposition, $det(D) \equiv 0 \mod(x^n - 1)$ yields (cf. (3.23)):

$$det(D) \equiv 0 \mod(x^{n} - 1)$$

$$\Rightarrow d_{11}d_{22} - d_{21}d_{12} \equiv 0 \mod(x^{n} - 1)$$

$$\Rightarrow d_{11}d_{22} \equiv d_{21}d_{12} \mod(x^{n} - 1))$$

$$\Rightarrow d_{11}d_{22} \equiv d_{k1}d_{1l} \mod(x^{n} - 1)), \qquad (3.47)$$

with the indices $j \neq k \in \mathcal{K}$ and $i \neq l \in \mathcal{K}$. Including (3.47) into condition (3.46) yields:

$$d_{ji}x^{p_{ki}} \equiv d_{jl}x^{p_{kl}} \mod(x^n - 1)$$
$$\Rightarrow d_{ki}x^{p_{ki}} \equiv d_{kl}x^{p_{kl}} \mod(x^n - 1).$$

Relabeling the indices $j \leftrightarrow k$ provides:

 $\Rightarrow d_{ji}x^{p_{ji}} \equiv d_{jl}x^{p_{jl}} \mod(x^n - 1),$

and contradicts (3.43) for any $n \in \mathbb{N}$.

(b) Necessity of n > 2 dimensions: Assume det $(D) \neq 0 \mod(x^n - 1)$ holds. We consider the right-hand sides of (3.43) and (3.44). (3.44) is expanded by d_{ii} .

$$(3.43): d_{ji}x^{p_{ji}} \neq d_{jl}x^{p_{jl}} \mod(x^n - 1), (3.44): d_{ij}x^{p_{ji}} \neq d_{ij}x^{p_{ki}} \mod(x^n - 1).$$

These right-hand side terms must also be pairwise distinct, since we can relabel the indices $i \leftrightarrow l$ in (3.48) to obtain (3.45):

$$d_{jl}x^{p_{jl}} \not\equiv d_{ji}x^{p_{ki}} \mod(x^n - 1)$$

$$\Leftrightarrow d_{ji}x^{p_{ji}} \not\equiv d_{jl}x^{p_{kl}} \mod(x^n - 1).$$
(3.48)

There is no solution to satisfy all three conditions on $d_{ji}x^{p_{ji}}$ with only n = 1 or n = 2 dimensions.

(c) Sufficiency of n = 3 and $\det(D) \not\equiv 0 \mod(x^n - 1)$: From the perfect IA condition in (3.46), the following holds:

$$x^{p_{12}} \equiv d_{22}^{-1} d_{21} x^{p_{11}} \mod(x^3 - 1), \tag{3.49}$$

$$x^{p_{21}} \equiv d_{11}^{-1} d_{12} x^{p_{22}} \mod(x^3 - 1).$$
(3.50)

Furthermore, the condition (3.44) must hold:

$$x^{p_{11}} \not\equiv x^{p_{21}} \mod(x^3 - 1), \tag{3.51}$$

$$x^{p_{12}} \not\equiv x^{p_{22}} \mod(x^3 - 1). \tag{3.52}$$

The insertion of (3.49) and (3.50) into condition (3.51) yields:

$$x^{p_{22}} \not\equiv d_{22}d_{11}d_{21}^{-1}d_{12}^{-1}x^{p_{12}} \mod(x^3-1), \tag{3.53}$$

Due to the determinant in (3.47), the following holds:

$$d_{12}^{-1}d_{21}^{-1}d_{22}d_{11} \not\equiv 1 \mod(x^3 - 1).$$

W.l.o.g., we can fix p_{11} and compute p_{12} using (3.49). We can determine a solution for p_{22} from (3.52) and (3.53) only if n > 2. For n = 3 the solution of p_{22} is unique. The remaining parameter p_{21} is derived using (3.52).

The validity of condition (3.43) is yet to check. Inserting (3.49) and (3.50) into (3.43) for all cases provides:

$$\begin{aligned} x^{p_{11}} & \neq d_{12}d_{21}d_{22}^{-1}d_{11}^{-1}x^{p_{11}} \mod(x^3-1), \\ x^{p_{21}} & \neq d_{22}d_{11}d_{12}^{-1}d_{21}^{-1}x^{p_{21}} \mod(x^3-1). \end{aligned}$$

Both conditions are satisfied by prerequisite, since (3.47) holds. Altogether, there is a solution for cyclic IA on the X-channel with n = 3 dimensions and M = 4 messages satisfying the separability conditions and achieving the upper bound of $\frac{4}{3}$ DoF. Note that this theorem also includes the example of the 2-user X-channel considered in [37, Figure 1] for $d_{11} = d_{21} = x^1$, $d_{12} = x^0$, $d_{22} = x^2$ and $p_{21} = p_{12} = 0$, $p_{11} = p_{22} = 1$.

Corollary 3.7. If $det(D) \equiv 0 \mod(x^n - 1)$ holds, only 1 DoF can be achieved on the cyclic polynomial 2-user X-channel.

Proof:

Theorem 3.6 (a) yields that interference signals can not be aligned if the condition $det(D) \equiv 0 \mod(x^n - 1)$ holds. Each message must be received distinctly within its own dedicated offset, i.e., a MA scheme demands $n \geq 4$. As in Lemma 3.3, only 1 DoF is achievable by decoding all four messages at each receiver.

Note that, if $\det(D) \equiv 0 \mod(x^n - 1)$ holds, the cyclic polynomial X-channel does not provide the *relativity of IA*. In other words, the influence of each subchannel to

the receivers does not provide a sufficiently diverse view of the transmitted signals, so that the observed signals from both transmitters behave identically and the benefits of cyclic IA are precluded. An according exceptional case is also observed in the symmetric 2-user X-channel in terms of the LDCM in [21, Theorems 3.1 & 3.2] at full symmetry.

Before we devise a more generalized cyclic IA scheme for the X-channel with an arbitrary number of submessages, we first elaborate an inherent symmetry property of IA. The observed property is a main tool for the derivation of the capacityachieving scheme.

3.3.2 Complementary Reciprocal Symmetry of Cyclic IA

Interestingly, aligning two messages from different transmitters at one receiver provides an inherent symmetry property in the given X-channel, if the necessary determinant condition $\det(D) \not\equiv 0 \mod (x^n-1)$ holds. To elaborate this, we consider two parameters $p_1, p_2 \in \mathbb{N}$ for messages \tilde{W}_1, \tilde{W}_2 from Tx_1 and Tx_2 , respectively. Let p_1 and p_2 be chosen such that these messages are aligned at Rx_2 at a fixed offset $\lambda_2 \in \mathbb{N}$:

$$d_{21}x^{p_1} \equiv d_{22}x^{p_2} \equiv x^{\lambda_2} \mod (x^n - 1). \tag{3.54}$$

This alignment results in the following received signals (denoted with tilde):

$$\tilde{r}_1(x) \equiv x^{\lambda_2} (d_{11} d_{21}^{-1} \tilde{W}_1 + d_{12} d_{22}^{-1} \tilde{W}_2) \mod (x^n - 1), \tag{3.55}$$

$$\tilde{r}_2(x) \equiv x^{\lambda_2}(\tilde{W}_1 + \tilde{W}_2) \mod (x^n - 1).$$
 (3.56)

Conversely, we may choose p_1, p_2 such that two messages \mathring{W}_1 , \mathring{W}_2 align at Rx_1 at a fixed offset $\lambda_1 \in \mathbb{N}$ instead:

$$d_{11}x^{p_1} \equiv d_{12}x^{p_2} \equiv x^{\lambda_1} \mod (x^n - 1).$$
(3.57)

The received signals of the second case yield (denoted with circle):

$$\mathring{r}_1(x) \equiv x^{\lambda_1}(\mathring{W}_1 + \mathring{W}_2) \mod (x^n - 1),$$
(3.58)

$$\mathring{r}_{2}(x) \equiv x^{\lambda_{1}} (d_{11}^{-1} d_{21} \mathring{W}_{1} + d_{12}^{-1} d_{22} \mathring{W}_{2}) \mod (x^{n} - 1).$$
(3.59)

By comparing (3.55) with (3.59) and (3.56) with (3.58), we observe that the offsets of the polynomials $\tilde{r}_i(x)$ and $\dot{r}_j(x)$ can be mutually converted by the following transformation³:

$$\mathring{r}_{j}(x) \equiv x^{\lambda_{i}+\lambda_{j}} \, \widetilde{r}_{i}(x^{-1}) \Big|_{\widetilde{W}_{1} \to \mathring{W}_{1}, \widetilde{W}_{2} \to \mathring{W}_{2}} \, \mod(x^{n}-1), \tag{3.60}$$

$$\tilde{r}_{j}(x) \equiv x^{\lambda_{i}+\lambda_{j}} \dot{r}_{i}(x^{-1}) \big|_{\dot{W}_{1} \to \tilde{W}_{1}, \dot{W}_{2} \to \tilde{W}_{2}} \mod (x^{n}-1),$$
(3.61)

for $i \neq j \in \mathcal{K}$. The coefficients with the messages are substituted correspondingly. We call these two alignments *complementary reciprocal symmetric*⁴, since the dedicated

³For the *reciprocal polynomial* of p(x), the sequence of coefficients is reversed, i.e., we obtain $p(x^{-1}) = \sum_{k=0}^{n-1} p^{[k]} x^{-k}$.

⁴This property should not be confused with the related concept of the *reciprocity of alignment* in [40], where IA on the reciprocal channel with swapped transmitter-receiver pairs is considered instead.

signals and the aligned interference signals are swapped at the receivers and the offsets are reciprocally inverted along λ_i and shifted to λ_j (or vice versa). In other words, the complementary reciprocal symmetry basically states that aligning signals from both transmitters at Rx_1 provides a particular interference pattern at Rx_2 , while vice versa, aligning two signals at Rx_2 provides the reciprocal interference pattern of the one observed before at Rx_1 .

3.3.3 X- Channel with Generalized Message Lengths

Now, we generalize the perfect IA scheme of Section 3.3.1 such that the messages \boldsymbol{w}_{ji} may have different lengths $m_{ji} \geq 1$. In contrast to the simplified case treated above, a scheme purely based on IA is not sufficient to achieve the upper bound. Similar to the 2IFC case, we also need to employ a LEaD scheme. Furthermore, we apply the complementary reciprocal symmetry property elaborated in the previous section.

The generalized messaging matrix of the 2-user X- channel is:

$$\boldsymbol{M} = \begin{pmatrix} m_{11} & m_{12} \\ m_{21} & m_{22} \end{pmatrix}.$$
 (3.62)

The total number of transmitted submessages is:

$$M = m_{11} + m_{21} + m_{12} + m_{22}. \tag{3.63}$$

In analogy to the MAC, BC and 2IFC, we use the vectors:

$$\boldsymbol{w}_{ji} = (W_{ji}^{[0]}, ..., W_{ji}^{[m_{ji}-1]}), \qquad (3.64)$$

$$\boldsymbol{p}_{ji} = (p_{ji}^{[0]}, \dots, p_{ji}^{[m_{ji}-1]}).$$
(3.65)

The transmitted polynomial from Tx_i is:

$$v_i(x) = \sum_{j=1}^2 \sum_{k=1}^{m_{ji}} W_{ji}^{[k]} x^{p_{ji}^{[k]}}.$$
(3.66)

The received polynomials derived from (2.18) yield:

$$r_1(x) \equiv d_{11}v_1(x) + d_{12}v_2(x) \mod (x^n - 1),$$

$$r_2(x) \equiv d_{21}v_1(x) + d_{21}v_2(x) \mod (x^n - 1).$$

The separability conditions defined in Section 3.3.1 are now generalized for generalized message lengths, i.e., each condition also applies to any pair of submessages as follows:

$$\begin{aligned} Multiple-access interference conditions with\\ i, j, l \in \mathcal{K}, \ i \neq l, \ t \in \{0, ..., m_{ji} - 1\}, \ t' \in \{0, ..., m_{jl} - 1\}:\\ d_{ji}x^{p_{ji}^{[t]}} \not\equiv d_{jl}x^{p_{jl}^{[t']}} \mod (x^n - 1). \end{aligned}$$
(3.67)

Intra-user interference conditions with

$$i, j, k \in \mathcal{K}, \ j \neq k, \ t \in \{0, ..., m_{ji} - 1\}, \ t' \in \{0, ..., m_{ki} - 1\}:$$

 $x^{p_{ji}^{[t]}} \not\equiv x^{p_{ki}^{[t']}} \mod (x^n - 1).$ (3.68)

Inter-user interference conditions with $i, j, k \in \mathcal{K}, i \neq l, k \neq j, t \in \{0, ..., m_{ji} - 1\}, t' \in \{0, ..., m_{ki} - 1\}:$ $d_{ji}x^{p_{ji}^{[t]}} \neq d_{jl}x^{p_{ki}^{[t']}} \mod (x^n - 1).$ (3.69)

3.3.4 Achievability of Perfect Cyclic IA with Linear Coding

Despite the vast amount of separability conditions for the X-channel as given above, a communication scheme that includes cyclic IA, LEaD and exploits the property of complementary reciprocal symmetry of IA, can achieve the corresponding upper bounds and the maximal sum-rate of up to $\frac{4}{3}$ DoF for equal message lengths. This is shown in the following theorem.

Theorem 3.8. A combined cyclic IA and LEaD scheme achieves the upper bound:

 $\mathrm{DoF} \leq \frac{m_{11} + m_{21} + m_{12} + m_{22}}{\max_{i \neq j \in \mathcal{K}} (m_{ii} + m_{ij} + \max(m_{ji}, m_{jj}))},$

on the 2-user X-channel with arbitrary message lengths, if the total number of dimensions is $n = \max_{i \neq j \in \mathcal{K}} (m_{ii} + m_{ij} + \max(m_{ji}, m_{jj}))$, and if $\det(D) \not\equiv 0 \mod(x^n - 1)$ holds.

Proof:

(a) Necessity of $n \ge \max_{i \ne j \in \mathcal{K}} (m_{ii} + m_{ij} + \max(m_{ji}, m_{jj}))$:

This upper bound on n is a special case of the upper bound already shown in (3.38) for the given M in (3.62).

(b) Necessity of $\det(\mathbf{D}) \neq 0 \mod (x^n - 1)$: For $\det(\mathbf{D}) \equiv 0 \mod(x^n - 1)$, cyclic IA contradicts to the separability conditions as already proven in Theorem 3.6 (a).

(c) Sufficiency of cyclic IA for $det(D) \not\equiv 0 \mod (x^n - 1)$ and the given lower bound on n:

Our cyclic IA scheme with LEaD is outlined as follows:

- (i) Align the primary interference space at Rx_2 : Interference signals are pairwise aligned at Rx_2 beginning at offset λ_2 to form a primary interference space confined to a continuous frame⁵.
- (ii) Linear encoding and decoding extension at Rx₂: The LEaD extension ensures that dedicated signals are decodable at Rx₁. Fur-
 - The LEaD extension ensures that dedicated signals are decodable at Rx_1 . Furthermore, this extension provides a frame of sufficiently many unused dimensions at Rx_1 such that the next step (iii) can be applied.
- (iii) Complementary reciprocal symmetric alignment at Rx₁: Using the complementary reciprocal symmetry of cyclic IA, the remaining signals are reciprocally aligned at Rx₁ such that the already aligned dedicated signals still remain decodable.

⁵We call a continuous sequence of offsets a *frame*.

First, let $m_{11} = m_{12} = m_{21} = m_{22} = k$, with $k \in \mathbb{N}$, so that lower bound on n yields n = 3k with a maximal number of dedicated submessages k. Clearly, it suffices to prove three cases: n = 3k, n = 3k + 1, and n = 3k + 2.

Step (i) - Alignment of the Primary Interference Space:

In this initial step, k parameters $p_{11}^{[0]}, ..., p_{11}^{[k-1]}$ and k parameters $p_{12}^{[0]}, ..., p_{12}^{[k-1]}$ for the messages \boldsymbol{w}_{11} and \boldsymbol{w}_{12} are fixed such that the interfering submessages $W_{11}^{[l]}$ and $W_{12}^{[l]}$ are pairwise aligned at Rx_2 for l = 0, ..., k - 1 in a frame of k distinct dimensions. We choose the offset of the first aligned submessages $W_{11}^{[0]}, W_{12}^{[0]}$ at Rx_2 as $\lambda_2 \in \{0, ..., n-1\}$, so that:

$$d_{21}x^{p_{11}^{[l]}} \equiv d_{22}x^{p_{12}^{[l]}} \equiv x^{\lambda_2 + l} \mod (x^n - 1).$$
(3.70)

So far, the transmitted signals from Tx_i , for $i \in \mathcal{K}$, are:

$$\tilde{v}_i(x) = \sum_{l=0}^{k-1} W_{1i}^{[l]} x^{p_{1i}^{[l]}}.$$
(3.71)

The following signals are received at Rx_1 and Rx_2 :

$$\tilde{r}_{1}(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_{2}+l} (d_{11}^{-1}d_{21}W_{11}^{[l]} + d_{12}^{-1}d_{22}W_{12}^{[l]}) \mod (x^{n} - 1),$$
(3.72)

$$\tilde{r}_2(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_2+l} (W_{11}^{[l]} + W_{12}^{[l]}) \mod (x^n - 1)$$
(3.73)

This allocation ensures that the aligned interference at Rx_2 is confined to a frame of k dimensions at offsets $x^{\lambda_2}, ..., x^{\lambda_2+k-1}$. We call this frame of aligned interference the primary interference space $\mathcal{I}_2 = \mathcal{I}_{21} \cup \mathcal{I}_{22}$ at Rx_2 .

Let a frame of $\tau \in \mathbb{N}$ dedicated submessages transmitted from Tx_i and received at Rx_j be denoted by the set:

$$\mathcal{S}_{ji}^{[\tau]} = \{ d_{ji} W_{ji}^{[0]} x^{p_{ji}^{[0]}}, ..., d_{ji} W_{ji}^{[\tau-1]} x^{p_{ji}^{[\tau-1]}} \}.$$
(3.74)

Currently, two different frames, each of k dedicated submessages, occupy the signal space at Rx_1 as illustrated by the white areas of Figure 3.7. Let the parameters τ_{11} and τ_{12} denote the offset of the first submessage for the frames of S_{11} and S_{12} as received at Rx_1 , respectively:

$$\tau_{11} \equiv \lambda_2 - \delta_{21} + \delta_{11} \pmod{n},\tag{3.75}$$

$$\tau_{12} \equiv \lambda_2 - \delta_{22} + \delta_{12} \pmod{n}. \tag{3.76}$$

The offset which is next to the end of such a k-dimensional frame is at:

$$\tau_1 = \tau_{11} + k, \tag{3.77}$$

$$\tau_2 = \tau_{12} + k. \tag{3.78}$$

We deliberately avoid the (mod *n*)-operation in (3.77) and (3.78) to maintain a linear ordering relationship over a cyclically unrolled period. In order to determine the size of the yet unused offsets at Rx_1 (grey shading), we need to compute the distance Δ_1

between τ_1 and τ_{12} , and also the distance Δ_2 between τ_2 and τ_{11} (cf. clock-wise distance metric in [60]):

$$\Delta_1 = \begin{cases} \tau_{12} - \tau_1, & \text{if } \tau_{11} < \tau_{12}, \\ n + \tau_{12} - \tau_1, & \text{if } \tau_{11} > \tau_{12}, \end{cases}$$
(3.79)

$$\Delta_2 = \begin{cases} n + \tau_{11} - \tau_2, & \text{if } \tau_{11} < \tau_{12}, \\ \tau_{11} - \tau_2, & \text{if } \tau_{11} > \tau_{12}. \end{cases}$$
(3.80)

If Δ_1 or Δ_2 yields a negative value, then the frames S_{11} and S_{12} overlap. An example of the parameters τ_{11} , τ_1 , τ_{12} , τ_2 , and distances Δ_1 , Δ_2 is illustrated in Figure 3.7.

Case (a): If there is no intersection at Rx_1 , i. e., if $S_{11} \cap S_{12} = \emptyset$, then we will obtain two unused frames of non-negative length $0 \leq \Delta_1 < k$ and $0 \leq \Delta_2 < k$, respectively. We define the following auxiliary parameters for the maximal and the minimal distance of Δ_1 and Δ_2 :

$$\underline{\Delta}_3 = \min(\Delta_1, \Delta_2), \tag{3.81}$$

$$\overline{\Delta}_3 = \max(\Delta_1, \Delta_2). \tag{3.82}$$

In this non-intersecting case, $|S_{11} \cup S_{12}| = 2k$ dimensions are occupied by two disjoint frames of k dimensions each. A number of n-2k = k dimensions at Rx_1 remains unused yet. The unused offsets cover exactly $\underline{\Delta}_3 + \overline{\Delta}_3 = \Delta_1 + \Delta_2 = n - 2k = k$ dimensions in this case (cf. Figure 3.7(a)). Both parameters $\underline{\Delta}_3$ and $\overline{\Delta}_3$ are non-negative integers here.

Case (b): Otherwise, the dedicated signals must intersect at Rx_1 , i. e., $S_{11} \cap S_{12} \neq \emptyset$. We obtain that either Δ_1 or Δ_2 is negative. $\underline{\Delta}_3$ also becomes negative. The absolute value $|\underline{\Delta}_3|$ indicates the number of overlapping dimensions. $2k - |\underline{\Delta}_3|$ dimensions are occupied by (partially overlapping) dedicated signals. A number of $n - \overline{\Delta}_3$ dimensions at Rx_1 remains unused yet. The fully overlapping case $S_{11} = S_{12}$ is already excluded by $\det(D) \not\equiv 0 \mod (x^n - 1)$.

Step (ii) - Linear Encoding and Decoding (LEaD) Extension:

We propose a LEaD scheme in order to provide the following three properties for the received signals at Rx_1 and Rx_2 :

- I All dedicated signals at Rx₁ are linearly decodable.
- II A frame of at least k dimensions is available for the yet unallocated primary interference space \mathcal{I}_1 at Rx₁.
- III A frame of 2k unused dimensions is available at Rx_2 . This frame is retained for the remaining dedicated messages w_{21} and w_{22} .

We repeat the first $|\underline{\Delta}_3|$ submessages $W_{11}^{[l]}, W_{12}^{[l]}$, for $l = 0, ..., |\underline{\Delta}_3| - 1$ at Rx₂, by aligning them next to \mathcal{I}_2 :

$$d_{21}x^{q_{11}^{[l]}} \equiv d_{22}x^{q_{12}^{[l]}} \equiv x^{\lambda_2 + k + l} \mod(x^n - 1), \tag{3.83}$$



(b) Overlapping frames in n dimensions: $S_{11} \cap S_{12} \neq \emptyset$.

Figure 3.7: The illustration shows an example for the case of non-overlapping frames $S_{11} \cap S_{12} = \emptyset$ at the top, and for the case of overlapping frames $S_{11} \cap S_{12} \neq \emptyset$ at the bottom, as allocated by step (i) at Rx₁. Yet unused dimensions are shown in grey. For step (ii), the linear encoding and decoding extension is depicted for $\tau_{11} < \tau_{12}$, and $0 < \Delta_1 < \Delta_2$. The cancellation of redundant or overlapping signals is indicated by dashed arrows. Overlapping offsets are dotted in grey.

with the additional offset parameters $q_{1i}^{[l]}$ for the repeated submessages $W_{1i}^{[l]}$. The extended transmission signals from Tx_i , for $i \in \mathcal{K}$, become:

$$\tilde{v}_{i}(x) = \sum_{l=0}^{k-1} W_{1i}^{[l]} x^{p_{1i}^{[l]}} + \sum_{l'=0}^{|\Delta_{3}|-1} W_{1i}^{[l']} x^{q_{1i}^{[l']}},$$
(3.84)

and the corresponding received signals are:

$$\tilde{r}_{1}(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_{2}+l} (d_{11}^{-1} d_{21} W_{11}^{[l]} + d_{12}^{-1} d_{22} W_{12}^{[l]}) +$$

$$(3.85)$$

$$\sum_{l'=0}^{|\underline{\Delta}_3|-1} x^{\lambda_2+k+l'} (d_{11}^{-1} d_{21} W_{11}^{[l']} + d_{12}^{-1} d_{22} W_{12}^{[l']}) \mod (x^n - 1),$$
(3.86)

$$\tilde{r}_{2}(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_{2}+l} (W_{11}^{[l]} + W_{12}^{[l]}) + \sum_{l'=0}^{|\Delta_{3}|-1} x^{\lambda_{2}+k+l'} (W_{11}^{[l']} + W_{12}^{[l']}) \mod (x^{n}-1).$$
(3.87)

This extended alignment scheme produces a secondary interference space \mathcal{I}'_2 at Rx₂ with $|\underline{\Delta}_3|$ submessages. The secondary interference space \mathcal{I}'_2 at Rx₂ is cancelled by using the first 0, ..., $|\underline{\Delta}_3| - 1$ dimensions of the primary interference space \mathcal{I}_2 . This cancellation procedure is depicted by the dashed arrows in the lower graphs of Figure 3.7. Thus,

the effective interference space left over at Rx_2 after cancellation is again \mathcal{I}_2 , with only k dimensions.

In the case (a) with $S_{11} \cap S_{12} = \emptyset$ at Rx_1 , the smaller gap of $|\underline{\Delta}_3|$ unused dimensions is filled up with the corresponding repeated dedicated signals $S_{ij}^{[|\underline{\Delta}_3|]}$. This repetition in the smaller gap is used to cancel out the corresponding dedicated signals as depicted in Figure 3.7(a). In the larger gap with $|\overline{\Delta}_3|$ dimensions, the additional repetition of signals are cancelled by using the corresponding dedicated signals.

In the contrary case (b), we have $S_{11} \cap S_{12} \neq \emptyset$ at Rx₁. Now, a number of $|\underline{\Delta}_3|$ dimension overlaps at Rx₁. Likewise, we extend the signal as in (3.83) with the first $|\underline{\Delta}_3|$ submessages. At Rx₁, the repeated signals $S_{1j}^{[|\underline{\Delta}_3|]}$ occur at τ_j , correspondingly. The overlapping parts are cancelled using the corresponding interference-free signals. This extension and cancellation procedure for this second case is depicted in Figure 3.7(b).

In both cases, the dedicated signals and the repeated signals form a continuous frame of 2k dimensions at Rx_1 after performing the cancellation scheme as described above. The dedicated signals allocated so far are decodable. Moreover, a frame of k dimensions is yet left unused after cancellation at Rx_1 and a frame of 2k dimensions is left unused at Rx_2 . Hence, the LEaD-extension provides all three desired properties I, II, and III.

Step (iii) - Complementary Reciprocal Symmetric Alignment:

In this step, we derive the complementary alignment of interfering signals at Rx₁. We align the interference at Rx₁, caused by $W_{2l}^{[l]}$ and $W_{22}^{[l]}$, at an offset $\lambda_1 \in \{0, ..., n-1\}$ by:

$$d_{11}x^{p_{21}^{[l]}} \equiv d_{12}x^{p_{22}^{[l]}} \equiv x^{\lambda_1 - l} \mod(x^n - 1), \tag{3.88}$$

for l = 0, ..., k - 1. This primary interference space \mathcal{I}_1 is aligned in the reciprocal direction, when compared to the IA at Rx_2 in (3.70). The offset λ_1 is uniquely chosen by:

$$\lambda_1 = \begin{cases} \tau_{11} - 1 + [\Delta_1]^+, & \text{if } \Delta_1 \le \Delta_2, \\ \tau_{12} - 1 + [\Delta_2]^+, & \text{if } \Delta_2 < \Delta_1, \end{cases}$$
(3.89)

so that \mathcal{I}_1 is located exactly within the frame of k yet unused (or cancelled) offsets.

Analogously to step (ii), we also apply a LEaD-extension to provide decodability for the dedicated signals $W_{22}^{[l]}$ and $W_{22}^{[l]}$. Again, the first $|\underline{\Delta}_3|$ submessages $W_{21}^{[l]}, W_{22}^{[l]}$, with $l = 0, ..., |\underline{\Delta}_3| - 1$, are repeated next to the primary interference space \mathcal{I}_1 , also in the reciprocal direction, so that they are aligned at:

$$d_{11}x^{q_{21}^{[l]}} \equiv d_{12}x^{q_{22}^{[l]}} \equiv x^{\lambda_1 - k - l} \mod (x^n - 1), \tag{3.90}$$

with the offsets $q_{2i}^{[l]}$ for the repeated submessages $W_{2i}^{[l]}$, for $i \in \mathcal{K}$. The linearly extended transmission signal $\mathring{v}_i(x)$ from Tx_i is:

$$\mathring{v}_{i}(x) = \sum_{l=0}^{k-1} W_{2i}^{[l]} x^{p_{2i}^{[l]}} + \sum_{l'=0}^{|\underline{\Delta}_{3}|-1} W_{2i}^{[l']} x^{q_{2i}^{[l']}}.$$
(3.91)

	x ⁰	x^1	x^2	x^3	x^4	x^5	x^6
$v_1(x)$	$W_{11}^{[2]} + W_{21}^{[0]}$	$W_{21}^{[1]}$	$W_{21}^{[0]}$	$W_{11}^{[0]}$	$W_{11}^{[1]}$	$W_{11}^{[0]} + W_{11}^{[2]}$	0
$v_2(x)$	0	$W_{12}^{[0]}$	$W_{12}^{[1]}$	$W_{12}^{[0]}$	$W_{22}^{[0]}$	$W_{22}^{[1]}$	$W_{22}^{[0]}$
$r_1(x)$	$W_{11}^{[2]} + \underline{W_{21}^{[0]}} + W_{22}^{[0]}$	$W_{22}^{[1]} + W_{21}^{[1]}$	$W_{21}^{[0]} + W_{22}^{[0]}$	$W_{11}^{[0]}$	$W_{11}^{[1]} + W_{12}^{[0]}$	$W_{12}^{[1]} + W_{11}^{[0]} + W_{11}^{[0]}$	$W_{12}^{[0]}$
$r_2(x)$	$W_{11}^{[0]} + W_{12}^{[0]}$	$W_{11}^{[1]} + W_{12}^{[1]}$	$W_{11}^{[0]} + W_{12}^{[0]} + W_{11}^{[2]}$	$W_{22}^{[0]}$	$W_{22}^{[1]} + W_{21}^{[\theta]} + W_{11}^{[\theta]}$	$W_{21}^{[1]} + W_{22}^{[0]}$	$W_{21}^{[0]}$

Figure 3.8: An example of cyclic IA with the LEaD-extension for the 2-user X- channel with $n = 7 = 3 \cdot 2 + 1$ dimensions ($\Rightarrow k = 2$). Let the channel matrix **D** impose the following cyclic 'right'-shifts $d_{11} = x^0$, $d_{12} = x^3$, $d_{21} = x^4$, $d_{22} = x^6$ and let $m_{11} = 3$, $m_{12} = m_{21} = m_{22} = 2$. The parameters from the scheme are $\lambda_2 = 0$, $\lambda_1 = 2$, $\tau_{11} = 3$, $\tau_{12} = 4$, $\tau_1 = 5$, $\tau_2 = 6$, $\Delta_1 = -1$, and $\Delta_2 = 4$. The primary interference spaces are highlighted in red and the secondary interference spaces in green. The complementary reciprocal symmetry is satisfied, when ignoring $W_{11}^{[2]}$, as highlighted in blue. The cancellation of interference by the linear decoder is indicated by the terms crossed out. Altogether, all M = 9 dedicated submessages are decodable within 7 dimensions achieving the proposed upper bound.

This provides the following received signals at Rx_1 and Rx_2 :

$$\mathring{r}_{1}(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_{1}-l} (W_{21}^{[l]} + W_{22}^{[l]}) + \sum_{l'=0}^{|\Delta_{3}|-1} x^{\lambda_{1}-k-l'} (W_{21}^{[l']} + W_{22}^{[l']}) \mod (x^{n} - 1),$$
(3.92)
$$\mathring{r}_{2}(x) \equiv \sum_{l=0}^{k-1} x^{\lambda_{1}-l} (d_{11}d_{21}^{-1}W_{21}^{[l]} + d_{12}d_{22}^{-1}W_{21}^{[l]}) +$$
(3.93)

$$\mathring{r}_{2}(x) \equiv \sum_{l=0}^{\kappa-1} x^{\lambda_{1}-l} (d_{11}d_{21}^{-1}W_{21}^{ll} + d_{12}d_{22}^{-1}W_{21}^{ll}) +$$
(3.93)

$$\sum_{\substack{l'=0\\l'=0}}^{\lfloor \underline{\Delta}_3 \rfloor - 1} x^{\lambda_1 - k - l'} (d_{11} d_{21}^{-1} W_{21}^{\lfloor l' \rfloor} + d_{12} d_{22}^{-1} W_{22}^{\lfloor l' \rfloor}) \mod (x^n - 1).$$
(3.94)

Now, we combine the transmitted signals (3.84) and (3.91) of Tx_i :

$$v_i(x) = \tilde{v}_i(x) + \dot{v}_i(x). \tag{3.95}$$

At each Rx_i , the combined received signals yield from $\tilde{r}_i(x)$ and $\dot{r}_i(x)$, respectively:

$$r_j(x) \equiv \tilde{r}_j(x) + \mathring{r}_j(x) \mod (x^n - 1).$$
 (3.96)

For $r_1(x)$ all dedicated signals are linearly decoded as shown in step (ii). Recall that the primary interference space \mathcal{I}_1 fits perfectly into the formerly unused (or cancelled) signal space at Rx₁. The secondary interference space \mathcal{I}'_1 is cancelled using \mathcal{I}_1 . The proposed alignment scheme satisfies the following complementary reciprocal symmetry relationship for $r_2(x)$ (cf. (3.60), (3.61)) by construction:

$$\mathring{r}_{2}(x) \equiv x^{\lambda_{1}+\lambda_{2}} \, \widetilde{r}_{1}(x^{-1}) \big|_{w_{11} \to w_{21}, \, w_{12} \to w_{22}} \mod (x^{n} - 1), \tag{3.97}$$

$$\tilde{r}_{2}(x) \equiv x^{\lambda_{1}+\lambda_{2}} \dot{r}_{1}(x^{-1}) \big|_{w_{21} \to w_{11}, w_{22} \to w_{12}} \mod (x^{n}-1).$$
(3.98)

The resulting $r_2(x) = \tilde{r}_2(x) + \mathring{r}_2(x)$ is complementary reciprocal symmetric w.r.t. $r_1(x)$. Due to the corresponding relabelling of submessages, all dedicated signals for Rx_2 are decodable now, as well. At this point, the case n = 3k, for $k \in \mathbb{N}$, is already proven.

For the case that the minimal necessary n is lower bounded by n = 3k + 1, we may consider $m_{11} = k + 1$, $m_{12} = m_{21} = m_{22} = k$, w.l.o.g. As in the previous scheme, k pairs of submessages are aligned in the same way as in steps (i) to (iii) at both receivers. But now, the lower bound on n is limited by 3k + 1 instead of only 3k dimensions. Thus, there is still one unused offset left over at both receivers after the cancellation of known interference. Tx₁ allocates its remaining k + 1-th submessage $W_{11}^{[k]}$ in $v_1(x)$ at those two offsets, such that one $W_{11}^{[k]}$ is received within the unused dimensions of the receivers Rx₁ and Rx₂ each. Both receivers will receive one interference-free $W_{11}^{[k]}$ and one interfering $W_{11}^{[k]}$ in general. Clearly, linear decoders can each cancel the interference at both receivers. Case n = 3k + 2 is treated analogously and hence it is omitted here.

An example of the proposed scheme for case n = 3k + 1 is provided in the table of Figure 3.8. As a result, all $m_{11} + m_{21} + m_{12} + m_{22}$ submessages are successfully conveyed within a total number of $n = \max_{i \neq j \in \mathcal{K}} (m_{ii} + m_{ij} + \max(m_{ji}, m_{jj}))$ dimensions for all cases.

The optimal choice for the number of submessages is $m_{ji} = k$ for a non-zero constant $k \in \mathbb{N}$, so that interference is perfectly aligned and the DoF are maximized by $\frac{4}{3}$.

3.4 Cyclic IA on the Cyclic Polynomial *K*-User Interference Channel

In the following section, a cyclic polynomial K-user interference channel is considered for $K \ge 3$ user-pairs. The K-user interference channel is of notable interest, since IA is capable to provide a number of $\frac{K}{2}$ DoF [3]. In contrast to the conventional approaches such a channel is not user-limited, since the DoF linearly scale with the number of user-pairs K.

3.4.1 Perfect Cyclic Interference Alignment

Similar to the initial exposition of the 2-user X-channel in Section 3.3, we use a symmetric messaging matrix $\mathbf{M} = \mathbf{I}_{K \times K}$. There is a number of M = K independent messages $W_i = \mathbf{w}_{ii}$ dedicated to be conveyed from transmitter Tx_i to receiver Rx_i in disjoint pairs with indices $i \in \mathcal{K}$ for $\mathcal{K}_{\operatorname{Tx}} = \mathcal{K}_{\operatorname{Rx}} = \mathcal{K} = \{1, ..., K\}$. Note that in [3, Appendix I], the closely related example for IA by propagation delay yields a special case of this problem. We will not cover the general case with arbitrary message lengths in this dissertation. The K-user interference channel is depicted in Figure 3.9.

The channel matrix between is defined as $\mathbf{D} = (d_{ji})_{1 \le i,j \le K}$ with $d_{ji} \in \mathcal{D}$. This channel is also called *fully-connected* as each transmitter has a non-zero subchannel to each receiver. The polynomial $v_i(x)$ only contains the single (sub-)message W_i intended for Rx_i . The offset parameter $p_i \in \mathbb{N}_0$, for $i \in \mathcal{K}$, allocates the message W_i within $v_i(x)$:

$$v_i(x) = W_i x^{p_i}.\tag{3.99}$$

The input vector of the transmitted polynomials is denoted by $\boldsymbol{v} = (v_1(x), ..., v_K(x))$. The transfer function of channel yields the received vector $\boldsymbol{r} = (r_1(x), ..., r_K(x))$:

$$\boldsymbol{r}^{\mathsf{T}} = \boldsymbol{D}\boldsymbol{v}^{\mathsf{T}} \mod(x^n - 1).$$



Figure 3.9: The fully-connected cyclic polynomial interference channel with K userpairs and M = K submessages $W_1, ..., W_K$, so that there is one message between each pair of transmitters $\operatorname{Tx}_1, ..., \operatorname{Tx}_K$ and receivers $\operatorname{Rx}_1, ..., \operatorname{Rx}_K$ for the transfer matrix D.

And the received polynomial at Rx_j yields:

$$r_j(x) = \sum_{i=1}^{K} d_{ji} W_i x^{p_i} \mod(x^n - 1).$$

The task of cyclic IA for the given K-user interference channel is to convey and decode the M = K dedicated messages $W_1, ..., W_K$ interference-free within n dimensions. The scheme is optimal in the sense of cyclic IA if the number of dimensions n is minimal and still feasible.

In contrast to the 2 × 2 cyclic polynomial X-channel, neither the *multiple-access* interference nor the intra-user interference conditions are to be considered for this K-user interference channel, since there is only one message W_i per dedicated userpair i anyway. Thus, only the following inter-user interference conditions are to be considered for $j \neq i \in \mathcal{K}$:

$$d_{jj}x^{p_j} \not\equiv d_{ji}x^{p_i} \mod(x^n - 1).$$
 (3.100)

We count K(K-1) of these separability conditions in total for K users, i.e., a single receiver observes K-1 inter-user interference signals from the undesired transmitters. In the given case of K user-pairs, *perfect IA* means to align all K-1 interfering signals received at each Rx_i into a single dimension:

$$d_{ji}x^{p_i} \equiv d_{jk}x^{p_k} \mod(x^n - 1), \tag{3.101}$$

with pairwise distinct $i, j, k \in \mathcal{K}$.

For notational convenience, we define auxiliary 2×2 submatrices of D denoted as $D_{j,k,i}$, if the following structure is satisfied for pairwise distinct indices $i, j, k \in \mathcal{K}$:

$$\boldsymbol{D}_{j,k,i} = \begin{pmatrix} d_{jj} & d_{ji} \\ d_{kj} & d_{ki} \end{pmatrix}.$$
(3.102)

Theorem 3.9. A perfect cyclic IA scheme for the K-user interference channel with messaging matrix $\mathbf{M} = \mathbf{I}_{K \times K}$ exists, if the three conditions:

 $det(\mathbf{D}_{j,k,i}) \neq 0 \mod(x^n - 1),$ $d_{ji}d_{kj}d_{ik} \equiv d_{ij}d_{jk}d_{ki} \mod(x^n - 1),$ and n = 2 dimensions,

hold for pairwise distinct $i, j, k \in \mathcal{K}$. Then, cyclic IA achieves $\frac{K}{2}$ DoF.

Proof:

(a) Necessity of $d_{ji}d_{kj}d_{ik} \equiv d_{ij}d_{jk}d_{ki} \mod(x^n-1), n \in \mathbb{N}$: Let $i, j, k \in \mathcal{K}$ be pairwise distinct. By relabelling the indices in (3.101) to $i \to j, j \to k$ and $k \to i$, we obtain (3.103). And by relabelling the indices in (3.101) to $i \to k, j \to i$ and $k \to j$, we obtain (3.104), respectively:

$$(3.101): \ d_{ji}x^{p_i} \equiv d_{jk}x^{p_k} \mod(x^n - 1), d_{kj}x^{p_j} \equiv d_{ki}x^{p_i} \mod(x^n - 1),$$
(3.103)

 $d_{ik}x^{p_k} \equiv d_{ij}x^{p_j} \mod(x^n - 1).$ (3.104)

These three equivalences are solvable, if and only if the following condition holds:

$$d_{ji}d_{kj}d_{ik} \equiv d_{jk}d_{ki}d_{ij} \mod(x^n - 1). \tag{3.105}$$

Otherwise, perfect IA cannot be applied.

(b) Necessity of $\det(\mathbf{D}_{j,k,i}) \not\equiv 0 \mod(x^n - 1), n \in \mathbb{N}$: Inserting the relabelled (3.103) into the inter-user interference (3.100) condition yields:

$$\begin{aligned} d_{jj}d_{ki}d_{kj}^{-1}x^{p_i} &\neq d_{ji}x^{p_i} \mod(x^n-1) \\ &\Rightarrow d_{jj}d_{ki} \neq d_{kj}d_{ji} \mod(x^n-1) \\ &\Rightarrow 0 \neq \det(\boldsymbol{D}_{j,k,i}) \mod(x^n-1), \end{aligned}$$
(3.106)

for pairwise distinct $i, j, k \in \mathcal{K}$. If the contraposition $\det(\mathbf{D}_{j,k,i}) \equiv 0 \mod(x^n - 1)$ is assumed, the separability conditions can not be fulfilled by perfect IA.

(c) Necessity of n > 1 dimensions:

Only n = 1 dimension would preclude any separation of desired and interfering messages necessary for the inter-user interference conditions in (3.100).

(d) Sufficiency the given separability conditions with n = 2 to achieve $\frac{K}{2}$ DoF: Firstly, we consider the valid channel matrix $\mathbf{D} = x^1 \mathbf{I}_{K \times K} + (\mathbf{1}_{K \times K} - \mathbf{I}_{K \times K})x^2$ as in [3, Appendix I]. The condition (3.105) holds since any non-diagonal entry is x^2 , i.e., $d_{ji}d_{kj}d_{ki} \equiv d_{ij}d_{jk}d_{ki} \equiv x^2x^2x^2 \equiv 1 \mod(x^2-1)$. The condition (3.106) also holds since $\det(\mathbf{D}_{j,k,i}) \equiv x^3(x^1-1) \not\equiv 0 \mod(x^2-1)$.

There are many other valid channel matrices D: If all entries of a row in the given D are cyclically shifted by x^m , $m \in \mathbb{N}$, the two conditions still hold. For a shifted row j, we obtain $(d_{ji}x^m)d_{kj} \equiv d_{ki}(d_{jj}x^m) \mod(x^2-1) \Rightarrow d_{ji}d_{kj} \equiv d_{ki}d_{jj} \mod(x^2-1)$, leading to (3.106). An analogous argument holds for (3.105).

For such a valid D, we fix p_1 w.l.o.g. and determine all other p_i , with i = 2, ..., K, by applying the perfect IA condition in (3.101):

$$x^{p_i} \equiv d_{j_i}^{-1} d_{j_1} x^{p_1} \mod(x^2 - 1).$$

A solution that perfectly aligns all K-1 interference signals at each receiver exists for the fixed offset parameters, since (3.105) holds by prerequisite on the given D for n = 2. The (aligned) interference signals do not overlap with the dedicated signals at all receivers, since the same perfect IA used for the derivation of the required (3.106) is also used here.

Altogether, K messages are conveyed interference-free in n = 2 dimensions and $\frac{K_2}{2}$ DoF are achieved. This corresponds to the upper bound (3.38) for messaging matrix $M = I_{K \times K}$.

The conditions on the K-user interference channel can also be expressed alternatively. We define yet another kind of auxiliary 2×2 -matrices of D with the following structure:

$$\boldsymbol{D}_{i,j} = \begin{pmatrix} d_{ii} & d_{ij} \\ d_{ji} & d_{jj} \end{pmatrix},$$
(3.107)

for $i \neq j \in \mathcal{K}$. In contrast to the submatrices in (3.102), there are two diagonal elements of D in $D_{i,j}$ instead of only one diagonal element in $D_{j,k,i}$.

Corollary 3.10. For perfect cyclic IA with n = 2, the following alternative conditions:

$$\det(\boldsymbol{D}_{i,j}) \equiv 0 \mod(x^2 - 1), \tag{3.108}$$

must hold for $i \neq j \in \mathcal{K}$.

Proof:

Since n = 2, a perfect IA scheme can be alternatively formulated by allocating dedicated and (aligned) interference signals side-by-side with:

$$d_{ii}x^{p_i} \equiv d_{ij}x^{p_j}x^1 \mod(x^2 - 1), \tag{3.109}$$

$$d_{ii}x^{p_i} \equiv d_{ik}x^{p_k}x^1 \mod(x^2 - 1), \tag{3.110}$$

for pairwise distinct $i, j, k \in \mathcal{K}$. On the one hand, we may substitute $d_{ii}x^{p_i}$ in both congruences and obtain (3.101) to derive condition (3.105) as above. On the other hand, we may consider a relabelled version of (3.109) with swapped indices instead:

$$(3.109): \ d_{ii}x^{p_i} \equiv d_{ij}x^{p_j}x^1 \mod(x^2 - 1), d_{jj}x^{p_j} \equiv d_{ji}x^{p_i}x^1 \mod(x^2 - 1).$$
(3.111)

Substituting these two congruences leads to the condition:

$$d_{ii}d_{jj} \equiv d_{ij}d_{ji} \mod(x^2 - 1) \tag{3.112}$$

$$\Leftrightarrow \det(\mathbf{D}_{i,j}) \equiv 0 \mod(x^2 - 1), \tag{3.113}$$

for $i \neq j \in \mathcal{K}$.

3.4.2 Common Eigenvectors in Separate Subspaces

It is shown in [3, Sec. IV-D], that perfect IA⁶ is infeasible for the constant MIMO K-user interference channel with single antennas per user. The infeasibility of perfect IA is due to common eigenvectors in the dedicated and the interference subspaces. Interestingly, an analogous problem is recognizable in the CPCM, but it merely imposes a constraint on the D. To briefly elaborate this, we consider a perfect cyclic IA scheme for the 3-user interference channel mimicking the MIMO IA scheme in [3, Section IV-D: Equations (25)-(27)]:

$$d_{13}x^{p_3} \equiv d_{12}x^{p_2} \mod(x^2 - 1), \tag{3.114}$$

$$d_{13}x^{p_3} \equiv d_{31}x^{p_1} \mod(x^2 - 1), \tag{3.115}$$

$$d_{32}x^{p_2} \equiv d_{31}x^{p_1} \mod(x^2 - 1). \tag{3.116}$$

By resolving these equations w.r.t. x^{p_1} , we obtain:

$$x^{p_1} \equiv d_{13}^{-1} d_{23} d_{21}^{-1} d_{12} d_{32}^{-1} d_{31} x^{p_1} \mod(x^2 - 1).$$
(3.117)

An analogous formulation arises in [3, Section IV-D], but with corresponding diagonal MIMO channel matrices and beam-forming vectors. This corresponding equation leads to the problem of common eigenvectors when applying perfect MIMO IA. In the cyclic IA scheme, (3.117) only implies a constraint like (3.105).

3.4.3 On the Feasibility Conditions of the *K*-User Interference Channel

To elaborate further common properties of the CPCM and the GMCM, we consider the feasibility conditions of the K-user MIMO interference channel with M_{Tx} transmit antennas and M_{Rx} receive antennas (cf. [39]). In the GMCM, let the channel matrices \boldsymbol{H}_{ji} for each link from Tx_i to Rx_j have $M_{\text{Rx}} \times M_{\text{Tx}}$ dimensions. The transmit beamforming matrices \boldsymbol{V}_i have $M_{\text{Tx}} \times d_i$ dimensions and the receive filtering matrices \boldsymbol{N}_i have $M_{\text{Rx}} \times d_i$ dimensions. We have $\text{rank}(\boldsymbol{V}_i) = d_i = \text{rank}(\boldsymbol{N}_i) = d_i \leq \min(M_{\text{Tx}}, M_{\text{Rx}})$ for both matrices. The feasibility conditions for this case are compactly formulated by:

$$\boldsymbol{N}_{i}\boldsymbol{H}_{ji}\boldsymbol{V}_{i} = \boldsymbol{0}_{d_{j}\times d_{i}}, \text{ for all } i \neq j \in \mathcal{K},$$

$$(3.118)$$

$$\operatorname{rank}(\boldsymbol{N}_{i}\boldsymbol{H}_{ii}\boldsymbol{V}_{i}) = d_{i}, \text{ for all } i \in \mathcal{K},$$
(3.119)

for some properly chosen beam-forming and zero-forcing matrices, as in [39], [40]. The first condition (3.118) ensures that the interference signals from Tx_i are projected into the null space of the receiver at Rx_j for distinct i, j. The second condition (3.119) ensures that the dedicated signals from Tx_i are projected into a dedicated signal space of d_i dimensions at Rx_i . Analogous to the demanded properties in Section 2.2.3, satisfying these feasibility conditions also provides linear interference-free communication and linear decodability of all dedicated signals.

⁶However, asymptotically perfect IA is feasible using symbol-extensions for instance [3].

We now translate these feasibility conditions to an analogous expression for the CPCM. Let a zero-forcing polynomial at Rx_i be defined by:

$$z_j(x) = \sum_{k=0}^{n-1} z_j^{[k]} x^k, \ z_j^{[k]} \in \{0, 1\}.$$
(3.120)

By element-wise multiplying the zero-forcing polynomial $z_j(x)$ with the received polynomial $r_j(x)$, the resulting polynomial $z_j(x) \circ r_j(x)$ is filtered correspondingly. A filter coefficient $z_j^{[k]} = 1$ keeps the received coefficient $r_j^{[k]}$ at offset k unchanged, and a filter coefficient $z_j^{[k]} = 0$ zero-forces the received coefficient at offset k.

With such zero-forcing polynomials, the feasibility conditions for the cyclic polynomial *K*-user interference channel are analogously formalized as follows:

$$z_j(x) \circ d_{ji}v_i(x) = 0, \text{ for } i \neq j \in \mathcal{K},$$

$$(3.121)$$

$$\operatorname{rank}(z_i(x) \circ d_{ii}v_i(x)) = m_{ii}. \tag{3.122}$$

The first condition provides that interference space is zero-forced at each receiver, and the second condition provides that there are sufficiently dimensions available for the m_{ii} dedicated submessages from Tx_i to Rx_i , for all $i \in \mathcal{K}$. These conditions are fulfilled by the cyclic IA scheme proven in Theorem 3.9 using the zero-forcing polynomial:

$$z_j(x) \equiv d_{ji} x^{p_i} \equiv d_{jk} x^{p_k} \mod(x^2 - 1).$$
(3.123)

3.5 Infeasibility of Perfect Cyclic IA in Multi-User X-Networks

In this section, we consider a 3-user X-network, an extension of the 2-user X-channel. Similar to the 2-user X-network, each transmitter has one dedicated message for each receiver, i.e., three messages per transmitter. The investigation of this channel particularly addresses an interesting open problem that was stated in the discussion of [16, Section V]:

Problem: Is perfect IA by propagation delay feasible on K-user X-networks with K > 2 users?

We will show in this section that, in terms of the CPCM, perfect cyclic IA in a K-user X-network for K > 2 users is indeed overconstrained and hence infeasible. We observe that this property even goes beyond the closely related problem of common invariant subspaces of constant MIMO IA [37].

Then, in order to tackle this infeasibility problem, we extend the capabilities of the 3-user X-network by using a cooperative *interference alignment and cancellation* (IAC) scheme over rate-limited *backhaul network* (BHN). The concept of IAC is initially introduced in [61] and also applied in [62] for instance. Therein, a BHN provides a limited exchange of messages at the receiver-side to support the cancellation of known interference by the aid of cognitive messages from the other cooperating receivers.

We first analyze a simple feedforward scheme achieving $\frac{9}{5}$ DoF within 5 dimensions and only one message over an interference-free feedforward BHN. In other words, we permit a single cognitive receiver that has interference-free access to one message of the transmitters.

Our second step leads us to a related cyclic IAC scheme, but its cognitive messages are restricted to a receiver-sided backhaul network (Rx-BHN). In that case, only receivers can cooperate and exchange a limited number of messages. Cyclic IAC achieves the $\frac{9}{5}$ DoF with a minimum of only 2 messages over the Rx-BHN.

For a transmitter-sided backhaul network (Rx-BHN), we employ the concept of IN as discussed in [47] and [63] to inhibit interference at the receivers. IN is a communication scheme cancelling interference 'over the air' by aligning complementary versions of the same message within the same signalling space. Since IN is not in the main focus of this particular section, we point the reader to Sections 4.1 and 4.2 in the subsequent chapter. We observe a duality relationship between IAC on the Rx-BHN and IN on the dual Tx-BHN. This insight also generalizes the related observations in [23] for a 2IFC with cooperation.

3.5.1 Cyclic Polynomial 3-User X-Network

Note that, we already derived the DoF for the 2-user X-channel in Section 3.3, and in [27, Section III]. Here, each transmitter intends to convey K = 3 dedicated messages, one to each receiver. The set of user-indices is $\mathcal{K}_{Tx} = \mathcal{K}_{Rx} = \mathcal{K} = \{1, 2, 3\}$. For the sake of simplicity, we constrain our model to $M = \mathbf{1}_{3\times3}$, i.e., $m_{ji} = 1$ submessages for all transmitter-receiver pairs. We set our focus on perfect IA. There are $M = K^2 = 9$ unicast messages in the system in total for these given assumptions. The transmitted signal from Tx_i is a polynomial with (sub-)messages W_{ji} dedicated for receiver Rx_j :

$$v_i(x) = \sum_{j \in \mathcal{K}} W_{ji} x^{p_{ji}}.$$
(3.124)

Each receiver must decode three dedicated messages, and receives six interfering signals, i.e., they must cope with two interfering messages per transmitter Tx_i . The received signal $r_i(x)$ at Rx_i is:

$$r_j(x) \equiv \sum_{i \in \mathcal{K}} d_{ji} v_i(x) \mod (x^n - 1).$$
(3.125)

The 3-user X- network is depicted in Figure 3.10.

Separability Conditions

In order to guarantee decodability, all three types of *separability conditions* as discussed in Section 3.3.1 must be considered for the 3-user X-network. Including all nine messages, the intra-user interference conditions at Tx_i , for pairwise distinct p_{ii} , p_{ji} , p_{ki} , are:

$$x^{p_{ji}} \not\equiv x^{p_{ki}} \mod (x^n - 1),$$
 (3.126)

 $x^{p_{ii}} \not\equiv x^{p_{ji}} \mod (x^n - 1),$ (3.127)

$$x^{p_{ii}} \not\equiv x^{p_{ki}} \mod (x^n - 1).$$
 (3.128)



Figure 3.10: The fully-connected 3-user X- network with three transmitters Tx_1, Tx_2 and Tx_3 , three receivers Rx_1, Rx_2, Rx_3 , nine independent messages W_{ji} and nine estimated messages \widehat{W}_{ji} . The influence of the channel is parameterized by d_{ji} .

The multiple-access interference conditions at Rx_i for distinct dedicated signals $d_{ii}x^{p_{ii}}$, $d_{ij}x^{p_{ij}}$, $d_{ik}x^{p_{ik}}$, are:

$$d_{ij}x^{p_{ij}} \not\equiv d_{ik}x^{p_{ik}} \mod (x^n - 1),$$
 (3.129)

$$d_{ii}x^{p_{ii}} \neq d_{ij}x^{p_{ij}} \mod (x^n - 1), \tag{3.130}$$

$$d_{ii}x^{p_{ii}} \neq d_{ik}x^{p_{ik}} \mod (x^n - 1).$$
(3.131)

And the inter-user interference conditions at Rx_i are:

$$d_{ii}x^{p_{ii}} \not\equiv d_{ij}x^{p_{kj}} \mod (x^n - 1),$$
 (3.132)

$$d_{ii}x^{p_{ii}} \not\equiv d_{ij}x^{p_{jj}} \mod (x^n - 1), \tag{3.133}$$

$$d_{ii}x^{p_{ii}} \not\equiv d_{ik}x^{p_{jk}} \mod (x^n - 1), \tag{3.134}$$

$$d_{ii}x^{p_{ii}} \not\equiv d_{ik}x^{p_{kk}} \mod (x^n - 1), \tag{3.135}$$

$$d_{ij}x^{p_{ij}} \not\equiv d_{ii}x^{p_{ji}} \mod (x^n - 1), \tag{3.136}$$

$$d_{ij}x^{p_{ij}} \not\equiv d_{ii}x^{p_{ki}} \bmod (x^n - 1), \tag{3.137}$$

$$d_{ij}x^{p_{ij}} \neq d_{ik}x^{p_{jk}} \mod (x^n - 1), \tag{3.138}$$

$$d_{ij}x^{p_{ij}} \not\equiv d_{ik}x^{p_{kk}} \mod (x^n - 1),$$
 (3.139)

for distinct indices $i, j, k \in \mathcal{K}$, respectively. Note that by a circular relabelling of indices, these conditions can also be expressed for Tx_j , Tx_k , Rx_j and Rx_k (cf. Appendix A).

3.5.2 Infeasibility Problem of Perfect Cyclic Interference Alignment

A receiver Rx_i will receive a superposition of six interfering signals in total: $d_{ii}x^{p_{ji}}$, $d_{ij}x^{p_{kj}}$, $d_{ij}x^{p_{kj}}$, $d_{ij}x^{p_{jj}}$, $d_{ik}x^{p_{jk}}$, $d_{ik}x^{p_{kk}}$. Two interfering signals from the same transmitter can not be aligned due to the intra-user interference conditions (3.126) to (3.128). As three dimensions are reserved for dedicated signals and at least two dimensions must be reserved for interference, we demand $n \ge 5$ dimensions.

Perfect cyclic IA is optimal and requires n = 5 dimensions. Three interference signals, i.e., one from each transmitter, must be aligned to a single dimension reserved for interference only. A potentially perfect cyclic IA scheme at Rx_i is constructed by choosing one element from each of these three sets as implied by curly brackets:

$$\begin{cases} d_{ii}x^{p_{ji}} \\ d_{ii}x^{p_{ki}} \end{cases} \equiv \begin{cases} d_{ij}x^{p_{jj}} \\ d_{ij}x^{p_{kj}} \end{cases} \equiv \begin{cases} d_{ik}x^{p_{jk}} \\ d_{ik}x^{p_{kk}} \end{cases} \mod (x^5 - 1),$$
(3.140)

e.g., by taking the elements in the first row, we obtain:

$$d_{ii}x^{p_{ji}} \equiv d_{ij}x^{p_{jj}} \equiv d_{ik}x^{p_{jk}} \mod (x^5 - 1), \tag{3.141}$$

for a fully symmetric perfect IA scheme similar to the one in [37, Section V-C]. IA in one interference dimension directly implicates the complementary alignment of the other interference dimension at the same receiver. For the example given in (3.141), the complementary alignment is given by the second row of (3.140):

$$d_{ii}x^{p_{ki}} \equiv d_{ij}x^{p_{kj}} \equiv d_{ik}x^{p_{kk}} \mod (x^5 - 1).$$

For notational convenience, we will denote auxiliary submatrices of D by:

$$\boldsymbol{D}_{i,k,j,l} = \left(\begin{array}{cc} d_{ij} & d_{il} \\ d_{kj} & d_{kl} \end{array}\right).$$

Note that the determinant of $D_{i,k,j,l}$ also implies the following symmetries:

$$\mathsf{det}(\boldsymbol{D}_{i,k,j,l}) \equiv \mathsf{det}(\boldsymbol{D}_{k,i,l,j}) \equiv -\mathsf{det}(\boldsymbol{D}_{k,i,j,l}) \equiv -\mathsf{det}(\boldsymbol{D}_{i,k,l,j}) \mod (x^n - 1)$$

Theorem 3.11. Perfect cyclic IA with the messaging matrix $M = \mathbf{1}_{3\times 3}$ is infeasible on the 3-user X-network.

Proof:

Each $D_{i,k,j,l}$ corresponds to a subordinate 2×2 X-channel matrix with distinct transmitters Tx_j , Tx_l , and distinct receivers Rx_i , Rx_k . Note that for the 2×2 X-channel, a non-zero determinant of the channel matrix is necessary to perform IA, as we have already shown in [27, Theorem 1 (a)].

Now, we assume that, e. g., $\det(D_{i,j,i,j}) \equiv 0 \mod(x^5-1)$ holds and consider some particular implications on (3.140). On the one hand, if $d_{ii}x^{p_{ji}} \equiv d_{ij}x^{p_{jj}} \mod(x^5-1)$ holds, the above assumption leads to $d_{jj}x^{p_{jj}} \equiv d_{ji}x^{p_{ji}} \mod(x^5-1)$. But this contradicts the multiple-access interference conditions. In analogy, aligning $d_{jj}x^{p_{ij}} \equiv d_{ji}x^{p_{ii}} \mod(x^5-1)$ would imply $d_{ii}x^{p_{ii}} \equiv d_{ij}x^{p_{ij}} \mod(x^5-1)$, yielding another violation of the separability conditions.

Contrariwise, by aligning $d_{ii}x^{p_{ki}} \equiv d_{ij}x^{p_{kj}}$ with the initial assumption $\det(D_{i,j,i,j}) \equiv 0$ would implies $d_{ji}x^{p_{ki}} \equiv d_{jj}x^{p_{kj}}$. This is not a contradiction so far. Beyond that, the above assumption is even a necessary condition when both of these two alignments are used for the IA scheme.

However, the complementary alignment at Rx_i to $d_{ii}x^{p_{ki}} \equiv d_{ij}x^{p_{kj}} \mod (x^5-1)$ is again the previously considered alignment $d_{ii}x^{p_{ji}} \equiv d_{ij}x^{p_{jj}} \mod (x^5-1)$. Thus, the separability conditions are violated for both $\det(\mathbf{D}_{i,j,i,j}) \equiv 0$ and $\det(\mathbf{D}_{i,j,i,j}) \neq 0$. In other words, there is no feasible D available. This conflict carries over to all minors of D analogously. $\hfill\blacksquare$

Theorem 3.11 entails that perfect cyclic IA is also infeasible for the K- user X- network with $K \ge 3$ users since there are $\binom{K}{3}$ embedded 3- user X- networks.

Nonetheless, this problem does not exclude imperfect cyclic IA schemes with a number of n > 2K - 1 dimensions. But imperfect cyclic IA schemes do not achieve the upper bound (3.38) exactly. A corresponding approach will not be discussed in this dissertation.

3.5.3 Common Eigenvectors in Separate Subspaces - Revisited

The infeasibility of perfect cyclic IA shown here is also linked to the problem of the common eigenvectors in perfect spatial IA schemes as discussed in Section 3.4.2 and in the works [3, Section IV-D], [64], and [37, Section V-C].

But the common eigenvector problem is only a subordinate part of the infeasibility problem presented in Theorem 3.11 above. To briefly elaborate this, we consider a symmetric perfect cyclic IA scheme which is analogous to the spatial IA scheme in [37, Equations (10)-(12)]:

$$d_{ii}x^{p_{ji}} \equiv d_{ij}x^{p_{jj}} \equiv d_{ik}x^{p_{jk}} \mod (x^5 - 1), \tag{3.142}$$

$$d_{ii}x^{p_{ki}} \equiv d_{ij}x^{p_{kj}} \equiv d_{ik}x^{p_{kk}} \mod (x^5 - 1), \tag{3.143}$$

for pairwise distinct indices $i, j, k \in \mathcal{K}$. Due to symmetry, the alignment of (3.142) at receiver Rx_k corresponds to:

$$d_{kk}x^{p_{jk}} \equiv d_{kj}x^{p_{jj}} \equiv d_{ki}x^{p_{ji}} \mod (x^5 - 1), \tag{3.144}$$

by a simple relabelling of indices. With (3.142), (3.144), we obtain:

$$\begin{aligned} d_{ii}x^{p_{ji}} &\equiv d_{ij}x^{p_{jj}} \bmod (x^{5} - 1), \\ d_{ki}x^{p_{ji}} &\equiv d_{kj}x^{p_{jj}} \bmod (x^{5} - 1), \\ \Rightarrow d_{ii}d_{ij}^{-1}d_{kj}d_{ki}^{-1}x^{p_{ji}} &\equiv x^{p_{ji}} \bmod (x^{5} - 1). \end{aligned}$$
(3.145)

An analogous formulation arises in [37, Equation (15)] with corresponding diagonal MIMO channel matrices and leads to the problem of common eigenvectors in perfect spatial IA. But in the case of cyclic IA, the result of (3.145) only implies the additional constraint $\det(\boldsymbol{D}_{i,k,j,i}) \equiv 0 \mod (x^5 - 1)$.

3.6 3-User X- Networks with Minimal Backhaul

Since perfect cyclic IA is already shown to be an overconstrained problem, our upcoming approach is to relax the restraining conditions by providing a limited number of cognitive messages over a BHN to achieve sufficient feasibility.

3.6.1 Cyclic IA with Minimal Feedforward Backhaul Networks

A first and very intuitive approach is to include a *feedforward* (*FF*) backhaul network between some transmitters and receivers. A single FF-link $\theta_{\text{FF},ji}$ between Tx_i and Rx_j with rate $\Theta_{\text{FF},ji} = 1$ simply bypasses the channel d_{ji} for a single message so that the actual transmission of the forwarded message may be omitted. The FF-BHN is also shown at the bottom of Figure 5.4. Note that this approach corresponds to using one cognitive receiver Rx_j knowing W_{ji} .

We propose the following alignment scheme and prove its optimality w.r.t. the minimal necessary sum-rate Θ_{FF} :

At Rx_i , interference from Rx_j and Rx_k is perfectly aligned within two dimensions:

$$d_{ii}x^{p_{ji}} \equiv d_{ij}x^{p_{kj}} \equiv d_{ik}x^{p_{jk}} \mod (x^5 - 1), \tag{3.146}$$

$$d_{ii}x^{p_{ki}} \equiv d_{ij}x^{p_{jj}} \equiv d_{ik}x^{p_{kk}} \mod (x^5 - 1).$$
(3.147)

The dedicated and interfering signals at Rx_i are aligned by:

$$d_{ji}x^{p_{ki}} \equiv d_{jj}x^{p_{kj}} \equiv d_{jk}x^{p_{ik}} \mod (x^5 - 1), \tag{3.148}$$

$$d_{ji}x^{p_{ii}} \equiv d_{jk}x^{p_{kk}} \mod (x^5 - 1), \tag{3.149}$$

$$d_{jj}x^{p_{ij}} \equiv d_{jk}x^{p_{jk}} \mod (x^5 - 1), \tag{3.150}$$

and similarly, we use the following Cyclic IA scheme at Rx_k :

$$d_{ki}x^{p_{ii}} \equiv d_{kj}x^{p_{jj}} \equiv d_{kk}x^{p_{ik}} \mod (x^5 - 1), \tag{3.151}$$

$$d_{kj}x^{p_{ij}} \equiv d_{ki}x^{p_{ji}} \mod (x^5 - 1), \tag{3.152}$$

$$d_{ki}x^{p_{ki}} \equiv d_{kk}x^{p_{jk}} \mod (x^5 - 1). \tag{3.153}$$

A relabelling of indices is not permitted in this asymmetric IA scheme. Note that (3.150) and (3.153) explicitly violate the separability conditions. Independent of channel matrix D, the dedicated messages W_{ik} and W_{ki} can not be decoded yet.

Theorem 3.12. The upper bound of $\frac{9}{5}$ DoF for n = 5 on the 3-user X - network is achievable by Cyclic IA with feedforward for $\min(\Theta_{\rm FF}) \ge 1$.

For the considered Cyclic IA scheme, we assume that the following conditions:

- (i) $d_{ij}d_{ki}d_{jk} \equiv d_{ji}d_{ik}d_{kj} \mod (x^5 1),$
- (*ii*) $d_{ii}d_{jk}d_{kj} \equiv d_{jj}d_{ik}d_{ki} \equiv d_{kk}d_{ij}d_{ji} \mod (x^5 1),$
- (*iii*) det $(D_{i,k,i,k}) \not\equiv 0 \mod (x^5 1)$,
- (*iv*) det($D_{i,j,i,j}$) $\not\equiv 0 \mod (x^5 1)$,
- $(v) \det(\boldsymbol{D}_{i,j,i,k}) \not\equiv 0 \mod (x^5 1),$
- (vi) $\det(\mathbf{D}_{i,j,j,k}) \not\equiv 0 \mod (x^5 1)$,
- (vii) $\det(\mathbf{D}_{k,j,k,i}) \not\equiv 0 \mod (x^5 1),$
- (viii) $\det(\boldsymbol{D}_{k,j,k,j}) \not\equiv 0 \mod (x^5 1),$

 $(ix) \ d_{ii}d_{jj}d_{kk} \not\equiv d_{ij}d_{jk}d_{ki} \equiv d_{ji}d_{kj}d_{ik} \ \operatorname{mod}(x^5-1),$

 $(x) \quad d_{kk}d_{ii}d_{jj}d_{kk} \neq d_{jk}d_{kj}d_{ik}d_{ki} \mod (x^5 - 1), \\ d_{ii}d_{ii}d_{jj}d_{kk} \neq d_{ij}d_{ji}d_{kk}d_{ki} \mod (x^5 - 1), \\ d_{ij}d_{ii}d_{jj}d_{kk} \neq d_{ij}d_{ii}d_{ik}d_{kj} \mod (x^5 - 1).$

hold for distinct indices $i, j, k \in \mathcal{K}$.

Proof:

(a) Necessity of $\Theta_{FF} \ge 1$:

Since $\Theta_{FF} = 0$ would correspond to using no FF-link at all, the necessity of $\Theta_{FF} \ge 1$ follows from Theorem 3.11 evidently.

(b) Necessity of constraints (i) to (x) for the given scheme:

First, we consider the constraints (i), (ii) that are implied by the cyclic IA scheme given in (3.146) to (3.153). We depict how the parameters are interlinked by the adjacency graph shown in Figure A.1 of Appendix A. Other potentially valid alignment schemes will imply a different set of constraints and hence another adjacency graph.

Constraint (i) is obtained by substituting the parameters p_{jj} , p_{kk} , p_{ii} in (3.147), (3.148) and (3.151).

One part of constraint (ii) yields from substituting p_{jk} , p_{ji} , p_{ij} in (3.146), (3.150), (3.149). Another part from (ii) yields from substituting p_{jj} , p_{kj} , p_{ik} in (3.147), (3.151), (3.148), and the remaining part from substituting p_{kj} , p_{ki} , p_{jk} in (3.148), (3.153), (3.146).

Now, we consider the impact of the separability conditions. Note that the feasibility of (3.150) and (3.153) is provided by the FF-BHN, as we will show in part (b), i.e., these particular violations are excluded. Constraint (iii) is derived from (3.131) and (3.151). Constraint (iv) is derived by substituting p_{ki} , p_{ji} , p_{kj} with (3.126), (3.146), (3.148).

The remaining constraints (v) to (x) are proven analogously and w.r.t. all separability conditions at each receiver. The extensive proof of those constraints is given in Appendix A. Note that the three constraints of (x) are equivalent due to (ii).

(b) Sufficiency of Cyclic IA with $\Theta_{\rm FF} = 1$:

It suffices to bypass the transmission of the dedicated message W_{jk} over d_{ji} through a single FF-link $\theta_{\text{FF},jk}$ with sum-rate $\Theta_{FF} = 1$. Then, Tx_k may omit the transmission of W_{jk} over D, and Rx_j can still decode W_{jk} from the FF-BHN. As W_{jk} is not transmitted over D at all, Rx_k can also decode W_{ki} interference-free.

To show that the proposed cyclic IA scheme with FF is feasible now, all nine transmission parameters must be resolved. As indicated by the adjacency graph in Figure A.1, we fix the topmost parameter p_{ki} , w.l.o.g. With (3.147), we obtain p_{jj} , p_{kk} , (3.148) provides p_{kj} , p_{ik} , and (3.153) yields p_{jk} . With (3.149), p_{ii} yields from p_{kj} . With (3.146), p_{ji} yields from p_{kj} . And with (3.150), p_{ij} yields from p_{jk} .

A valid matrix, normalized w.r.t. the main diagonal is, e.g.:

$$\boldsymbol{D} = \left(\begin{array}{ccc} 1 & x^4 & x^2 \\ x^4 & 1 & x^2 \\ x & x & 1 \end{array} \right),$$

as all 10 constraints (i) $\equiv x^2$, (ii) $\equiv x^3$, (iii) $\equiv 1 - x^3$, (iv) $\equiv 1 - x^3$, (v) $\equiv x^2 - 1$, (vi) $\equiv x^1 - x^2$, (vii) $\equiv x^4 - x^3$, (viii) $\equiv 1 - x^3$, (ix) $1 \neq x^2$, (x) $1 \neq x^1$ for i = 1, j = 2, k = 3, are fulfilled. A valid set of transmission parameters satisfying all conditions with a fixed $p_{31} = 4$ is:

$$(p_{11}, p_{21}, p_{31}, p_{12}, p_{22}, p_{32}, p_{13}, p_{23}, p_{33}) = (0, 2, 4, 2, 0, 3, 1, \emptyset, 2).$$

The transmitted signals are:

$$\begin{split} v_1(x) &= W_{11}x^0 + W_{21}x^2 + W_{31}x^4, \\ v_2(x) &= W_{12}x^2 + W_{22}x^0 + W_{32}x^3, \\ v_3(x) &= W_{13}x^1 + W_{33}x^2, \end{split}$$

and the received signals are:

$$\begin{split} r_1(x) &\equiv W_{11}x^0 + W_{12}x^1 + (W_{21} + W_{32})x^1 + W_{13}x^3 + (W_{31} + W_{22} + W_{13})x^4 \mod (x^n - 1), \\ r_2(x) &\equiv W_{22}x^0 + W_{21}x^1 + W_{12}x^2 + (W_{13} + W_{32} + W_{31})x^3 + (W_{11} + W_{33})x^4 \mod (x^n - 1), \\ r_3(x) &\equiv W_{31}x^0 + (W_{11} + W_{22} + W_{13})x^1 + W_{33}x^2 + (W_{12} + W_{21})x^3 + W_{32}x^4 \mod (x^n - 1). \end{split}$$

Altogether, including the forwarded message W_{23} for Rx_2 over the FF-BHN, $\frac{9}{5}$ DoF are achieved by cyclic IA with $\Theta_{FF} = 1$.

Note that a delayed FF transmission only delays the decoding time, but it does not affect the feasibility.

We would like to emphasize, that the given constraints on D are profoundly interdependent with the proposed IA scheme. Nonetheless, the analysis of comparable Cyclic IA schemes on this channel can be performed analogously to Theorem 3.12.

3.6.2 Cyclic IAC over Minimal Receiver Backhaul Networks

Now, instead of a FF-BHN, we consider a *receiver backhaul network* (Rx-BHN). The Rx-BHN only permits that receivers may exchange messages to resolve leaking interference in order to satisfy all separability conditions. The Rx-BHN is depicted in Figure 5.4 on the right-hand side. A single link with rate $\Theta_{\text{Rx},ij}$ in the Rx-BHN from Rx_i to Rx_j is denoted by $\theta_{\text{Rx},ji}$.

Similar to the previous section, our aim is to characterize the minimal sum-rate Θ_{Rx} on the Rx-BHN, that is necessary to achieve the upper bound of $\frac{9}{5}$ DoF on the 3- user X- network.

Lemma 3.13. The upper bound of $\frac{9}{5}$ DoF for n = 5 on the 3-user X-network is achievable by cyclic IAC with $\Theta_{Rx} \ge 2$ for the conditions given in Theorem 3.12.

Proof:

(a) Necessity of $\Theta_{Rx} \ge 2$:

As cyclic IA without cancellation is precluded by Theorem 3.11 for n = 5, it follows that $\Theta_{\text{Rx}} > 0$. In contrast to Theorem 3.12, no message can be neglected and bypassed so that all must be sent over the channel **D**. Thus the case $\Theta_{\text{Rx}} = 1$ demands that

the interference at two receivers, say, Tx_i and Tx_j , must be perfectly aligned and only one interfering signal may leak at the remaining receiver Tx_k . However, simultaneous perfect cyclic IA at two receivers is already precluded by Theorem 3.11.

(b) Sufficiency of Cyclic IAC for $\Theta_{Rx} = 2$:

We consider the same cyclic IA scheme as provided in (3.146) to (3.153) subject to constraints (i) to (x) of Theorem 3.12. Note that message W_{jk} is indeed transmitted here. But now, the leaking interference in (3.150) and (3.153) is resolved by $\Theta_{Rx} = 2$ messages over the Rx-BHN. In particular W_{ij} is conveyed over $\theta_{Rx,ji}$, so that W_{ij} can be cancelled from the aligned $W_{ij} + W_{jk}$ to decode the dedicated message W_{jk} at Rx_j . In a subsequent step, W_{jk} is conveyed over $\theta_{Rx,kj}$, so that W_{jk} is cancelled from the aligned $W_{ik} + W_{ki}$ to decode the dedicated message W_{ki} at Rx_k .

A delayed Rx-BHN transmission does not affect feasibility as long as the backhaul messages W_{ij} and W_{jk} for cancellation adhere to the proposed sequence.

3.6.3 Cyclic IN over Minimal Transmitter Backhaul Networks

We now consider the reversed case: Transmitters are connected via a *transmitter* backhaul network (Tx-BHN) instead. A backhaul link from Tx_i to Tx_j is described by $\theta_{Tx,ji}$ correspondingly. The sum-rate over the Tx-BHN is denoted by Θ_{Tx} . The Tx-BHN is depicted in Figure 5.4 on the left-hand side.

Lemma 3.14. The upper bound of $\frac{9}{5}$ DoF for n = 5 on the 3-user X-network is achievable by cyclic IN with $\Theta_{Tx} \ge 2$ for the conditions given in Theorem 3.12.

Proof:

This scheme is a dual to Lemma 3.13 for the Rx-BHN as considered above, so that the necessity of $\Theta_{Tx} \geq 2$ is analogous. Again, we use the alignment scheme of (3.146) to (3.153) subject to constraints (i) to (x) of Theorem 3.12. But in contrast to Lemma 3.13, W_{ij} is firstly conveyed over $\theta_{Tx,kj}$ with $\Theta_{Tx,kj} = 1$ and then the combined message $W_{ij} - W_{jk}$ is conveyed over $\theta_{Tx,ik}$ with $\Theta_{Tx,ik} = 1$. Tx_k transmits the superposition $W_{jk} - W_{ij}$ instead of W_{jk} only, and Tx_i transmits the superposition $W_{ij} - W_{ki} - W_{jk}$ instead of W_{ki} only. This change does not only maintain the decodability of the dedicated signals received at Rx_i or Rx_j, but rather neutralizes the previously leaking interference observed at Rx_k.

In contrast to cyclic IAC over the Rx-BHN, the exchange of signals over the Tx-BHN must be performed at any time before the actual transmission. This relationship also endorses a related IAC-IN duality property reported in [23] for a linear deterministic 2-user interference channel with cooperation.

3.6.4 Combined Cyclic IAC and IN over Minimal Tx/Rx-BHNs

Moreover, if transmitters and receivers are each connected to a disjoint Tx-BHN and Rx-BHN, the IAC and IN schemes of Sections 3.6.2 and 3.6.3 can be combined. The sum-rate over both BHNs is denoted by $\Theta_{TR} = \Theta_{Rx} + \Theta_{Tx}$.

	x^0	x^1	x^2	x^3	x^4
$v_1(x)$	W11	0	W_{21}	0	W ₃₁
$v_2(x)$	W_{22}	0	W_{12}	W_{32}	0
$v_3(x)$	W_{23}	W_{13}	W_{33}	0	0
$r_1(x)$	W_{11}	W_{12}	$W_{21}\!+\!W_{32}\!+\!W_{23}$	W_{13}	$W_{31} + W_{22} + W_{33}$
$r_2(x)$	W_{22}	W_{21}	$W_{23} + W_{12}$	$W_{31} + W_{32} + W_{13}$	$W_{11} + W_{33}$
$r_3(x)$	$W_{31} + W_{23}$	$W_{11} + W_{22} + W_{13}$	W_{33}	$W_{21} + W_{12}$	W_{32}

Figure 3.11: Example of cyclic IAC: The transmitted signals are $v_i(x)$ and the received signals are $r_i(x)$, $i \in \mathcal{K}$. Dedicated signals are highlighted in blue and interference signals in red. Cyclic right-shifts are used here.

Corollary 3.15. The upper bound of $\frac{9}{5}$ DoF for n = 5 on the 3-user X-network is achievable by cyclic IAC/IN with $\Theta_{\text{TR}} \ge 2$.

Proof:

Using (3.146) to (3.153), W_{ij} is provided over $\theta_{Tx,kj}$ and $W_{jk}-W_{ij}$ replaces W_{ij} at Tx_k so that W_{ij} is neutralized at Rx_j . Rx_k receives $W_{ki}+W_{jk}-W_{ij}$. Then, $W_{jk}+W_{ji}$ is provided over $\theta_{Rx,kj}$. Rx_k decodes $(W_{ki}+W_{jk}-W_{ij})-(W_{jk}+W_{ji})+(W_{ji}+W_{ij}) = W_{ki}$ using its interfered signal and (3.152).

3.6.5 Examples for the Given Cyclic IA, IAC and IN Schemes

We use the following valid channel matrix for all examples in this section:

$$\boldsymbol{D} = \begin{pmatrix} 1 & x^4 & x^2 \\ x^4 & 1 & x^2 \\ x & x & 1 \end{pmatrix},$$
(3.154)

and we fix the indices i = 1, j = 2, k = 3 and the parameter vector:

p = (0, 2, 4, 2, 0, 3, 1, 0, 2),

as in part (b) of the proof for Theorem 3.12. The resulting transmitted and received signals for this example are depicted in the table of Figure 3.11. It is easy to see that all dedicated messages are decodable at Rx_1, Rx_2 and Rx_3 , except the messages W_{23} at Rx_2 and W_{31} at Rx_3 .

Cyclic IAC with a FF-BHN

According to Theorem 3.12, Tx_3 provides W_{23} to Rx_2 via the FF-BHN, so that W_{23} is decodable at Rx_2 now. As the message W_{23} is not transmitted over the channel D at all, W_{31} is also decodable at Rx_3 .

Cyclic IAC with a Rx-BHN

According to Lemma 3.13, Rx_1 provides W_{12} to Rx_2 via the Rx-BHN, so that W_{22} is decodable at Rx_2 now. After decoding W_{23} at Rx_2 , Rx_2 provides W_{23} to Rx_3 via the Rx-BHN, so that W_{31} is decodable at Rx_3 .

Cyclic IN with a Tx-BHN

According to Lemma 3.14, Tx₂ provides W_{12} to Tx₃ via the Tx-BHN, and then Tx₃ provides the combined message $W_{12}-W_{23}$ to Tx₁ before the actual transmission of these messages over **D**. Then, Tx₃ transmits $W_{23} - W_{12}$ instead of W_{23} , and Tx₁ transmits $W_{12} - W_{31} - W_{23}$ instead of W_{31} . As a result, all interfering messages neutralize each other and all dedicated messages are decodable.

Cyclic IAC and IN with Tx/Rx-BHNs

According to Corollary 3.15, Tx₂ provides W_{12} to Tx₃ over the Tx-BHN before transmission. Tx₃ transmits $W_{23} - W_{12}$ instead of W_{23} . On the receiver-side, Rx₂ can decode W_{23} as the interference is neutralized. Then Rx₂ provides the combined message $W_{23} + W_{21}$ to Rx₃. As Rx₃ receives $W_{31} + W_{23} - W_{12}$, it adds the known interference $W_{21} + W_{12}$ and substracts $W_{23} + W_{21}$ to finally decode W_{31} . All dedicated messages are decodable.

3.7 Summary

In this chapter, we have presented the three classical unidirectional communication channels, the multiple-access, broadcast and interference channel, for an introductory exposition of the cyclic polynomial channel model. Accordingly, three different and elementary types of *separability conditions* have been defined to analyze the feasibility of these channels: the multiple-access interference conditions, the intra-user interference conditions, and the inter-user interference conditions. Those conditions strongly depend on the interaction of dedicated and interfering messages between multiple users, and also on the number of incident and outgoing subchannels. Given the fulfillment of these channels. We have shown that it suffices to use orthogonal multiple-access schemes and linear encoding and decoding schemes to achieve the upper bounds on the Degrees-of-Freedom for the 2-user X- channel and the K-user interference channel. The results are in accordance with the Degrees-of-Freedom of their counterpart Gaussian channels with single antennas.

In particular, we have observed a *complementary reciprocal symmetry property* of interference alignment in the 2-user X-channel. This property basically states that aligning signals from both transmitters at a receiver Tx_1 provides a particular interference pattern at receiver Tx_2 , and vice versa, aligning the same signals at receiver Tx_1 provides the reciprocal interference pattern observed before at receiver Tx_1 . This property is shown to be significant for the optimal interference alignment scheme of the 2-user X-channel with arbitrary message lengths. To the best of our knowledge, such a property has not been discussed in the literature yet.

Furthermore, we have related the separability conditions to the common eigenvector problem of separate subspaces in MIMO interference alignment. We have observed that a set of feasibility conditions for the cyclic polynomial *K*-user interference channel can be formulated in analogy to the feasibility conditions for the MIMO channel model. Moreover, we have shown that perfect cyclic interference alignment is infeasible in the 3-user X-network and hence also for K-user X-networks with $K \geq 3$ userpairs. In order to counteract the infeasibility we allowed a minimal manipulation: We approached this infeasibility problem by providing a minimal number of cognitive messages to a subset of users via limited backhaul networks. Therein, we observed a duality relationship between cyclic interference alignment and cancellation with a backhaul network at the receivers versus cyclic interference neutralization with a backhaul network at the transmitters.

Altogether, the present chapter has provided Degrees-of-Freedom-achieving for several unidirectional multi-user networks in terms of the cyclic polynomial channel model using multiple-access, linear encoding and decoding and cyclic interference alignment schemes.

4 Multi-User Two-Hop and Two-Way Relay-Networks

In this chapter, we extend our focus to the CPCM for a unidirectional two-hop interference network with two transmitters, two relays and two receivers:

The $2 \times 2 \times 2$ relay-interference channel¹,

and introduce the concept of *cyclic interference neutralization* (IN). Furthermore, we extend the CPCM and cyclic IN to two-way (bidirectional) communications on:

The two-way $2 \times 2 \times 2$ relay-interference channel².

We also consider the following generalization of a multi-user two-way relay channel:

The cascaded two-way relay channel³.

4.1 Cyclic Interference Neutralization

So far, we observed that IA in single-hop multi-user networks is necessary to achieve the maximal number of DoF. For communication scenarios with weak direct links between the dedicated users, however, the use of relays is beneficial to boost the data rate. We consider relays which forward their received signals to the dedicated destination within two hops and we assume that there is no direct link between the users. We assume for the sake of simplicity that relays have full-duplex capability and can perfectly cancel the loop-back self-interference. The considered relays are only intended to support communication, so that they do not desire to convey their own messages to other users.

If signals are forwarded in cooperative multi-hop networks with multiple interjacent relays, e.g., the $2 \times 2 \times 2$ relay-interference channel as depicted in Figure 4.1, IA can be extended to exploit interference neutralization (IN). Such a $2 \times 2 \times 2$ relay-interference channel comprises two sources Tx_1, Tx_2 , two parallel non-interfering interjacent relays R_1, R_2 , and two destinations Rx_1, Rx_2 . A source Tx_i intends to communicate a message W_i to its dedicated destination Rx_i with the aid of both relays R_1 and R_2 within two hops. In the first hop, the signals from Tx_1 and Tx_2 interfere at both relays. In the second hop, the relays forward and re-encode their superimposed messages which will also interfere at the destinations Rx_1 and Rx_2 .

¹Parts of this work have been published in [28].

²Parts of this work have been published in [29].

³Parts of this work have been published in [31].



Figure 4.1: Neutralization on the $2 \times 2 \times 2$ relay-interference channel: Blue solid lines describe the signal paths of message W₁. Red dotted lines describe the signal paths of message W₂.

IN is a cooperative signalling scheme at both the sources and relays such that the interfering signals at undesired destinations are literally 'erased over the air'. In other words, the two incident paths of the corresponding interference signals at a destination Rx_i must exactly coincide and erase each other, while the dedicated signals remain (cf. Figure 4.1). As both relays receive all signals from both transmitters, they forward theses signals such that only the interference *neutralizes* itself at each receivers. This approach generates an effective channel as if there were no interference present between the sources and their dedicated receivers. Thus, the communication from a source Tx_i to its destination Rx_i becomes interference-free over two hops. Hence, the cut-set upper bounds on the DoF of the relay-interference channel are achievable.

To our knowledge, the original idea of IN has been derived and discussed in [47] in terms of the conceptual LDCM at first. A generalization to a *Z*-chain relay-interference network, i. e., a *K*-fold concatenation of *Z*-shaped relay-interference channels, is performed in [48].

For a MIMO $2 \times 2 \times 2$ relay-interference channel with time-varying channel coefficients, the work [63] introduces a closely related *aligned IN* scheme. In [65], an IN scheme on a $2 \times 2 \times 2$ relay-interference channel with interfering relays is investigated. The approximate ergodic capacity of a $2 \times 2 \times 2$ relay-interference channel based on IN and ergodic IA is discussed in [66]. A finite-field $2 \times 2 \times 2$ relay-interference channel is considered in [56]. The authors of [67] provide a more generalized IN scheme by *aligned interference diagonalization* on a $K \times K \times K$ relay-interference channel to achieve the corresponding cut-set upper bound.

In the following, we begin with the investigation of the (unidirectional) $2 \times 2 \times 2$ relay-interference channel in terms of the CPCM. We apply a corresponding cyclic IN scheme that asymptotically achieves its maximal number of DoF.

4.1.1 Cyclic Polynomial 2 × 2 × 2 Relay-Interference Channel

The sets of user-indices and relay-indices are $\mathcal{K}_{Rx} = \mathcal{K}_{Tx} = \mathcal{K}_{R} = \mathcal{K} = \{1, 2\}$. Each source Tx_i desires to communicate a message \boldsymbol{w}_i to a dedicated destination Rx_i for $i \in \mathcal{K}$. The message from Tx_i is represented by a vector n submessages:

$$\boldsymbol{w}_{i} = (W_{i}^{[0]}, W_{i}^{[1]}, \dots, W_{i}^{[n-1]}).$$
(4.1)


Figure 4.2: The polynomial $2 \times 2 \times 2$ relay-interference channel with channel matrix $D = (d_{ji})^{2 \times 2}$ between transmitters Tx_i and relays R_j and channel matrix $E = (e_{ji})^{2 \times 2}$ between relays R_i and destinations Rx_j .

The messaging matrix (for end-to-end communication from Tx_1 , Tx_2 to Rx_1 , Rx_2) is:

$$\boldsymbol{M} = n\boldsymbol{I}_{2\times 2} = \begin{pmatrix} n & 0\\ 0 & n \end{pmatrix}.$$
(4.2)

There is no direct link between the sources and destinations. Thus, the communication is performed over two hops by the aid of two full-duplex relays R_1 and R_2 . The relays receive signals of the first hop at time-instant t. Concurrently, the relays apply a causal relaying function to forward their received signals of the previous first hop of time-instant t - 1.

The block of transmitted signals $u_i(x)$ in one hop has n dimensions. The subchannel of the first hop is denoted by $\mathbf{D} = (d_{ji})_{1 \le i, j \le 2}$ and the subchannel of the second hop by $\mathbf{E} = (e_{ji})_{1 \le i, j \le 2}$ with $d_{ji}, e_{ji} \in \mathcal{D}$. Furthermore, the offset exponents are denoted by δ_{ji} , $\eta_{ji} \in \mathbb{N}$, so that $d_{ji} = x^{\delta_{ji}}$ and $e_{ji} = x^{\eta_{ji}}$.

The sources Tx_i transmit *n* submessages in polynomials of *n* dimensions:

$$u_i(x) = \sum_{k=0}^{n-1} u_i^{[k]} x^k.$$
(4.3)

The received polynomial at a relay R_j is a cyclically shifted superposition of the transmitted polynomials from the sources S_i :

$$r_j(x) \equiv \sum_{i \in \mathcal{K}} d_{ji} u_i(x) \mod (x^n - 1).$$

$$(4.4)$$

Both relays R_i map and encode their received polynomials $r_i(x)$ to construct the forwarded polynomials $v_i(x)$. This mapping may involve a *permutation of coefficients*, a *change of sign*, and even *discarding some specified coefficients* of the received polynomials. After the signals are forwarded from R_1 and R_2 , the received polynomial at destination Rx_j yields:

$$t_j(x) \equiv \sum_i e_{ji} v_i(x) \mod (x^n - 1).$$

$$(4.5)$$

The whole model of the $2 \times 2 \times 2$ relay-interference channel is also illustrated in Figure 4.2. The receivers Rx_i can only resolve and decode their desired messages, if the coefficients of the received polynomials $t_i(x)$ are linear interference-free.

The transmission vector of the first hop is denoted by $\mathbf{u} = (u_1(x), u_2(x))$ and the received vector by $\mathbf{r} = (r_1(x), r_2(x))$ in the vectorized notation. Accordingly, the

transmission vector for the second hop is $\boldsymbol{v} = (v_1(x), v_2(x))$ and the received vector is $\boldsymbol{t} = (t_1(x), t_2(x))$. Then, the transfer functions of both hops can be compactly expressed by:

$$\boldsymbol{r}^{\mathsf{T}} = \boldsymbol{D}\boldsymbol{u}^{\mathsf{T}} \mod (\boldsymbol{x}^n - 1), \tag{4.6}$$

$$\boldsymbol{t}^{\mathsf{T}} = \boldsymbol{E}\boldsymbol{v}^{\mathsf{T}} \mod (\boldsymbol{x}^n - 1), \tag{4.7}$$

where the modulo-operation is component-wise.

In the following, we consider the upper bounds on the DoF of the $2 \times 2 \times 2$ relay-interference channel and the proposed achievable scheme.

4.1.2 Cut-Set Upper Bounds

The capacity is limited by the min-cut upper bounds [63] which are valid for both the CPCM and the time-variant MIMO channel model. Thus, presuming that each message is received interference-free at its dedicated receiver, each user-pair would achieve the capacity of the corresponding point-to-point link. In total there is a maximum number of M = 2n interference-free submessages, i.e., n submessages per user to be conveyed over n dimensions so that the maximal data rate is upper bounded by the cut-set bound of 2 DoF.

4.1.3 Achievability

Our main goal is to formulate an interference-free transmission for all messages from each source to each dedicated destination. Note that instead of decoding single messages at both relays, only functions of superimposed messages are decoded in a network-coded fashion. These functions of superimposed messages are forwarded to the destinations using a proper relaying function. In the following, let the superscript indices in squared brackets be reduced by modulo n for notational convenience.

1) First hop: Each of the *n* submessages $W_i^{[k]}$, k = 0, ..., n - 1, from source Tx_i is allocated to the corresponding dimension at offset x^k . The transmitted polynomial from source Tx_i yields:

$$u_i(x) = \sum_{k=0}^{n-1} W_i^{[k]} x^k.$$
(4.8)

As given by (4.4), the relays \mathbf{R}_j receive the following superposition of submessages per dimension:

$$r_{j}^{[k]} = W_{1}^{[k-\delta_{j1}]} + W_{2}^{[k-\delta_{j2}]}.$$

$$(4.9)$$

2) Second hop: The two relays forward their previously received polynomials by $v_1(x) = x^{\gamma_1}r_1(x) \mod (x^n - 1)$ and $v_2(x) = -x^{\gamma_2}r_2(x) \mod (x^n - 1)$, using the offset parameters $\gamma_1, \gamma_2 \in \{0, ..., n-1\}$, respectively. Using (4.5) and (4.9), the destinations Rx_1 and Rx_2 receive four submessages per dimension:

$$t_1^{[k]} = W_1^{[k-\delta_{11}-\gamma_1-\eta_{11}]} + W_2^{[k-\delta_{12}-\gamma_1-\eta_{11}]} - W_1^{[k-\delta_{21}-\gamma_2-\eta_{12}]} - W_2^{[k-\delta_{22}-\gamma_2-\eta_{12}]},$$
(4.10)

$$t_{2}^{[k]} = W_{1}^{[k-\delta_{11}-\gamma_{1}-\eta_{21}]} + W_{2}^{[k-\delta_{12}-\gamma_{1}-\eta_{21}]} - W_{1}^{[k-\delta_{21}-\gamma_{2}-\eta_{22}]} - W_{2}^{[k-\delta_{22}-\gamma_{2}-\eta_{22}]}.$$
 (4.11)

At both destinations, the dedicated submessages are superimposed by interference. The concept of IN is translated to aligning and combining two identical inter-user interference signals with complementary signs within the same dimension k, so that their sum is zero. To suppress the inter-user interference at both destinations, these two *interference-neutralization conditions* must hold:

$$\delta_{12} + \gamma_1 + \eta_{11} \equiv \delta_{22} + \gamma_2 + \eta_{12} \pmod{n}, \tag{4.12}$$

$$\delta_{11} + \gamma_1 + \eta_{21} \equiv \delta_{21} + \gamma_2 + \eta_{22} \pmod{n}. \tag{4.13}$$

In other words, the inter-user interference is aligned and *neutralized* over two hops along the two possible paths to the non-dedicated receiver.

On the other hand, we must also ensure that the desired signals remain intact and are not neutralized, i. e., the following *no-signal-neutralization conditions* must hold:

$$\delta_{11} + \gamma_1 + \eta_{11} \not\equiv \delta_{21} + \gamma_2 + \eta_{12} \pmod{n}, \tag{4.14}$$

$$\delta_{12} + \gamma_1 + \eta_{21} \not\equiv \delta_{22} + \gamma_2 + \eta_{22} \pmod{n}. \tag{4.15}$$

Let $\Gamma = \text{diag}(x^{\gamma_1}, -x^{\gamma_2})$ denote the relaying function as performed by both relays. The above conditions (4.12) to (4.15) indicate that the matrix product $E\Gamma D$ must be a diagonal matrix of full rank, similar to the expression in [66, Section IV]. If these conditions are satisfied, the superposition of submessages given by (4.10) and (4.11) is reduced to:

$$t_1^{[k]} = W_1^{[k-\delta_{11}-\gamma_1-\eta_{11}]} - W_1^{[k-\delta_{21}-\gamma_2-\eta_{12}]},$$
(4.16)

$$t_2^{[k]} = W_2^{[k-\delta_{12}-\gamma_1-\eta_{21}]} - W_2^{[k-\delta_{22}-\gamma_2-\eta_{22}]}.$$
(4.17)

The superposition of desired submessages received at destination Rx_j as given by (4.16) and (4.17) is compactly expressed by:

$$t_j(x) = (\boldsymbol{X}\boldsymbol{C}_j)\boldsymbol{w}_j^{\mathsf{T}},\tag{4.18}$$

for $X = \text{diag}(x^0, x^1, x^2, ..., x^{n-1})$ and the $n \times n$ coefficient matrix $C_j = (c_{j,lm})_{0 \le l,m \le n-1}$ with:

$$c_{j,lm} = \begin{cases} 1 & , \text{if } m - l \equiv \delta_{1j} + \gamma_1 + \eta_{j1} \pmod{n}, \\ -1 & , \text{if } m - l \equiv \delta_{2j} + \gamma_2 + \eta_{j2} \pmod{n}, \\ 0 & , \text{else.} \end{cases}$$
(4.19)

The superimposed submessages of w_j are resolvable by a linear decoding scheme, if the condition $\det(C_j) \neq 0$ holds.

Lemma 4.1. A linear decoding scheme at destination Rx_i can not resolve n desired submessages $W_i^{[k]}$ from the received vector \mathbf{t}_i for the given perfect interference neutralization scheme.

Proof:

 C_j corresponds to an $n \times n$ circulant matrix \widetilde{C}_j (cf. [68]) with entries $c_{j,lm} = \widetilde{c}_{j,(m-l \mod n)}$. Thus, we have n eigenvectors of C_j , namely $\boldsymbol{v}_i = \frac{1}{\sqrt{n}} (1, w_i, w_i^2, ..., w_i^{n-1})^{\mathsf{T}}$

for the indices i = 0, ..., n - 1 with the roots of unity $w_i = \exp(\frac{j2\pi i}{n})$ and the complex symbol $j = \sqrt{-1}$. The *n* corresponding eigenvalues are $\lambda_i = \sum_{k=0}^{n-1} \tilde{c}_k w_i^k$. Let $\nu_{ji} = \delta_{ij} + \gamma_i + \eta_{ji} \pmod{n}$. The determinant $\det(C_j)$ is computed by the multiplication of *n* eigenvalues:

$$det(C_j) = \prod_{j=0}^{n-1} \lambda_j = \prod_{j=0}^{n-1} (w_j^{\nu_{j1}} - w_j^{\nu_{j2}}) = (1^{\nu_{j1}} - 1^{\nu_{j2}}) \cdot \prod_{j=1}^{n-1} (w_j^{\nu_{j1}} - w_j^{\nu_{j2}}) = 0.$$
(4.20)

Thence, the messages \boldsymbol{w}_i can not be linearly resolved.

In other words, the conditions (4.12) to (4.15) are too strict for a perfect cyclic IN scheme with a total number of 2n submessages. Thence, we propose an *asymptotic IN* scheme for 2n - 1 submessages:

1) First hop: Let source Tx_1 transmit n submessages as in (4.8) and let Tx_2 transmit only n-1 submessages, discarding a single submessage $W_2^{[\tau]}$ with the parameter $\tau \in \{0, ..., n-1\}$:

$$u_1(x) = \sum_{k=0}^{n-1} W_1^{[k]} x^k, \tag{4.21}$$

$$u_2(x) = \sum_{k=0, k\neq\tau}^{n-1} W_2^{[k]} x^k.$$
(4.22)

Now, the k = 0, ..., n - 1 received dimensions at relays \mathbf{R}_i are:

1

$$r_{j}^{[k]} = \begin{cases} W_{1}^{[k-\delta_{j1}]}, & \text{if } k \equiv \tau + \delta_{j2}, \\ W_{1}^{[k-\delta_{j1}]} + W_{2}^{[k-\delta_{j2}]}, & \text{otherwise.} \end{cases}$$
(4.23)

2) Second hop: Relay R_1 forwards all signal in its *n* dimensions and R_2 forwards only n-1 of the *n* received dimensions. In particular, relay R_2 discards forwarding the dimension received at $k_2 \equiv \tau + \delta_{22} \pmod{n}$. One γ_1, γ_2 is arbitrarily chosen and the other is computed by (4.12). The transmitted polynomials are:

$$v_1(x) = x^{\gamma_1} r_1(x) \mod (x^n - 1), \tag{4.24}$$

$$v_2(x) = -x^{\gamma_2} \sum_{k=0, k\neq k_2}^{n-1} r_2^{[k]} x^k \mod (x^n - 1).$$
(4.25)

The received signals at Rx₁, Rx₂ correspond to (4.10), (4.11). The discarded messages for $\sigma_{ji} = \tau + \delta_{i2} + \gamma_i + \eta_{ji} \pmod{n}$ yield:

$$t_1^{[\sigma_{11}]} = W_1^{[\sigma_{11}-\delta_{11}-\gamma_1-\eta_{11}]} - W_1^{[\sigma_{11}-\delta_{21}-\gamma_2-\eta_{12}]} - W_2^{[\sigma_{11}-\delta_{22}-\gamma_2-\eta_{12}]},$$
(4.26)

$${}_{1}^{[\sigma_{12}]} = W_{1}^{[\sigma_{12}-\delta_{11}-\gamma_{1}-\eta_{11}]} + W_{2}^{[\sigma_{12}-\delta_{12}-\gamma_{1}-\eta_{11}]},$$

$$(4.27)$$

$$t_{2}^{[\sigma_{21}]} = W_{1}^{[\sigma_{21}-\delta_{11}-\gamma_{1}-\eta_{21}]} - W_{1}^{[\sigma_{21}-\delta_{21}-\gamma_{2}-\eta_{22}]} - W_{2}^{[\sigma_{21}-\delta_{22}-\gamma_{2}-\eta_{22}]},$$
(4.28)

$$t_2^{[\sigma_{22}]} = W_1^{[\sigma_{22} - \delta_{11} - \gamma_1 - \eta_{21}]} + W_2^{[\sigma_{22} - \delta_{12} - \gamma_1 - \eta_{21}]}.$$
(4.29)

Theorem 4.2. The asymptotic interference neutralization scheme achieves $\frac{2n-1}{n}$ DoF on the cyclic polynomial channel, if the interference-neutralization conditions (4.12), (4.13) and no-self-neutralization conditions (4.14), (4.15) hold.

Proof:

For the given conditions, the received signals at Rx_1 , Rx_2 further simplify to (4.16), (4.17) and to these special cases:

$$t_{1}^{[\sigma_{1j}]} = W_{1}^{[\sigma_{1j}-\delta_{11}-\gamma_{1}-\eta_{11}]}.$$
(4.30)

$$t_2^{[\sigma_{21}]} = -W_2^{[\sigma_{21}-\delta_{22}-\gamma_2-\eta_{22}]},\tag{4.31}$$

$$t_{2}^{[\sigma_{22}]} = W_{1}^{[\sigma_{22}-\delta_{11}-\gamma_{1}-\eta_{21}]} + W_{2}^{[\sigma_{22}-\delta_{12}-\gamma_{1}-\eta_{21}]}.$$
(4.32)

Note that $\sigma_{11} \equiv \sigma_{12} \pmod{n}$ holds here. Furthermore, the conditions (4.12) to (4.15) imply a proper choice of γ_1 and γ_2 . At destination Rx_1 , the coefficient matrix C_1 has almost the same structure as in (4.19). The exception is an additional zero-entry in C_1 at row σ_{1j} and column $\sigma_{1j} - \delta_{21} - \gamma_2 - \eta_{12}$ as given by (4.30). By Laplace's formula, we can recursively expand the determinant of C_1 along the rows with only one non-zero entry, i. e., row σ_{1j} in the first iteration. The determinant yields $\det(C_1) = 1$ and each submessage dedicated for Rx_1 is linear decodable.

Destination Rx_2 discards row σ_{22} and column τ in \mathbb{C}_2 since it only needs to decode the remaining n-1 submessages and not submessage $W_2^{[\tau]}$. Furthermore, the interfering submessage $W_1^{[\sigma_{22}-\delta_{11}-\gamma_1-\eta_{21}]}$ in (4.32) is not neutralized anyway. Thus, we consider a reduced coefficient matrix $\widehat{\mathbb{C}}_2$ which is a corresponding $(n-1) \times (n-1)$ matrix of \mathbb{C}_2 . $\widehat{\mathbb{C}}_2$ has a single row with only one non-zero entry at σ_{21} as given in (4.31). In analogy to \mathbb{C}_1 , the determinant yields $\det(\widehat{\mathbb{C}}_2) = 1$, so that each submessage dedicated for Rx_2 is also linear decodable.

Altogether, a total number of M = 2n - 1 submessages is conveyed interference-free over $n \ge 2$ dimensions using cyclic IN and linear decoding. The asymptotic scheme achieves $\lim_{n\to\infty} \frac{2n-1}{n} = 2$ DoF in the limit.

Note that valid channel parameters for $n \ge 2$ do exist, e.g., $d_{12} = e_{12} = x^1$, $d_{11} = d_{21} = d_{22} = e_{11} = e_{21} = e_{22} = x^0$.

Corollary 4.3. The conditions of Theorem 4.2 also imply that:

- (a) $\delta_{12} + \delta_{21} + \eta_{11} + \eta_{22} \equiv \delta_{11} + \delta_{22} + \eta_{12} + \eta_{21} \pmod{n}, \\ \delta_{12} + \delta_{21} + \eta_{12} + \eta_{21} \not\equiv \delta_{11} + \delta_{22} + \eta_{11} + \eta_{22} \pmod{n},$
- (b) $\det(\mathbf{D}) \not\equiv 0 \mod (x^n 1), \det(\mathbf{E}) \not\equiv 0 \mod (x^n 1),$
- (c) and $n \ge 2$ dimensions,

must hold for cyclic IN.

Proof:

(a) The first condition, is obtained by substituting (4.12) and (4.13) w.r.t. γ_1 or γ_2 . The same is done in (4.14) and (4.15) for the second condition, respectively.

(b) Assuming $\det(D) \equiv 0 \mod (x^n - 1)$, yields $\delta_{11} + \delta_{22} \equiv \delta_{12} + \delta_{21} \pmod{n}$. Inserting this into the first condition of (a), it follows $\det(E) \equiv 0 \mod (x^n - 1)$. Further inserting $\delta_{11} + \delta_{22} \equiv \delta_{12} + \delta_{21} \pmod{n}$ and $\eta_{11} + \eta_{22} \equiv \eta_{12} + \eta_{21} \pmod{n}$ into the second condition

of (a), leads to a contradiction.

(c) By assuming n = 1, $det(D) \equiv det(E) \equiv 0 \mod(x-1)$ always holds. This also leads to a contradiction as shown in (b).

An illustrative example of cyclic IN is given in Appendix B.

4.1.4 Aligned IN on MIMO $2 \times 2 \times 2$ Relay-Interference Channels

The proposed asymptotic cyclic IA scheme of the previous section is not restricted for the application on the CPCM. In this section, we elaborate how the scheme can be extended to a generalized version of the *aligned interference neutralization* (AIN) scheme of [63] for a time-varying $2 \times 2 \times 2$ relay-interference channel.

System Model

We briefly recapitulate the channel model of the time-varying $2 \times 2 \times 2$ relay-interference channel for the AIN scheme as given in [63]. The aforementioned MIMO channel model is depicted in Figure 4.3.

1) First hop: The subchannel from source Tx_i to relay R_j is characterized by a channel coefficient $F_{ji} \in \mathbb{C}$. The relay R_j receives a superposition of the signals transmitted by the two sources Rx_i plus additive i.i.d. Gaussian noise $Z_{\operatorname{R}_i}(t) \sim \mathcal{CN}(0,1)$:

$$Y_{\mathbf{R}_{j}}(t) = \sum_{i=1}^{2} F_{ji}(t) X_{\mathrm{Tx}_{i}}(t) + Z_{\mathbf{R}_{j}}(t).$$
(4.33)

2) Second hop: The subchannel from relay R_i to destination Rx_j is characterized by a channel coefficient $G_{ji} \in \mathbb{C}$. The destinations Rx_j receive a superposition of the signals from the relays R_i plus additive i.i.d. Gaussian noise, $Z_{Rx_j}(t) \sim \mathcal{CN}(0, 1)$:

$$Y_{\text{Rx}_{j}}(t) = \sum_{i=1}^{2} G_{ji}(t) X_{\text{R}_{i}}(t) + Z_{\text{Rx}_{j}}(t).$$
(4.34)

All sources, relays and destinations are equipped with single antennas only. The channel coefficients are generic and assumed to be time-varying in each discrete time-instant t. The values of these coefficients are bounded between a non-zero minimum and a finite maximum. Since the channel coefficients are generic, the matrices have full rank almost surely and are invertible, accordingly.

An *n*-symbol extension over n time-slots, as also utilized in [3] and [63], provides diagonal channel matrices over time-varying channel coefficients:

$$\boldsymbol{F}_{ji}(t) = \text{diag}(F_{ji}(nt+1), \dots, F_{ji}(nt+n)), \qquad (4.35)$$

$$G_{ji}(t) = \text{diag}(G_{ji}(nt+1), ..., G_{ji}(nt+n)).$$
(4.36)

From now on, the time-instant t is dropped for notational brevity. We obtain the following received signals at the relays R_j and at the destinations Rx_k , as also depicted in Figure 4.3:

$$\boldsymbol{Y}_{\mathrm{R}_{j}} = \sum_{i=1}^{2} \boldsymbol{F}_{ji} \boldsymbol{X}_{\mathrm{Tx}_{i}} + \boldsymbol{Z}_{\mathrm{R}_{j}}, \qquad (4.37)$$

$$\boldsymbol{Y}_{\mathrm{Rx}_{k}} = \sum_{i=1}^{2} \boldsymbol{G}_{ki} \boldsymbol{X}_{\mathrm{R}_{i}} + \boldsymbol{Z}_{\mathrm{Rx}_{k}}.$$
(4.38)



Figure 4.3: The channel model of [63] for the MIMO relay-interference channel with diagonal channel matrices F_{ji} between sources Tx_i , relays R_j and diagonal channel matrices G_{ji} between relays R_i and destinations Rx_j for $i, j \in \{1, 2\}$.

X, Y, Z are $n \times 1$ vectors, i.e., the *n*-symbol extensions of X, Y, Z. Sources and relays encode their messages into Gaussian codebooks of length n, with codeword symbols $w_{\mathrm{Tx}_i}^{[k]}, w_{\mathrm{R}_i}^{[k]}$, and use beam-forming vectors $v_{\mathrm{Tx}_i}^{[k]}$ and $v_{\mathrm{R}_i}^{[k]}$, each of $n \times 1$ dimensions, to transmit the codewords over the given channel. The transmitted signals⁴ from the Tx_i and R_i are:

$$\boldsymbol{X}_{\mathrm{Tx}_{i}} = \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{i}}^{[k]} \boldsymbol{w}_{\mathrm{Tx}_{i}}^{[k]}, \qquad (4.39)$$

$$\boldsymbol{X}_{\mathrm{R}_{i}} = \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{R}_{i}}^{[k]} \boldsymbol{w}_{\mathrm{R}_{i}}^{[k]}.$$
(4.40)

The average transmit power for each transmit vector is limited by P.

(Cyclic) Generalization of the Aligned Interference Neutralization Scheme

An explanatory example for AIN is given in [63, Section I-D] for a symbol extension of n = 2 symbols. A linear AIN scheme for general $n \ge 2$ is given in [63, Section III-A]. Therein, AIN discards the symbols $w_{\text{Tx}_2}^{[n-1]}$, $w_{\text{R}_2}^{[n-1]}$ and aligns the beam-forming vectors for i = 0, ..., n - 2 by:

$$\boldsymbol{F}_{11}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i+1]} = \boldsymbol{F}_{12}\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[i]}, \qquad (4.41)$$

$$\boldsymbol{F}_{21}\boldsymbol{v}_{\mathrm{Tx}_1}^{[i]} = \boldsymbol{F}_{22}\boldsymbol{v}_{\mathrm{Tx}_2}^{[i]}, \qquad (4.42)$$

$$G_{11}v_{R_1}^{[i+1]} = -G_{12}v_{R_2}^{[i]},$$
 (4.43)

$$\boldsymbol{G}_{21}\boldsymbol{v}_{\mathrm{R}_{1}}^{[i]} = -\boldsymbol{G}_{22}\boldsymbol{v}_{\mathrm{R}_{2}}^{[i]}.$$
(4.44)

There, it is proven that $\frac{2n-1}{n}$ DoF are achievable on the relay-interference channel.

We now show how this scheme can be interpreted as a special case of Theorem 4.2 with the cyclic IN scheme.

1) First hop:

Source Rx₁ sends n and Rx₂ sends n-1 symbols $w_{\text{Tx}_j}^{[0]}, ..., w_{\text{Tx}_j}^{[n-1]}$, along beam-forming vectors $\boldsymbol{v}_{\text{Tx}_j}^{[0]}, ..., \boldsymbol{v}_{\text{Tx}_j}^{[n-1]}$, for j = 1, 2, discarding $w_{\text{Tx}_2}^{[\tau]}$ along $\boldsymbol{v}_{\text{Tx}_2}^{[\tau]}$, respectively. In order

⁴Our notation of indices slightly differs from [63].

to imitate the separate dimensions of the CPCM, the beam-forming vectors align at the relays R_1 and R_2 , for i = 0, ..., n - 1, as follows:

$$\boldsymbol{F}_{11}\boldsymbol{v}_{\mathrm{Tx}_1}^{[i-\delta_{11}]} = \boldsymbol{F}_{12}\boldsymbol{v}_{\mathrm{Tx}_2}^{[i-\delta_{12}]}, \ i \not\equiv \tau + \delta_{12} \ (\mathrm{mod} \ n), \tag{4.45}$$

$$\boldsymbol{F}_{21}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i-\delta_{21}]} = \boldsymbol{F}_{22}\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[i-\delta_{22}]}, \ i \not\equiv \tau + \delta_{22} \pmod{n}.$$
(4.46)

First, we equivalently express (4.45) and (4.46) by:

$$\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[i]} = \boldsymbol{F}_{12}^{-1} \boldsymbol{F}_{11} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i+\delta_{12}-\delta_{11}]}, \ i \notin \tau \pmod{n},$$
(4.47)

$$\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[i]} = \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i+\delta_{22}-\delta_{21}]}, \ i \not\equiv \tau \pmod{n},$$
(4.48)

in order to substitute $v_{\text{Tx2}}^{[i]}$:

$$\boldsymbol{F}_{11}\boldsymbol{F}_{12}^{-1}\boldsymbol{v}_{\mathrm{Tx}_1}^{[i+\delta_{12}-\delta_{11}]} = \boldsymbol{F}_{22}^{-1}\boldsymbol{F}_{21}\boldsymbol{v}_{\mathrm{Tx}_1}^{[i+\delta_{22}-\delta_{21}]}, \ i \not\equiv \tau \pmod{n}.$$

Then, we resolve the resulting equation w.r.t. $v_{\text{Tx}_1}^{[i]}$:

$$\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i+\delta_{12}-\delta_{11}+\delta_{21}-\delta_{22}]} = \boldsymbol{F}_{11}^{-1} \boldsymbol{F}_{12} \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[i]}.$$

Now, we use the following variables for brevity:

$$\Delta_{\boldsymbol{D}} \equiv \delta_{12} - \delta_{11} + \delta_{21} - \delta_{22} \pmod{n}, \tag{4.49}$$

$$\tau_1 \equiv \tau + \delta_{12} - \delta_{11} \pmod{n},\tag{4.50}$$

$$F = F_{11}^{-1} F_{12} F_{22}^{-1} F_{21}.$$
(4.51)

Using these variables, we obtain the simpler expression:

$$v_{\text{Tx}_1}^{[i+\Delta_D]} = F v_{\text{Tx}_1}^{[i]}.$$
 (4.52)

We fix $\boldsymbol{v}_{\mathrm{Tx}_1}^{[\kappa]} = \mathbf{1}_{n \times 1}$ for some $\kappa \in \{0, \ldots, n-1\}$. To compute the corresponding vector from Tx₂ aligned to $\boldsymbol{v}_{\mathrm{Tx}_1}^{[\kappa]}$, we substitute $\boldsymbol{v}_{\mathrm{Tx}_1}^{[\kappa+\Delta_D]}$ on the left-hand side with (4.48), and obtain:

$$\begin{split} \boldsymbol{v}_{\text{Tx}_{2}}^{[\kappa+\Delta_{D}]} &= \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{v}_{\text{Tx}_{1}}^{[\kappa+\Delta_{D}+\delta_{22}-\delta_{21}]} \\ \Rightarrow \boldsymbol{v}_{\text{Tx}_{2}}^{[\kappa+\Delta_{D}-\delta_{22}+\delta_{21}]} &= \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{F} \boldsymbol{v}_{\text{Tx}_{1}}^{[\kappa]}. \end{split}$$

Then, by substituting $\boldsymbol{v}_{\text{Tx}_2}^{[\kappa+\Delta_D-\delta_{22}+\delta_{21}]} = \boldsymbol{F}_{12}^{-1} \boldsymbol{F}_{11} \boldsymbol{v}_{\text{Tx}_1}^{[\kappa+2\Delta_D]}$ with (4.47), we obtain:

$$\begin{split} \boldsymbol{F}_{12}^{-1} \boldsymbol{F}_{11} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa+2\Delta_{D}]} &= \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]} \\ \Rightarrow \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa+2\Delta_{D}]} &= \boldsymbol{F}^{2} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]}. \end{split}$$

Repeating these steps for all i = 1, ..., n - 1 vectors, yields:

$$\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa+i\Delta_{D}]} = \boldsymbol{F}^{i} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]}.$$

For the computation of each $v_{Tx_2}^{[\kappa+i\Delta_D-\delta_{22}+\delta_{21}]}$, we obtain:

$$\begin{aligned} & \boldsymbol{v}_{\text{Tx}_2}^{[\kappa+i\Delta_D-\delta_{22}+\delta_{21}]} = \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \boldsymbol{F}^i \boldsymbol{v}_{\text{Tx}_1}^{[\kappa]} \\ \Rightarrow & \boldsymbol{v}_{\text{Tx}_2}^{[\kappa+i\Delta_D-\delta_{12}+\delta_{11}]} = \boldsymbol{F}_{21}^{-1} \boldsymbol{F}_{21} \boldsymbol{F}^{i-1} \boldsymbol{v}_{\text{Tx}_1}^{[\kappa]} \end{aligned}$$

for general i = 1, ..., n - 1. Since only diagonal matrices are involved here, the multiplication is commutative, providing:

$$\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[\kappa+i\Delta_{D}-\delta_{12}+\delta_{11}]} = \boldsymbol{F}^{i-1}\boldsymbol{F}_{22}^{-1}\boldsymbol{F}_{21}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]}.$$
(4.53)

Hence, we have the following extension⁵ for the expressions in [63, Eqs. (48), (49)]:

$$\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa+i\Delta_{D}]} = \boldsymbol{F}^{i}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]}, \qquad (4.54)$$

$$\boldsymbol{v}_{\mathrm{Tx}_{2}}^{[\kappa+i\Delta_{D}+\delta_{21}-\delta_{22}]} = \boldsymbol{F}^{i}\boldsymbol{F}_{22}^{-1}\boldsymbol{F}_{21}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\kappa]}.$$
(4.55)

Since $det(D) \not\equiv 0 \mod (x^n - 1)$ is assumed for the CPCM, it is clear that also $\Delta_D \not\equiv 0 \pmod{n}$ holds. The initial offset for the allocation of the vectors from Tx_1 is chosen as:

$$\tau_1 = \tau + \delta_{12} - \delta_{11}. \tag{4.56}$$

This offset is aligned at R_1 , so that it exactly fits into the offset-gap which was intentionally left open by Tx_2 at offset τ . We consider the following composite matrix for the *n* transmission vectors from Tx_1 :

$$\boldsymbol{B} = (\boldsymbol{v}_{\text{Tx}_{1}}^{[\tau_{1}]}, \boldsymbol{v}_{\text{Tx}_{1}}^{[\tau_{1}+\Delta_{D}]}, \boldsymbol{v}_{\text{Tx}_{1}}^{[\tau_{1}+2\Delta_{D}]}, ..., \boldsymbol{v}_{\text{Tx}_{1}}^{[\tau_{1}+(n-1)\Delta_{D}]})$$
(4.57)

$$= (\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \boldsymbol{F}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \boldsymbol{F}^{2}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \dots, \boldsymbol{F}^{n-1}\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}).$$
(4.58)

To ensure that all *n* vectors are allocated, *n* and $\Delta_{\mathbf{D}}$ must be relatively prime, i.e., $gcd(n, \Delta_{\mathbf{D}}) = 1$. Then, $\Delta_{\mathbf{D}}$ is a generator of the abelian (cyclic) additive group \mathbb{Z}_n .

It is yet to show that these vectors are not colinear in n dimensions. Let B_m denote the *m*-th diagonal entry of F, so that (4.57) can also be represented by an $n \times n$ Vandermonde matrix:

$$\boldsymbol{B} = \begin{pmatrix} 1 & B_1 & B_1^2 & \cdots & B_1^{n-1} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 1 & B_{n-1} & B_{n-1}^2 & \cdots & B_{n-1}^{n-1} \end{pmatrix}.$$
 (4.59)

Its well-known determinant yields the Vandermonde polynomial:

$$\det(\boldsymbol{B}) = \prod_{0 \le i < j \le n-1} (B_j - B_i) \neq 0, \qquad (4.60)$$

since the B_m are distinct almost surely. Thus, the *n* beam-forming vectors in **B** are linear independent. The n-1 beam-forming vectors transmitted from Tx_2 are also

⁵The proposed IN scheme in [63, Equations (48), (49)] resembles the following choice of parameters: $\delta_{21} = \delta_{22} = \delta_{11} = 0, \delta_{12} = 1, \tau = 0 \implies \Delta_D = \kappa = 1.$

linear independent by an analogous computation. We consider the following composite matrix C for the n-1 transmission vectors from Tx₂: (with $v_{\text{Tx}_2}^{[\tau_1-\delta_{12}+\delta_{11}+i\Delta_D]} = v_{\text{Tx}_2}^{[\tau+i\Delta_D]}$)

$$\boldsymbol{C} = (\boldsymbol{v}_{\text{Tx}_2}^{[\tau+\Delta_D]}, \boldsymbol{v}_{\text{Tx}_2}^{[\tau+2\Delta_D]}, ..., \boldsymbol{v}_{\text{Tx}_2}^{[\tau+(n-1)\Delta_D]})$$
(4.61)

$$= \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} (\boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \boldsymbol{F} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \boldsymbol{F}^{2} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}, \dots, \boldsymbol{F}^{n-2} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[\tau_{1}]}).$$
(4.62)

Let C_m denote the *m*-th diagonal element of C. The according representation as an $n \times n - 1$ Vandermonde matrix is:

$$\boldsymbol{C} = \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \left(\begin{array}{cccc} 1 & C_1 & C_1^2 & \cdots & C_1^{n-2} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 1 & C_{n-1} & C_{n-1}^2 & \cdots & C_{n-1}^{n-2} \end{array} \right).$$
(4.63)

Likewise, its determinant is non-zero, almost surely, and the vectors are linear independent:

$$\det(\boldsymbol{F}_{22}^{-1}\boldsymbol{F}_{21}\boldsymbol{C}) = \det(\boldsymbol{F}_{22}^{-1}) \cdot \det(\boldsymbol{F}_{21}) \cdot \prod_{0 \le i < j \le n-1} (C_j - C_i) \neq 0.$$
(4.64)

Using the given pre-coding scheme, the received signal $\boldsymbol{Y}_{\mathrm{R}_{i}}$ is:

$$\boldsymbol{Y}_{\mathrm{R}_{j}} = \boldsymbol{F}_{j1} \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[k]} w_{\mathrm{Tx}_{1}}^{[k]} + \boldsymbol{F}_{j2} \sum_{k'=0, \, k'\neq\tau}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{2}}^{[k']} w_{\mathrm{Tx}_{2}}^{[k']} + \boldsymbol{Z}_{\mathrm{R}_{j}}.$$
(4.65)

Then, the received signal is filtered by $F_{jj}^{-1} Y_{R_j}$:

$$\begin{split} \boldsymbol{F}_{11}^{-1} \boldsymbol{Y}_{\mathrm{R}_{1}} &= \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[k]} \boldsymbol{w}_{\mathrm{Tx}_{1}}^{[k]} + \boldsymbol{F}_{11}^{-1} \boldsymbol{F}_{12} \sum_{k'=0,\,k'\neq\tau}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{2}}^{[k']} \boldsymbol{w}_{\mathrm{Tx}_{2}}^{[k']} + \boldsymbol{F}_{11}^{-1} \boldsymbol{Z}_{\mathrm{R}_{1}} \\ &= \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[k]} \boldsymbol{w}_{\mathrm{Tx}_{1}}^{[k]} + \sum_{k'=0,\,k'\neq\tau}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[k'-\delta_{11}+\delta_{12}]} \boldsymbol{w}_{\mathrm{Tx}_{2}}^{[k']} + \boldsymbol{F}_{11}^{-1} \boldsymbol{Z}_{\mathrm{R}_{1}}, \\ \boldsymbol{F}_{22}^{-1} \boldsymbol{Y}_{\mathrm{R}_{2}} &= \boldsymbol{F}_{22}^{-1} \boldsymbol{F}_{21} \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{1}}^{[k]} \boldsymbol{w}_{\mathrm{Tx}_{1}}^{[k]} + \sum_{k'=0,\,k'\neq\tau}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{2}}^{[k']} \boldsymbol{w}_{\mathrm{Tx}_{2}}^{[k']} + \boldsymbol{F}_{22}^{-1} \boldsymbol{Z}_{\mathrm{R}_{2}} \\ &= \sum_{k=0}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{2}}^{[k-\delta_{22}+\delta_{21}]} \boldsymbol{w}_{\mathrm{Tx}_{1}}^{[k]} + \sum_{k'=0,\,k'\neq\tau}^{n-1} \boldsymbol{v}_{\mathrm{Tx}_{2}}^{[k']} \boldsymbol{w}_{\mathrm{Tx}_{2}}^{[k']} + \boldsymbol{F}_{22}^{-1} \boldsymbol{Z}_{\mathrm{R}_{2}}. \end{split}$$

Note that the filtered noise is negligible here due to the high SNR assumption. The resulting symbols are analogously ordered (and cyclically shifted) for each dimension as in the cyclic IN scheme proposed in Section 4.1.3.

2) Second hop:

An analogous construction to obtain the *n* separate dimensions is performed in the second hop. The relays amplify and forward their received signals from the previous hop. Furthermore, the forwarded symbols are also index-shifted by the offsets γ_1, γ_2 given in (4.24) and (4.25).

Relay R_1 sends n and R_2 sends a number of n-1 symbols $x_{R_j}^{[0]}, ..., x_{R_j}^{[n-1]}$ along beam-forming vectors $\boldsymbol{v}_{R_j}^{[0]}, ..., \boldsymbol{v}_{R_j}^{[n-1]}$. Relay R_2 discards to forward $x_{R_2}^{[\tau+\delta_{22}+\gamma_2]}$ along $\boldsymbol{v}_{R_2}^{[\tau+\delta_{22}+\gamma_2]}$. The beam-forming vectors are aligned at the destinations Rx_1 and Rx_2 , for i = 1, ..., n-2, as follows:

$$\begin{split} & \boldsymbol{G}_{11} \boldsymbol{v}_{\mathrm{R}_{1}}^{[i-\eta_{11}]} = - \, \boldsymbol{G}_{12} \boldsymbol{v}_{\mathrm{R}_{2}}^{[i-\eta_{12}]}, \ i \not\equiv \tau + \delta_{22} + \gamma_{2} + \eta_{12} \ (\mathrm{mod} \ n), \\ & \boldsymbol{G}_{21} \boldsymbol{v}_{\mathrm{R}_{1}}^{[i-\eta_{21}]} = - \, \boldsymbol{G}_{22} \boldsymbol{v}_{\mathrm{R}_{2}}^{[i-\eta_{22}]}, \ i \not\equiv \tau + \delta_{22} + \gamma_{2} + \eta_{22} \ (\mathrm{mod} \ n). \end{split}$$

The dependencies of the beam-forming vectors are similarly resolved w.r.t. $v_{R_1}^{[\tau_2]} = \mathbf{1}_{n-1}$ as in the first hop, for i = 0, ..., n-2:

$$\boldsymbol{v}_{\mathrm{B}_{1}}^{[\tau_{2}+i\Delta_{\boldsymbol{E}}]} = \boldsymbol{G}^{i}\boldsymbol{v}_{\mathrm{B}_{1}}^{[\tau_{2}]},\tag{4.66}$$

$$\boldsymbol{v}_{\mathrm{R}_{2}}^{[\tau_{2}+i\Delta_{E}+\eta_{21}-\eta_{22}]} = -\boldsymbol{G}^{i}\boldsymbol{G}_{22}^{-1}\boldsymbol{G}_{21}\boldsymbol{v}_{\mathrm{R}_{1}}^{[\tau_{2}]}, \qquad (4.67)$$

using the parameters:

$$\Delta_{\boldsymbol{E}} \equiv \eta_{12} - \eta_{11} + \eta_{21} - \eta_{22} \pmod{n}, \tag{4.68}$$

$$\tau_2 \equiv \tau + \delta_{22} + \gamma_2 + \eta_{12} - \eta_{11} \pmod{n}, \tag{4.69}$$

$$G = (G_{11}^{-1}G_{12}G_{22}^{-1}G_{21}).$$
(4.70)

Note that (4.66) and (4.67) are basically analogous to (4.54) and (4.55).

Since $\det(E) \neq 0 \mod (x^n - 1)$ is assumed, $\Delta_E \neq 0 \pmod{n}$ holds, as well. As in the first hop, n and Δ_E must be coprime. Then, the linear independence of vectors can be shown analogously to the scheme given for the first hop. The received signals Y_{Rs_j} are filtered with G_{jj}^{-1} . The resulting symbols are comparably ordered as in (4.10) and (4.11) with the special cases of (4.30) to (4.32).

As a result, we can apply the cyclic IA framework for IN of Section 4.1.3 on the transmitted symbols, and achieve $\frac{2n-1}{n}$ DoF by Theorem 4.2.

The linear scheme of [63] expressed by (4.41) to (4.44) is translated to the CPCM using $n \in \mathbb{N}$ dimensions, parameter $\tau = n - 1$, and channel matrices D and E with the following entries, $d_{12} = e_{12} = x^1$, $d_{11} = d_{21} = d_{22} = e_{11} = e_{22} = x^0$.

4.2 Two-Way Cyclic Interference Neutralization

Now, we further generalize the concept of cyclic IN to a two-way version of the fullduplex $2 \times 2 \times 2$ relay-interference channel. The arrangement of users in this channel is similar to [69] and [70].



Figure 4.4: The 2 × 2 × 2 full-duplex two-way relay-interference channel: Transceivers T_i transmit their signals $u_i(x)$ to relays R_j over the uplink channel matrix $\boldsymbol{D} = (d_{ji})_{1 \le j \le 2, 1 \le i \le 4}$ and the relays receive $r_j(x)$. Relays R_j forward signals $v_j(x)$, as functions of $r_j(x)$, over the downlink channel matrix $\boldsymbol{E} = (e_{ij})_{1 \le i \le 4, 1 \le j \le 2}$ to transceivers T_i , who receive the corresponding $t_i(x)$.

4.2.1 Cyclic Polynomial Two-Way $2 \times 2 \times 2$ Relay-Interference Channel

Such a channel comprises four full-duplex transceivers T_1, T_2, T_3, T_4 , and two fullduplex relays R_1, R_2 , as depicted in Figure 4.4. Each transceiver T_i transmits a message $\boldsymbol{w}_i, i \in \mathcal{K}_T$. The user-pair (T_1, T_3) desires to exchange messages \boldsymbol{w}_1 and \boldsymbol{w}_3 over the given channel, and the user-pair (T_2, T_4) desires to exchange messages \boldsymbol{w}_2 and \boldsymbol{w}_4 , respectively. There are no direct links between the transceivers and no direct links between the two relays. The messaging matrix (for end-to-end communication from all T_i to all T_j) is:

$$\boldsymbol{M} = n \begin{pmatrix} \boldsymbol{I}_{2\times 2} & \boldsymbol{0}_{2\times 2} \\ \boldsymbol{0}_{2\times 2} & \boldsymbol{I}_{2\times 2} \end{pmatrix} = \begin{pmatrix} 0 & n & 0 & 0 \\ n & 0 & 0 & 0 \\ 0 & 0 & 0 & n \\ 0 & 0 & n & 0 \end{pmatrix}.$$
 (4.71)

Note that this channel can also be interpreted as a cognitive unidirectional $4 \times 2 \times 4$ relay-interference channel with 4 dedicated transmitter-receiver pairs with cognitive messages between pairs of Tx_i and Rx_i for each $i \in \mathcal{K}$. The related cognitive $3 \times 2 \times 3$ relay-interference channel is included by simply removing a dedicated transmitter-receiver pair, e. g., Tx_4 and Rx_2 . Furthermore, the two-way $2 \times 2 \times 2$ relay-interference channel is a superposition of four (unidirectional) $2 \times 2 \times 2$ relay-interference channels:

- 1) The dedicated links are $T_1 \rightarrow T_3$ and $T_2 \rightarrow T_4$
- 2) The dedicated links are $T_3 \rightarrow T_1$ and $T_4 \rightarrow T_2$
- 3) The dedicated links are $T_1 \rightarrow T_3$ and $T_4 \rightarrow T_2$
- 4) The dedicated links are $T_3 \rightarrow T_1$ and $T_2 \rightarrow T_4$.

The set of dedicated transmitter-indices for a receiver T_i is denoted by \mathcal{D}_i , i.e., the singleton sets $\mathcal{D}_1 = \{3\}$, $\mathcal{D}_2 = \{4\}$, $\mathcal{D}_3 = \{1\}$ and $\mathcal{D}_4 = \{2\}$. We combine the indices of the two communicating user-pairs in the sets $\mathcal{G}_{13} = \{1,3\}$ and $\mathcal{G}_{24} = \{2,4\}$. The set of interfering transmitter-indices at a receiver R_i is denoted by \mathcal{I}_i , i.e., $\mathcal{I}_1 = \mathcal{I}_3 = \{2,4\}$ and $\mathcal{I}_2 = \mathcal{I}_4 = \{1,3\}$.

In both two-hop and two-way relay communication systems, the channel access is usually described by two different access phases: The *multiple-access phase* or *first hop* describes the communication from transceivers to relays and the *broadcast phase* or *second hop* describes the communication from relays to transceivers, accordingly. For multiple-relays, we will term these phases as the *uplink-phase* and the *downlink-phase*.

To model individual cyclic shifts for each transmitter-receiver link, the uplink (UL) from the four transceivers to the two relays is described by the *UL channel matrix* $\boldsymbol{D} = (d_{j_i})_{1 \leq j \leq 2, 1 \leq i \leq 4}$ and the downlink (DL) from the relays is described by the *DL channel matrix* $\boldsymbol{E} = (e_{ij})_{1 \leq i \leq 4, 1 \leq j \leq 2}$ with $d_{j_i}, e_{j_i} \in \mathcal{D} = \{x^k \mid k \in \mathbb{N}\}$, respectively. We denote the offsets by $\delta_{j_i}, \eta_{j_i} \in \mathbb{N}$, i. e., $d_{j_i} = x^{\delta_{j_i}}$ and $e_{ij} = x^{\eta_{ij}}$.

1) UL-phase: As in (4.3), the sources T_i , $i \in \mathcal{K}_T$, map their messages w_i to a polynomial $u_i(x)$ with n dimensions:

$$u_i(x) = \sum_{k=0}^{n-1} W_i^{[k]} x^k.$$
(4.72)

The polynomials $u_i(x)$ are transmitted to the relays \mathbf{R}_j , $j \in \mathcal{K}_{\mathbf{R}}$, over the UL matrix D so that the two relays \mathbf{R}_j receive a superposition of interfering polynomials:

$$r_j(x) = \sum_{i=1}^4 d_{ji} u_i(x) \mod (x^n - 1).$$
(4.73)

2) *DL-phase*: The relays R_j use a causal relaying function on their received polynomials $r_j(x)$. These polynomials are mapped to the polynomials $v_j(x)$. Then, the relays R_j forward their $v_j(x)$:

$$v_1(x) = x^{\gamma_1} r_1(x) \mod (x^n - 1), \tag{4.74}$$

$$v_2(x) = -x^{\gamma_2} r_2(x) \mod (x^n - 1), \tag{4.75}$$

to the four destinations T_i , $i \in \mathcal{K}$, over the DL matrix E. A destination T_i receive the following superposition:

$$t_i(x) = \sum_{j=1}^{2} e_{ij} v_j(x) \mod (x^n - 1).$$
(4.76)

The superposition of the three non-dedicated messages at destination T_i causes undesired interference. Only if all dedicated signals are received interference-free, the four destinations can decode their dedicated messages $\widehat{\boldsymbol{w}}_1, \widehat{\boldsymbol{w}}_2, \widehat{\boldsymbol{w}}_3$ and $\widehat{\boldsymbol{w}}_4$, respectively.

The submessages \boldsymbol{w}_i , which return from the relays to their original transceiver T_i during the DL-phase, are back-propagated self-interference [71]. Since the transceivers T_j know their own signals transmitted in the previous UL-phase, they can completely cancel their corresponding back-propagated self-interference.

The transmission vectors of the UL-phase are denoted by

$$\boldsymbol{u} = (u_1(x), u_2(x), u_3(x), u_4(x)), \tag{4.77}$$

$$\boldsymbol{r} = (r_1(x), r_2(x)),$$
 (4.78)

respectively. In the DL-phase, we utilize the vectors:

$$v = (v_1(x), v_2(x)),$$

$$t = (t_1(x), t_2(x), t_3(x), t_4(x)).$$

Then, the transfer functions of the channel are:

$$\boldsymbol{r}^{\mathsf{T}} = \boldsymbol{D}\boldsymbol{u}^{\mathsf{T}} \mod (\boldsymbol{x}^n - 1), \tag{4.79}$$

$$\boldsymbol{t}^{\mathsf{T}} = \boldsymbol{E}\boldsymbol{v}^{\mathsf{T}} \mod (x^n - 1), \tag{4.80}$$

where the modulo operation is taken component-wise.

4.2.2 Cut-Set Upper Bounds

For an interference-free transmission between four users with n dimensional signals, a total number of M = 4n independent submessages must be decodable, i. e., n messages per user. A total number of exactly 4 DoF would be achieved for four interference-free subchannels. Such a result describes the cut-set upper bound on the DoF of the $2 \times 2 \times 2$ two-way relay-interference channel. In the following, we propose a combined cyclic SA and cyclic IN scheme that achieves the upper bound in the asymptotic limit for $n \to \infty$.



Figure 4.5: The interference-neutralization conditions in (4.81) demand that the identical signals transmitted by T_i coincide at each undesired transceivers T_j with complementary signs, so that interference is 'erased over the air'.

4.2.3 Achievability

We begin with the separability conditions of the two-way cyclic IN scheme by extending the previous separability conditions of the unidirectional $2 \times 2 \times 2$ relay-interference channel. First, we omit back-propagated self-interference, since it is cancelled at each transceiver. Thus, no additional self-interference condition comes up in this regard.

However, on the one hand, we demand that the *inter-user interference* caused by the two undesired transceivers in \mathcal{I}_i is neutralized. The essential idea is again to combine two identical inter-user interference signals with complementary signs within the same dimension k, such that their sum vanishes. Thence, the *interference-neutralization conditions* for all interfering pairs, with $i \in \mathcal{K}_T$, and $j \in \mathcal{I}_i$, are:

$$\delta_{1i} + \gamma_1 + \eta_{j1} \equiv \delta_{2i} + \gamma_2 + \eta_{j2} \pmod{n}. \tag{4.81}$$

This condition is also illustrated in Figure 4.5. The dedicated submessages on the other hand, may not be neutralized and must remain decodable. Accordingly, the *no-signal-neutralization conditions* for all dedicated pairs, with $i \in \mathcal{K}_{T}$, and $j \in \mathcal{D}_{i}$, are:

$$\delta_{1i} + \gamma_1 + \eta_{j1} \not\equiv \delta_{2i} + \gamma_2 + \eta_{j2} \pmod{n}. \tag{4.82}$$

Altogether, assuming that the back-propagated self-interference signals are removed and the conditions (4.81) and (4.82) are fulfilled, the transceivers T_j only receive a superposition of two dedicated submessages per dimension k in the DL-phase:

$$t_{j}^{[k]} = W_{i}^{[k-\delta_{1i}-\gamma_{1}-\eta_{j1}]} - W_{i}^{[k-\delta_{2i}-\gamma_{2}-\eta_{j2}]}, j \in \mathcal{D}_{i}.$$
(4.83)

Using the vectorized notation as in 4.1.3:

$$\boldsymbol{w} = (W_i^{[0]}, W_i^{[1]}, \dots, W_i^{[n-1]}), \tag{4.84}$$

$$\boldsymbol{X} = \mathsf{diag}(x^0, x^1, \dots, x^{n-1}), \tag{4.85}$$

we can consider all *n* components of (4.83) together, yielding the received polynomial $t_j(x) = (\mathbf{X}\mathbf{C}_j)\mathbf{w}_j^{\mathsf{T}}$ with an $n \times n$ coefficient matrix $\mathbf{C}_j = (c_{j,lm})_{0 \leq l,m \leq n-1}$ with rows

indexed by l and columns by m. For $i \in \mathcal{D}_j$, we obtain the following *circulant matrix* with two non-zero bands:

$$c_{j,lm} = \begin{cases} 1 & , \text{if } m - l \equiv \delta_{1i} + \gamma_1 + \eta_{j1} \pmod{n}, \\ -1 & , \text{if } m - l \equiv \delta_{2i} + \gamma_2 + \eta_{j2} \pmod{n}, \\ 0 & , \text{else.} \end{cases}$$
(4.86)

The received submessages are linear decodable, only if $\det(C_j) \neq 0$ holds. In Lemma 4.1, we have already shown that cyclic IN is infeasible if all transmitting users allocate n submessages in n dimensions in a unidirectional $2 \times 2 \times 2$ relay-interference channel. Thus, as none of the four contained unidirectional relay-interference channels supports perfect cyclic IN for the case of n submessages per user, the two-way $2 \times 2 \times 2$ relay-interference channel does not support perfect cyclic IN either.

In order to enable cyclic IN with linear decoding nonetheless, we propose an extended *asymptotic* cyclic IN scheme that generalizes the cyclic IN scheme in Section 4.1.3 to the given $2 \times 2 \times 2$ two-way relay-interference channel:

1) UL-phase:

The transceivers T_1, T_3 transmit n submessages whereas T_2, T_4 only transmit n-1 submessages, discarding the submessages $W_2^{\lceil \tau_2 \rceil}, W_4^{\lceil \tau_1 \rceil}$ for $\tau_2, \tau_4 \in \{1, ..., n-1\}$:

$$u_i(x) = \sum_{k=0}^{n-1} W_i^{[k]} x^k, \ i = 1, 3,$$
(4.87)

$$u_i(x) = \sum_{k=0, k\neq\tau_i}^{n-1} W_i^{[k]} x^k, \ i = 2, 4.$$
(4.88)

The submessages received at the two relays per dimension k correspond to:

$$r_j^{[k]} = \sum_{i=1}^4 W_i^{[k-\delta_{ji}]},\tag{4.89}$$

except for the following cases with $j \in \mathcal{K}_{\mathrm{R}}$, $m \in \{2, 4\}$:

$$r_{j}^{[\tau_{m}+\delta_{jm}]} = \sum_{i=1,i\neq m}^{4} W_{i}^{[\tau_{m}+\delta_{jm}-\delta_{ji}]}.$$
(4.90)

We choose the parameters τ_2 and τ_4 such that:

$$\kappa_2 = \tau_2 + \delta_{22} \equiv \tau_4 + \delta_{24} \pmod{n} \tag{4.91}$$

holds, i.e., both the discarded submessages will affect exactly one dimension κ_2 at receiver R_2 . Accordingly, we define $\kappa_{12} \equiv \tau_2 + \delta_{12} \pmod{n}$ and $\kappa_{14} \equiv \tau_4 + \delta_{14} \pmod{n}$ to describe the dimensions at R_1 that are affected by the discarded submessages from T_2 and T_4 . Due to the interference-neutralization conditions, the dimensions of these discarded submessages are likewise aligned at R_1 . To show this, we consider (4.81) for $i \in \{2, 4\}$ and j = 1:

$$\begin{split} \delta_{12} + \gamma_1 + \eta_{11} &\equiv \delta_{22} + \gamma_2 + \eta_{12} \pmod{n}, \\ \delta_{14} + \gamma_1 + \eta_{11} &\equiv \delta_{24} + \gamma_2 + \eta_{12} \pmod{n}. \end{split}$$

By substituting γ_1 , we easily obtain:

$$\delta_{12} - \delta_{14} \equiv \delta_{22} - \delta_{24} \pmod{n}.$$
 (4.92)

It follows from (4.91) and (4.92) that $\tau_2 + \delta_{12} \equiv \tau_4 + \delta_{14} \pmod{n}$ holds, i.e., we may set $\kappa_1 = \kappa_{12} \equiv \kappa_{14} \pmod{n}$.

2) DL-phase:

Relay R_1 forwards its received polynomial $r_1(x)$ according to:

$$v_1(x) = x^{\gamma_1} r_1(x) \mod (x^n - 1). \tag{4.93}$$

 R_2 forwards only n-1 dimensions of the received polynomial $r_2(x)$ and discards $r_2^{[\kappa_2]}$:

$$v_2(x) = -x^{\gamma_2} \sum_{k=0, k\neq\kappa_2}^{n-1} r_2^{[k]} x^k \mod (x^n - 1).$$
(4.94)

Let $\sigma_{ji} = \kappa_i + \gamma_i + \eta_{ji}$ for notational brevity. The received dimensions at the destinations T_j are as given in (4.83). The following cases result from the discarded coefficients, and from the self-interference cancellation for $j \in \mathcal{G}_{13}, i \in \mathcal{D}_j$:

$$\begin{split} t_{j}^{[\sigma_{j1}]} &= W_{i}^{[\sigma_{j1}-\delta_{1i}-\gamma_{1}-\eta_{j1}]} - W_{i}^{[\sigma_{j1}-\delta_{2i}-\gamma_{2}-\eta_{j2}]} - W_{2}^{[\sigma_{j1}-\delta_{22}-\gamma_{2}-\eta_{j2}]} \\ &+ W_{4}^{[\sigma_{j1}-\delta_{14}-\gamma_{1}-\eta_{j1}]} - W_{4}^{[\sigma_{j1}-\delta_{24}-\gamma_{2}-\eta_{j2}]}, \end{split} \tag{4.95} \\ t_{j}^{[\sigma_{j2}]} &= W_{i}^{[\sigma_{j1}-\delta_{1i}-\gamma_{1}-\eta_{j1}]} - W_{i}^{[\sigma_{j2}-\delta_{2i}-\gamma_{2}-\eta_{j2}]} + W_{2}^{[\sigma_{j1}-\delta_{12}-\gamma_{1}-\eta_{j1}]} \\ &+ W_{4}^{[\sigma_{j1}-\delta_{14}-\gamma_{1}-\eta_{j1}]} - W_{4}^{[\sigma_{j1}-\delta_{24}-\gamma_{2}-\eta_{j2}]}, \end{aligned} \tag{4.96}$$

and for $j \in \mathcal{G}_{24}, i \in \mathcal{D}_j$:

$$t_{j}^{[\sigma_{j_{1}}]} = W_{1}^{[\sigma_{j_{1}}-\delta_{11}-\gamma_{1}-\eta_{j_{1}}]} - W_{1}^{[\sigma_{j_{1}}-\delta_{21}-\gamma_{2}-\eta_{j_{2}}]} - W_{i}^{[\sigma_{j_{1}}-\delta_{2i}-\gamma_{2}-\eta_{j_{2}}]} + W_{3}^{[\sigma_{j_{1}}-\delta_{13}-\gamma_{1}-\eta_{j_{1}}]} - W_{3}^{[\sigma_{j_{1}}-\delta_{23}-\gamma_{2}-\eta_{j_{2}}]},$$
(4.97)

By further including the interference-neutralization conditions from (4.81), the equations (4.95) and (4.96) reduce to:

$$t_{j}^{[\sigma_{j1}]} = W_{i}^{[\sigma_{j1}-\delta_{1i}-\gamma_{1}-\eta_{j1}]}, i \neq j \in \mathcal{G}_{13},$$
(4.99)

$$t_{j}^{[\sigma_{j2}]} = W_{i}^{[\sigma_{j2}-\delta_{2i}-\gamma_{2}-\eta_{j2}]}, i \neq j \in \mathcal{G}_{13}.$$
(4.100)

By definition of σ_{ji} and by condition (4.81), we observe that (4.97) and (4.98) coincide for each $j \in \mathcal{G}_{13}$. According simplifications also apply to (4.97) and (4.98):

$$t_{j}^{[\sigma_{j1}]} = -W_{i}^{[\sigma_{j1}-\delta_{2i}-\gamma_{2}-\eta_{j2}]}, \ i \neq j \in \mathcal{G}_{24},$$

$$(4.101)$$

$$t_{j}^{[\sigma_{j2}]} = W_{i}^{[\sigma_{j2}-\delta_{1i}-\gamma_{1}-\eta_{j1}]} + W_{1}^{[\sigma_{j2}-\delta_{11}-\gamma_{1}-\eta_{j1}]} + W_{3}^{[\sigma_{j2}-\delta_{13}-\gamma_{1}-\eta_{j1}]}, \ i \neq j \in \mathcal{G}_{24}.$$
(4.102)

The following theorem generalizes the (unidirectional) cyclic IN scheme in Theorem 4.2 of Section 4.1.3 to the present two-way case.

Theorem 4.4. Asymptotic cyclic interference neutralization on the $2 \times 2 \times 2$ full-duplex two-way relay-interference channel achieves $\frac{4n-2}{n}$ DoF if all the following conditions hold:

(a) back-propagated self-interference is cancelled at each T_i ,

(b) the separability conditions (4.81) and (4.82) hold,

(c) and the number of signalling dimensions is $n \ge 2$.

Proof:

The *n* dedicated submessages received at T_j , $j \in \mathcal{G}_{13}$, are described by (4.83) and by the exceptions in (4.99) and (4.100). Now, the corresponding coefficient matrices C_j have almost the same structure as (4.86), except that the single entry in row σ_{j1} and column $\sigma_{j1} - \delta_{1i} - \gamma_1 - \eta_{j1}$ is zero. In this case, all *n* submessages at T_j with $j \in \mathcal{G}_{13}$ are decodable, since $\det(C_j) = 1$ holds as in Theorem 4.2 of Section 4.1.3 for the unidirectional case.

The n-1 dedicated submessages at T_j , $j \in \mathcal{G}_{24}$, are also decodable. In this case, it suffices to consider a reduced $(n-1) \times (n-1)$ coefficient matrices \widetilde{C}_j , since only n-1 submessages per transceiver must be decoded. Moreover, the interference in the remaining dimension is not neutralized anyway. In particular, the entry in row σ_{j2} and column with $W_i^{[\tau_i]}$, $j \in \mathcal{D}_i$, $j \neq i \in \mathcal{G}_{24}$ is discarded. Then, $\det(\widetilde{C}_j) = 1$ for $j \in \mathcal{G}_{13}$, as analogously shown in Theorem 4.2.

By considering the derivation of (4.92), we observe that the proposed interferenceneutralization conditions demand a particular symmetry of the considered channel. We subsume the symmetry for all analogous cases by the parameters $\alpha_1, \alpha_2, \beta_1, \beta_2 \in \mathbb{N}$:

$$\alpha_1 = \delta_{11} - \delta_{21} \equiv \delta_{13} - \delta_{23} \pmod{n}, \tag{4.103}$$

$$\alpha_2 = \delta_{12} - \delta_{22} \equiv \delta_{14} - \delta_{24} \pmod{n}, \tag{4.104}$$

$$\beta_1 = \eta_{11} - \eta_{12} \equiv \eta_{31} - \eta_{32} \pmod{n}, \tag{4.105}$$

$$\beta_2 = \eta_{21} - \eta_{22} \equiv \eta_{41} - \eta_{42} \pmod{n}. \tag{4.106}$$

Using this parameterization, the interference-neutralization conditions yield:

$$\alpha_1 + \gamma_1 \equiv \gamma_2 - \beta_2 \pmod{n},\tag{4.107}$$

$$\alpha_2 + \gamma_1 \equiv \gamma_2 - \beta_1 \pmod{n},\tag{4.108}$$

and the no-signal-neutralization conditions are:

$$\alpha_1 + \gamma_1 \not\equiv \gamma_2 - \beta_1 \pmod{n}, \tag{4.109}$$

$$\alpha_2 + \gamma_1 \not\equiv \gamma_2 - \beta_2 \pmod{n}. \tag{4.110}$$

Substituting (4.107) and (4.108) into (4.109) and (4.110) yields:

$$\alpha_1 - \alpha_2 \not\equiv \beta_2 - \beta_1 \pmod{n}. \tag{4.111}$$

in both cases. Valid matrices that fulfil these simplified conditions clearly exist, if $n \ge 2$ as demanded by condition (c).

Altogether, 4n - 2 submessages are conveyed interference-free over n dimensions. In the asymptotic limit, the cyclic IN scheme achieves $\lim_{n\to\infty} \frac{4n-2}{n} = 4$ DoF on the two-way $2 \times 2 \times 2$ relay-interference channel.

4.2.4 Discussion

In contrast to Section 4.1 on the (unidirectional) 2×2 relay-interference channel, the present cyclic IN scheme can not be translated to a corresponding *aligned IN* scheme for a MIMO system with time-varying channel coefficients in the same manner.

The main problem, we encounter here is that the signals from the transceivers T_2 and T_4 aligned at R_1 are exactly the same ones aligned at R_2 , since $\kappa_{12} \equiv \kappa_{14} \pmod{n}$ holds as given in Section 4.2.2. Although this problem is neither an issue for in the representation of the CPCM, nor for the closely related representation in the LDCM, this might be an overconstrained problem for the aligned IN framework [63] which is based on spatial IA and IN [3].

The related IN scheme given in [70] provides, among other results, that for all four terminals with M antennas each and N antennas at each relay, a total number of $\min(2M, 4N)$ DoF is achievable for $N \leq M$. The authors also mention the corresponding cut-set upper bounds of $\min(4M, 4N)$ DoF. However, it is shown that the cut-set upper bound of 4M DoF is indeed achievable, but under the condition that the relays are equipped with a greater number of $N > \frac{4}{3}M$ antennas by using the particular IN scheme provided in [70]. An interesting problem that remains to be solved is whether there is a feasible aligned IN scheme that is capable to achieve the cut-set upper bound of 4M DoF on the related Gaussian MIMO channel model with only M = N antennas at the relays.

4.3 Cascaded Two-Way Relaying

We now consider two closely related two-way relay communication systems with fullduplex users and full-duplex intermediate relays. The elementary network problem of the well-known two-way relay channel [57] considers a pair of two users communicating via one relay, that is used to forward the bidirectional messages. Achievable rate regions for cooperation schemes are provided in [71] and [72] for two-way decodeand-forward or two-way compress-and-forward, respectively. An approximate capacity analysis of the two-way relay channel within 3 bits, involving the conceptual LDCM [9], is provided in [73].

Generalizations to $K \ge 2$ user-pairs communicating over a single relay are considered in [74] and [75]. A multi-hop scheme for a single user-pair over a a line of K twoway relays is presented in [76]. The given relaying scheme avoids the regeneration of back-propagated self-interference at the relays in particular. Another generalization of two-way relaying concerns multiple-unicast transmissions per user. For the case K = 3, this is called a Y-channel [77], [78], and [79]. In that case, each of the 3 users intends to transmit a single message to the other two users over a single two-way relay. Besides IA, the communication schemes achieving the upper bounds on the DoF also involve SA and the cancellation of back-propagated self-interference. Similar strategies are used for two-way relaying with network-coding.

Another kind of generalizations for two-way relaying concerns multiple-unicast transmissions per user. For the case K = 3, this is called a Y-channel as discussed in [77], [78], and [79]. In that case, each of the 3 users intends to transmit a single message to the other two users over a single two-way relay.

In this section, we investigate K cascaded concatenated two-way relay channels w.r.t. the CPCM as depicted in Figure 4.6. To the best of our knowledge, this kind of generalization of a two-way relay network has not been considered in the literature so far. More precisely, we consider two closely related systems with a number of K users and K, or K-1, relays forming either an open-loop line network or a closed-loop ring network, respectively. In contrast to the ordinary two-way relay channel, the transmissions of each user are involved in up to two neighbouring two-way relay channels. We provide an upper bound and an achievable scheme that is based on both SA and IA and the cancellation of back-propagated self-interference. Furthermore, we observe for the closed-loop case that switching the roles of users and relays yields a dual network with reciprocal channels.



Figure 4.6: A cascaded two-way relay network: K users and K relays are alternately arranged. User T_i , is only connected to its predecessor T_{i-1} and its successor T_{i+1} via two-way relays R_{i-1} and R_i , respectively, for each i = 1, ..., K. The UL channel \boldsymbol{E} is described by solid lines and the DL channel \boldsymbol{F} by dashed lines. The open-loop case neglects Relay R_K , as indicated in grey.

4.3.1 Cyclic Polynomial Cascaded Two-Way Relay Networks

We now describe the two cascaded two-way relay networks with K-users in terms of the CPCM. We consider a full-duplex two-way relay network with K users T_i , $i \in \mathcal{K}_T$, and K relays R_j , $j \in \mathcal{K}_R$, for $K \ge 3$, as depicted in Figure 4.6. The sets of user-indices \mathcal{K}_T and relay-indices \mathcal{K}_R are both defined by $\mathcal{K}_T = \mathcal{K}_R = \mathcal{K}$.

Each user $T_i, i \in \mathcal{K}$, intends to convey one message $W_{i-1,i}$ to its predecessor T_{i-1} over relay R_{i-1} and another message $W_{i+1,i}$ to its successor T_{i+1} over relay R_i . Thus, there are M = 2K independent messages W_{ji} in the system in total. For notational simplicity, indices K + i correspond to i for a closed-loop of K users in a circular indexation. The messages \widehat{W}_{ji} to be decoded at each T_j are denoted with a hat. To compactly describe the number of messages $m_{j,i} \in \mathbb{N}$ from T_i to T_j for each communication involved, we define a messaging matrix $\mathbf{M} = (m_{j,i})_{j,i\in\mathcal{K}}$ and set its entries to $m_{i,i+1} = m_{i,i-1} = 1$ and $m_{j,i} = 0$, else:

$$\boldsymbol{M} = \begin{pmatrix} \boldsymbol{0}_{1 \times K-1} & 1 \\ \boldsymbol{I}_{K-1 \times K-1} & \boldsymbol{0}_{K-1 \times 1} \end{pmatrix} + \begin{pmatrix} \boldsymbol{0}_{K-1 \times 1} & \boldsymbol{I}_{K-1 \times K-1} \\ 1 & \boldsymbol{0}_{1 \times K-1} \end{pmatrix}.$$
(4.112)

In other words, almost all elements are zero, except the side-diagonals and the top-right and the bottom-left entries. The total number of messages is M = 2K for the given $K \times K$ -matrix.

Next, we take a closer look at the signalling and the communication over the channel itself. The transmitted signal from T_i is a polynomial with messages W_{ji} for each intended receiver T_{i-1} and T_{i+1} :

$$u_i(x) \equiv W_{i-1,i}x^{p_{i-1,i}} + W_{i+1,i}x^{p_{i+1,i}} \mod (x^n - 1).$$
(4.113)

The parameter $p_{j,i} \in \{0,...,n-1\}$ allocates the message W_{ji} to a particular offset within n dimensions.

The UL channel matrix describes all subchannels from the users T_i to the relays R_i . It is defined by $\boldsymbol{E} = (e_{j,i})_{j,i\in\mathcal{K}}$ and has the elements $e_{j,i} \in \mathcal{D}$ with the set of monomials $\mathcal{D} = \{x^k \mid k \in \mathbb{Z}^+\}$ describing the individual shifts from T_i to R_j . In \boldsymbol{E} , most elements are zero, except $e_{i,i+1}$ and $e_{i+1,i}$ for all $i \in \mathcal{K}$. The zero entries are similar to those in the messaging matrix \boldsymbol{M} . The DL channel matrix $\boldsymbol{F} = (f_{j,i})_{j,i\in\mathcal{K}}$ for subchannels from R_i to T_j is defined accordingly for $f_{j,i} \in \mathcal{D}$ with non-zero $f_{i,i+1}$ and $f_{i+1,i}$, for all $i \in \mathcal{K}$. In Figure 4.6, the UL subchannels are depicted by solid lines and the DL subchannels by dashed lines. The transfer function of the UL channel is the congruence:

$$\boldsymbol{r}^{\mathsf{T}} \equiv \boldsymbol{E}\boldsymbol{u}^{\mathsf{T}} \mod (x^n - 1), \tag{4.114}$$

with the $1 \times K$ input vector \boldsymbol{u} and the $1 \times K$ output vector \boldsymbol{r} :

$$\boldsymbol{u} = (u_1(x), \dots, u_K(x)), \tag{4.115}$$

$$\boldsymbol{r} = (r_1(x), \dots, r_K(x)). \tag{4.116}$$

The modulo operation is taken element-wise. The signals $r_i(x)$ received at R_i are further processed as follows. For filtering certain dimensions from the received polynomial $r_i(x)$, we define the element-wise product of two polynomials $r_i(x)$ and $z_i(x)$, with maximal degree n - 1, by:

$$r_i(x) \circ z_i(x) = \sum_{k=0}^{n-1} r_i^{[k]} z_i^{[k]} x^k.$$
(4.117)

The yet unspecified filter polynomial $z_i(x)$ has filtering coefficients $z_i^{[k]} \in \{0, 1\}$ for all offsets $k \in \{0, 1, ..., n-1\}$. The entries in $z_i(x)$ are chosen such that the dimensions with the undesired interference terms in $r_i(x)$ are removed (by multiplying zero) in the element-wise product $r_i(x) \circ z_i(x)$, whereas the dimensions of the dedicated signals in $r_i(x)$ remain unchanged (by multiplying one). The filtered polynomial in (4.117) may furthermore be cyclically shifted by x^{γ_i} with offset $\gamma_i \in \mathbb{N}$. The resulting polynomial after filtering and cyclic shifting is forwarded by \mathbf{R}_i and denoted:

$$v_i(x) \equiv x^{\gamma_i}(r_i(x) \circ z_i(x)) \mod (x^n - 1).$$
 (4.118)

The transfer function of the DL channel is the congruence:

$$\boldsymbol{t}^{\mathsf{T}} \equiv \boldsymbol{F} \boldsymbol{v}^{\mathsf{T}} \mod (\boldsymbol{x}^{n} - 1), \tag{4.119}$$

for the $1 \times K$ input vector \boldsymbol{v} and the $1 \times K$ output vector \boldsymbol{t} :

$$\boldsymbol{v} = (v_1(x), ..., v_K(x)),$$
 (4.120)

$$\boldsymbol{t} = (t_1(x), \dots, t_K(x)). \tag{4.121}$$

To ensure interference-free decodability of all dedicated signals, the *separability con*ditions for $i \in \mathcal{K}$ as defined in Chapter 3 are adapted to this particular communication problem as follows:

Multiple-access interference conditions: Dedicated messages to T_i transmitted by different sources T_{i-1} and T_{i+1} must be separable from each other at destination T_i :

$$f_{i,i-1}x^{\gamma_{i-1}}e_{i,i-1}x^{p_{i,i-1}} \neq f_{i,i+1}x^{\gamma_i}e_{i,i+1}x^{p_{i,i+1}} \mod (x^n - 1).$$

$$(4.122)$$

Intra-user interference conditions: Messages from the same source T_i , but dedicated for different destinations, must be distinct:

$$x^{p_{i+1,i}} \not\equiv x^{p_{i-1,i}} \mod (x^n - 1). \tag{4.123}$$

Inter-user interference conditions: The dedicated messages from T_{i-1} to T_{i-2} and from T_{i+1} to T_{i+2} may not interfere with any of the dedicated signals for T_i :

$$f_{i,i-1}x^{\gamma_{i-1}}e_{i,i-1}x^{p_{i-2,i-1}} \neq f_{i,i+1}x^{\gamma_i}e_{i,i+1}x^{p_{i,i+1}} \mod (x^n - 1), \tag{4.124}$$

$$f_{i,i-1}x^{\gamma_{i-1}}e_{i,i-1}x^{p_{i,i-1}} \not\equiv f_{i,i+1}x^{\gamma_i}e_{i,i+1}x^{p_{i+2,i+1}} \mod (x^n - 1).$$
(4.125)

4.3.2 Upper Bounds

For a fixed messaging matrix M the minimal number of dimensions necessary is bounded by (cf. the lower bound in (3.38)):

$$n \ge \max_{m_{ji}} \left(\sum_{i=1}^{K} m_{ji} + \sum_{j=1}^{K} m_{ji} - m_{ji} \right).$$
(4.126)

In other words, the lower bound on n is determined by the row j and column i of messaging matrix M that maximizes the sum in (4.126). For $K \ge 3$, the given messaging matrix M will always provide n = 2 for each $i, j \in \mathcal{K}$ and M = 2K, so that:

$$DoF \le \frac{M}{n} = \frac{2K}{2} = K.$$
 (4.127)

Note that the case K = 2 corresponds to the elementary two-way relay channel which is also covered by this bound, since for $\mathbf{M} = \mathbf{1}_{2\times 2} - \mathbf{I}_{2\times 2}$, (4.126) provides n = 1 and hence the upper bound yields $\text{DoF} \leq \frac{M}{n} = \frac{2}{1} = 2$.



Figure 4.7: UL-phase: Signal alignment (solid arrows) of dedicated signals and interference alignment (dashed arrows) is performed at each relay R_i . DL-phase: Superimposed dedicated messages at T_i are allocated by

multiple-access and the self-interference is successively cancelled at T_i .

4.3.3 Achievability

In order to achieve the upper bound on the DoF as stated above, we propose a two-way relaying scheme including both network-coded cyclic SA and cyclic IA.

UL-phase: We consider the alignment scheme at the receiving relay R_i for all $i \in \mathcal{K}$. The two dedicated signals to be exchanged between T_i , T_{i+1} are aligned at R_i by SA:

$$e_{i+1,i}x^{p_{i+1,i}} \equiv e_{i,i+1}x^{p_{i,i+1}} \mod (x^n - 1).$$
(4.128)

The dedicated signals, superimposed at R_i , yield a network-coded messages, i.e., $W_{i+1,i} + W_{i,i+1}$. Likewise, the corresponding interference at R_i is aligned by IA:

$$e_{i+1,i}x^{p_{i-1,i}} \equiv e_{i,i+1}x^{p_{i+2,i+1}} \mod (x^n - 1).$$
(4.129)

DL-phase: We now consider the alignment scheme at the receiving user T_i for all $i \in \mathcal{K}$. The interference aligned in (4.129) is zero-forced by (4.118) with the zero-forcing particular polynomial:

$$z_i(x) = e_{i,i+1} x^{p_{i,i+1}} \equiv e_{i+1,i} x^{p_{i+1,i}} \mod (x^n - 1).$$
(4.130)

As a result, no inter-user interference is forwarded by R_i , so that the inter-user interference conditions (4.124) and (4.125) always hold. This is in contrast to the SA and IA schemes for the Y-channel in [77], where the inter-user interference is yet forwarded by the relays, but zero-forced at the destinations. The remaining dedicated networkcoded signals from T_{i-1} to T_i are forwarded by R_{i-1} and the signals from T_{i+1} to T_i are forwarded by R_i .

This is a multicast problem, since each relay demands to convey a single albeit network-coded message to two transceivers simultaneously. The two forwarded signals received at T_i must yet satisfy the multiple-access conditions (4.122). To satisfy these conditions, a tuple of feasible parameters $(\gamma_1, ..., \gamma_K)$ is to be determined, if it exists.

 \mathbf{T}_i also receives back-propagated self-interference from \mathbf{R}_{i-1} and from \mathbf{R}_i . We apply self-interference cancellation at each \mathbf{T}_i to remove the corresponding self-interference $W_{i-1,i}$ and $W_{i+1,i}$. Then, only the dedicated signals $W_{i,i-1}$ and $W_{i,i+1}$ remain, and can be decoded interference-free, if they satisfy the multiple-access conditions.

Case 1 - Open-Loop: Line Network

We first discuss the simpler case of the *open-loop line network*, discarding relay R_K w.l.o.g., so that the links between T_1 and T_K are severed and the parameters $p_{K,1}$, $p_{1,K}$ are neglected. In Figure 4.6, the affected links are highlighted in grey.

Theorem 4.5. Cyclic IA achieves K - 1 DoF on the given line network of K transceivers and K - 1 relays a number of n = 2 dimensions.

Proof:

(a) Necessity of $n \ge 2$:

The multiple-access interference conditions (4.122) demand that two signals are decodable in the DL at each T_i , $i \in \mathcal{K}$, so that $n \ge 2$ is clearly necessary. The edge-transceivers T_1 and T_K are an exception and demand only $n \ge 1$, accordingly.

(b) Sufficiency of Cyclic IA with n = 2:

UL-phase: We may fix the UL-transmission parameter $p_{2,1}$ at T_1 w.l.o.g. At T_2 , we use the given SA in (4.128), to obtain the parameter $p_{1,2}$. As T_2 also has a dedicated message for transmission to T_3 , we fix the parameter $p_{3,2}$ at T_2 , satisfying the intrauser interference conditions in (4.123). At T_3 , we use the given SA in (4.128) and the IA-condition in (4.129) to compute the parameters $p_{2,3}$ and $p_{4,3}$, respectively. Both satisfy the intra-user interference conditions in (4.123).

This is an iterative allocation procedure and it is analogously performed for each subsequent user $i \leq K - 1$. At T_K , only the SA of (4.128) is needed to determine the parameters $p_{K-1,K}$ as $p_{1,K} = 0$ produces no interference at T_K . There are no constraints on the channel matrix \boldsymbol{E} of the UL and each parameter can be determined uniquely.

DL-phase: The zero-forcing polynomial of \mathbb{R}_1 in (4.117) only contains the dimension of the dedicated signals, i. e., $z_1(x) \equiv e_{1,2}x^{p_{1,2}} \equiv e_{2,1}x^{p_{2,1}}$. The other coefficients in $z_1(x)$ are zero. As a result, the content of the complementary offset $e_{1,2}x^{p_{3,2}}$, with the interfering message $W_{2,1}$ in $r_1(x)$, is removed by $r_1(x) \circ z_1(x)$.

We can fix the parameter γ_1 for \mathbb{R}_1 w.l.o.g. Due to the multiple-access interference conditions in (4.122), the signal from \mathbb{R}_1 to \mathbb{T}_2 may not align to the forwarded signal from \mathbb{R}_2 to \mathbb{T}_2 , i. e., $x^{\gamma_1} f_{2,1} \neq x^{\gamma_2} f_{2,3}$ must hold. A unique solution exists for γ_2 if n = 2. Analogously, the γ_i for all other $3 \leq i \leq K - 1$ can also be determined uniquely. There are no constraints on the channel matrix \mathbf{F} of the DL either. The back-propagated self-interference is known at the transceivers and cancelled from the received signal.

As each T_i , with $2 \le i \le N - 1$, sends two dedicated messages, and the edge-transceivers T_1 and T_K send one dedicated message each, a total number of 2K-2 messages is conveyed interference-free over n = 2 dimensions, yielding $\frac{2K-2}{2} = K-1$ DoF.

Case 2 - Closed-Loop: Ring Network

In this case, we consider the *closed-loop ring network* as depicted in Figure 4.6 (including the grey links) with an active relay R_K and non-zero messages between T_1 and T_K . We observe that this network imposes a constraint on \boldsymbol{E} and \boldsymbol{F} .

Furthermore, UL and DL channels are called *reciprocal*, if:

$$e_{i,j}^{-1} \equiv f_{j,i} \mod (x^n - 1), \tag{4.131}$$

holds for all $i \neq j \in \mathcal{K}$.

Theorem 4.6. A cyclic IA/SA scheme achieves $DoF \le K$ for n = 2 dimensions on the ring network with K transceiver and K relays, if the following UL- and DL-conditions for the channel matrices hold:

$$\prod_{i=1}^{K} e_{i+1,i}^{-1} e_{i,i+1} \equiv 1 \mod (x^n - 1), \tag{4.132}$$

$$\prod_{i=1}^{K} f_{i,i+1}^{-1} f_{i+1,i} \equiv 1 \mod (x^n - 1).$$
(4.133)

Proof:

(a) Necessity of $n \ge 2$:

Cf. proof in Theorem 4.6(a), but without the edge-transceivers.

(b) Necessity of the UL-condition (4.132): For a fixed p_{1K} in the SA-condition (4.128), we obtain:

$$x^{p_{1,K}} \equiv e_{1,K}^{-1} e_{K,1} x^{p_{K,1}}.$$
(4.134)

Then, we substitute the parameter $x^{p_{K,1}}$ by $x^{p_{K,1}} \equiv e_{2,1}^{-1}e_{1,2}x^{p_{3,2}}$ from the IA-condition (4.129) into (4.134):

$$x^{p_{1,K}} \equiv e_{1,K}^{-1} e_{K,1} e_{2,1}^{-1} e_{1,2} x^{p_{3,2}}.$$

In the next step, $x^{p_{3,2}}$ is substituted by using the SA-condition (4.128). By a simple unrolling of the dependencies with an alternating application of the IA and SA conditions in (4.128) and (4.129), for all 1 < i < K, we obtain an additional product-term of $e_{i,i+1}^{-1}e_{i+1,i}$ for each *i*. In the last step, $x^{p_{1,K}}$ appears on both sides of the congruence. Clearly, the congruence is only true if the UL-condition (4.132) on E holds.

(c) Sufficiency of Cyclic SA and IA in the UL with n = 2:

The achievable scheme is analogous to Theorem 4.5 for the allocation of the parameter values $p_{i,i+1}$ and $p_{i+1,i}$ for all $i \in \{1, 2, ..., K-1\}$ for n = 2. A feasible solution for the two remaining parameters $p_{K,1}$ and $p_{1,K}$, satisfying the separability conditions, only exists if (4.132) holds.

(d) Necessity of the DL-condition (4.133):

For an arbitrary DL-matrix \boldsymbol{F} , a reciprocal UL-matrix $\boldsymbol{E}_{\rm rpr}$ can be computed easily by (4.131). If $\boldsymbol{E}_{\rm rpr}$ satisfies the UL-condition (4.132), then \boldsymbol{F} will satisfy (4.133). As already given by parts (a), (b), and (c) of this proof, a feasible cyclic SA/IA scheme for the reciprocal UL with $\boldsymbol{E}_{\rm rpr}$ exists. For the multicast transmission in the reciprocal DL channel \boldsymbol{F} , the superimposed dedicated signals are send back over exactly the same dimensions over which they have been received in the reciprocal UL with $\boldsymbol{E}_{\rm rpr}$. As the intra-user interference conditions already hold in the reciprocal UL, the multipleaccess conditions hold analogously in the DL by channel reciprocity. And conversely, if (4.133) is violated, the parameters γ_i can not be determined uniquely for all i = 1, ..., K.

(e) Sufficiency of Cyclic SA in the DL with n = 2:

The inter-user interference conditions (4.124), (4.125) always hold since the inter-user interference is removed at each R_i by using the corresponding filtering polynomial. Now, the superimposed dedicated signals are multicast by each R_i and received at T_i and T_{i+1} . We may fix parameter γ_1 . Again, $x^{\gamma_i}f_{i+1,i} \neq x^{\gamma_i}f_{i+1,i+1}$ must hold for all γ_i with i = 1, ..., K, to ensure that the multiple-access conditions are not violated. This is valid for any F that satisfies (4.133). In the last step, the back-propagated self-interference is cancelled. Altogether, a total number of $\frac{2K}{2} = K$ DoF is achieved.

If the given UL- and DL-matrices are reciprocal, the conditions (4.132) and (4.133) become equivalent.

In case that at least one condition of Theorem 4.6 does not hold, the first straightforward approach is to demand at least n = 3 dimensions. Then, only a total number of $\frac{2K}{3}$ DoF is achievable. However a second and more efficient approach is to exploit the unconstrained Theorem 4.5 to achieve K-1 DoF by neglecting the dedicated messages between a single arbitrary pair (T_i, T_{i+1}) in both the UL and DL phases. For each transmission, a different pair of messages chosen by a scheduled round-robin scheme or chosen i.i.d. is neglected. The second approach clearly outperforms the first approach, since $\frac{2K}{3} < K - 1$ holds for K > 3.

4.3.4 User-Relay Duality

A dual network of the closed-loop case is physically the same network as the primary network with \boldsymbol{E} and \boldsymbol{F} , but with former users operating as relays and former relays operating as users instead. Although the indexation of such a dual network is not unique for the closed-loop case of the cascaded two-way channel, it is obvious that the resulting dual network is indeed unique, since the equivalent indexation do not change the setup of dedicated messages between neighbouring users. W.l.o.g., we may relabel $\mathbf{R}_i \to \mathbf{T}_i$ and $\mathbf{T}_{i+1} \to \mathbf{R}_i$, i.e., the labels are cyclically right-shifted by one position. Let $\tilde{\boldsymbol{Z}}_K$ denote a particular circulant matrix with $\tilde{z}_{K-1} = 1$, and all other $\tilde{z}_j = 0$. The UL and DL matrices of the dual network are cyclically rotated versions \boldsymbol{F} and \boldsymbol{E} from the primary network:

$$\boldsymbol{E}_{\text{dual}} = \tilde{\boldsymbol{Z}}_{K} \boldsymbol{F}, \qquad (4.135)$$

$$\boldsymbol{F}_{\text{dual}} = \boldsymbol{E} \tilde{\boldsymbol{Z}}_{K}^{\mathsf{T}}.$$
(4.136)

The corresponding dual UL and DL conditions yield:

$$\prod_{i=1}^{K} f_{i,i+1} f_{i+1,i}^{-1} \equiv 1 \mod (x^n - 1), \tag{4.137}$$

$$\prod_{i=1}^{K} e_{i+1,i} e_{i,i+1}^{-1} \equiv 1 \mod (x^n - 1).$$
(4.138)

Hence, E_{dual} satisfies the dual UL condition, if F satisfied the primary DL condition and F_{dual} satisfies the dual DL condition, if E satisfied the primary UL condition. Any cyclically right-shifted relabelling by an odd number of positions will yield the same dual conditions as given in (4.137) and (4.138). As a result, we observe that the following relationship:

User-Relay Duality: If a cyclic SA and IA scheme is feasible on the primary network, it will also be feasible on any dual network.

4.4 Summary

In this chapter, we have included full-duplex relays in order to extend the unidirectional communication schemes of the previous chapter and to enhance the achievable Degrees-of-Freedom.

First, we have considered an interference channel which was extended by two intermediate relays - a $2 \times 2 \times 2$ relay-interference channel. To achieve the cut-set upper bounds on the Degrees-of-Freedom, we have applied interference neutralization as introduced by Mohajer et al. in [47] in terms of the proposed cyclic polynomial channel model.

A particular gain of our provided problem description was the emphasis on the *interference-neutralization conditions* and *no-signal-neutralization conditions*. The approach has also allowed us to generalize the asymptotic MIMO interference alignment and neutralization scheme of Gou et al. in [63]. Afterwards, we have further equipped the users with perfect full-duplex capabilities and we have considered a two-way generalization of cyclic neutralization scheme. The given scheme has also incorporated the concept of *signal alignment* (dedicated signals are aligned), and of *back-propagated self-interference cancellation*.

In the last part of this chapter, we have focussed on a cascaded version of the wellknown two-way relay channel. A user-relay duality and a unicast-multicast duality, which was exploited for the optimal DoF-achieving communication scheme, has been discussed in particular.

5 Multi-Way Communications with Three Users

In this chapter, we pursue the following goals:

Characterizing DoF-achieving and capacity-achieving schemes for the 3-way channel and the 3-user *Y*-channel in terms of:

- The cyclic polynomial channel model (CPCM),
- the linear deterministic channel model¹ (LDCM), and
- the Gaussian MIMO channel model² (GMCM).

Device-to-Device Communications: 3-Way versus Y- Channels

A natural communication scenario with multiple users is multi-way device-to-device (D2D) conferencing. Especially in wireless multi-user communication networks, this will involve multiple simultaneous transmissions causing interference that impairs the maximal achievable data rates per user. The D2D approach in [80], [81], and [82] is intended to increase the spectral efficiency of multi-way networks without the utilization of base stations for data transmission (except for low-rate top-level control mechanisms). As a countermeasure to deal with the impairment caused by interference, all transmission signals must be carefully designed so that mutual interference is minimized.

The 3-way channel (or Δ -channel) is an instance of a D2D-communication network with 3 users mutually communicating to each other, i.e., each user transmits 2 messages, one per receiver. We assume that all participating users employ perfect full-duplex devices. Note that the related *conferencing* 3 *-way channel* in [83] considers a single message per user that is multicast to the two other receivers.

A 3-user Y- channel as in [77], [78], and [79], is a closely related 3-way communication system, but it has an intermediate relay and no direct links between the users. Each transceiver T_i sends two messages to the relay R, and the relay forwards three network-coded messages back to the dedicated users. The DoF of the MIMO 3-user Y- channel with an M_i antennas per user T_i (There is an equal number of transmit and receive antennas, but a different number of antennas among different users) and M_R relay antennas are provided in [78]. Its generalization to a MIMO 4-user Y- channel is considered in [84] and [85]. In [79], the capacity region of the related 3-user Y- channel in the LDCM is derived.

¹Parts of this work have been published in [34].

²Parts of this work have been published in [30] and [33].

The schemes proposed in the works on the Y-channel mainly rely on SA. Recall from the previous chapter, that SA is a particular adaptation of IA to multi-way relay networks, such that bidirectional dedicated signals (and not the interference signals) are aligned at the intermediate two-way relay.

Interestingly, the 3-way channel demands a combined scheme of various important communication techniques: IA, SA, IN, and backward decoding (BD).

Comparison of the GMCM and LDCM to the CPCM

Moreover, we have already pointed to several interesting properties that were also observed in the LDCM and the GMCM in the previous chapters: Up to now, we already enlightened the analogous representation of the upper bound on the DoF for $K_{\rm R} \times K_{\rm T} X$ - networks in Section 3.2, the problem of common eigenvectors in invariant subspaces discussed in Section 3.4.2 and 3.5.3, and the duality relationship of cyclic IAC and cyclic IN in Section 3.6.4, for instance. We will further elaborate conceptual connections between those different models within this chapter for the Y- channel and the 3-way channel.

5.1 Cyclic Interference Alignment for 3-Way and *Y*- Channels

We begin with the consideration of the 3-user Y-channel and the 3-way channel in terms of the CPCM. This approach serves us to provide a first-order understanding of the two aforementioned multi-way networks. Since the achievable scheme for the Y-channel is quite similar to the one in [79] for the LDCM, we would like to emphasize, that the provided scheme covers the case of arbitrary non-reciprocal shifts. Then, we investigate the Y- Δ product relationship as it occurs for the CPCM. To the best of our knowledge, such a product relationship has not been reported in terms of information theory for multi-user communications yet.

5.1.1 Cyclic Polynomial 3-User Y- Channel

In the Y-channel there are three users T_1, T_2 , and T_3 linked to an intermediate fullduplex relay R. However, there is no direct link between the three transceivers. This arrangement is also depicted in Figure 5.1. The set of user-indices is $\mathcal{K}_T = \mathcal{K} = \{1, 2, 3\}$. The number of submessages per dedicated user-pair (end-to-end) is given by the messaging matrix:

$$\boldsymbol{M} = \begin{pmatrix} 0 & m_{12} & m_{13} \\ m_{21} & 0 & m_{23} \\ m_{31} & m_{32} & 0 \end{pmatrix}.$$
 (5.1)

Since there are no messages to the transmitting user itself, the main diagonal is assigned with zero values. The total number of submessages amounts to:

$$M = m_{12} + m_{21} + m_{13} + m_{31} + m_{23} + m_{32}.$$



Figure 5.1: The Y-channel with three transceivers T_1, T_2 and T_3 and one intermediate relay R_1 , has six independent dedicated messages \boldsymbol{w}_{ji} . The influence of the channel is parameterized by the corresponding e_{Ri} and f_{jR} . The solid arrows describe the UL and the dashed arrows the DL subchannels.

The UL-phase denotes the transmission in the first hop from the transceivers T_i to the relay R. Signals transmitted from T_i are represented by a polynomial $u_i(x)$, with messages allocated to distinct offset parameters $p_{i}^{[i]} \in \mathbb{N}$:

$$u_i(x) \equiv \sum_{j=1, j \neq i}^3 u_{ji}(x),$$
 (5.2)

$$u_{ji}(x) \equiv \sum_{t=0}^{m_{ji}-1} W_{ji}^{[t]} x^{p_{ji}^{[t]}}.$$
(5.3)

The UL channel vector is $e = (e_{R1}, e_{R2}, e_{R3})$, with $e_{R1}, e_{R2}, e_{R3} \in \mathcal{D}$. At relay R, the received signal is a shifted superposition of the three signals from all T_i :

$$r_{\mathrm{R}}(x) \equiv \boldsymbol{e}\boldsymbol{u}^{\mathsf{T}} \equiv \sum_{i=1}^{3} e_{\mathrm{R}i} u_{i}(x).$$
(5.4)

The DL-phase denotes the transmission from relay R back to the three transceivers T_j in the second hop. The DL channel is denoted by a vector $\mathbf{f} = (f_{1R}, f_{2R}, f_{3R})$ with the entries $f_{1R}, f_{2R}, f_{3R} \in \mathcal{D}$. The relay forwards its previously received superimposed signals back to all transceivers:

$$\boldsymbol{r}^{\mathsf{T}} \equiv \boldsymbol{f}^{\mathsf{T}} \boldsymbol{r}_{\mathsf{R}}(x) \equiv (f_{1R}, f_{2R}, f_{3R})^{\mathsf{T}} \boldsymbol{r}_{\mathsf{R}}(x).$$
(5.5)

The two channel vectors e and f are independent w.l.o.g. With the perfect full-duplex operation, the UL-phase and DL-phase is performed simultaneously in each time-step. But due to causality in time, a DL transmission is delayed by one time-instant w.r.t. its previous UL transmission. For distinct user-indices $i, j, k \in \mathcal{K}$ and submessage-indices t and t', the separability conditions of the 3-user Y- channel are given as follows:

Multiple-access interference conditions, for $t \in \{0, ..., m_{ji} - 1\}$, $t' \in \{0, ..., m_{jk} - 1\}$:

$$e_{\rm Ri} f_{j\rm R} x^{p_{ji}^{[t]}} \neq e_{\rm Rk} f_{j\rm R} x^{p_{jk}^{[t']}} \mod (x^n - 1)$$
(5.6)

$$\Rightarrow e_{\mathrm{R}i} x^{p_{j_i}^{[t]}} \neq e_{\mathrm{R}k} x^{p_{j_k}^{[t']}} \mod (x^n - 1), \tag{5.7}$$

intra-user interference conditions, for $t \in \{0, ..., m_{ji} - 1\}, t' \in \{0, ..., m_{ki} - 1\}$:

$$x^{p_{ji}^{[t]}} \not\equiv x^{p_{ki}^{[t']}} \mod (x^n - 1), \tag{5.8}$$

inter-user interference conditions, for $t \in \{0, ..., m_{ji} - 1\}$, $t' \in \{0, ..., m_{ik} - 1\}$:

$$e_{\mathrm{R}i} f_{j\mathrm{R}} x^{p_{j_{i}}^{[t]}} \not\equiv e_{\mathrm{R}k} f_{j\mathrm{R}} x^{p_{i_{k}}^{[t']}} \mod (x^{n} - 1),$$

$$\Rightarrow e_{\mathrm{R}i} x^{p_{j_{i}}^{[t]}} \not\equiv e_{\mathrm{R}k} x^{p_{i_{k}}^{[t']}} \mod (x^{n} - 1)$$
(5.9)

Before approaching an optimal communication scheme for the Y- channel, we introduce the closely related 3-way channel.

5.1.2 Cyclic Polynomial 3-Way Channel

For the 3-way channel, the messaging matrix M is identical to (5.1). However, there are only direct links between the users and there is no intermediate relay involved. It has only one channel matrix which is defined by $D = (d_{ji})_{1 \le j, i \le 3}$ with independent elements $d_{ji} \in D$. The diagonal elements of D are zero. Such a 3-way channel is depicted in Figure 5.2.



Figure 5.2: The 3-way channel with three transceivers T_1, T_2 and T_3 has six independent dedicated messages \boldsymbol{w}_{ji} . The influence of the channel is parameterized by the corresponding d_{ji} .

In this case, the received signal at T_j , $j \in \mathcal{K}$, is a superposition of shifted polynomials $u_i(x)$:

$$r_j(x) \equiv \sum_{i=1}^3 d_{ji} u_i(x) \mod (x^n - 1).$$
 (5.10)

We can express this compactly by $\mathbf{r}^{\mathsf{T}} \equiv \mathbf{D}\mathbf{u}^{\mathsf{T}} \mod (x^{n}-1)$, with $\mathbf{r} = (r_{1}(x), r_{2}(x), r_{3}(x))$ and $\mathbf{u} = (u_{1}(x), u_{2}(x), u_{3}(x))$.

In analogy to the 3 user Y- channel, we also define the set of separability conditions with pairwise distinct $i, j, k \in \mathcal{K}$:

Multiple-access interference conditions for $t \in \{0, ..., m_{ji} - 1\}$, $t' \in \{0, ..., m_{jl} - 1\}$:

$$d_{ji}x^{p_{ji}^{[t]}} \not\equiv d_{jl}x^{p_{jl}^{[t']}} \mod(x^n - 1), \tag{5.11}$$

intra-user interference conditions for $t \in \{1, ..., m_{ji} - 1\}, t' \in \{1, ..., m_{ki} - 1\}$:

$$x^{p_{ji}^{[t]}} \not\equiv x^{p_{ki}^{[t']}} \mod(x^n - 1), \tag{5.12}$$

inter-user interference conditions for $t \in \{0, ..., m_{ji} - 1\}, m' \in \{0, ..., m_{ki} - 1\}$:

$$d_{ji}x^{p_{ji}^{[t]}} \not\equiv d_{jl}x^{p_{ki}^{[t']}} \mod(x^n - 1)$$
(5.13)

Having defined the channel models and the separability conditions of the Y- channel and the 3-way channel, we are now ready to investigate their upper bounds on the DoF.

5.1.3 Upper Bounds

In the case of a 3-way channel, we consider $K_{\rm T} = K_{\rm R} = 3$ and hence $\mathcal{K} = \{1, 2, 3\}$. At each receiver T_j , the minimal necessary number of dimensions is lower bounded by $\max(n_1, n_2, n_3)$ for $n_j = m_{ji} + m_{jk} + \max(m_{ik}, m_{ki})$. The two dedicated messages received at T_j must demand at least $m_{ji} + m_{jk}$ dimensions. The intra-user interference of both transmitters T_i and T_j must cover at least $n \ge \max(m_{ik}, m_{ki})$ dimensions, since each T_i must allocate its two different dedicated signals linearly independent, such that $m_{ji} + m_{ki}$ dimensions is necessary. This leads to the following expression for the upper bound on the DoF:

$$DoF \le \frac{m_{12} + m_{21} + m_{13} + m_{31} + m_{23} + m_{32}}{\max(n_1, n_2, n_3)},$$
(5.14)

for distinct $i, j, k \in \mathcal{K}$. Note that this term coincides with the general upper bound on the DoF for $K_{\mathrm{T}} \times K_{\mathrm{R}} X$ -channels (cf. (3.38)) for the specific M given in (5.1):

$$\text{DoF} \le \frac{m_{12} + m_{21} + m_{13} + m_{31} + m_{23} + m_{32}}{\max_{i\neq j\in\mathcal{K}} \left(\sum_{k=1,k\neq j}^{3} m_{jk} + \sum_{l=1,l\neq i}^{3} m_{li} - m_{ji}\right)}$$

For the 3-user Y-channel, the upper bound is actually the same: The transmitters T_i and T_j must satisfy the multiple-access conditions, and transmit $m_{ji} + m_{ki}$ and $m_{ij} + m_{kj}$ submessages, respectively. At relay R, the signals dedicated to be forwarded to T_k must be distinct from the intra-user interference from T_i and T_j . These dedicated signals must cover at least $m_{ki}+m_{kj}$ dimensions to be linearly decodable. The intra-user interference of T_i and T_j is aligned at R and covers at least $\max(m_{ji}, m_{ij})$ dimensions. Then, relay R must forward and broadcast all received messages to all three users, since each submessage is dedicated for one receiver. This does not preclude that the received signals at T_i may be interfered with back-propagated self-interference. Hence, in order to provide a decodable signal for T_k , a number of $n_k = m_{ki}+m_{kj}+\max(m_{ji}, m_{ij})$ dimensions is necessary. Combining these conditions for all users $k \in \mathcal{K}$, we obtain the lower bound:

$$n \ge \max(n_1, n_2, n_3),$$
 (5.15)

which is obviously the same lower bound on the dimensions n as for the 3-way channel.

5.1.4 Achievable Scheme for the Y- Channel

First, we investigate an achievable scheme for the non-reciprocal 3-user Y-channel with general message lengths. A closely related scheme for a reciprocal Y-channel in the LDCM is given in [79]. Both schemes are mainly based on SA and LEaD.

Theorem 5.1. On the 3-user Y-channel, a cyclic SA scheme achieves:

$$\mathrm{DoF} \leq \frac{m_{12} + m_{21} + m_{13} + m_{31} + m_{23} + m_{32}}{\max(n_1, n_2, n_3)} \leq 2,$$

within $n \ge \max(n_1, n_2, n_3)$ dimensions for distinct $i, j, k \in \mathcal{K}$ for:

$$n_i = m_{ii} + m_{ik} + \max(m_{ik}, m_{ki}).$$

Proof:

(a) Necessity of $n \ge \max(n_1, n_2, n_3)$ dimensions:

The lower bound on the number of dimensions n has already been provided in Section 5.1.3.

(b) Sufficiency of the Cyclic SA scheme: In the UL-phase, the signals from each T_i are aligned at R within 3 main blocks:

A) Bidirectional: There are non-zero pairs of m_{ii}, m_{ij} for two users $i \neq j$.

- **B)** Cyclic: There are non-zero triples m_{ii}, m_{ik}, m_{ki} for three distinct users $i, j, k \in \mathcal{K}$.
- C) Unidirectional: None of the above cases holds.

The communication scheme for the signal alignment at the relay R is briefly outlined as follows. For submessages satisfying (a), the bidirectional communication scheme allocates two submessages per dimension. Then, the remaining submessages satisfying (b) are allocated by the cyclic communication scheme with $\frac{3}{2}$ submessages per dimension. The yet remaining submessages are allocated by the unidirectional scheme with one submessage per dimension.

In the DL, these received signals aligned at R are simply forwarded to all T_i . The T_i use linear decoders and self-interference cancellation to resolve their dedicated signals.

Uplink

A) Bidirectional Communication

We define three parameters a_1 , a_2 , a_3 , and their cumulative sum a, for distinct indices $i, j, k \in \mathcal{K}$:

$$a_i = \min(m_{ik}, m_{ki}), \tag{5.16}$$

$$a = a_1 + a_2 + a_3. \tag{5.17}$$

The transmitted signals of each T_i are aligned in bidirectional pairs of submessages with the following offsets of relay R:

offset SA at Relay R index t

$$x^t: e_{R2}x^{p_{32}^{[t]}} \equiv e_{R3}x^{p_{23}^{[t]}} \mod (x^n - 1), \quad t = 0, ..., a_1 - 1,$$

 $x^{a_1+t}: e_{R1}x^{p_{31}^{[t]}} \equiv e_{R3}x^{p_{13}^{[t]}} \mod (x^n - 1), \quad t = 0, ..., a_2 - 1,$
 $x^{a_1+a_2+t}: e_{R1}x^{p_{21}^{[t]}} \equiv e_{R2}x^{p_{12}^{[t]}} \mod (x^n - 1), \quad t = 0, ..., a_3 - 1.$

This allocation provides a unique choice for the subset the parameters $p_{ji}^{[t]}$ with the given ranges of t.

The matrix ${\cal M}'$ contains the residual submessages yet to be allocated and is denoted by:

$$M' = M - A, \tag{5.18}$$

with the symmetric matrix:

$$\boldsymbol{A} = \begin{pmatrix} 0 & a_3 & a_2 \\ a_3 & 0 & a_1 \\ a_2 & a_1 & 0 \end{pmatrix}.$$
 (5.19)

After this allocation procedure, there are at most three non-zero elements left over in M'.

B) Cyclic Communication

For two mutually exclusive cases of M', a cyclic communication scheme is utilized. We define two parameters b_1 , b_2 , and their cumulative sum b:

$$b_1 = \min(m'_{21}, m'_{13}, m'_{32}), \tag{5.20}$$

$$b_2 = \min(m'_{12}, m'_{23}, m'_{31}), \tag{5.21}$$

$$b = b_1 + b_2, \tag{5.22}$$

The term *cyclic* communication is due to the clock-wise/counter clock-wise cyclic indexation of the indices. Parameter $b_1 > 0$ indicates clock-wise cyclic communication, and $b_2 > 0$ indicates counter clock-wise cyclic communication. From the bidirectional communication applied before in scheme A), these two cases exclude each other:

$$\text{if } b_1 = b > 0 \Rightarrow b_2 = 0, \tag{5.23}$$

$$if b_2 = b > 0 \Rightarrow b_1 = 0, \tag{5.24}$$

otherwise
$$b_1 = b_2 = 0.$$
 (5.25)

For clock-wise cyclic communication, we align the signals at the offsets of R by:

offset SA at Relay R index t

$$x^{a+t}: e_{R1}x^{p_{21}^{[a_3+t]}} \equiv e_{R2}x^{p_{32}^{[a_1+t]}} \mod (x^n - 1), \quad t = 0, ..., b_1 - 1,$$

 $x^{a+b_1+t}: e_{R2}x^{p_{13}^{[a_2+t]}} \equiv e_{R2}x^{p_{32}^{[a_1+t]}} \mod (x^n - 1), \quad t = 0, ..., b_1 - 1.$

For counter clock-wise cyclic communication, we similarly align:

offset SA at Relay R index t

$$x^{a+t}: e_{R2}x^{p_{13}^{[a_3+t]}} \equiv e_{R1}x^{p_{31}^{[a_2+t]}} \mod (x^n - 1), \quad t = 0, ..., b_2 - 1,$$

 $x^{a+b_2+t}: e_{R2}x^{p_{23}^{[a_1+t]}} \equiv e_{R2}x^{p_{31}^{[a_2+t]}} \mod (x^n - 1), \quad t = 0, ..., b_2 - 1.$

In both of these cyclic communication schemes, the inter-user interference conditions seem to be violated. However, the receivers can linearly decode the dedicated signals in the DL using known back-propagated self-interference for cancellation. We will briefly elaborate this decoding scheme when discussing the DL-phase of this scheme. With the circulant matrix \boldsymbol{B} for (either clock-wise or counter clock-wise) cyclic communication:

$$\boldsymbol{B} = \begin{pmatrix} 0 & b_2 & b_1 \\ b_1 & 0 & b_2 \\ b_2 & b_1 & 0 \end{pmatrix},$$
(5.26)

the matrix M'' of the residual submessages yet to be allocated is:

$$M'' = M' - B = M - A - B.$$
 (5.27)

C) Unidirectional Communication

There are still at most 3 non-zero submessages in M'', if scheme b = 0, or at most 2 non-zero submessages, if b > 0. Based on the previous allocation, there are six possible cases with non-zero submessages that are not yet covered by the previous schemes A and B. We define the six parameters for unidirectional communication and their cumulative sum by:

Case 1:
$$c_{1,12} = m_{12}'', c_{1,13} = m_{13}'', c_{1,23} = m_{23}'' \Rightarrow b_2 \ge 0, b_1 = 0$$
 (5.28)

Case 2:
$$c_{2,12} = m_{12}'', c_{2,13} = m_{13}'', c_{2,32} = m_{32}'' \Rightarrow b_1 \ge 0, b_2 = 0$$
 (5.29)

Case 3:
$$c_{3,21} = m_{21}'', \ c_{3,23} = m_{23}'', \ c_{3,13} = m_{13}'' \Rightarrow b_1 \ge 0, b_2 = 0$$
 (5.30)

Case 4:
$$c_{4,21} = m_{21}'', c_{4,23} = m_{23}'', c_{4,31} = m_{31}'' \Rightarrow b_2 \ge 0, b_1 = 0$$
 (5.31)

Case 5:
$$c_{5,31} = m_{31}'', c_{5,32} = m_{32}'', c_{5,12} = m_{12}'' \Rightarrow b_2 \ge 0, b_1 = 0$$
 (5.32)

Case 6:
$$c_{6,31} = m_{31}'', \ c_{6,32} = m_{32}'', \ c_{6,21} = m_{21}'' \Rightarrow b_1 \ge 0, b_2 = 0,$$
 (5.33)

$$c = \sum_{i=1}^{6} c_{i,12} + c_{i,13} + c_{i,21} + c_{i,23} + c_{i,31} + c_{i,32}.$$
(5.34)

These six cases are pairwise exclusive. The non-zero submessages in M'' are each allocated to an individual frame in a multiple-access scheme. E.g., in *Case 1*, we align:

offset R index t

$$x^{a+2b+t}: e_{R2}x^{p_{12}^{(a_3+b+t)}}, \quad t = 0, ..., c_{1,12} - 1, \\
 x^{a+2b+c_{1,12}+t}: e_{R3}x^{p_{13}^{(a_2+b+t)}}, \quad t = 0, ..., c_{1,13} - 1, \\
 x^{a+2b+c_{1,12}+c_{1,13}+t}: e_{R3}x^{p_{23}^{(a_1+b+t)}}, \quad t = 0, ..., c_{1,23} - 1.$$
(5.35)

Note that c has at most three non-zero components in each case, e.g., $c = c_{1,12} + c_{1,13} + c_{1,23}$ in *Case 1*. The other five cases for the unidirectional communication scheme are similarly allocated at R within a frame of c dimensions. We consider all six cases in a combined view, and define the matrix C for unidirectional communication by:

$$\boldsymbol{C} = \begin{pmatrix} 0 & c_{1,12} + c_{2,12} + c_{5,12} & c_{1,13} + c_{2,13} + c_{3,13} \\ c_{6,21} + c_{4,21} + c_{3,21} & 0 & c_{4,23} + c_{1,23} + c_{3,23} \\ c_{6,31} + c_{4,31} + c_{5,31} & c_{6,32} + c_{2,32} + c_{5,32} & 0 \end{pmatrix}.$$
 (5.36)

All submessages in the original messaging matrix M are completely allocated, if we set C = M''. The auxiliary matrices of all three schemes are combined to:

$$\boldsymbol{M} = \boldsymbol{A} + \boldsymbol{B} + \boldsymbol{C}. \tag{5.37}$$

We expand n in its explicit form so that it is the maximum of all 6 lower bounds:

$$n = \max(m_{12} + m_{13} + m_{23}, m_{12} + m_{13} + m_{32}, m_{21} + m_{23} + m_{13}, m_{21} + m_{23} + m_{13}, m_{31} + m_{32} + m_{12}, m_{31} + m_{32} + m_{21}).$$
(5.38)

It remains to show that the separability conditions of the Y-channel hold. Since each T_i aligns its transmitted submessages to specific dimensions at R only, we do not have to take signals into account, that are aligned at other receivers. Hence, the corresponding transmit signals can be easily derived from the (aligned) received signal at r(x).

As both the aligned interference signals and the dedicated signals are distinctly allocated within successively arranged frames at R, valid transmission signals always exist. The separability conditions hold by construction, if the total number of used dimensions satisfies the lower bound of n.

We now show that the proposed scheme achieves the upper bound on the DoF:

$$\begin{split} n &= \min(m_{23}, m_{32}) + \min(m_{13}, m_{31}) + \min(m_{12}, m_{21}) + \\ &2\min(m'_{21}, m'_{13}, m'_{32}) + 2\min(m'_{12}, m'_{31}, m'_{23}) + \\ & \begin{pmatrix} m''_{12} + m''_{13} + m''_{23}, & \text{if Case 1 holds} \\ m''_{12} + m''_{13} + m''_{32}, & \text{if Case 2 holds} \\ m''_{21} + m''_{23} + m''_{13}, & \text{if Case 3 holds} \\ m''_{21} + m''_{23} + m''_{13}, & \text{if Case 4 holds} \\ m''_{31} + m''_{32} + m''_{21}, & \text{if Case 5 holds} \\ m''_{31} + m''_{32} + m''_{21}, & \text{if Case 6 holds} \\ &= a_1 + a_2 + a_3 + 2b_1 + 2b_2 + \\ & \begin{pmatrix} m_{12} - a_3 - b_2 + m_{13} - a_2 - b_1 + m_{23} - a_1 - b_2, & \text{if Case 1 holds} & (b_1 = 0) \\ m_{12} - a_3 - b_2 + m_{13} - a_2 - b_1 + m_{32} - a_1 - b_1, & \text{if Case 2 holds} & (b_2 = 0) \\ m_{21} - a_3 - b_1 + m_{23} - a_1 - b_2 + m_{13} - a_2 - b_1, & \text{if Case 3 holds} & (b_1 = 0) \\ m_{31} - a_2 - b_2 + m_{32} - a_1 - b_1 + m_{12} - a_3 - b_2, & \text{if Case 5 holds} & (b_1 = 0) \\ m_{31} - a_2 - b_2 + m_{32} - a_1 - b_1 + m_{12} - a_3 - b_2, & \text{if Case 6 holds} & (b_1 = 0) \\ m_{31} - a_2 - b_2 + m_{32} - a_1 - b_1 + m_{12} - a_3 - b_1, & \text{if Case 6 holds} & (b_2 = 0) \\ &= \max(m_{12} + m_{13} + m_{23}, m_{12} + m_{13} + m_{32}, m_{21} + m_{23} + m_{13}, \\ m_{21} + m_{23} + m_{31}, m_{31} + m_{32} + m_{12}, m_{31} + m_{32} + m_{21}). \end{split}$$

In other words, the number of dimensions n used for the proposed communication scheme is sufficient to convey all submessages in M for all six cases in (5.38).

Downlink

The received signal r(x) of the UL at R is forwarded over the subchannels $f_{i\rm R}$ to all receivers T_i . The signals received at each T_i are only cyclically shifted versions

of the forwarded signal r(x) without further superimposed interference from other transmitters. Each receiver T_i cancels its back-propagated self-interference and also cancels the aforementioned interference from the cyclic communication.

For instance, T₁ cancels the known self-interference $e_{R3}x^{p_{21}^{[a_3+t]}}$ and decodes the remaining non-dedicated signal $e_{R2}x^{p_{32}^{[a_1+t]}}$ for $t = 0, ..., b_1 - 1$. Next, this decoded interference signal is cancelled from the other received signal $e_{R2}x^{p_{13}^{[a_2+t]}} + e_{R2}x^{p_{32}^{[a_1+t]}}$, so that the dedicated signal $e_{R2}x^{p_{13}^{[a_2+t]}}$ remains and can be decoded successfully. The other users perform analogous procedures for the interference in the cyclic communication schemes.

Altogether, it is shown that the given upper bound on the DoF is achievable. The DoF are only maximized if the bidirectional scheme A) can already align all dedicated messages. Hence, if $m_{23} = m_{32} = m_1, m_{13} = m_{31} = m_2$ and $m_{12} = m_{21} = m_3$ hold, we obtain $n = n_1 = n_2 = n_3 = m_1 + m_2 + m_3$, so that at most $\frac{2(m_1+m_2+m_3)}{m_1+m_2+m_3} = 2$ DoF are achievable.

5.1.5 Δ -Y Transformation

We have already seen that the upper bounds on the DoF of the considered 3-user Y-channel and the 3-way channel coincide exactly, even for arbitrary message lengths in M. The upper bounds for the 3-user Y-channel have been shown to be achievable in the theorem above. In the following, we will show that the upper bounds of the 3-way channel are also achievable.

From the perspective of an electrical engineer, a similar equivalence occurs in wired resistor-networks to some extend. There is a so-called Δ -Y transformation³ which is a well-known method from elementary circuit theory as, e.g., in [86], [87]. The basic statement of this transformation is that the external resistance behaviour of a Δ -shaped resistor circuit can be equivalently described by a Y-shaped resistor circuit and vice versa.

In this section, we explore a conceptually similar Δ -Y transformation for the 3-user Y-channel and the 3-way channel. To the best of our knowledge, such a transformation has not been considered in terms of multi-user interference channels in the literature yet.

To elaborate the transformation, we commence with considering a Y-channel of three users \tilde{T}_1 , T_2 , T_3 and a relay R. In the first step, this channel is extended by an additional bidirectional link between two users, e.g., T_2 and T_3 . The shape of a 3-way channel occurs between T_2 , R and T_3 . Next, we unite user \tilde{T}_1 and relay R to a single device T_1 by permitting full cooperation for them. Now, each subchannel from T_i to T_j is renamed to d_{ji} , correspondingly. As a result, we obtain the 3-way channel.

This transformation procedure is depicted in Figure 5.3. We assume that the (virtual) subchannel between the relay R and \tilde{T}_1 imposes no further shift w.l.o.g. The subchannels d_{23} and d_{32} represent the main difference between an ordinary Y-channel and this extended Y-channel.

³This transformation is also known as Π -T transform in [86], [87].


Figure 5.3: The 3-way channel is transformed into an extended (upside-down) Y-channel including an additional bidirectional link between T_2 and T_3 . The additional link is drawn with dotted lines.

5.1.6 Achievable Scheme for the 3-Way Channel

Theorem 5.2. Cyclic SA on the 3-way channel achieves the upper bound:

$$\mathrm{DoF} \le \frac{m_{12} + m_{21} + m_{13} + m_{31} + m_{23} + m_{32}}{\max(n_1, n_2, n_3)} \le 2,$$

within $n \ge \max(n_1, n_2, n_3)$ for $n_j = m_{ji} + m_{jk} + \max(m_{ik}, m_{ki})$ with distinct $i, j, k \in \mathcal{K}$.

Proof:

(a) Necessity of $n \ge \max(n_1, n_2, n_3)$ dimensions:

The lower bound on the number of dimensions n is already provided in Section 5.1.3.

(b) Sufficiency of Cyclic IA for $n = \max(n_1, n_2, n_3)$:

We basically apply the same achievable scheme that was used for the 3-user Y-channel to the extended Y-channel. We only include some minor adaptations to compensate the additional interfering signals conveyed over d_{23} and d_{32} . We consider a finite communication scheme over N + 1 time-instants with N UL-transmissions and N DLtransmissions. The entire decoding procedure at the receivers is postponed to the last, i.e., the (N + 1)-th time-instant. The last time-instant is only a DL-transmission from R and there will be no additional interference over d_{23} and d_{32} , since the ULtransmitters of T₂ and T₃ are already silent. There are only two classes of additional interference over d_{23} and d_{32} to be analyzed:

- (a) The interference over d_{ij} , for $i \neq j \in \{2,3\}$, received at T_i is a *dedicated signal* from T_i , which will also be forwarded from R to T_i in the next time instant.
- (b) The interference over d_{ij} , for $i \neq j \in \{2, 3\}$, received at T_i is an *interfering signal* from T_j that is dedicated for T_1 .

Class (a): The interference of class (a) only contains messages that are dedicated for the receiver of T_2 (or of T_3). After the (N+1)-th time-instant, the receivers T_2 (and T_3) can completely decode all the forwarded dedicated signals that were transmitted in the UL of the last, N-th, time-instant without interference over d_{23} and d_{32} . These dedicated signals can now be used to cancel the additional interference of class (a) from the received signals of T_2 (and T_3) that were both transmitted and received during the *N*-th time-instant. If the interference of class (b) (this will be explained in the next paragraph) is fully removed as well, all the dedicated messages from the *N*-th hop can also be decoded completely. These steps of decoding and cancellation are continued in a backward decoding procedure until all interfering signals of class (a) are cancelled. Note that in the first time-instant, no DL-transmission has been performed yet. Thus, the class (a) interference is not harmful in the first time-instant and simply ignored.

Class (b): The interference of class (b) only contains messages that are dedicated for receiver T_1 and hence they represent interference for T_2 (or T_3), respectively. In this case, we apply an IN scheme. In particular, T_2 (or T_3) additionally pre-transmits the corresponding interference signals with a complementary sign over d_{32} (or d_{23}) of time instant l one time-instant in advance, i.e., at time-instant l-1. These pre-transmitted signals are aligned at the relay R, such that the forwarded signals from R in timeinstant l will exactly coincide with the original interference signals of class (b) over d_{32} (or d_{23}) at time instant l. This is clearly feasible for T_2 (and T_3), since it can align its signals to any of the n levels at R. As a result, the two complementary interference signals of class (b) coincide and neutralize each other completely. Note that the pretransmitted signals which are back-propagated to their original senders are known and can be cancelled. Moreover, since the pre-transmitted signals are dedicated signals for T_1 , T_1 can cancel those signals via a backward decoding procedure as discussed analogously for the interference of class (a). The pre-transmitted signals over d_{32} (or d_{23}) arriving at T₃ (or T₂) are again neutralized by another pre-transmission. This pre-transmission procedure is iterated backwards to the first transmission, where the signals over d_{32} (or d_{23}) may be discarded at T₃ (or T₂).

Altogether, we have shown that the additional interference over the subchannels d_{32} or d_{23} can be compensated and thence we have proven that the upper bounds on the 3-way channel are achievable.

We conjecture that there might also be an DoF-achieving scheme for the 3-way channel that is based on IA and not on SA. Then, the subchannels d_{32} or d_{23} are also used to convey some dedicated signals directly, without relaying over T₁. Such an optimal scheme would certainly also rely on closely related bidirectional, cyclic and unidirectional schemes as discussed similarly for the 3-user Y-channel. A symmetric case of this problem is investigated in Section 5.2.7 in terms of the LDCM.

5.1.7 Discussion

In contrast to most of the previously considered channels, e.g., the 2-user X-channel, the K-user interference channel, or the $2 \times 2 \times 2$ relay-interference channel, the channel matrices of the 3-way channel and the Y-channel are not subject to further conditions, i.e., the schemes can be applied on any channel matrix D, E, and F, with entries in \mathcal{D} . It turns out, that this is a useful property when regarding the the corresponding Gaussian MIMO 3-way channel. A direct consequence is that the problem of common eigenvectors in separate subspaces, as discussed in 3.4.2, vanishes, so that it is possible to formulate closed-form IA schemes for the MIMO 3-way channel with constant chan-

nel coefficients that are capable to achieve the upper bounds on the DoF. Note that these formulations do not resort to infinite symbol-extensions as in [3] for instance. Subsequently, we consider the 3-way channel in terms of the LDCM and GMCM.

5.2 Capacity Region of the Reciprocal Linear Deterministic 3-Way Channel

Similar to the CPCM, the linear deterministic 3-way channel and the linear deterministic 3-user Y-channel [79] are closely related. Motivated by this, we consider the reciprocal 3-way channel in terms of the LDCM. We first provide cut-set bounds and genie-aided upper bounds describing the outer bound on the capacity region. Achieving this capacity region interestingly relies upon SA and on BD, as well. Our main tool is again a Δ -Y transformation of the 3-way channel to an (extended) Y-channel, as motivated by elementary electrical circuit theory. However, due to the linear shift in the LDCM, the approach differs at several points. Furthermore, we integrate a capacityachieving scheme for a symmetric 3-way subchannel that is also based on signal-scale interference alignment (IA) as in [3], [88], and [22], without resorting to BD.



Figure 5.4: The linear deterministic 3-way channel (or Δ -channel) with three transceivers T_1, T_2 and T_3 has six independent messages $W_{j,i}$ transmitted and six corresponding estimated messages $\widehat{W}_{j,i}$ received by the nodes, for $i \neq j \in \mathcal{K}$. The channel gains are parameterized by $n_j \in \mathbb{N}, j \in \mathcal{K}$.

5.2.1 System Model: Linear Deterministic 3-Way Channel

In the linear deterministic 3-way channel, a user T_i is a combined full-duplex transmitter Tx_i and receiver Rx_i . We consider six independent messages W_{ji} dedicated to be conveyed from T_i to T_j with $W_{ji} \in \mathcal{W}_{ji} = \{1, ..., 2^{nR_{ji}}\}$, $R_{ji} \in \mathbb{R}^+$, for distinct $i, j \in \mathcal{K} = \{1, 2, 3\}$. The vector of all messages is denoted by:

$$\boldsymbol{w} = (W_{12}, W_{21}, W_{13}, W_{31}, W_{23}, W_{32}).$$
(5.39)

The rate tuple \mathbf{R} and the total sum-rate R_{Σ} are defined for rates $R_{ji} \in \mathbb{R}^+$ between Tx_i and Rx_j by:

$$\boldsymbol{R} = (R_{12}, R_{21}, R_{13}, R_{31}, R_{23}, R_{32}), \tag{5.40}$$

$$R_{\Sigma} = R_{12} + R_{21} + R_{13} + R_{31} + R_{23} + R_{32}.$$
(5.41)

 T_j encodes its messages into a codeword⁴ \boldsymbol{x}_j^N . The *l*-th symbol of \boldsymbol{x}_j^N is an element of an alphabet \mathcal{X} encoded as $\boldsymbol{x}_j(l) = f_{j,l}(W_{ij}, W_{kj}, \boldsymbol{y}_j^{l-1})$ for distinct $i, j, k \in \mathcal{K}$. Therein, \boldsymbol{y}_j^{l-1} are all received symbols at T_j until time-instant l-1 with encoding function $f_{j,l}(\cdot)$. A receiving T_i decodes $(\hat{W}_{ij}, \hat{W}_{ik}) = g_i(\boldsymbol{y}_i^N, W_{ij}, W_{ik})$ with the decoding function $g_i(\cdot)$. An error occurs if $(\hat{W}_{ij}, \hat{W}_{ik}) \neq (W_{ij}, W_{ik})$. The collection of messages, encoders, and decoders defines a code for the 3-way channel. Furthermore, rate tuple \boldsymbol{R} is called *achievable* if there is a sequence of codes such that the average error probability ϵ_N becomes arbitrarily small by increasing N. The set of all achievable rate tuples is the capacity region \mathcal{C}_{Δ} .

In the LDCM, the physical channel between T_i and T_j is modelled by $n_{ji} \in \mathbb{N}$ bit pipes, and the transmitted symbols $\boldsymbol{x}_j(l)$ are bit-vectors in $\mathcal{X} = \mathbb{F}_2^q$ with a number of $q = \max_{i \neq j \in \mathcal{K}}(n_{ji})$ dimensions. The received signals \boldsymbol{y}_i at receivers Rx_i , $i \in \mathcal{K}$, are deterministic functions of the transmitted signals for distinct $i, j, k \in \mathcal{K}$:

$$\boldsymbol{y}_i = \boldsymbol{S}^{q-n_{ij}} \boldsymbol{x}_j \oplus \boldsymbol{S}^{q-n_{ik}} \boldsymbol{x}_k, \tag{5.42}$$

where S is a $q \times q$ lower shift matrix, having unit entries on the lower side-diagonal. The effect of noise is mimicked by clipping linearly shifted symbols. Note that loopback self-interference is entirely cancelled from (5.42) due to the perfect full-duplex operation.

In a wireless channel, it is valid to assume reciprocity for the bidirectional links such that we may use the following parametrization $n_{ij} = n_{ji} =: n_k$ holds for distinct $i, j, k \in \mathcal{K}$. We also assume:

$$n_3 \ge n_2 \ge n_1,\tag{5.43}$$

as an ordering of parameters w.l.o.g., and obtain $n_3 = q$. We denote this linear deterministic reciprocal 3-way channel by $\text{D3C}(n_1, n_2, n_3)$.

5.2.2 Outer Bounds of Capacity Region: 3-Way Channel

Cut-set upper bounds

The cut-set bounds of broadcast and multiple-access channels as in [9] state that users T_i can not receive more bits than the number of incoming bit-levels and they cannot transmit more bits than the number of outgoing bit-levels available:

$$R_{ij} + R_{ik} \le \max(n_k, n_j), \tag{5.44}$$

$$R_{ji} + R_{ki} \le \max(n_k, n_j), \tag{5.45}$$

for distinct $i, j, k \in \mathcal{K}$. These cut-set bounds already provide a loose upper bound $\overline{\mathcal{C}}_{\Delta, \text{cut-set}}$ on the actual capacity region \mathcal{C}_{Δ} . To obtain a tight capacity characterization, we include further genie-aided upper bounds similar to those derived in [79].

⁴We use \boldsymbol{v}^N denote a sequence of N-vectors $(\boldsymbol{v}(1),...,\boldsymbol{v}(N))$.

Genie-aided upper bounds

Receiver T₁ intends to decode the dedicated messages W_{12} and W_{13} using its received signal \boldsymbol{y}_1^N and its own messages W_{21}, W_{31} , with a reliable decoding strategy. Let the interfering message W_{23} be provided to node T₁ as genie-aided side-information. Since T₁ already knows W_{13} and W_{23} , it can reconstruct $\boldsymbol{x}_3(1)$. From \boldsymbol{y}_1^N and $\boldsymbol{x}_3(1)$, T₁ can derive $\boldsymbol{x}_2(1)$ from the deterministic function (5.42). T₁ then constructs $\boldsymbol{y}_3(1)$ from $\boldsymbol{x}_1(1)$ and $\boldsymbol{x}_2(1)$. With W_{13}, W_{23} and $\boldsymbol{y}_3(1)$, T₁ can generate $\boldsymbol{x}_3(2)$. These steps are repeated accordingly for all time-instants from 2 to N until \boldsymbol{y}_3^N is completely constructed. Therefore, by knowing $\boldsymbol{y}_1^N, W_{21}, W_{31}$ and W_{23} at T₁, it can reliably decode W_{12} and W_{13} , and then reconstruct \boldsymbol{y}_3^N to reliably decode W_{32} . All messages are known at node T₁ now. Thus, for the genie-aided channel, any reliable code allows decoding W_{32} . From Fano's inequality [45], we can derive:

$$N(R_{12} + R_{13} + R_{32})$$

$$\leq I(W_{12}, W_{13}, W_{32}; \boldsymbol{y}_1^N, W_{21}, W_{31}, W_{23}) + N\epsilon_N$$

$$\leq H(\boldsymbol{y}_1^N) - H(\boldsymbol{y}_1^N | \boldsymbol{w}) + N\epsilon_N$$

$$\leq H(\boldsymbol{y}_1^N) + N\epsilon_N$$

$$\leq N(\max(n_3, n_2) + \epsilon_N)$$

$$= N(n_3 + \epsilon_N),$$
(5.46)

where $\epsilon_N \to 0$ as $N \to \infty$. By letting $N \to \infty$, we get $R_{12} + R_{13} + R_{32} \le n_3$. Similar bounds can be derived by considering different receivers and side-information (cf. Appendix C).

Outer Bound Region

By considering the genie-aided and non-redundant cut-set bounds jointly, we obtain the following set of upper bounds on the capacity region C_{Δ} :

$$R_{31} + R_{32} \le n_2,\tag{5.47}$$

$$R_{13} + R_{23} \le n_2, \tag{5.48}$$

$$R_{12} + R_{13} + R_{32} \le n_3, \tag{5.49}$$

$$R_{12} + R_{13} + R_{23} \le n_3, \tag{5.50}$$

$$R_{21} + R_{23} + R_{13} \le n_3 + n_2 - n_1, \tag{5.51}$$

$$R_{21} + R_{23} + R_{31} \le n_3, \tag{5.52}$$

$$R_{31} + R_{32} + R_{21} \le n_3, \tag{5.53}$$

$$R_{31} + R_{32} + R_{12} \le n_3 + n_2 - n_1. \tag{5.54}$$

From (5.49) and (5.52) for instance, the sum-capacity upper bound yields $R_{\Sigma} \leq 2n_3$. This set of bounds leads to the following lemma. Note that this sum-rate upper bound coincides with the sum-rate upper bound of the (reciprocal) deterministic two-way channel [89].

Lemma 5.3. The capacity region C_{Δ} of the $D3C(n_1, n_2, n_3)$ is outer bounded by \overline{C}_{Δ} , *i.e.*, $C_{\Delta} \subseteq \overline{C}_{\Delta}$, where:

$$\overline{\mathcal{C}}_{\Delta} = \{ \boldsymbol{R} \in \mathbb{R}^6_+ \mid \boldsymbol{R} \text{ satisfies } (5.47) \text{-} (5.54) \}.$$

This outer bound is in fact achievable. The achievability of this bound is proven via a Δ -Y transformation utilizing the optimal scheme for the Y-channel as a building block. Next, we briefly recapitulate the related Y-channel and its capacity region in terms of the LDCM as characterized in [79].

5.2.3 System Model: Linear Deterministic *Y*- Channel

The linear deterministic reciprocal 3-user Y-channel⁵ DYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3$) is depicted in Figure 5.5. The definitions of the message vector, the rate tuple, the transmission sym-



Figure 5.5: The reciprocal Y-channel with three transceivers T_1, T_2 , and T_3 has six independent messages W_{ji} transmitted and six corresponding estimated messages \widehat{W}_{ji} received by the nodes, $i \neq j \in \mathcal{K}$. The channel gains are parameterized by $\tilde{n}_i \in \mathbb{N}$, for $j \in \mathcal{K}$.

bols and the encoding/decoding functions carry over from those given in Section 5.2.1, but are denoted with the tilde-notation to distinguish between the two models. In contrast to the 3-way channel, all users T_j are connected via bidirectional reciprocal links to an intermediate relay R. The channel gain from R to user T_j is denoted by \tilde{n}_j . The gains are ordered w.l.o.g. by:

$$\tilde{n}_1 \ge \tilde{n}_2 \ge \tilde{n}_3,\tag{5.55}$$

so that $q = \max_{i \in \mathcal{K}}(\tilde{n}_i) = \tilde{n}_1$. Note that this ordering is reversely oriented when comparing it with (5.43). The transmitted signals are vectors $\boldsymbol{x}_j, \boldsymbol{x}_{\mathrm{R}} \in \mathbb{F}_2^q$ from T_j and R, respectively. The received signal at R and the received signals at T_j are given by

$$\boldsymbol{y}_{\mathrm{R}} = \sum_{j=1}^{3} \boldsymbol{S}^{q-n_{j}} \boldsymbol{x}_{j}, \qquad (5.56)$$

$$\boldsymbol{y}_j = \boldsymbol{S}^{q-n_j} \boldsymbol{x}_{\mathrm{R}},\tag{5.57}$$

respectively. Next, we re-state the capacity region of the linear shift deterministic Y-channel, which will be an essential part of the proof of the achievability of Lemma 5.3.

⁵Our notation slightly differs from [79] w.r.t. swapped indexation and tilde.

5.2.4 Outer Bounds of Capacity Region: Y- Channel

The capacity region C_Y of the DYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3$) has already been characterized in [79], and is given by the set of rate tuples $\tilde{\boldsymbol{R}} = (\tilde{R}_{12}, \tilde{R}_{21}, \tilde{R}_{13}, \tilde{R}_{31}, \tilde{R}_{23}, \tilde{R}_{32})$, satisfying:

$$\tilde{R}_{31} + \tilde{R}_{32} \le \tilde{n}_3,$$
 (5.58)

$$\tilde{R}_{13} + \tilde{R}_{23} \le \tilde{n}_3,$$
 (5.59)

$$\tilde{R}_{12} + \tilde{R}_{13} + \tilde{R}_{32} \le \tilde{n}_2, \tag{5.60}$$

$$\tilde{R}_{12} + \tilde{R}_{13} + \tilde{R}_{23} \le \tilde{n}_2, \tag{5.61}$$

$$\tilde{R}_{21} + \tilde{R}_{23} + \tilde{R}_{13} \le \tilde{n}_1, \tag{5.62}$$

$$\tilde{R}_{21} + \tilde{R}_{23} + \tilde{R}_{31} \le \tilde{n}_2, \tag{5.63}$$

$$\tilde{R}_{31} + \tilde{R}_{32} + \tilde{R}_{21} \le \tilde{n}_2,$$
 (5.64)

$$\tilde{R}_{31} + \tilde{R}_{32} + \tilde{R}_{12} \le \tilde{n}_1. \tag{5.65}$$

There is an interesting resemblance between the bounds (5.58)-(5.65) and (5.47)-(5.54). This resemblance will be exploited to design an optimal scheme for the 3-way channel next.

5.2.5 Δ -Y Transformation

Equating the upper bounds of the D3C (n_1, n_2, n_3) in (5.47)-(5.54) and the upper bounds of DYC $(\tilde{n}_1, \tilde{n}_2, \tilde{n}_3)$ in (5.58)-(5.65) yields:

$$\tilde{n}_1 = n_2 + n_3 - n_1 \tag{5.66}$$

$$\tilde{n}_2 = n_3,$$
 (5.67)

$$\tilde{n}_3 = n_2. \tag{5.68}$$

In other words, the outer bound for the $D3C(n_1, n_2, n_3)$ coincides with the capacity region of a $DYC(\tilde{n}_1, \tilde{n}_2, \tilde{n}_3)$. Note that the ordering of the channel gains in (5.43) and (5.55) still holds.

In order to show the achievability of the outer bound in Lemma 5.3, we first express the 3-way channel in terms of an extended Y-channel as depicted in Figure 5.6 (cf. Section 5.1.5 for the CPCM). User T_1 is extended such that it behaves like a virtual relay R which is also connected to a virtual user \tilde{T}_1 via an artificial sub-channel parametrized by \tilde{n}_1 . At R, the topmost levels $\tilde{n}_2 + 1, ..., \tilde{n}_1$ are only accessible by \tilde{T}_1 and not visible for T_2 and T_3 (see Figure 5.7) and hence only virtual within T_1 . The residual link with n_1 from the previous $D3C(n_1, n_2, n_3)$ remains as a weak bidirectional link between T_2 and T_3 in the extended Y-channel $eDYC(\tilde{n}_1, \tilde{n}_2, \tilde{n}_3, n_1)$. The channel gain n_1 is still the weakest one, since:

$$n_1 \le n_2 = \tilde{n}_3 \le \tilde{n}_2 = n_3 \le n_3 + n_2 - n_1 = \tilde{n}_1.$$
(5.69)

The optimal scheme for the Y-channel already achieves the outer bound $\overline{\mathcal{C}}_{\Delta}$ for $n_1 = 0$. However, in general we have $n_1 \ge 0$. Hence, we have to modify our scheme to deal with the additional interference signals over n_1 .



Figure 5.6: The reciprocal 3-way channel $D3C(n_1, n_2, n_3)$ is transformed into an extended (upside-down) Y-channel $eDYC(\tilde{n}_1, \tilde{n}_2, \tilde{n}_3, n_1)$ including an additional bidirectional link n_1 between T_2 and T_3 .

5.2.6 Achievability of $\overline{\mathcal{C}}_{\Delta}$

Let all (virtual) users \tilde{T}_1 , T_2 , T_3 and R apply the capacity-achieving SA scheme described in [79] as if it would be applied on a DYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3$), but without decoding at the receivers yet. We consider N uplink and N transmissions, over N + 1 timeinstants. We call this scheme the 'original' scheme. In Figures 5.7 and 5.8, we depict a signal-level representation of the eDYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3, n_1$) with the uplink on top, the downlink at the bottom, and with the additional interference links with channel gain n_1 crossing in the middle.

In contrast to the original scheme for the DYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3$), the achievable scheme for the eDYC($\tilde{n}_1, \tilde{n}_2, \tilde{n}_3, n_1$) must be adapted to deal with the signals inherently transmitted over n_1 . We will overlay the adapted scheme on top of the original scheme.

In particular, any signal transmitted by T_i , $i \in \{2,3\}$, appearing on the topmost levels $\tilde{n}_1 - n_1 + 1, ..., \tilde{n}_1$, will interfere at receiver T_j on the lowermost levels $1, ..., n_1$. We discern three classes of interference over n_1 that are potentially received at T_2 and T_3 when applying the original scheme:

- (a) The interference over n_1 received at T_i is a dedicated signal from T_j to T_i , which will also be forwarded from R to T_i in the next time-instant.
- (b) The interference over n_1 received at T_3 is a dedicated signal from T_2 to T_1 .
- (c) The interference over n_1 received at T_2 is a dedicated signal from T_3 to T_1 .

The signal levels of the uplink to R for instance comprise three components: the topmost $\{\tilde{n}_2 + 1, ..., \tilde{n}_1\}$ levels accessible by T₁ only, the levels $\{\tilde{n}_3 + 1, ..., \tilde{n}_2\}$ accessible by T₁ and T₂, and the lowermost $\{1, ..., \tilde{n}_3\}$ levels accessible by all three uplink users.

Class (a): To compensate interference of class (a), we postpone decoding until the last signals of the (N + 1)-th time-instant are received. The transmission scheme does not change w.r.t. the original one. Since the (uplink) transmitters of T₂ and T₃ are silent on the (N + 1)-th time-instant, the (downlink) receivers of T₂ and T₃ receive no signal over n_1 at the (N + 1)-th time-instant. Hence, T₂ and T₃ can decode their dedicated signals as received in the last hop. In fact, the downlink signals of

the (N + 1)-th time-instant behave analogously to the (N + 1)-th hop of the original scheme. Since the class (a) interference of the N-th hop is a subset of the dedicated signals in the (N + 1)-th hop, it is cancelled after decoding the dedicated signals of the (N + 1)-th hop. With such a BD scheme, the interference of class (a) is cancelled analogously for all preceding time-instants N - 1, ..., 2, 1.



Figure 5.7: Auxiliary illustration for interference of class (b).

Class (b): To compensate the interference of class (b), i.e., those bits received at T_3 over n_1 carrying a dedicated signal from T_2 to T_1 , say \boldsymbol{x}_{12} , we apply an IN scheme. In detail, T_2 pre-transmits the interference signal $(\boldsymbol{x}_{12}(l))$ one time-instant in advance (in time-instant l-1) as follows. Assume that T_3 receives $[\boldsymbol{x}_{R,3}^T(l), \boldsymbol{x}_{R,3}^T(l)]^T$ from R in the downlink at time-instant l, where $\boldsymbol{x}'_{R,3}(l)$ and $\boldsymbol{x}_{R,3}(l)$ are binary vectors of lengths n_1 and $\tilde{n}_3 - n_1$, respectively (see Figure 5.7). Moreover, assume that T_3 receives interference from $\boldsymbol{x}_{12}(l)$ over some bits of $\boldsymbol{x}_{R,3}(l)$. To deal with this interference, T_2 pre-transmits $\boldsymbol{x}_{12}(l)$ in time-instant l-1 in the uplink, over exactly the same levels where $\boldsymbol{x}_{R,3}(l)$ is received in the uplink⁶. By doing so, T_3 receives $\boldsymbol{x}_{12}(l)$ twice over $\boldsymbol{x}_{R,3}(l)$ in the downlink, once from T_2 and once from R. Since $\boldsymbol{x}_{12}(l)$ is a binary vector, the addition of $\boldsymbol{x}_{12}(l)$ to itself results in interference neutralization.

It remains to make sure that the pre-transmission does not disturb any other node. Clearly, \boldsymbol{x}_{12} does not disturb T₂ since it originates from the same node T₂. Also, \boldsymbol{x}_{12} does not disturb T₁ since \boldsymbol{x}_{12} is a desired signal at T₁, and thus the interfering \boldsymbol{x}_{12} is removed by BD.

One more problem remains. Our approach using IN only works if $x_{R,3}(l)$ is received over levels that are accessible by T_2 in the uplink, i. e., the levels $1, ..., \tilde{n}_2$ at R. However,

⁶Note that $x_{R,3}(l)$ is received at R at time-instant l-1, and transmitted at time-instant l.

 $\boldsymbol{x}_{R,3}(l)$ might contain information from T_1 , say \boldsymbol{x}_{31} , which might not be accessible by T_2 in the uplink. This is exactly the case if T_1 sends \boldsymbol{x}_{31} over levels $\tilde{n}_2 + 1, ..., \tilde{n}_1$ at R (blue area in Figure 5.7). However, the given problem can be solved easily by noting that the number of levels in the blue area in Figure 5.7 is $\tilde{n}_1 - \tilde{n}_2$. We have $\tilde{n}_1 - \tilde{n}_2 = \tilde{n}_3 - n_1$ by (5.66), i. e., the same number of levels in the non-interfered downlink levels at T_3 (green area in Figure 5.7). Therefore, we exploit this interesting equality of the transformation: R forwards \boldsymbol{x}_{31} over the non-interfered downlink levels at T_3 and the given problem is avoided. By pursuing such an approach, the impact of class (b) interference is completely eliminated.



Figure 5.8: Auxiliary illustration for interference of class (c).

Class (c): To compensate the interference of class (c) at T_2 received over n_1 , i.e., a dedicated signal \mathbf{x}_{31} from T_3 to T_1 , we apply a similar IN scheme. T_3 likewise additionally pre-transmits class (c) interference one time-instant in advance. T_3 can access \tilde{n}_3 levels in the uplink to R that are potentially forwarded to T_2 in the downlink during the next time-instant.

However, the levels $\tilde{n}_3 + 1, ..., \tilde{n}_1$ (blue area in Figure 5.8) are not accessible by T_3 in the uplink. Thus, if the signal received by T_2 from R in the downlink over levels $1, ..., n_1$ are sent over relay levels $\tilde{n}_3 + 1, ..., \tilde{n}_1$ in the uplink, then, T_3 can not perform IN. However this scenario can be avoided by sending all signals received in the uplink on the blue levels in Figure 5.8, over the green levels in the downlink. This is possible since $\tilde{n}_2 - n_1 = \tilde{n}_1 - \tilde{n}_3$ holds by (5.66). In this case, these signals do not interfere with the levels $1, ..., n_1$ at T_2 which renders T_3 capable of performing IN.

For the downlink from R to T_3 , the pre-transmitted signals which are back-propagated to T_3 are known self-interference and cancelled. For the downlink of T_1 , these pre-transmitted signals are dedicated for T_1 and cancelled by BD. Thus, the interference of class (c) is eliminated as well.

In all three classes, the interference over the bidirectional link n_1 is cancelled or neutralized and all dedicated signals are decodable by BD after N + 1 time-instants. Altogether, this proves the achievability of $\overline{\mathcal{C}}_{\Delta}$ leading to the following Theorem.

Theorem 5.4. The capacity region C_{Δ} of the D3C (n_1, n_2, n_3) is given by \overline{C}_{Δ} defined in Lemma 5.5.

At this point, the achievability of the capacity region is fully proven by means of SA, IN and BD.

5.2.7 Capacity Region of the Symmetric Case by IA

Interestingly, signals conveyed over the weak link n_1 are not used for direct communication. The interfering signals over n_1 are cancelled or neutralized by the communication scheme proposed in Section 5.2.6, so that the impact of the link n_1 is effectively eliminated. A certain drawback of our previous scheme is that the receivers must wait for N + 1 time-instants to apply the BD procedure. This is a very limiting property, especially for delay-limited communications [90].

As a contrary approach, we now propose a purely IA-based communication scheme for the symmetric D3C(m, m, m) that achieves the corresponding capacity region. The communication scheme for the D3C(m, m, m) based on IA is proven with similar methods as the one in [79]. In this case, there is no need for BD and IN.

Theorem 5.5. An interference alignment scheme based on bidirectional, cyclic and unidirectional communication suffices to achieve the outer bounds on the capacity region of a symmetric D3C(m,m,m) with $m \in \mathbb{N}$.

We consider a communication scheme of three components (cf. Theorem 5.1):

- A) Bidirectional: For distinct $i, j \in \mathcal{K}$, the pair of rates R_{ji}, R_{ij} is non-zero.
- **B)** Cyclic: For distinct $i, j, k \in \mathcal{K}$, the triple of rates R_{ji}, R_{jk}, R_{ki} is non-zero, whereas $R_{ij} = R_{kj} = R_{ik} = 0$.
- C) Unidirectional: None of the above cases holds.

We now outline our proposed IA scheme based on the components A, B and C. We will begin with scheme A on the D3C(m, m, m) operating at 2 bits per level. Pairs of users communicate bidirectionally. Then, we reduce the channel to D3C(m', m', m') by removing the already used levels from scheme A. Next, scheme B with $\frac{3}{2}$ bits per level is applied. Again we reduce the channel to D3C(m'', m'', m'') removing the levels occupied by scheme B. In the last step, we apply scheme C allocating 1 bit per level. If the rate tuple to be achieved does not satisfy one of conditions A, B, and C, the corresponding scheme is merely discarded. We will show in the following that these schemes suffice to achieve the outer bounds of the capacity region for the D3C(m, m, m).

Proof:

A) Bidirectional Communication on the D3C(m, m, m)

We define the following transmission parameters $a, b, c \in \mathbb{N}$:

$$a = \min(R_{12}, R_{21}), \ b = \min(R_{13}, R_{31}), \ c = \min(R_{23}, R_{32}).$$
 (5.70)

If a = b = c = 0 holds, scheme A is skipped and we continue with scheme B. We propose a signal allocation such that 2 bits per level are achieved. The signals are $\boldsymbol{x}_{12}, \boldsymbol{x}_{21} \in \mathbb{F}_2^a, \boldsymbol{x}_{31}, \boldsymbol{x}_{13} \in \mathbb{F}_2^b$, and $\boldsymbol{x}_{32}, \boldsymbol{x}_{23} \in \mathbb{F}_2^c$. To transmit these signals, $a + b + c \leq m$ levels are allocated as depicted in Figure 5.9. The interference signals \boldsymbol{x}_{ji} and \boldsymbol{x}_{ij} are aligned at T_k with pairwise distinct $i, j, k \in \mathcal{K}$.

k	$oldsymbol{x}_k$	$oldsymbol{y}_k$	intervals of levels
1	0	$m{x}_{32}$ + $m{x}_{23}$	a+b+1,, a+b+c
1	$oldsymbol{x}_{31}$	$oldsymbol{x}_{13}$	a + 1,, a + b
1	$oldsymbol{x}_{21}$	$oldsymbol{x}_{12}$	1,, a
2	$oldsymbol{x}_{32}$	$oldsymbol{x}_{23}$	a+b+1,, a+b+c
2	0	$m{x}_{13}$ + $m{x}_{31}$	a + 1,, a + b
2	$oldsymbol{x}_{12}$	$oldsymbol{x}_{21}$	1,, a
3	$oldsymbol{x}_{23}$	$oldsymbol{x}_{32}$	a+b+1,, a+b+c
3	$oldsymbol{x}_{13}$	$oldsymbol{x}_{31}$	a+1,, a+b
3	0	$m{x}_{21}$ + $m{x}_{12}$	1,, a

Figure 5.9: A) Allocation of signals to bit-levels for the bidirectional scheme over the D3C(m,m,m). The 1st column denotes the considered user, the 2nd column the transmitted signal, the 3rd column its received signal, and the 4th column describes the interval of levels concerned. The lowest bit-level is indexed by 1.

This allocation scheme is only feasible if enough levels are available at the transmitters and receivers for all bidirectional streams. For $\mathbf{R} \in \overline{C}$, the following must hold on a, b, c:

$$a + b + c \stackrel{(5.70)}{\leq} R_{12} + R_{13} + R_{32} \stackrel{(5.49)}{\leq} m.$$
 (5.71)

This is also true for all other upper bounds. For the yet unused levels, we still need to achieve the residual rate-vector:

$$\mathbf{R}' = (R_{12} - a, R_{21} - a, R_{13} - b, R_{31} - b, R_{23} - c, R_{32} - c)$$

= $(R'_{12}, R'_{21}, R'_{13}, R'_{31}, R'_{23}, R'_{32}).$ (5.72)

So far, at least three components will already be zero due to the min-expressions in (5.70). We remove the allocated levels so that the reduced D3C(m', m', m') is parameterized by:

$$m' = m - a - b - c. \tag{5.73}$$

Clearly, the reduced channel remains symmetric.

i	$oldsymbol{x}_i$	$oldsymbol{y}_i$	intervals of levels
1	0	$oldsymbol{x}_{13}$ + $oldsymbol{x}_{32}$	d + 1,, 2d
1	$oldsymbol{x}_{21}$	$oldsymbol{x}_{32}$	1,, d
2	$oldsymbol{x}_{32}$	$oldsymbol{x}_{13}$	d + 1,, 2d
2	$oldsymbol{x}_{32}$	$oldsymbol{x}_{21}$	1,, d
3	x_{13}	$oldsymbol{x}_{32}$	d + 1,, 2d
3	0	$oldsymbol{x}_{21}$ + $oldsymbol{x}_{32}$	1,, d

Figure 5.10: E	B) Allocation	of signals	to	bit-levels f	for	$\mathrm{D3C}(m',m',m')$	for	clock-wise
с	yclic commun	ication.						

i	$oldsymbol{x}_i$	$oldsymbol{y}_i$	intervals of levels
1	0	x_{12} + x_{23}	e + 1,, 2e
1	$oldsymbol{x}_{31}$	$oldsymbol{x}_{12}$	1,, e
2	$oldsymbol{x}_{12}$	$oldsymbol{x}_{23}$	e + 1,, 2e
2	$oldsymbol{x}_{12}$	$oldsymbol{x}_{23}$	1,, e
3	$m{x}_{23}$	$oldsymbol{x}_{12}$	e + 1,, 2e
3	0	$m{x}_{12}$ + $m{x}_{31}$	1,, e

Figure 5.11: B) Allocation of signals to bit-levels for D3C(m', m', m') for counter clockwise cyclic communication.

B) Cyclic Communication on D3C(m', m', m')

Given that the conditions for scheme B hold, and depending on the residual rate-vector \mathbf{R}' computed in (5.72), we apply either clock-wise cyclic communication $1 \rightarrow 2 \rightarrow 3 \rightarrow 1$ with parameter d or counter-clock-wise cyclic communication $1 \rightarrow 3 \rightarrow 2 \rightarrow 1$ with parameter e. The parameters $d, e \in \mathbb{N}$ are:

$$d = \min(R'_{21}, R'_{13}, R'_{32}), \ e = \min(R'_{12}, R'_{31}, R'_{23}).$$
(5.74)

Note that either d or e must be zero, since bidirectional communication is already taken care of by the previous scheme A. The definitions in (5.74) provide two cases:

$$d > 0 \Rightarrow e = 0, \ a = R_{12}, \ b = R_{31}, \ c = R_{23},$$
 (5.75)

$$e > 0 \Rightarrow d = 0, \ a = R_{21}, \ b = R_{13}, \ c = R_{32}.$$
 (5.76)

If both d = e = 0, this section is skipped and we continue with scheme C. In the following, we propose a signal allocation scheme such that $\frac{3}{2}$ bits per level are achieved.

For case (5.75), $R'_{12} = R'_{31} = R'_{23} = 0$ and $R'_{21}, R'_{13}, R'_{32}$ are non-zero. The signals are $\boldsymbol{x}_{21}, \boldsymbol{x}_{13}, \boldsymbol{x}_{32} \in \mathbb{F}_2^d$, allocated in blocks of d levels as depicted in Figure 5.10. The constraint $2d \leq m'$ must hold at each user to provide a feasible allocation. Signal \boldsymbol{x}_{32} is transmitted by T_2 on both intervals of size d and T_1 applies interference cancellation to decode \boldsymbol{x}_{13} from $\boldsymbol{x}_{13} + \boldsymbol{x}_{32}$. The interference signals \boldsymbol{x}_{21} and \boldsymbol{x}_{32} are aligned at T_3 .

Sufficiently many levels are available for scheme B on the reduced D3C(m', m', m')

if $\mathbf{R} \in \overline{\mathcal{C}}$, since:

$$2d \stackrel{(5.74)}{\leq} R'_{13} + R'_{32}$$

$$\stackrel{(5.72)}{=} R_{13} + R_{32} - b - c$$

$$\stackrel{(5.47)}{\leq} m - R_{12} - b - c$$

$$\stackrel{(5.75)}{=} m - a - b - c \stackrel{(5.73)}{=} m'.$$
(5.77)

The second case (5.76) for counter clock-wise communication is derived analogously, but with the indices swapped and with an adapted allocation (cf. Figure 5.11). In particular, we have $R'_{21} = R'_{13} = R'_{32} = 0$ and $R'_{12}, R'_{31}, R'_{23}$ are non-zero. In analogy to (5.77), the allocations for counter-clock-wise communication with parameter e satisfies $\mathbf{R} \in \overline{C}$:

$$2e \stackrel{(5.74)}{\leq} R'_{31} + R'_{23} \leq m - a - b - c = m'.$$

For the yet unused levels, the residual rate-vector is:

$$\begin{aligned} \boldsymbol{R}'' &= (R'_{12} - d, R'_{21} - e, R'_{13} - e, R'_{31} - d, R'_{23} - d, R'_{32} - e) \\ &= (R''_{12}, R''_{21}, R''_{13}, R''_{31}, R''_{23}, R''_{32}), \end{aligned}$$
(5.78)

over the D3C(m'', m'', m'') (where either d or e is zero) with:

$$m'' = m' - 2d - 2e. \tag{5.79}$$

C) Unidirectional Communication on the D3C(m'', m'', m'')

Six possible non-zero rate tuples remain that are not yet covered by the previous schemes A and B:

$$(R_{21}'', R_{31}'', R_{32}') \neq \mathbf{0}_{1\times3}, \ (R_{21}'', R_{31}'', R_{23}'') \neq \mathbf{0}_{1\times3}, (R_{12}'', R_{13}'', R_{23}'') \neq \mathbf{0}_{1\times3}, \ (R_{12}'', R_{13}'', R_{32}'') \neq \mathbf{0}_{1\times3}, (R_{12}'', R_{31}'', R_{32}'') \neq \mathbf{0}_{1\times3}, \ (R_{21}'', R_{13}'', R_{23}'') \neq \mathbf{0}_{1\times3}.$$

These cases pairwise exclude each other. W.l.o.g., we only consider the unidirectional case $(R_{12}'', R_{13}'', R_{23}'') \neq \mathbf{0}_{1\times 3}$, here. The remaining cases are derived by analogous steps. We have $R_{21}'' = R_{31}'' = R_{32}'' = 0$ and we parameterize the 3 non-zero rates by $f, g, h \in \mathbb{N}$:

$$R_{12}'' = f, \ R_{13}'' = g, \ R_{23}'' = h.$$
 (5.80)

The signals are $\mathbf{x}_{12} \in \mathbb{F}_2^f$, $\mathbf{x}_{13} \in \mathbb{F}_2^g$, $\mathbf{x}_{23} \in \mathbb{F}_2^h$. A number of $f + g + h \leq m''$ levels are allocated as depicted in Figure 5.12.

Since we demand $\mathbf{R}'' \in \overline{C}$, we discern two cases depending on the previous scheme B. In the first case, we assume that clock-wise cyclic communication was applied before.

k	$oldsymbol{x}_k$	$oldsymbol{y}_k$	intervals of levels
3	0	$oldsymbol{x}_{23}$	f+g+1,,f+g+h
3	0	$oldsymbol{x}_{13}$	$f+1, \ldots, f+g$
3	0	$m{x}_{12}$	1,, f
1	0	$oldsymbol{x}_{23}$	f+g+1,,f+g+h
1	0	$oldsymbol{x}_{13}$	$f+1, \ldots, f+g$
1	$oldsymbol{x}_{12}$	0	1,, f
2	$oldsymbol{x}_{23}$	$oldsymbol{x}_{12}$	f+g+1,,f+g+h
2	$m{x}_{13}$	0	$f+1, \ldots, f+g$
2	0	0	1,, f

Figure 5.12: C) Allocation of signals to bit-levels in unidirectional communication over D3C(m'', m'', m'').

Recall that either d or e must be zero. If d > 0 and e = 0, then:

$$f + g + h = R_{12}'' + R_{13}'' + R_{23}''$$

$$\stackrel{(5.78)}{=} R_{12}' + R_{13}' + R_{23}' - 2d$$

$$\stackrel{(5.72)}{=} R_{12} + R_{13} + R_{23} - a - b - c - 2d$$

$$\stackrel{(5.47)}{\leq} m - a - b - c - 2d$$

$$\stackrel{(5.73)}{=} m' - 2d.$$
(5.81)

Otherwise, if e > 0 (counter clock-wise) and d = 0, then:

$$f + h + \binom{(5.78)}{g} \stackrel{(5.72)}{=} R_{12} + R_{13} + R_{23} - a - b - c - e$$

$$\stackrel{(5.52)}{\leq} R_{12} + R_{13} - a - b - e$$

$$\stackrel{(5.52)}{\leq} m - R_{23} - a - b - e$$

$$\leq m - (e + c) - a - b - e \qquad (5.82)$$

$$= m' - 2e. \qquad (5.83)$$

For (5.82), we use $e \le R'_{23} = R_{23} - c$ from (5.80) and (5.72). Since either d or e is zero, combining (5.81) and (5.83) yields:

$$f + g + h \le m' - 2d - 2e \stackrel{(5.79)}{=} m''. \tag{5.84}$$

Hence, sufficiently many levels are also available for the D3C(m'',m'',m''). Altogether, there are enough levels to communicate all a + b + c + 2d + 2e + f + g + h bits. The application of schemes A to C achieves the upper bounds of the capacity region for the D3C(m,m,m) proving Theorem 5.5. The remaining steps, showing that each corner point of the capacity region is achievable is analogous to [79, Theorems 3 & 4] and omitted here.

Note that, by considering the CPCM of the 3-way channel first, we were able to facilitate the analysis of the 3-way channel within the LDCM.

5.3 Degrees of Freedom of the MIMO $M_i \times M_i$ 3-Way Channel

In the following, we investigate MIMO IA on the Gaussian 3-way channel with multiple antennas. The DoF of several unidirectional multi-user interference networks with particular antenna configurations have already been studied thoroughly in the literature (cf. References of [12]). Therein, a particular focus concerns the DoF for MIMO IA with constant channel coefficients. For instance, the DoF of the 2-user MIMO interference channel using zero-forcing are provided in [35], the sum-DoF and the DoF-region of the 2-user MIMO X- channel are considered in [11] and [38], respectively, where MIMO IA was used. The DoF of the general MIMO K-user interference channel with an arbitrary number of antennas at the transmitters and receivers are yet unknown and a full derivation of the remains quite challenging so far.

In this section, we study the DoF of the MIMO 3-way channel with constant channel coefficients. In this first case, we assume an arbitrary number of M_i transmit antennas and also M_i receive antennas at each transceiver T_i . We derive cut-set and genie-aided upper bounds and on the sum-DoF of the MIMO 3-way channel. A MIMO IA and zero-forcing scheme is proposed to show that the derived upper bound is achievable. We observe that the sum-DoF are limited by the strongest channel (the one with the largest rank), and therefore, the sum-DoF are achievable by just letting the two strongest users communicate similar to the single-input single-output (SISO) case [91]. Since this does not serve all users in a fair manner, we propose an alternative scheme which also achieves the sum-DoF upper bound while serving all three users simultaneously. The achievable DoF of this alternative scheme is expressed in terms of a linear programming optimization problem which can be solved by using the well-known simplex method. As in the cases of the CPCM and the LDCM, the considered MIMO 3-way channel is closely related to the MIMO 3-user Y- channel [77], [78].



Figure 5.13: The MIMO 3 -Way Δ - channel with an equal number of $M_{\text{Tx}_i} = M_i$ transmit antennas and $M_{\text{Rx}_i} = M_i$ receive antennas at each user i = 1, 2, 3.

5.3.1 MIMO $M_i \times M_i$ 3-Way Channel

The MIMO 3-way channel comprises three full-duplex users T_i with user-indices i in the set $\mathcal{K} = \{1, 2, 3\}$. A message from T_i to T_j is denoted by W_{ji} and has rate R_{ji} for $i \neq j \in \mathcal{K}$. Each user desires to communicate a message to the two other users. In

general, a transceiver T_i is equipped with an arbitrary number of $M_{\mathrm{Tx}_i} \in \mathbb{N}$ transmit antennas and a number of number of $M_{\mathrm{Rx}_i} \in \mathbb{N}$ receive antennas. However in this section, we consider the symmetric special case, where the number of transmit and receive antennas is assumed to be equal to $M_{\mathrm{Tx}_i} = M_{\mathrm{Rx}_i} = M_i$ per T_i , for $i \in \mathcal{K}$. We may assume w.l.o.g. that the number of antennas is ordered among the three users by:

$$M_1 \ge M_2 \ge M_3.$$
 (5.85)

As defined in Section 2.2, the received signal for the 3-way channel is particularly defined by:

$$y_1(n) = H_{12}x_2(n) + H_{13}x_3(n), \qquad (5.86)$$

$$y_2(n) = H_{21}x_1(n) + H_{23}x_3(n), \qquad (5.87)$$

$$y_3(n) = H_{31}x_1(n) + H_{32}x_2(n).$$
 (5.88)

The considered channel coefficients of each H_{ji} are assumed to be constant throughout the whole duration of the transmission and the coefficients are fully known at each user.

Encoding and Decoding Functions

After receiving $\boldsymbol{y}_i(n)$, T_i constructs $\boldsymbol{x}_i(n+1)$ as:

$$\boldsymbol{x}_{j}(n+1) = \mathcal{E}_{j,n}(W_{ij}, W_{kj}, \boldsymbol{y}_{j}^{n}), \qquad (5.89)$$

where $\mathcal{E}_{j,n}$ is the encoding function of T_j at time-instant n, and sends $\boldsymbol{x}_j(n+1)$ in the next transmission. After N transmissions, where N is the length of one transmission block (codeword), T_j decodes W_{ji} and W_{jk} as follows:

$$(W_{ji}, W_{jk}) = \mathcal{D}_j(W_{ij}, W_{kj}, \boldsymbol{y}_j^N), \qquad (5.90)$$

where \mathcal{D}_j is the decoding function of T_j . We will neglect the time-index *n* for notational simplicity unless necessary.

5.3.2 Intersection Subspace of Random Matrices

For the derivation of the capacity-achieving schemes, we need to compute the intersection subspaces of the spaces spanned by the channel matrices. This is accomplished by the following lemma⁷.

Lemma 5.6. If A_1 and A_2 are complex $L \times M_1$ and $L \times M_2$ random matrices, respectively, whose entries are drawn randomly *i*. *i*. *d*., then there exists an intersection subspace of $[\min(M_1, L) + \min(M_2, L) - L]^+$ dimensions between the two column spaces of A_1 and A_2 , almost surely.

Proof:

⁷The proof is a slightly generalized version of [77, Lemma 1].

Let an $L \times 1$ vector \boldsymbol{q} lie in span $(\boldsymbol{A}_1) \cap \text{span}(\boldsymbol{A}_2)$. Then, there exists $\boldsymbol{q}_i \in \mathbb{C}^{M_i \times 1}$, with i = 1, 2, such that:

$$\boldsymbol{q} = \boldsymbol{A}_1 \boldsymbol{q}_1 = \boldsymbol{A}_2 \boldsymbol{q}_2. \tag{5.91}$$

In matrix form this yields:

$$\begin{bmatrix} I_{L\times L} & -A_1 & \mathbf{0}_{L\times M_2} \\ I_{L\times L} & \mathbf{0}_{L\times M_1} & -A_2 \end{bmatrix} \begin{pmatrix} q \\ q_1 \\ q_2 \end{pmatrix} = M\mathbf{x} = \mathbf{0}_{2L\times \mathsf{rank}(M)}.$$
(5.92)

Note that $\operatorname{rank}(A_i) = \min(M_i, L)$ holds, almost surely. We compute the dimension of $\operatorname{span}(A_1) \cap \operatorname{span}(A_2)$ by computing the dimension of the nullity of M. Since:

$$\mathsf{rank}(\boldsymbol{M}) = \min(2L, \min(M_1, L) + \min(M_2, L) + L)$$

holds for i.i.d. matrices A_1 and A_2 almost surely, we can conclude with the rank-nullity theorem of linear algebra, that:

$$dim(null(M)) = \min(M_1, L) + \min(M_2, L) + L - rank(M) = [\min(M_1, L) + \min(M_2, L) - L]^+$$
(5.93)

holds, almost surely.

5.3.3 Main Result

The main result of Section 5.3 is a sum-DoF characterization for the MIMO 3-way channel as provided in the following theorem.

Theorem 5.7. The DoF of the MIMO 3-way channel with M_i antennas at user T_i , and $M_1 \ge M_2 \ge M_3$, are given by:

$$d_{\Sigma} = d_{12} + d_{21} + d_{13} + d_{31} + d_{23} + d_{32} = 2M_2.$$
(5.94)

The converse of this theorem is provided in Section 5.4.2 and the achievability in Section 5.4.3. This theorem states that the sum-DoF in this case is given by twice the rank of the channel matrix between T_1 and T_2 , which is the channel of largest rank. Therefore, this DoF is achievable by letting these two users communicate while leaving T_3 silent. Albeit this achieves $2M_2$ DoF, it completely excludes T_3 and it does not distribute the resources fairly between the three users. In Section 5.4.3, we provide an alternative scheme which achieves the DoF while maintaining non-zero DoF for all users.

5.3.4 Upper Bounds

Cut-Set Bounds

We begin with considering the cut-set bounds for the MIMO 3-way channel:

$$d_{ji} + d_{ki} \le \min(M_i, M_j + M_k), \tag{5.95}$$

$$d_{ij} + d_{ik} \le \min(M_j + M_k, M_i).$$
(5.96)

The right-hand side of (5.95) is the rank of the MIMO channel between T_i and a receiver formed by enabling full cooperation between T_j and T_k , with channel matrix $[\boldsymbol{H}_{ii}^T \boldsymbol{H}_{ki}^T]^{\mathsf{T}}$. A similar interpretation holds for the second bound.

Similar to [78], the cut-set bounds provide bounds on the sum of the DoF of two messages at a time. However, using genie-aided arguments, it is possible to establish bounds on the sum of the DoF of three messages, which are tighter than the cut-set bounds. The key idea is to allow some user to decode one more message in addition to its two desired messages by enhancing this user with some side-information.

Genie-Aided Bounds

Assume every node can obtain its dedicated messages with an arbitrary small probability of error. This means that T_2 for instance can decode its dedicated messages W_{21} and W_{23} reliably from its available information, i.e., from its own transmitted W_{12}, W_{32} , and from its received signal \boldsymbol{y}_2^N . Now let us enhance T_2 by providing the message W_{31} as side-information. We also provide T_2 with the correction-noise signal:

$$\tilde{z}_{2}^{N} = z_{1}^{N} - H_{13}H_{23}z_{2}^{N}, \qquad (5.97)$$

as side-information 8 .

At this point, T₂ knows W_{21} (decoded) and W_{31} (side-information). With W_{21} , W_{31} , T₂ can generate $\boldsymbol{x}_1(1)$. By subtracting $\boldsymbol{H}_{21}\boldsymbol{x}_1(1)$ from $\boldsymbol{y}_2(1)$, and multiplying the result with \boldsymbol{H}_{23} , T₂ can recover a noisy observation of $\boldsymbol{x}_3(1)$ given by $\boldsymbol{x}_3(1)+\boldsymbol{H}_{23}\boldsymbol{z}_2(1)$. Next, T₂ multiplies this noisy observation by \boldsymbol{H}_{13} , and adds $\boldsymbol{H}_{12}\boldsymbol{x}_2(1)$ and $\tilde{\boldsymbol{z}}_2(1)$ to it to obtain $\boldsymbol{y}_1(1)$. Thus, T₂ obtains the first instance of \boldsymbol{y}_1^n . Knowing $\boldsymbol{y}_1(1)$, W_{21} and W_{31} , T₂ can generate $\boldsymbol{x}_1(2)$ (cf. (5.89)). Using $\boldsymbol{x}_1(2)$ again with $\boldsymbol{y}_2(2)$, T₂ can generate $\boldsymbol{y}_1(2)$ and $\boldsymbol{x}_1(3)$. T₂ proceeds this way until all instances (up to the *N*-th instance) of \boldsymbol{y}_1^N have been generated. Now, having \boldsymbol{y}_1^N , W_{21} , and W_{31} , i.e., the same information as T₁, T₂ can decode W_{13} (cf. (5.90)). Therefore, given W_{13} and $\tilde{\boldsymbol{z}}_2^N$ as side-information, T₂ can decode W_{21} , W_{23} and W_{13} . Hence, the DoF of these messages are almost surely upper bounded by:

$$d_{21} + d_{23} + d_{31} \le \operatorname{rank}([\boldsymbol{H}_{21} \, \boldsymbol{H}_{23}]) \tag{5.98}$$

$$= \min(M_2, M_1 + M_3) \stackrel{(5.85)}{=} M_2. \tag{5.99}$$

We can apply a similar approach to bound $d_{31} + d_{32} + d_{12}$ by M_2 . However, in this case, we need to enhance T_3 with $M_2 - M_3$ antennas to make it as strong as T_2 . The

⁸ \boldsymbol{H}_{23} exists since \boldsymbol{H}_{23} is an $M_2 \times M_3$ matrix with $M_2 \ge M_3$.

effective channel output at T₃ after this enhancement becomes:

$$\tilde{\boldsymbol{y}}_{3}(n) = \tilde{\boldsymbol{H}}_{31}\boldsymbol{x}_{1}(n) + \tilde{\boldsymbol{H}}_{32}\boldsymbol{x}_{2}(n) + \tilde{\boldsymbol{z}}_{3}(n), \qquad (5.100)$$

for n = 1, ..., N, where \tilde{H}_{31} and \tilde{H}_{32} are $M_2 \times M_1$ and $M_2 \times M_2$ matrices with rank M_2 , respectively, and \tilde{z}_3 is a Gaussian noise vector with M_2 dimensions. T₃ can decode W_{31}, W_{32} having $\tilde{y}_3^n, W_{13}, W_{23}$. By providing W_{21} and:

$$\tilde{\boldsymbol{z}}_{3}^{N} = \boldsymbol{z}_{1}^{N} - \boldsymbol{H}_{12} \tilde{\boldsymbol{H}}_{32}^{-1} \boldsymbol{z}_{3}^{N}$$
 (5.101)

to the enhanced T_3 with M_2 antennas, it can generate $\boldsymbol{x}_1(1)$. We use analogous operations as applied for (5.99) to obtain \boldsymbol{y}_1^N and to decode W_{12} . This leads to the upper bound:

$$d_{31} + d_{32} + d_{12} \le \operatorname{rank}([\tilde{H}_{31} \tilde{H}_{32}]) \tag{5.102}$$

$$= \min(M_2, M_1 + M_2) = M_2, \tag{5.103}$$

almost surely. Concluding the converse proof by combining (5.99) and (5.103) yields the sum-DoF upper bound of Theorem 5.7:

$$d_{\Sigma} = d_{12} + d_{21} + d_{13} + d_{31} + d_{23} + d_{32} \le 2M_2. \tag{5.104}$$

To achieve the upper bound on the sum-DoF, we propose a beam-forming and zeroforcing scheme using MIMO interference alignment [78] in the following sections.

5.3.5 Achievability

Pre-coding

We consider the receive signal space at T_1 at first. Note that as T_2 and T_3 each have less antennas than T_1 , they can not beam-form interference into the null space of T_1 . Instead of zero-forcing beam-forming, we use interference alignment. In order to minimize the number of dimensions spanned by the interference caused by T_2 and T_3 at T_1 , we align the interference caused by the bidirectional communication between T_2 and T_3 (signals u_{32} and u_{23} , respectively) in the intersection subspace of the spaces spanned by the columns of H_{12} and H_{13} . From Lemma 5.6, the columns of H_{12} and H_{13} intersect in an \tilde{M}_1 -dimensional subspace, where $\tilde{M}_1 = [M_2 + M_3 - M_1]^+$. To achieve this alignment, T_2 and T_3 pre-code the signal streams $u_{32} \in \mathbb{C}^{\tilde{d}_{32}}$ and $u_{23} \in \mathbb{C}^{\tilde{d}_{23}}$ with:

$$0 \le \tilde{d}_{32} = \tilde{d}_{23} \le \tilde{M}_1$$
 (5.105)

dimensions, into $V_{32}u_{32}$ and $V_{23}u_{23}$, respectively, where the beam-forming matrices $V_{32} \in \mathbb{C}^{M_2 \times \tilde{d}_{32}}$ and $V_{23} \in \mathbb{C}^{M_3 \times \tilde{d}_{32}}$ satisfy the following alignment at T_1 :

$$span(H_{13}V_{23}) = span(H_{12}V_{32}).$$
 (5.106)

This accounts for a total of $2\tilde{d}_{32}$ streams that can be exchanged by T_2 and T_3 while causing interference in only \tilde{d}_{32} dimensions at T_1 .

Now, we consider the receive signal space at T_2 . As T_1 has more antennas than T_2 , T_1 can send a signal $\bar{\boldsymbol{u}}_{31} \in \mathbb{C}^{\bar{d}_{31}}$ to T_3 in the null space of \boldsymbol{H}_{21} . The maximal number of such streams that can be beam-formed to this null space is bounded by $\min(M_1 - M_2, M_3)$. Thus, T_1 sends streams of:

$$0 \le \bar{d}_{31} \le \min(M_1 - M_2, M_3) \tag{5.107}$$

dimensions beam-formed into the null space of H_{21} . To realize this, T_1 designs a zero-forcing beam-forming matrix $\bar{V}_{31} \in \mathbb{C}^{M_1 \times \bar{d}_{31}}$ that satisfies:

$$\boldsymbol{H}_{21}\bar{\boldsymbol{V}}_{31} = \boldsymbol{0}_{M_2 \times \bar{d}_{31}},\tag{5.108}$$

and pre-codes $\bar{\boldsymbol{u}}_{31}$ by $\bar{\boldsymbol{V}}_{31}\bar{\boldsymbol{u}}_{31}$. The remaining streams sent from T_1 to T_3 (if any) can be aligned to the streams sent from T_3 to T_1 within the receive signal space of T_2 . This alignment is possible since the columns of \boldsymbol{H}_{21} and \boldsymbol{H}_{23} intersect in an M_3 -dimensional subspace as given by Lemma 5.6. To this end, T_1 and T_3 construct $\tilde{\boldsymbol{V}}_{31}\tilde{\boldsymbol{u}}_{31}$ and $\boldsymbol{V}_{13}\boldsymbol{u}_{13}$, respectively, where $\tilde{\boldsymbol{u}}_{31} \in \mathbb{C}^{\tilde{d}_{31}}$ and $\boldsymbol{u}_{13} \in \mathbb{C}^{\tilde{d}_{13}}$ have:

$$0 \le \tilde{d}_{31} = \tilde{d}_{13} \le M_3 \tag{5.109}$$

dimensions, and where the beam-forming matrices defined by $V_{13} \in \mathbb{C}^{M_3 \times \tilde{d}_{31}}$, and $\tilde{V}_{31} \in \mathbb{C}^{M_1 \times \tilde{d}_{31}}$ satisfy:

$$span(H_{23}V_{13}) = span(H_{21}\tilde{V}_{31}).$$
 (5.110)

The aligned interference of \tilde{u}_{31} and u_{13} occupies $\tilde{d}_{31} \leq M_3$ dimensions at the receive signal space of T_2 .

Considering the interference space at T_3 , we see that T_3 has less antennas than T_1 and than T_2 . Thus, T_1 beam-forms a signal $\bar{\boldsymbol{u}}_{21} \in \mathbb{C}^{\bar{d}_{21}}$ into the null space of \boldsymbol{H}_{31} of size min $(M_2, M_1 - M_3)$, which requires:

$$0 \le \bar{d}_{21} \le \min(M_2, M_1 - M_3) \tag{5.111}$$

dimensions. This is accomplished by designing a zero-forcing beam-forming matrix $\bar{V}_{21} \in \mathbb{C}^{M_1 \times \bar{d}_{21}}$ such that:

$$\boldsymbol{H}_{31} \bar{\boldsymbol{V}}_{21} = \boldsymbol{0}_{M_3 \times \bar{d}_{21}}, \tag{5.112}$$

and by pre-coding \bar{u}_{21} with $\bar{V}_{21}\bar{u}_{21}$. Then, T_2 beam-forms $\bar{u}_{12} \in \mathbb{C}^{\bar{d}_{12}}$ into the null space at T_3 of size $M_2 - M_3$, where:

$$0 \le \bar{d}_{12} \le M_2 - M_3. \tag{5.113}$$

To realize this, we design a zero-forcing beam-forming matrix $\bar{V}_{12} \in \mathbb{C}^{M_2 \times \bar{d}_{12}}$ such that:

$$\boldsymbol{H}_{32}\bar{\boldsymbol{V}}_{12} = \boldsymbol{0}_{M_3 \times \bar{d}_{12}},\tag{5.114}$$

and pre-code $\bar{\boldsymbol{u}}_{12}$ by $\bar{\boldsymbol{V}}_{12}\bar{\boldsymbol{u}}_{12}$. The remaining streams from T_1 to T_2 and vice versa (if any) are aligned at T_3 . The spaces spanned by \boldsymbol{H}_{31} and \boldsymbol{H}_{32} intersect in M_3 dimensions

as given by Lemma 5.6. We choose the beam-forming matrices $\tilde{V}_{21} \in \mathbb{C}^{M_1 \times \tilde{d}_{21}}$ and $\tilde{V}_{12} \in \mathbb{C}^{M_2 \times \tilde{d}_{21}}$ such that:

$$span(H_{32}\tilde{V}_{12}) = span(H_{31}\tilde{V}_{21}),$$
 (5.115)

and use them to pre-code \tilde{u}_{21} and \tilde{u}_{12} with:

$$0 \le \tilde{d}_{21} = \tilde{d}_{12} \le M_3 \tag{5.116}$$

dimensions into $\tilde{V}_{21}\tilde{u}_{21}$ and $\tilde{V}_{12}\tilde{u}_{12}$.

Finally, the transmitters send the following signals:

 $\boldsymbol{x}_{1} = [\tilde{\boldsymbol{V}}_{21} \ \bar{\boldsymbol{V}}_{21}] \begin{bmatrix} \tilde{\boldsymbol{u}}_{21} \\ \bar{\boldsymbol{u}}_{21} \end{bmatrix} + \begin{bmatrix} \tilde{\boldsymbol{V}}_{31} \ \bar{\boldsymbol{V}}_{31} \end{bmatrix} \begin{bmatrix} \tilde{\boldsymbol{u}}_{31} \\ \bar{\boldsymbol{u}}_{31} \end{bmatrix}, \tag{5.117}$

$$\boldsymbol{x}_{2} = \left[\boldsymbol{V}_{12} \ \bar{\boldsymbol{V}}_{12} \right] \left[\frac{\bar{\boldsymbol{u}}_{12}}{\bar{\boldsymbol{u}}_{12}} \right] + \boldsymbol{V}_{32} \boldsymbol{u}_{32}, \tag{5.118}$$

$$\boldsymbol{x}_3 = \boldsymbol{V}_{13} \boldsymbol{u}_{13} + \boldsymbol{V}_{23} \boldsymbol{u}_{23}. \tag{5.119}$$

In total, T_1 sends $d_{21} = \tilde{d}_{21} + \bar{d}_{21}$ and $d_{31} = \tilde{d}_{31} + \bar{d}_{31}$ streams to T_2 and T_3 , respectively, T_2 sends $d_{12} = \tilde{d}_{12} + \bar{d}_{12}$ and $d_{32} = \tilde{d}_{32}$ streams to T_1 and T_3 , respectively, and T_3 sends $d_{13} = \tilde{d}_{12}$ and $d_{23} = \tilde{d}_{23}$ streams to T_1 and T_2 , respectively.

Post-coding

The received signal at T_1 can be written as:

$$\boldsymbol{y}_{1} = \boldsymbol{H}_{12} \begin{bmatrix} \tilde{\boldsymbol{V}}_{12} \ \bar{\boldsymbol{V}}_{12} \end{bmatrix} \begin{bmatrix} \tilde{\boldsymbol{u}}_{12} \\ \bar{\boldsymbol{u}}_{12} \end{bmatrix} + \begin{bmatrix} \boldsymbol{H}_{12} \boldsymbol{V}_{32} \boldsymbol{u}_{32} + \boldsymbol{H}_{13} \boldsymbol{V}_{23} \boldsymbol{u}_{23} \end{bmatrix} + \boldsymbol{H}_{13} \boldsymbol{V}_{13} \boldsymbol{u}_{13} + \boldsymbol{z}_{1}. \quad (5.120)$$

The desired signals from T_2 occupy $\tilde{d}_{21} + \tilde{d}_{21}$ dimensions. The aligned interference in the second part of the summation, i.e., $H_{12}V_{32}u_{32} + H_{13}V_{23}u_{23}$, occupies \tilde{d}_{32} dimensions, and the desired signal from T_3 occupies \tilde{d}_{13} dimensions. The desired signals can be resolved from the interference as long as they are linearly independent of the interference and also among each other. Namely, the columns of the following $M_1 \times (\tilde{d}_{12} + \tilde{d}_{12} + \tilde{d}_{32} + \tilde{d}_{13})$ matrix must be linearly independent:

$$\left[\boldsymbol{H}_{12}\tilde{\boldsymbol{V}}_{12} \; \boldsymbol{H}_{12}\bar{\boldsymbol{V}}_{12} \; \boldsymbol{H}_{12}\boldsymbol{V}_{32} \; \boldsymbol{H}_{13}\boldsymbol{V}_{13}\right], \tag{5.121}$$

which requires:

$$0 \le \tilde{d}_{12} + \bar{d}_{12} + \tilde{d}_{32} + \tilde{d}_{13} \le M_1.$$
(5.122)

Under this condition, this linear independence can be guaranteed (almost surely) by designing \bar{V}_{12} according to (5.114), and choosing \tilde{V}_{12} , V_{32} , and V_{13} randomly.

Given this linear independence, T_1 can use zero-forcing matrices N_{12} and N_{13} of $d_{12} \times M_1$ and $d_{13} \times M_1$ dimensions, to zero-force the interference and to separate the two dedicated information signals. These zero-forcing matrices must satisfy:

$$N_{12}H_{13}(V_{13}+V_{23}) = \mathbf{0}_{d_{12}\times(d_{13}+d_{23})}, \tag{5.123}$$

$$N_{13}H_{12}(\bar{V}_{12} + \bar{V}_{12} + V_{32}) = \mathbf{0}_{d_{13} \times (d_{12} + d_{32})}.$$
(5.124)

Note that by zero-forcing $H_{13}V_{23}$, also $H_{12}V_{32}$ is zero-forced (and vice-versa) by (5.106). By using the proposed null-space beam-forming and zero-forcing, receiver T_1 obtains:

$$N_{12}y_1 = N_{12}H_{12}(\tilde{V}_{12}\tilde{u}_{12} + \bar{V}_{12}\bar{u}_{12}) + N_{12}z_1,$$
 (5.125)

$$N_{13}y_1 = N_{13}H_{13}V_{13}u_{13} + N_{13}z_1.$$
 (5.126)

Thus, T_1 recovers d_{12} linearly independent noisy observations of $\tilde{\boldsymbol{u}}_{12}$ and $\bar{\boldsymbol{u}}_{12}$, and also d_{13} linearly independent noisy observations of \boldsymbol{u}_{13} as $\boldsymbol{N}_1 = [\boldsymbol{N}_{12}^T \boldsymbol{N}_{13}^T]^T$ has sufficient row rank $d_{12} + d_{13}$ almost surely. Thus T_1 can decode all dedicated signals and achieves a number of $d_{12} + d_{13}$ DoF.

On the receiver side of T_2 , we have:

$$\boldsymbol{y}_{2} = \boldsymbol{H}_{21} [\tilde{\boldsymbol{V}}_{21} \ \bar{\boldsymbol{V}}_{21}] [\frac{\tilde{\boldsymbol{u}}_{21}}{\tilde{\boldsymbol{u}}_{21}}] + [\boldsymbol{H}_{21} \tilde{\boldsymbol{V}}_{31} \tilde{\boldsymbol{u}}_{31} + \boldsymbol{H}_{23} \boldsymbol{V}_{13} \boldsymbol{u}_{13}] \boldsymbol{H}_{23} \boldsymbol{V}_{23} \boldsymbol{u}_{23} + \boldsymbol{z}_{2}.$$
(5.127)

Note that \bar{u}_{31} is not observed by T₂ due to (5.108). Similarly to T₁, we need the following constraint to guarantee the linear independence of the desired signals and the interference:

$$0 \le \tilde{d}_{21} + \bar{d}_{21} + \tilde{d}_{31} + \tilde{d}_{23} \le M_2. \tag{5.128}$$

We use zero-forcing matrices N_{21} and N_{23} of $d_{21} \times M_2$ and $d_{23} \times M_2$ dimensions, respectively, satisfying:

$$N_{21}H_{23}(V_{23}+V_{13}) = \mathbf{0}_{d_{21}\times(d_{23}+d_{13})}, \tag{5.129}$$

$$N_{23}H_{21}(V_{21} + \bar{V}_{21} + V_{31}) = \mathbf{0}_{d_{23} \times (d_{21} + \tilde{d}_{31})},$$
(5.130)

to zero-force the interference and to separate the two dedicated information signals. By zero-forcing $H_{23}V_{13}$, also $H_{21}\tilde{V}_{31}$ is zero-forced (and vice-versa) by (5.110). With this scheme, receiver T₂ obtains:

$$N_{21}y_2 = N_{21}H_{21}(\tilde{V}_{21}\tilde{u}_{21} + \bar{V}_{21}\bar{u}_{21}) + N_{21}z_2, \qquad (5.131)$$

$$N_{23}y_2 = N_{23}H_{23}V_{23}u_{23} + N_{23}z_2.$$
(5.132)

T₂ recovers d_{21} linearly independent noisy observations of $\tilde{\boldsymbol{u}}_{21}$ and $\bar{\boldsymbol{u}}_{21}$, and d_{23} linearly independent noisy observations of \boldsymbol{u}_{23} from \boldsymbol{y}_2 since $\boldsymbol{N}_2 = [\boldsymbol{N}_{21}^T \boldsymbol{N}_{23}^T]^T$ has sufficient row rank $d_{21} + d_{23}$ almost surely. Hence, T₂ achieves a number of $d_{21} + d_{23}$ DoF.

On the receiver side of T_3 , we have:

$$\boldsymbol{y}_{3} = \boldsymbol{H}_{31} [\tilde{\boldsymbol{V}}_{31} \ \bar{\boldsymbol{V}}_{31}] [\frac{\tilde{\boldsymbol{u}}_{31}}{\tilde{\boldsymbol{u}}_{31}}] + [\boldsymbol{H}_{31} \tilde{\boldsymbol{V}}_{21} \tilde{\boldsymbol{u}}_{21} + \boldsymbol{H}_{32} \tilde{\boldsymbol{V}}_{12} \tilde{\boldsymbol{u}}_{12}] + \boldsymbol{H}_{32} \boldsymbol{V}_{32} \boldsymbol{u}_{32} + \boldsymbol{z}_{3}.$$
(5.133)

At T₃, the signals \bar{u}_{21} and \bar{u}_{21} are not observed due to (5.112) and (5.114). We need the following constraint to guarantee the linear independence of the desired signals and the interference:

$$0 \le \tilde{d}_{31} + \bar{d}_{31} + \tilde{d}_{21} + \tilde{d}_{32} \le M_3. \tag{5.134}$$

We use zero-forcing matrices N_{31} and N_{32} of dimensions $d_{31} \times M_3$ and $d_{32} \times M_3$, satisfying:

$$N_{31}H_{32}(V_{32} + \tilde{V}_{12}) = \mathbf{0}_{d_{31} \times (d_{32} + \tilde{d}_{12})}, \qquad (5.135)$$

$$N_{32}H_{31}(V_{31} + \bar{V}_{31} + V_{21}) = \mathbf{0}_{d_{23} \times (d_{31} + \tilde{d}_{21})}, \qquad (5.136)$$

to zero-force the interference space and to separate the two dedicated information signals. With this scheme, receiver T_3 obtains:

$$N_{31}y_3 = N_{31}H_{31}(\tilde{V}_{31}\tilde{u}_{31} + \bar{V}_{31}\bar{u}_{31}) + N_{31}z_3, \qquad (5.137)$$

$$N_{32}y_3 = N_{32}H_{32}V_{32}u_{32} + N_{32}z_3.$$
 (5.138)

Thus, T_3 can recover d_{31} linearly independent noisy observations of $\tilde{\boldsymbol{u}}_{31}$ and $\bar{\boldsymbol{u}}_{31}$, and d_{32} linearly independent noisy observations of \boldsymbol{u}_{32} from \boldsymbol{y}_3 since $\boldsymbol{N}_3 = [\boldsymbol{N}_3^{\mathsf{T}} \boldsymbol{N}_3^{\mathsf{T}} \boldsymbol{2}]^{\mathsf{T}}$ has sufficient row rank $d_{31} + d_{32}$ almost surely. Hence, T_2 can decode its dedicated signals and achieves $d_{31} + d_{32}$ DoF.

Assembling all constraints on the achievable DoF, yields:

$$\begin{split} \tilde{d}_{32} &= \tilde{d}_{23} \leq [M_2 + M_3 - M_1]^+, \\ &= \bar{d}_{31} \leq \min(M_3, M_1 - M_2), \\ &= \bar{d}_{21} \leq \min(M_2, M_1 - M_3), \\ &= \bar{d}_{12} \leq M_2 - M_3, \\ \tilde{d}_{12} &= \bar{d}_{12} + \tilde{d}_{32} + \tilde{d}_{13} \leq M_1, \\ &= \tilde{d}_{21} + \bar{d}_{21} + \tilde{d}_{31} + \tilde{d}_{23} \leq M_2, \\ &= \tilde{d}_{31} + \bar{d}_{31} + \tilde{d}_{21} + \tilde{d}_{32} \leq M_3. \end{split}$$

Note that real-valued DoF can be approximated by using signal-extensions over multiple time-slots [38, 78]. By maximizing d_{Σ} subject to these non-negative constraints, we get the maximum achievable sum-DoF of this scheme. This maximization is a linear optimization problem which can be solved by using the simplex method. The maximization yields a sum-DoF of $2M_2$. To verify this, we set:

$$\tilde{d}_{32} = \tilde{d}_{23} = (M_2 + M_3 - M_1)^+,$$
 (5.139)

$$\bar{d}_{31} = \min(M_3, M_1 - M_2),$$
 (5.140)

$$\bar{d}_{21} = \min(M_2, M_1 - M_3),$$
 (5.141)

$$\bar{d}_{12} = M_2 - M_3. \tag{5.142}$$

This allocation satisfies all the DoF constraints above, and leads to:

$$d_{\Sigma} = 2\tilde{d}_{32} + \bar{d}_{31} + \bar{d}_{21} + \bar{d}_{12} = 2M_2.$$

5.4 Symmetric Degrees of Freedom of the MIMO $M_{\rm Tx} \times M_{\rm Rx}$ 3-Way Channel

In the previous section, it has been shown that the DoF of the $M_i \times M_i$ 3-way channel are limited by M_2 , i.e., the rank of the strongest subchannel. In this section, we investigate how the DoF are limited when using a symmetrical number of M_{Tx} transmit and M_{Rx} receive antennas. For multi-way conferencing situations with eminently high and almost symmetric rate demands, as in video conferences for instance, using such homogeneously equipped devices is clearly beneficial when compared to heterogeneous devices. Another particular gain from this homogeneous setup is that the symmetric DoF allocation providing complete fairness among all users is sum-DoF optimal.



Figure 5.14: The homogeneous MIMO 3-way channel (or Δ -channel) with M_{Tx} transmit and M_{Rx} receive antennas at each user T_i , with i = 1, 2, 3.

5.4.1 MIMO $M_{Tx} \times M_{Rx}$ 3-Way Channel

We study the DoF of the homogeneous $M_{\mathrm{Tx}} \times M_{\mathrm{Rx}}$ MIMO 3-way channel with constant channel coefficients and with a number of $M_{\mathrm{Tx}_i} = M_{\mathrm{Tx}}$ transmit antennas at each transceiver and $M_{\mathrm{Rx}_i} = M_{\mathrm{Rx}}$ receive antennas.

The system model of the $M_{\rm Tx} \times M_{\rm Rx}$ 3-way channel is closely related to the model defined in Section 5.3.1. The input-output relationship is analogously defined by (5.86) to (5.88), but for the given homogeneous antenna configurations. The encoding and decoding functions defined in Section 5.3.1 also carry over to this channel model.

5.4.2 Upper Bounds

Cut-set bounds

We begin with considering the cut-set bounds for the MIMO 3-way channel:

$$d_{ji} + d_{ki} \le \min(M_{\mathrm{Tx}}, 2M_{\mathrm{Rx}}),$$
 (5.143)

$$d_{ij} + d_{ik} \le \min(2M_{\mathrm{Tx}}, M_{\mathrm{Rx}}).$$
 (5.144)

The right-hand side of (5.143) is the rank of the MIMO channel between T_i and a receiver formed by enabling full cooperation between T_j and T_k , with channel matrix

 $[\boldsymbol{H}_{ji}^{\mathsf{T}}\boldsymbol{H}_{ki}^{\mathsf{T}}]^{\mathsf{T}}$. A similar interpretation holds for the second bound. Combining (5.143) and (5.144) provides the sum-DoF bound:

$$d_{\Sigma} \le \min(3M_{\mathrm{Tx}}, 3M_{\mathrm{Rx}}).$$
 (5.145)

Genie-aided bounds

We first assume that $M_{\text{Rx}} \ge M_{\text{Tx}}$. Assume every node can obtain its dedicated messages with an arbitrary small probability of error. Hence, T_2 can decode W_{21}, W_{23} reliably from its available information, i.e., from \boldsymbol{y}_2^N , W_{12} and W_{32} , as shown in (5.90). Furthermore, we provide W_{31} to T_2 as side-information. We also provide T_2 with the correction-noise signal:

$$\tilde{z}_{2}^{N} = z_{1}^{N} - H_{13}H_{23}z_{2}^{N},$$
 (5.146)

as side-information⁹. Now, T₂ knows its decoded W_{21} and W_{31} by side-information. With W_{21} , W_{31} , user T₂ can generate $\boldsymbol{x}_1(1)$. By subtracting $\boldsymbol{H}_{21}\boldsymbol{x}_1(1)$ from $\boldsymbol{y}_2(1)$, and multiplying the result with \boldsymbol{H}_{23} , T₂ can recover a noisy observation of $\boldsymbol{x}_3(1)$ given by $\boldsymbol{x}_3(1) + \boldsymbol{H}_{23}\boldsymbol{z}_2(1)$. Next, T₂ multiplies this noisy observation by \boldsymbol{H}_{13} , and adds $\boldsymbol{H}_{12}\boldsymbol{x}_2(1)$ and $\boldsymbol{\tilde{z}}_2(1)$ to it to obtain $\boldsymbol{y}_1(1)$. Thus, T₂ obtains the first instance of \boldsymbol{y}_1^N . Knowing $\boldsymbol{y}_1(1)$, W_{21} and W_{31} , T₂ can generate $\boldsymbol{x}_1(2)$ (cf. (5.89)). Using $\boldsymbol{x}_1(2)$ again with $\boldsymbol{y}_2(2)$, T₂ can generate $\boldsymbol{y}_1(2)$ and $\boldsymbol{x}_1(3)$. T₂ proceeds this way until all instances (up to the N-th instance) of \boldsymbol{y}_1^N have been generated. Now, having \boldsymbol{y}_1^N , W_{21} , and W_{31} , i.e., the same information as T₁, T₂ can decode W_{13} (cf. (5.90)). Therefore, given W_{13} and $\boldsymbol{\tilde{z}}_2^N$ as side-information, T₂ can decode W_{21} , W_{23} and W_{13} . Hence, the DoF of these messages are almost surely upper bounded by:

$$d_{21} + d_{23} + d_{31} \le \operatorname{rank}([\boldsymbol{H}_{21} \, \boldsymbol{H}_{23}]) \tag{5.147}$$

$$= \min(M_{\rm Rx}, 2M_{\rm Tx}).$$
 (5.148)

We can apply a similar approach to bound:

$$d_{31} + d_{32} + d_{12} \le \operatorname{rank}([\boldsymbol{H}_{31} \, \boldsymbol{H}_{32}]) \tag{5.149}$$

$$= \min(M_{\rm Rx}, 2M_{\rm Tx}),$$
 (5.150)

by providing W_{21} and the correction noise-signal:

$$\tilde{\boldsymbol{z}}_{3}^{N} = \boldsymbol{z}_{1}^{N} - \boldsymbol{H}_{12}\boldsymbol{H}_{32}\boldsymbol{z}_{3}^{N}$$
 (5.151)

to T₃. As a result, T₃ can construct \boldsymbol{y}_1^N and decode W_{12} reliably. Combining (5.148) and (5.150), bounds the sum-DoF to:

$$d_{\Sigma} \le \min(2M_{\mathrm{Rx}}, 4M_{\mathrm{Tx}}). \tag{5.152}$$

For the contrary case, we assume that $M_{\text{Rx}} < M_{\text{Tx}}$ holds. We enhance the number of receive antennas at all receivers to $\mathring{M}_{\text{Rx}} = M_{\text{Tx}}$. The effective channel output at T₃ becomes:

$$\mathbf{\mathring{y}}_{3}(n) = \mathbf{\mathring{H}}_{31}\mathbf{x}_{1}(n) + \mathbf{\mathring{H}}_{32}\mathbf{x}_{2}(n) + \mathbf{\mathring{z}}_{3}(n),$$
(5.153)

⁹The pseudo-inverse H_{23} exists almost surely, since H_{23} is an $M_{\text{Rx}} \times M_{\text{Tx}}$ matrix with $M_{\text{Rx}} \ge M_{\text{Tx}}$.

with the extended $M_{\text{Tx}} \times M_{\text{Tx}}$ matrices¹⁰ \mathring{H}_{31} , \mathring{H}_{32} , and the extended $M_{\text{Tx}} \times 1$ noise vector $\mathring{z}_3(n)$. We can apply the upper bounds derived in (5.148) and (5.150) now, leading to:

$$d_{21} + d_{23} + d_{31} \le M_{\rm Tx},\tag{5.154}$$

$$d_{31} + d_{32} + d_{12} \le M_{\rm Tx},\tag{5.155}$$

$$d_{\Sigma} = 2\min(\mathring{M}_{\mathrm{Rx}}, 2M_{\mathrm{Tx}}) = \min(2M_{\mathrm{Tx}}, 4M_{\mathrm{Tx}}) = 2M_{\mathrm{Tx}}.$$
 (5.156)

Combining these bounds with the cut-set upper bounds yields:

$$d_{\Sigma} \le \min(2M_{\rm Rx}, 4M_{\rm Tx}, 3M_{\rm Rx}, 3M_{\rm Tx}) = \min(2M_{\rm Rx}, 3M_{\rm Tx}), \text{ if } M_{\rm Tx} \le M_{\rm Rx}, d_{\Sigma} \le \min(2M_{\rm Tx}, 3M_{\rm Rx}, 3M_{\rm Tx})$$
(5.157)

$$= \min(2M_{\rm Tx}, 3M_{\rm Rx}, 0, M_{\rm Tx}) = \min(2M_{\rm Tx}, 3M_{\rm Rx}), \text{ if } M_{\rm Tx} > M_{\rm Rx}.$$
(5.158)

Theorem 5.8. The DoF of the $M_{\text{Tx}} \times M_{\text{Rx}}$ MIMO 3 -way channel with M_{Tx} transmit antennas and M_{Rx} receive antennas at each user T_i are:

$$d_{\Sigma} = \begin{cases} \min(2M_{\rm Rx}, 3M_{\rm Tx}), & \text{if } M_{\rm Tx} \le M_{\rm Rx}, \\ \min(2M_{\rm Tx}, 3M_{\rm Rx}), & \text{if } M_{\rm Tx} > M_{\rm Rx}. \end{cases}$$
(5.159)

5.4.3 Achievability

The following two communication schemes provide achievability of the upper bounds in Theorem 5.8. Note that, symbol-extensions on the constant MIMO channels over multiple time-slots are used to achieve non-integer DoF per user, cf. [38], [78]. We discern two cases: either $M_{\text{Tx}} \leq M_{\text{Rx}}$ or $M_{\text{Tx}} > M_{\text{Rx}}$ holds.

Case $M_{\mathrm{Tx}} \leq M_{\mathrm{Rx}}$ with $d_{\Sigma} = 3M_{\mathrm{Tx}}$

The dominant term in (5.159) yields $3M_{\text{Tx}}$ if $3M_{\text{Tx}} \leq 2M_{\text{Rx}}$ holds. We use the following symmetric DoF allocation:

$$d = d_{ij} = d_{ji}.$$
 (5.160)

We further decompose the symmetric DoF d further for IA (tilde-notation, \bar{d}) and for beam-forming (bar-notation, \bar{d}):

$$\bar{d} = \bar{d}_{ij} = \bar{d}_{ji},\tag{5.161}$$

$$\tilde{d} = \tilde{d}_{ij} = \tilde{d}_{ji}, \tag{5.162}$$

$$d = \bar{d}_i + \tilde{d}_i. \tag{5.163}$$

In other words, we demand that bidirectional signals pairwise occupy the same number of DoF. According to the assumptions on M_{Tx} and M_{Rx} , the following bounds must hold:

$$0 \le 2d \le M_{\text{Tx}},$$
 (5.164)

$$0 \le 3d \le \min(M_{\rm Rx}, 2M_{\rm Tx}),$$
 (5.165)

¹⁰The inverses exist almost surely.

so that all upper bounds provided in Section 5.4.2 are satisfied.

Messages W_{ji} are encoded into complex-valued symbol streams $\tilde{\boldsymbol{u}}_{ji} \in \mathbb{C}^{\bar{d}\times 1}$ and $\bar{\boldsymbol{u}}_{ji} \in \mathbb{C}^{\bar{d}\times 1}$. These symbol streams are pre-coded at the transmitters and post-coded at the receivers, so that the proposed sum-DoF are achieved. For pre-coding, we use beamforming matrices $\tilde{\boldsymbol{V}}_{ji} \in \mathbb{C}^{M_{Tx} \times \bar{d}}$ and $\bar{\boldsymbol{V}}_{ji} \in \mathbb{C}^{M_{Tx} \times \bar{d}}$. Transmit signals \boldsymbol{x}_i are constructed from the pre-coded symbol streams as:

$$\boldsymbol{x}_{i} = \begin{bmatrix} \tilde{\boldsymbol{V}}_{ji} \ \bar{\boldsymbol{V}}_{ji} \end{bmatrix} \begin{bmatrix} \tilde{\boldsymbol{u}}_{ji} \\ \bar{\boldsymbol{u}}_{ji} \end{bmatrix} + \begin{bmatrix} \tilde{\boldsymbol{V}}_{ki} \ \bar{\boldsymbol{V}}_{ki} \end{bmatrix} \begin{bmatrix} \tilde{\boldsymbol{u}}_{ki} \\ \bar{\boldsymbol{u}}_{ki} \end{bmatrix}.$$
(5.166)

First, we consider the intersection space of the two incident subchannels at the receiver of T_j . The number of dimensions for $\text{span}(H_{ji}) \cap \text{span}(H_{jk})$ is computed by Lemma 5.6 as given in the appendix. It has $0 \leq [2M_{\text{Tx}} - M_{\text{Rx}}]^+ \leq \frac{1}{3}M_{\text{Rx}}$ dimensions, since $3M_{\text{Tx}} \leq 2M_{\text{Rx}}$. We fix:

$$\tilde{d} = [2M_{\rm Tx} - M_{\rm Rx}]^+,$$
 (5.167)

and design \tilde{V}_{ji} such that the two dedicated signals remain distinct, while the undesired interference is aligned at each undesired receiver:

$$\operatorname{span}(\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ki}) = \operatorname{span}(\boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{ik}). \tag{5.168}$$

After this first part of pre-coding, a number of:

$$\bar{M}_{Tx} = M_{Tx} - 2\tilde{d} \ge 0,$$
 (5.169)

$$\bar{M}_{Rx} = M_{Rx} - 3\bar{d} \ge 0,$$
 (5.170)

transmit and receive dimensions remain available at each user, respectively. Since the complete intersection space is already consumed by IA, we have $[2\bar{M}_{Tx} - \bar{M}_{Rx}]^* = 0$ and hence, $2\bar{M}_{Tx} \leq \bar{M}_{Rx}$ holds. The remaining DoF are allocated by:

$$\bar{d} = \frac{1}{2}\bar{M}_{Tx}$$
, (5.171)

for all $k \in \mathcal{K}$. The beam-forming matrices \overline{V}_{ji} and \overline{V}_{jk} are chosen such that the two signals x_i and x_k received are linearly independent at T_j . This allocation satisfies both upper bounds. The received signals at T_j yield:

$$\begin{aligned} \boldsymbol{y}_{j} &= \left(\boldsymbol{H}_{ji} [\tilde{\boldsymbol{V}}_{ji} \ \bar{\boldsymbol{V}}_{ji}] [\frac{\tilde{\boldsymbol{u}}_{ji}}{\tilde{\boldsymbol{u}}_{ji}}] + \boldsymbol{H}_{jk} [\tilde{\boldsymbol{V}}_{jk} \ \bar{\boldsymbol{V}}_{jk}] [\frac{\tilde{\boldsymbol{u}}_{k}}{\tilde{\boldsymbol{u}}_{jk}}] \right) + \\ & \left(\boldsymbol{H}_{ji} [\tilde{\boldsymbol{V}}_{ki} \ \bar{\boldsymbol{V}}_{ki}] [\frac{\tilde{\boldsymbol{u}}_{ki}}{\tilde{\boldsymbol{u}}_{ki}}] + \boldsymbol{H}_{jk} [\tilde{\boldsymbol{V}}_{ik} \ \bar{\boldsymbol{V}}_{ik}] [\frac{\tilde{\boldsymbol{u}}_{ik}}{\tilde{\boldsymbol{u}}_{ki}}] + \boldsymbol{z}_{j}. \end{aligned}$$

The first sum in brackets describes the dedicated signals. The second sum in brackets describes the interfering signals at T_j (with $H_{ji}\tilde{V}_{ki}$ and $H_{jk}\tilde{V}_{ik}$ aligned). The signal and interference subspaces are linearly independent, since the composite $M_{Rx} \times (3\tilde{d}+4\bar{d})$ matrix

$$[\boldsymbol{H}_{ji}[\boldsymbol{\tilde{V}}_{ji}\;\boldsymbol{\bar{V}}_{ji}\;\boldsymbol{\tilde{V}}_{ki}\;\boldsymbol{\bar{V}}_{ki}]\boldsymbol{H}_{jk}[\boldsymbol{\tilde{V}}_{jk}\;\boldsymbol{\bar{V}}_{jk}\;\boldsymbol{\bar{V}}_{ik}]], \qquad (5.172)$$

has full column rank, almost surely, due to:

$$3\tilde{d} + 4\bar{d} = 2M_{\rm Tx} - [2M_{\rm Tx} - M_{\rm Rx}]^+ < M_{\rm Rx}.$$
(5.173)

In the post-coding step, each receiver T_j uses a composite zero-forcing matrix $N_j = [N_{ji}^T N_{jk}^T]^T$ to separate and decode its two dedicated signals and to eliminate the interfering signals. The received signals y_j are filtered by the two corresponding zero-forcing matrices $N_{ji}, N_{jk} \in \mathbb{C}^{d \times M_{\text{Rx}}}$, with:

$$\boldsymbol{N}_{ji} [\boldsymbol{H}_{jk} (\tilde{\boldsymbol{V}}_{jk} + \bar{\boldsymbol{V}}_{jk} + \tilde{\boldsymbol{V}}_{ik} + \bar{\boldsymbol{V}}_{ik}) + \boldsymbol{H}_{ji} \bar{\boldsymbol{V}}_{ki}] = \boldsymbol{0}_{d \times (2d + \bar{d})}, \qquad (5.174)$$

$$\boldsymbol{N}_{jk} [\boldsymbol{H}_{ji}(\tilde{\boldsymbol{V}}_{ji} + \bar{\boldsymbol{V}}_{ji} + \tilde{\boldsymbol{V}}_{ki} + \bar{\boldsymbol{V}}_{ki}) + \boldsymbol{H}_{jk} \bar{\boldsymbol{V}}_{ik}] = \boldsymbol{0}_{d \times (2d + \bar{d})},$$
(5.175)

so that filtering with $N_{ji}y_j$ and $N_{jk}y_j$ provides $d_{ji} + d_{jk} = 2d$ noisy interference-free streams of dedicated signals at T_j :

$$\boldsymbol{N}_{ji}\boldsymbol{y}_{j} = \boldsymbol{N}_{ji}\boldsymbol{H}_{ji}\left[\tilde{\boldsymbol{V}}_{ji}\tilde{\boldsymbol{u}}_{ji} + \bar{\boldsymbol{V}}_{ji}\bar{\boldsymbol{u}}_{ji}\right] + \boldsymbol{N}_{ji}\boldsymbol{z}_{j}, \qquad (5.176)$$

$$\boldsymbol{N}_{jk}\boldsymbol{y}_{j} = \boldsymbol{N}_{jk}\boldsymbol{H}_{jk}\left[\tilde{\boldsymbol{V}}_{jk}\tilde{\boldsymbol{u}}_{jk} + \bar{\boldsymbol{V}}_{jk}\bar{\boldsymbol{u}}_{jk}\right] + \boldsymbol{N}_{jk}\boldsymbol{z}_{j}.$$
(5.177)

Thus, each user T_i can decode its two dedicated streams with:

$$d_{ji} + d_{jk} = 2d = 2(\tilde{d} + \bar{d}) = M_{\mathrm{Tx}}.$$
(5.178)

Altogether, $3M_{\text{Tx}}$ DoF in the first term of (5.159) are achieved:

$$d_{\Sigma} \leq 6d = 3M_{\mathrm{Tx}}.$$

Case $M_{\mathrm{Tx}} \leq M_{\mathrm{Rx}}$ with $d_{\Sigma} = 2M_{\mathrm{Rx}}$

On the other hand, the dominant term in (5.159) yields $2M_{\text{Rx}}$ if $2M_{\text{Rx}} \leq 3M_{\text{Tx}}$ holds. Then, $\text{span}(\boldsymbol{H}_{ji}) \cap \text{span}(\boldsymbol{H}_{jk})$ at T_j has $2M_{\text{Tx}} - M_{\text{Rx}} > \frac{1}{3}M_{\text{Rx}}$ dimensions. We allocate:

$$d = \tilde{d} = \frac{1}{3}M_{\text{Rx}}.$$
 (5.179)

for all $i \in \mathcal{K}$. This allocation satisfies all upper bounds:

$$0 \le 2d \le M_{\rm Tx},$$
 (5.180)

$$0 \le 3d \le M_{\text{Rx}},$$
 (5.181)

and no remaining dimensions are left at the receivers. The symbol streams $\tilde{\boldsymbol{u}}_{ji} \in \mathbb{C}^{\bar{d}\times 1}$ are pre-coded by the beam-forming matrices $\tilde{\boldsymbol{V}}_{ji} \in \mathbb{C}^{M_{T_x} \times \bar{d}}$ and are aligned analogously to (5.168). We have no symbol streams $\bar{\boldsymbol{u}}_{ji} \in \mathbb{C}^{\bar{d}\times 1}$, since $\bar{d} = 0$. Hence the received signal at T_j is:

$$\boldsymbol{y}_{j} = \left(\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ji}\tilde{\boldsymbol{u}}_{ji} + \boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{jk}\tilde{\boldsymbol{u}}_{jk}\right) + (5.182)$$
$$\left(\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ki}\tilde{\boldsymbol{u}}_{ki} + \boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{ik}\tilde{\boldsymbol{u}}_{ik}\right) + \boldsymbol{z}_{j}.$$

The composite $M_{\text{Rx}} \times 3\tilde{d}$ - dimensional matrix:

$$[\boldsymbol{H}_{ji}\boldsymbol{\tilde{V}}_{ji} \ \boldsymbol{H}_{jk}\boldsymbol{\tilde{V}}_{jk} \ \boldsymbol{H}_{ji}\boldsymbol{\tilde{V}}_{ki}], \qquad (5.183)$$

has full column rank, almost surely, so that dedicated and interfering signals are linearly independent. For post-coding, the zero-forcing matrices are chosen as:

$$N_{ji}(\boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{jk} + \boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ki}) = \boldsymbol{0}_{\tilde{d}\times 2\tilde{d}}, \qquad (5.184)$$

$$\boldsymbol{N}_{jk}(\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ji}+\boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{ik})=\boldsymbol{0}_{\tilde{d}\times 2\tilde{d}},$$
(5.185)

such that the filtered signals yield:

$$\boldsymbol{N}_{ji}\boldsymbol{y}_{j} = \boldsymbol{N}_{ji}\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ji}\tilde{\boldsymbol{u}}_{ji} + \boldsymbol{N}_{ji}\boldsymbol{z}_{j}, \qquad (5.186)$$

$$N_{jk}\boldsymbol{y}_{j} = N_{jk}\boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{jk}\tilde{\boldsymbol{u}}_{jk} + N_{jk}\boldsymbol{z}_{j}.$$
(5.187)

Each user \mathbf{T}_j can decode noisy but interference-free versions of its two dedicated streams and achieves:

$$d_{ji} + d_{jk} = 2\tilde{d} = \frac{2}{3}M_{\text{Rx}},$$
 (5.188)

so that the sum-DoF are:

$$d_{\Sigma} = 6d = 2M_{\text{Rx}}.$$
 (5.189)

Thence, the upper bound $\min(2M_{\text{Rx}}, 3M_{\text{Tx}})$ is shown to be achievable.

Case $M_{\mathrm{Tx}} > M_{\mathrm{Rx}}$ with $d_{\Sigma} = 2M_{\mathrm{Tx}}$

The upper bound (5.159) yields $2M_{\text{Tx}}$ if $2M_{\text{Tx}} \leq 3M_{\text{Rx}}$ holds. Again we use the symmetric DoF allocation as defined in (5.160) to (5.163). In this case, the following upper bounds must hold:

$$0 \le 2d \le M_{\rm Rx},\tag{5.190}$$

$$0 \le 3d \le M_{Tx}$$
. (5.191)

Since $M_{\text{Tx}} > M_{\text{Rx}}$, zero-forcing beam-forming [77, 78] is applicable and we allocate:

$$\bar{d} = M_{Tx} - M_{Rx} < \frac{1}{2}M_{Rx}$$
 (5.192)

dimensions. Analogous to Section 5.4.3, we pre-code the symbol-streams $\bar{\boldsymbol{u}}_{ji}$ and $\tilde{\boldsymbol{u}}_{ji}$ to construct the transmit signal \boldsymbol{x}_i as in (5.166). The beam-forming matrix $\bar{\boldsymbol{V}}_{ki}$ has $M_{\text{Tx}} \times \bar{d}$ dimensions and is designed to cast the interfering signal into the $(M_{\text{Tx}} - M_{\text{Rx}})$ -dimensional null-space of T_j :

$$\boldsymbol{H}_{ji} \bar{\boldsymbol{V}}_{ki} = \boldsymbol{0}_{M_{\text{Bx}} \times \bar{d}}.$$
(5.193)

For the next step, the number of remaining transmit and receive dimensions per user available for IA are:

$$M_{\rm Tx} = M_{\rm Tx} - 2(M_{\rm Tx} - M_{\rm Rx}) = 2M_{\rm Rx} - M_{\rm Tx},$$
 (5.194)

$$M_{\rm Rx} = M_{\rm Rx} - 2(M_{\rm Tx} - M_{\rm Rx}) = 3M_{\rm Rx} - 2M_{\rm Tx}.$$
 (5.195)

Since $2\tilde{M}_{Tx} > \tilde{M}_{Rx}$ holds, the remaining dimensions suffice for IA. Furthermore, since $3\tilde{M}_{Tx} > 2\tilde{M}_{Rx}$, more than $\tilde{M}_{Rx}/3$ dimensions are available for IA between each user pair (cf. Lemma 5.6). To establish a fair scheme, we set:

$$\tilde{d} = \frac{1}{3}\tilde{M}_{\rm Rx} = M_{\rm Rx} - \frac{2}{3}M_{\rm Tx}.$$
 (5.196)

The beam-forming matrices $\tilde{\boldsymbol{V}}_{ki}$ and $\tilde{\boldsymbol{V}}_{ik}$, each with $M_{\text{Tx}} \times \tilde{d}$ dimensions, are chosen such that the bidirectional interference signals are aligned at receiver T_j , as analogously done in (5.168). Due to zero-forcing beam-forming, the symbol streams $\bar{\boldsymbol{u}}_{ki}$ and $\bar{\boldsymbol{u}}_{ik}$ are not received at T_j , so that we obtain:

$$\boldsymbol{y}_{j} = \left(\boldsymbol{H}_{ji} [\boldsymbol{\tilde{V}}_{ji} \boldsymbol{\bar{V}}_{ji}] \begin{bmatrix} \boldsymbol{\tilde{u}}_{ji} \\ \boldsymbol{\tilde{u}}_{ji} \end{bmatrix} + \boldsymbol{H}_{jk} [\boldsymbol{\tilde{V}}_{jk} \boldsymbol{\bar{V}}_{jk}] \begin{bmatrix} \boldsymbol{\tilde{u}}_{jk} \\ \boldsymbol{\tilde{u}}_{jk} \end{bmatrix} \right) +$$

$$\left(\boldsymbol{H}_{ji} \boldsymbol{\tilde{V}}_{ki} \boldsymbol{\tilde{u}}_{ki} + \boldsymbol{H}_{jk} \boldsymbol{\tilde{V}}_{ik} \boldsymbol{\tilde{u}}_{ik} \right) + \boldsymbol{z}_{j}.$$

$$(5.197)$$

The signal and interference subspaces are linearly independent, almost surely, since the composite matrix:

$$[\boldsymbol{H}_{ji}[\tilde{\boldsymbol{V}}_{ji}\;\bar{\boldsymbol{V}}_{ji}\;\tilde{\boldsymbol{V}}_{ki}]\;\boldsymbol{H}_{jk}[\tilde{\boldsymbol{V}}_{jk}\;\bar{\boldsymbol{V}}_{jk}]], \qquad (5.198)$$

of $M_{\text{Rx}} \times (3\tilde{d} + 2\bar{d})$ dimensions $(\boldsymbol{H}_{ji}\tilde{\boldsymbol{V}}_{ki} \text{ and } \boldsymbol{H}_{jk}\tilde{\boldsymbol{V}}_{ik}$ are aligned) has full column rank.

For post-coding at the receivers, we use zero-forcing matrices N_{ji} of $d \times M_{\text{Rx}}$ dimensions as given in (5.174) and (5.175), but for differently allocated d according to (5.197). Analogously, the following signals are obtained after filtering:

$$\begin{split} \boldsymbol{N}_{ji}\boldsymbol{y}_{j} &= \boldsymbol{N}_{ji}\boldsymbol{H}_{ji}\left[\tilde{\boldsymbol{V}}_{ji}\tilde{\boldsymbol{u}}_{ji} + \bar{\boldsymbol{V}}_{ji}\bar{\boldsymbol{u}}_{ji}\right] + \boldsymbol{N}_{ji}\boldsymbol{z}_{j}, \\ \boldsymbol{N}_{jk}\boldsymbol{y}_{j} &= \boldsymbol{N}_{jk}\boldsymbol{H}_{jk}\left[\tilde{\boldsymbol{V}}_{jk}\tilde{\boldsymbol{u}}_{jk} + \bar{\boldsymbol{V}}_{jk}\bar{\boldsymbol{u}}_{jk}\right] + \boldsymbol{N}_{jk}\boldsymbol{z}_{j}. \end{split}$$

 T_i decodes two noisy but interference-free dedicated streams:

$$d_{ji} + d_{jk} = 2d = 2(\tilde{d} + \bar{d}) = \frac{2}{3}M_{\mathrm{Tx}},$$
(5.199)

so that the sum-DoF of $2M_{\text{Tx}}$ are achieved:

$$d_{\Sigma} = 6d = 2M_{\rm Tx}.$$
 (5.200)

Case $M_{\rm Tx} > M_{\rm Rx}$ with $d_{\Sigma} = 3M_{\rm Rx}$

In the case $2M_{\text{Tx}} \ge 3M_{\text{Rx}}$, the upper bound (5.159) yields $3M_{\text{Rx}}$. Now it suffices to use zero-forcing beam-forming only. IA is actually not necessary for this case. We allocate the DoF:

$$d = \bar{d} = \frac{1}{2}M_{\text{Rx}}$$
 (5.201)

for all $k \in \mathcal{K}$, satisfying (5.190) and (5.191). We use the beam-forming matrices $\bar{\mathbf{V}}_{ji}$ with $M_{\text{Tx}} \times \bar{d}$ dimensions and cast interference to the null-space of the undesired receivers:

$$\boldsymbol{H}_{ji} \bar{\boldsymbol{V}}_{ki} = \boldsymbol{0}_{M_{\mathrm{Rx}} \times \bar{d}}.$$
 (5.202)



Figure 5.15: Sum-DoF d_{Σ} for the parameter plane of $0 \leq M_{\text{Tx}} \leq 6$ transmit antennas and $0 \leq M_{\text{Rx}} \leq 6$ receive antennas.

The received signal at receiver T_j is:

$$\boldsymbol{y}_{j} = \boldsymbol{H}_{ji} \bar{\boldsymbol{V}}_{ji} \bar{\boldsymbol{u}}_{ji} + \boldsymbol{H}_{jk} \bar{\boldsymbol{V}}_{jk} \bar{\boldsymbol{u}}_{jk} + \boldsymbol{z}_{j}.$$
(5.203)

The dedicated signals are linearly independent, almost surely, since the composite $M_{\text{Rx}} \times 2\bar{d}$ matrix has full column rank:

$$[\boldsymbol{H}_{ji}\boldsymbol{\bar{V}}_{ji} \ \boldsymbol{H}_{jk}\boldsymbol{\bar{V}}_{jk}]. \tag{5.204}$$

For post-coding at receiver T_j , we use the zero-forcing matrices N_{ji} and N_{jk} of $\bar{d} \times M_{Rx}$ dimensions each so that (5.184) and (5.185) hold. We obtain the following filtered signals:

$$\boldsymbol{N}_{ji}\boldsymbol{y}_{j} = \boldsymbol{N}_{ji}\boldsymbol{H}_{ji}\bar{\boldsymbol{V}}_{ji}\bar{\boldsymbol{u}}_{ji} + \boldsymbol{N}_{ji}\boldsymbol{z}_{j}, \qquad (5.205)$$

$$\boldsymbol{N}_{jk}\boldsymbol{y}_{j} = \boldsymbol{N}_{jk}\boldsymbol{H}_{jk}\bar{\boldsymbol{V}}_{jk}\boldsymbol{\bar{u}}_{jk} + \boldsymbol{N}_{jk}\boldsymbol{z}_{j}.$$
(5.206)

Each receiver T_j can decode:

$$d_{ji} + d_{jk} = 2d = 2\bar{d} = M_{\rm Rx},\tag{5.207}$$

and achieves the sum-DoF of:

$$d_{\Sigma} = 6d = 3M_{\rm Rx}.$$
 (5.208)

Altogether, the upper bound $\min(2M_{\text{Tx}}, 3M_{\text{Rx}})$ is also shown to be achievable. Note that complete fairness is maintained in each case. This concludes the proof of Theorem 5.8.

5.4.4 Discussion

The parameter plane of the symmetric DoF depicted in Figure 5.15 provides a symmetry along the intersecting line $M_{\rm T} = M_{\rm R}$ for all parameters $M_{\rm T}$ and $M_{\rm R}$. At that line, the antenna parameters of the achieved DoF are swapped since null-space beamforming and linear independent beam-forming are swapped.

5.5 Extending the CPCM by Multi-Antenna Constraints

In order to expose further analogies between the CPCM and the GMCM, additional constraints are imposed to the CPCM of the 3-way channel, for distinct $i, j, k \in \mathcal{K}$:

$$m_{ji} \le \min(M_{\mathrm{Tx}}, M_{\mathrm{Rx}}), \tag{5.209}$$

$$m_{ji} + m_{ki} \le M_{\mathrm{Tx}},\tag{5.210}$$

$$m_{ij} + m_{ik} \le M_{\rm Rx}.\tag{5.211}$$

These inequalities are intended to mimic the MIMO antenna constraints for a number of M_{Tx} transmit antennas and M_{Rx} receive antennas at each user T_i . We now investigate the intersection subspace of two transmitted polynomials within a received polynomial (cf. Lemma 5.6) in terms of the CPCM.

Let $q_i(x)$ denote a polynomial of M_{Rx} dimensions for each transmitter $\text{Tx}_i, i \in \{1, 2\}$:

$$q_i(x) = \sum_{k=0}^{M_{\rm Rx}-1} q_i^{[k]} x^k.$$
 (5.212)

The two polynomials $q_1(x)$ and $q_2(x)$ from the transceivers Tx_1 and Tx_2 have M_{Rx} dimensions, since the intersection space of all dedicated signals must be decodable within the received signal space of a receiver Rx. We use binary coefficients $q_i^{[0]}, ..., q_i^{[M_{\operatorname{Rx}}-1]} \in \{0, 1\}$, to simplify the computation of the number of intersecting dimensions at Rx.

At each Tx_i , a number of only $\min(M_{Tx}, M_{Rx})$ coefficients in $q_i^{[\cdot]}$ is set to value one at random positions, to indicate that these dimensions are occupied by information. The remaining $[M_{Rx} - \min(M_{Tx}, M_{Rx})]^+$ unused coefficients are set to zero at each Tx_i .

Next, we compute the intersection space of the non-zero coefficients from the shifted polynomials $q_1(x)$ and $q_2(x)$ as received at Rx. The received polynomial q(x) has also M_{Rx} dimensions and is expressed by:

$$q(x) = d_{11}q_1(x) \wedge d_{21}q_2(x) \tag{5.213}$$

$$=\sum_{k=0}^{M_{\rm Rx}-1} q^{[k]} x^k, \tag{5.214}$$

with the bit-wise and-operation (symbol: \wedge). The sought intersection space corresponds to the number of non-zero coefficients in q(x) which is simply computed by the accumulative sum of the received coefficients $q^{[k]} \in \{0, 1\}$:

$$q(1) = \sum_{k=0}^{M_{\mathrm{Rx}}-1} q^{[k]}.$$

Lemma 5.9. The intersection space of two M_{Rx} -dimensional interfering polynomials $q_1(x)$ and $q_2(x)$ of $q_i(1) = \min(M_{\text{Tx}}, M_{\text{Rx}})$ non-zero coefficients, with $i \in \{1, 2\}$, is bounded by:

$$0 \le [2\min(M_{\rm Tx}, M_{\rm Rx}) - M_{\rm Rx}]^+ \le q(1) \le \min(M_{\rm Rx}, M_{\rm Tx})$$
(5.215)

non-zero coefficients in the M_{Rx} -dimensional received polynomial q(x).

Proof:

- (a) Upper bound: $\min(M_{\text{Tx}}, M_{\text{Rx}}) \ge q(1)$ dimensions: The maximal number of non-zero coefficients in the intersection $d_{11}q_1(x) \land d_{21}q_2(x)$ is $q(1) = \min(M_{\text{Tx}}, M_{\text{Rx}})$ and it is achieved if the non-zero coefficients of both shifted polynomials $d_{11}q_1(x)$ and $d_{12}q_2(x)$, each with $q_i(x) = \min(M_{\text{Tx}}, M_{\text{Rx}})$ non-zero coefficients, fully overlap at Rx. Then, it holds $q_1(x) = q_2(x) = q(x)$, and we obtain $q(1) = \min(M_{\text{Tx}}, M_{\text{Rx}})$.
- (b) Lower bound: 0 ≤ [2min(M_{Rx}, M_{Tx}) M_{Rx}]⁺ ≤ q(1) dimensions: At least min(M_{Rx}, M_{Tx}) coefficients are non-zero in both polynomials q₁(x) and q₂(x). Let M_{Tx} < M_{Rx}, so that min(M_{Tx}, M_{Rx}) = M_{Tx} < max(M_{Tx}, M_{Rx}) = M_{Rx}. Let Tx₁ fix its M_{Tx} non-zero coefficients first. This leaves (M_{Rx} - M_{Tx})⁺ dimensions available for M_{Tx} non-zero coefficients in q(x). In case M_{Rx} ≥ 2M_{Tx} holds, it is possible that all M_{Tx} non-zero coefficients are distinct from the M_{Tx} already fixed coefficients at Tx₁, yielding q(1) = 0. Otherwise, if M_{Rx} ≤ 2M_{Tx} holds, there will always be an intersection of non-zero coefficients in at least [2min(M_{Tx}, M_{Rx}) - M_{Rx}]⁺ dimensions. Thus, we obtain the lower bound [2min(M_{Tx}, M_{Rx}) - M_{Rx}]⁺.

As a result, the present lemma is proven.

Note that the lower bound provided here exactly coincides with the number of dimensions of the intersection subspaces as in Lemma 5.6 for $M_{\text{Tx}_i} = M_{\text{Tx}}$. By imposing these antenna constraints, we established an interesting analogy in the CPCM and the GMCM regarding the intersection space of two transmitted signals.

5.6 Summary

In this chapter, we have investigated the 3-way channel and its relationship to the 3user Y-channel. We have observed that the Degrees-of-Freedom of the two aforementioned channels coincide when considering the cyclic polynomial channel model. Our main result was that these channels were interlinked via a Δ -Y relationship, which was inspired by the well-known transformation in the circuit theory of resistor networks. Furthermore, in contrast to most of the other previously considered channels, these two channels were not subject to further constraints on the channel matrices. In order to gain from these insights as given by our proposed channel model, we have examined the 3-way channel w.r.t. two other well-established channel models: the linear deterministic channel model and the Gaussian MIMO channel model with constant coefficients.

Within the linear deterministic channel model, the Δ -Y relationship was also extended to characterize the capacity region of the linear deterministic 3-way channel with reciprocal channel coefficients in the uplink and downlink. We have investigated a capacity-achieving signal alignment scheme that has performed two-way relaying over a single user, and also a capacity-achieving interference alignment scheme for a symmetric special case.

Then, we have considered two cases of the MIMO 3-way channel: The $M_i \times M_i$ 3-way channel and the $M_{\text{Tx}} \times M_{\text{Rx}}$ 3-way channel. Due to the absence of further constraints on the channel matrices, the problem of common eigenvectors in separate subspaces has disappeared for the related MIMO channels. Thus, we have discussed a 3-way channel with M_i transmit and receive antennas at each user. We have provided a scheme that achieved the upper bound of $2M_2$ Degrees-of-Freedom. Basically, it has already sufficed to ignore messages between on the weakest subchannel. But in order to maintain fairness between the users, we have employed a MIMO IA and zero-forcing scheme to provide a fair rate-allocation, that has achieved the sum-Degrees-of-Freedom of $2M_2$. In the second case, we have discussed a 3-way channel with symmetrically equipped users having $M_{\rm Tx}$ transmit and $M_{\rm Rx}$ receive antennas at each transceiver. Accordingly, we have provided IA schemes that have achieve the upper bounds on the symmetric Degrees-of-Freedom.

Altogether, this chapter has provided a comparison between the three different channel models of interest for the example of the 3-way channel.
6 Exploring Practical Applications for Cyclic Interference Alignment

Up to this point, we discussed optimal communication strategies for several multi-user channels described by the CPCM. Now, we turn our attention to some practical issues of IA and cyclic IA by propagation delay on the CPCM in particular:

Potential practical applications of cyclic IA, and

Node placement in Euclidean space for cyclic IA by propagation delay¹.

6.1 On Practical Implementations of IA

The high rate-gains theoretically achievable by IA as introduced in [3] allured numerous researchers in the past few years. There are plenty of works dealing with IA under idealized assumptions as those ones cited in the survey [12]. However, embedding the theoretical concepts of IA into practical systems remains a very challenging task [92]. The restrictions and demands of IA are manifold. The main requirements on IA predominantly discussed in the literature [12], [92] are briefly summarized here:

Approximating the channel capacity in terms of the DoF becomes accurate at high SNR only. However, a practical implementation of mobile devices providing high SNR communication is limited.

The necessity to obtain exact global CSI^2 for each user is a computationally extensive task [40] and increases exponentially with K users. Furthermore, a severe overhead is involved for the acquisition of CSI.

IA schemes are very fragile if there are imperfections in the acquired CSI, as shown in [94], [92] and [95] for instance. If the accuracy drops below a certain threshold, it leads to detrimental leakage interference [40] and thus to an immense performance loss at each receiver.

Most IA schemes demand full and highly exact synchronization (in time and frequency) among multiple users, which is very challenging in practical implementations.

Computing obtain optimal pre-coding matrices is extensive [95]. Algorithms converge rather slowly and residual imperfect alignment causes leakage interference.

¹Parts of this work have been published in [27].

²However, this does not concern blind IA [93].

The infinite time-extensions [3] in time-varying MIMO channels, and the pairing of complementary channels as in ergodic IA [15], cause exponential decoding delays.

Depending on the specific underlying applied IA scheme, there are yet many other fragile requirements to be considered that are not mentioned in the list above. Nonetheless, a few works already focus on the initial implementation of a realistic practical testbed for MIMO IA. For a first feasibility check, the work [96] uses previously measured channel coefficients of a real environment while the actual IA scheme works in an offline simulation. It shows that the basic assumptions on the channel coefficients in the theoretical works are indeed feasible. Among other works, [61], [95], [97], [98], and [99] discuss the implementations of various IA testbeds. The results are promising, but they do not yet achieve the full potential proposed by theory. This weakness is due to the limitations and extensive demands enlisted above.

An important aspect to note here is that not all multi-user scenarios must necessarily involve IA. Depending on the respective interference power at each user, a proper combination of treating interference as noise, cancelling strong interference, and aligning only particular interference signals is yet open to be formulated. We conjecture that the optimal strategies will involve classifying specific interference scenarios in order to determine to what extent the application of IA is beneficial. In the following part, we focus on the requirements for a potential application of cyclic IA by propagation delay.

6.2 Cyclic IA by Propagation Delay

If the cyclic shifts as imposed by the channel matrix are unrolled over time, we may interpret the received signals undergoing a discrete propagation delay. For a proper application of cyclic IA, the decoders must take a transient settling time for the initial phase of the communication into account. But this initial transient period is only limited by the subchannel with the longest propagation delay. Afterwards, a stationary state follows that shows the same behaviour as the CPCM with n dimensions. If the transmitters are switched off at the end of the communication, a final transient period must be taken into account, as well. But these transient periods are negligible for permanent communication. With this interpretation, we the discrete propagation delay-based examples provided in [3, Appendix I], [37, Sect. 1] are generalized on the one hand, and on the other hand, the discussions of [51] and [16] are extended. There are some beneficial properties, but also a few challenging requirements, that must be taken into account when considering IA by propagation delay:

Advantages: It suffices to use very simple user-nodes with a single antenna, for omnidirectional signal propagation (i.e., no beam-forming). Only the CSI of the dedicated link must be known at the intended receivers. Although, the exact knowledge of all propagation delays between each user is obligatory, it can be measured simpler than the CSI (e.g., using a synchronized atomic clock, as in the global positioning system (GPS)). Minor variations in the propagation delay, i.e., jitter, can be compensated by using guard intervals if the frame durations are sufficiently long. For long frame durations, we can assume that the conveyed messages may also be protected by a conventional inner code to counteract noise.

Disadvantages: The propagation delays must be fairly static. Especially synchronization errors and multi-path propagation will impose leakage interference. For multiple users, exceeding $K \geq 3$, the total number of feasible channel delay matrices reduces quickly due to the separability conditions. This detrimental effect uncovers an interesting parallel to the *bandwidth-scaling* property discussed for IA by propagation delay with LoS paths in [51]. In other words, the bandwidth must scale sufficiently with the number of users K to provide a sufficiently large space to accommodate the dedicated signals. This property is actually conflicting with the fundamental demand of IA to include many users. However, note that in many of the different IA schemes proposed in the literature [12], comparable conflicting effects occur. It is clear that propagation delays are not discrete, but real-valued of course. However, this does not impose a major problem, since scaling-up the message demands in M, inherently scales with period-length n, so that real-valued delays in D can be approximated with sufficient accuracy.

Two communication scenarios, that seem to fit for (cyclic) IA by propagation delay at best, are: *Deep-space* communications and *underwater* communications. Both these setups may provide very specific delay and multi-path properties that are advantageous to implement cyclic IA by propagation delay.

6.2.1 Deep-Space Communications

Signals in a wireless deep-space scenario travel at the speed of light with $c_{\rm L} \approx 3 \cdot 10^8 m/s$. This scenario naturally scales with very long propagation delays and LoS signalling paths without multi-path scattering. For a distance of multiples of 300m between each transmitter and receiver, a sufficiently short frame-duration would be $T_{\Delta} \approx 1 \mu s$ as proposed in [16]. This implies quite a high frequency of $f_{\Delta} = T_{\Delta}^{-1} = 10$ MHz, and even higher carrier frequencies for multiple symbols within each frame. Since multiple communicating users are naturally distributed at different locations, each user-pair will experience different propagation delays. In the work [100], schemes for IA by propagation delay are discussed for a K-user interference channel. The given schemes are modelled for a corresponding satellite system with real-valued propagation delays.

6.2.2 Underwater Communications

As propagation delay³ is also a significant property observed in underwater communications with acoustic signals, another interesting opportunity for cyclic IA by propagation delay is available in [102]. From a simplified perspective, we may assume that acoustic signals travel at the speed of sound in the medium of seawater with $c_{\rm S} \approx 1.5 \cdot 10^3$ m/s. For a distance of multiples of 300 m between each transmitter and receiver, a sufficiently short frame duration would be $T_{\Delta} \approx 200$ ms, respectively. In this light, slight synchronization inaccuracies will be less degrading to the communication scheme than the previous case.

However, a more accurate and realistic acoustic underwater channel model undergoes some other severe detrimental effects as discussed in [103]. Especially potential multi-

³Propagation delay is sometimes called *latency* in oceanic communications engineering (e.g., [101]).

path effects caused by reflections on the surface and at the ocean floor and attenuation might weaken the LoS communication link. Furthermore, high SNR transmissions are an impractical option, since the available energy for the mobile transmitters (e.g., ships, submarines, buoys, aquanauts, sensors) is usually limited.

6.2.3 Node Placement in Euclidean Space

Since the number of feasible channel matrices is restrictive, a contrary approach is to design the propagation delays a wireless communication channel itself. *Node placement* terms a procedure of how to place user-nodes in Euclidean space enabling the discrete delays for (Cyclic) IA by propagation delay [16]. For the sake of simplicity, we assume that propagation delay is proportional to the Euclidean distance between each user. We neglect further wireless effects as, e.g., multi-path propagation, path loss and fading and consider a static LoS environment. These particular assumptions are also used in [16], [104], [105], [102] and [100] for the derivation of node placement schemes in multi-user interference channels.

The Euclidean distances between each user are denoted by a symmetric dissimilarity matrix $\mathbf{\Delta} = (\delta_{ji})_{1 \leq j, i \leq 4}$ with the entries $\delta_{ji} \in \mathbb{N}_{>0}$ and a zero diagonal [101]. We normalize the relationship between the propagation delay d_{ji} and the Euclidean distance δ_{ji} by $d_{ji} = x^{\delta_{ji}}$. The dissimilarity matrix $\mathbf{\Delta}$ of a communication channel with transmitters and receivers can be decomposed into four blocks:

$$\boldsymbol{\Delta} = \begin{pmatrix} \boldsymbol{B} & \boldsymbol{C}^{\mathsf{T}} \\ \boldsymbol{C} & \boldsymbol{A} \end{pmatrix},\tag{6.1}$$

with fixed entries in the matrix C and variable entries $a, b \in \mathbb{R}^+$ in A and B, respectively. The elements of the submatrix A correspond to distances between the transmitters and the elements of B to distances between the receivers. The elements of the submatrix C correspond to the fixed delay offset exponents of the channel matrix D from the CPCM.

The problem of node placement is analogous to an *Euclidean embedding* of the dissimilarity matrix Δ according to [104] and [105]. To derive the sought distances b and a, the receivers Rx_j and $k, j \neq k \in \{3, 4\}$, must be positioned such that all the given distances in C are fulfilled.

The objective of Euclidean embedding is to find a solution of *placement vectors* $\boldsymbol{x}_i \in \mathbb{R}^m$ for all users i = 1, ..., 4 within the lowest number of Euclidean dimensions m yet satisfying the fixed distances in $\boldsymbol{\Delta}$. In our case it is desirable to find a practically most relevant solution in only $m \leq 3$ dimensions for a spatial arrangement of users.

6.2.4 2-User *X*- Channel

A sufficiently simple example to describe node placement is the 2-user X-channel, since there are only four users and only minimal constraints to the channel matrix to be considered. Recall that this particular channel has been discussed in Section 3.3 in terms of the CPCM for general message lengths. For notational reasons, we define the following indexing for the 4 users Tx_1, Tx_2, Rx_1 and Rx_2 in the X-channel:

$$Tx_1 \cong 1, Tx_2 \cong 2, Rx_1 \cong 3, Rx_2 \cong 4.$$
 (6.2)

This change of indices will also be used for the delay exponents δ_{ji} . Note that the condition $\det(D) \neq 0 \mod(x^3 - 1)$ from Theorem 5.7 leads to an equivalent condition for matrix C:

$$\delta_{31} + \delta_{42} \not\equiv \delta_{41} + \delta_{32} \pmod{3}. \tag{6.3}$$

Figure 6.1 depicts an exemplary solution of node placement for cyclic IA by delay



Figure 6.1: A one-dimensional solution for the node placement of a given matrix C with $\delta_{31} = 2$, $\delta_{32} = 1$, $\delta_{41} = 4$ and $\delta_{42} = 1$ is shown. The numbered boxes indicate the users. The parameters of the dissimilarity matrix Δ are b = 2 and a = 3. Cyclic IA is possible since condition (6.3) is fulfilled.

on the X-channel in only one dimension. Another exemplary node placement in two dimensions is depicted in Figure 6.2.

We may set the variables $\delta_{12} = \delta_{21} = b \in \mathbb{R}^+$ and $\delta_{34} = \delta_{43} = a \in \mathbb{R}^+$ due to symmetry and $\delta_{ii} = 0$, for i = 1, ..., 4, and we obtain the sought submatrices:

$$B = b(1_{2\times 2} - I_{2\times 2}), \tag{6.4}$$

$$\boldsymbol{A} = a(\boldsymbol{1}_{2\times 2} - \boldsymbol{I}_{2\times 2}). \tag{6.5}$$

There is a feasible solution in two dimensions, if both transmitters satisfy all of the following (four) triangle inequalities with $j, k \in \{3, 4\}$ and $i, l \in \{1, 2\}$:

$$0 < \|\delta_{ji} - \delta_{jl}\|_2 \le b \le \|\delta_{ji}\|_2 + \|\delta_{jl}\|_2, \tag{6.6}$$

$$0 < \|\delta_{ji} - \delta_{ki}\|_2 \le a \le \|\delta_{ji}\|_2 + \|\delta_{ki}\|_2.$$
(6.7)

These inequalities must be non-zero because the users may not overlap at the same point in the Euclidean space. If the lower and upper bounds support a solution with $b, a \in \mathbb{R}$, then a 2-dimensional node placement exists.

The solution is easily derived from elementary geometry: W.l.o.g. we can position Tx_i on a reference point (0,0). To position Tx_l , $i \neq l$, some *b* satisfying condition (6.6) can be fixed on a straight line that originates in the reference point (0,0). We may fix Tx_l at (*b*,0) for instance. Then the valid positions for receivers Rx_j and Rx_k , $j \neq k$, can be determined, as well. Valid positions for Rx_j are the intersecting points of the two circles \mathcal{O}_{ji} and \mathcal{O}_{jl} with $j \neq k \in \{3,4\}$, $i \neq l \in \{1,2\}$:



Figure 6.2: A two-dimensional node placement solution is shown for a given matrix C with $\delta_{31} = 1$, $\delta_{32} = 2$, $\delta_{41} = 2$ and $\delta_{42} = 1$. The condition (6.3) is fulfilled, parameter b satisfies (6.6) and a satisfies (6.7).

The valid positions for Rx_k yield from the intersection points of the circles:

$$\mathcal{O}_{ki}: \qquad x^2 + y^2 = \delta_{ki}^2, \\ \mathcal{O}_{kl}: (x-b)^2 + y^2 = \delta_{kl}^2.$$
(6.9)

accordingly. Condition (6.7) is satisfied by construction, due to the fact that the circles \mathcal{O}_{ji} , \mathcal{O}_{ki} are concentric around node i and the circles \mathcal{O}_{jl} , \mathcal{O}_{kl} concentric around node l.

The resulting placement vectors are computed as:

$$\begin{aligned} \boldsymbol{x}_{i} &= \begin{pmatrix} 0\\0 \end{pmatrix}, \qquad \boldsymbol{x}_{l} &= \begin{pmatrix} b\\0 \end{pmatrix}, \\ \boldsymbol{x}_{j} &= \begin{pmatrix} \frac{\delta_{j_{i}}^{2} - \delta_{j_{l}}^{2} + b^{2}}{2b} \\ \pm \sqrt{\alpha_{j}} \end{pmatrix}, \qquad \boldsymbol{x}_{k} &= \begin{pmatrix} \frac{\delta_{k_{i}}^{2} - \delta_{k_{l}}^{2} + b^{2}}{2b} \\ \pm \sqrt{\alpha_{k}} \end{pmatrix}, \end{aligned}$$
(6.10)

with the indices $j \neq k \in \{3, 4\}$ and $i \neq l \in \{1, 2\}$, and the discriminants:

$$\alpha_j = \delta_{ji}^2 - \frac{\delta_{ji}^2 - \delta_{jl}^2 + b^2}{2b}, \tag{6.11}$$

$$\alpha_k = \delta_{ki}^2 - \frac{\delta_{ki}^2 - \delta_{kl}^2 + b^2}{2b}.$$
(6.12)

If both discriminants α_j , α_k are greater than zero, then a two-dimensional solution exists. If $\alpha_3 = \alpha_4 = 0$, the solution is one-dimensional. A negative discriminant states that there is no feasible solution for node placement. The remaining Euclidean distance *a* between the receivers Rx_j and Rx_k is computed by:

$$a = \|\boldsymbol{x}_j - \boldsymbol{x}_k\|_2. \tag{6.13}$$

with $j \neq k \in \{3, 4\}$. Moreover, such a positioning of users is invariant to rotation and translation w.r.t. a reference point.

An analogous dual solution is to fix receiver Rx_{j} at point (0,0) and Rx_{k} at point $\boldsymbol{x}_{k} = (a, 0)^{\mathsf{T}}$ satisfying condition (6.7) and finding the intersection points of the corresponding two circle pairs around Rx_{j} and Rx_{k} .

The extension to m = 3 dimensions would include an additional z-coordinate and the computation of two intersection circles for the four intersecting spheres. The 3-dimensional solution is also a rotational body around the connecting line δ_{li} between Tx_i and Tx_l of the 2-dimensional solution. Again, it is invariant to rotation and translation w.r.t. a fixed reference point.

6.3 Summary

In this chapter, we have given a brief overview of the practical challenges of the predominant interference alignment schemes in general. Next, we have discussed the opportunities and challenges of a potential practical application of *cyclic interference alignment by propagation delay.* We have observe that including the effect of long propagation delays in multi-user communication is indeed practically relevant, but quite limited so far. In the current stage, a fully practical application of cyclic IA by propagation delay has not been implemented yet. However, there are actually some very specific scenarios in deep-space and underwater communication systems that consider IA by propagation delay.

In the literature, the investigation of interference alignment by propagation delay is tightly connected to the problem of *node placement*, i. e., the positioning of transmitters and receivers in Euclidean space enabling interference alignment. In this light, have we investigated the node placement problem for the 2-user X-channel and we have provided conditions on its feasibility.

7 Conclusions

In this thesis, a novel channel model has been developed on the basis of a polynomial ring representation - the *cyclic polynomial channel model* (CPCM). The proposed model was mainly inspired by *interference alignment by propagation delay*, the algebraically convenient description of *cyclic codes*, and by the *linear deterministic channel model* (LDCM) as introduced in the seminal work of Avestimehr et al. The CPCM described multi-user communication systems with interference. Therein, each transmitted signal experienced an individual cyclic shift and at the receivers, those shifted signals interfered with each other.

Several different multi-user communication systems have been investigated from an information-theoretic perspective applying the CPCM. These systems ranged from elementary unidirectional multi-user channels to multi-way and multi-hop networks with relays. We have derived optimal communication strategies to achieve maximal datarates. These schemes involved a cyclic polynomial representation of *interference alignment*, signal alignment, interference neutralization, orthogonal multiple-access and linear coding schemes in particular. It has turned out that utilizing the CPCM is convenient for an approximation of optimal communication schemes when compared to the conventional Gaussian MIMO channel model (GMCM) and the LDCM. We have defined a set of separability conditions to allow for linear decodability at the receivers. This step has assisted us in classifying the intricate interference patterns of multiple users and it has facilitated the derivation of feasibility and optimality of our proposed communication schemes. As a result, even for channels with arbitrary asymmetric cyclic shifts, it has been shown that optimal solutions can be represented in a compact form.

Besides capacity characterization in terms of the CPCM, some interesting phenomena hidden within the discussed interference networks have been discovered. In particular, we have elaborated a *complementary reciprocal symmetry of alignment* that served as a key property for the capacity-achieving scheme of the 2-user X- network. It described how pairs of aligned signal and interference patterns at different receivers are related. Furthermore, we have described a duality between cyclic interference alignment and cyclic interference neutralization in a 3-user X- network with a minimal number of cognitive messages. Motivated by the well-known transformation in circuit theory, a Δ -Y duality relationship has been formulated for 3-way channels and 3-user Y- channels.

To further substantiate our proposed model, we have generalized the insights gained from the CPCM to design optimal schemes for the 3-way channel described by the LDCM and the GMCM. In both cases, upper bounds and achievable schemes have been derived w.r.t. the given model. In our last step, we have briefly explored potential applications of cyclic interference alignment in networks with long propagation delays, e.g., as in deep-space and underwater communication systems. Moreover, we have discussed a node placement scheme in Euclidean space for the 2-user X-channel.

In summary, this thesis has provided an elaborate theoretical analysis and optimal coding schemes for various multi-user interference networks modelled with arbitrary cyclic shifted communication links.

7.1 Outlook

There are still plenty of interesting interference network problems open to be considered using the CPCM.

In particular, we intend to generalize both the 3-way channel and the 3-user Y-channel, as discussed in Chapter 5, to K-users with $K \ge 3$. A solution would also answer the open question whether a generalization of the Δ -Y relationship exists for K users. Then, the corresponding results in the CPCM might facilitate the derivations for schemes on the more conventional GMCM and the LDCM. During our studies, we also encountered the concept of interference forwarding (IFWD), i. e., (aligned) interference signals are intentionally forwarded by a relay to provide side-information to the destinations, so that cancellation can be performed, as discussed in [106]. We already observed in some examples that IFWD is also useful for the achievable scheme of the 3-way channel and the multi-user extensions. This observation might eventually lead to some interesting duality relationships between cyclic interference alignment, interference neutralization and interference forwarding.

Furthermore, as perfect cyclic interference alignment schemes severely limit the ratio of feasible channel matrices for the K-user interference channel, we intend to study its generalization to arbitrary message lengths and investigate to what extend the conditions can be relaxed. In a subsequent step, we intend to permit imperfect interference alignment that will lead to limited leakage-interference. This approach can be used to characterize a trade-off for the achievable data-rate versus the channel-feasibility ratio.

There is also an opportunity to extend the underlying framework of the proposed channel model itself. A first step would be to use *channel matrices with sparse polynomials* instead of the matrices with monomials only. Such an approach would support modelling the interaction of a finite number of delayed multi-path signal echos, since they are of major concern for cyclic interference alignment by propagation delay. Some preliminary studies have already shown that, despite these echos, decoding schemes can be designed over the time-unrolled CPCM.

Appendix

A Constraints of the 3-User X-Network with Minimal Backhaul

In the following, we carefully prove that the IAC scheme in (3.146) to (3.153) satisfies all separability conditions (3.126) to (3.139) leading to the conditions of Theorem 3.12. The two particular exceptions given by (3.150) and (3.153) are neglected. The derivations are enlisted column-wise and congruences are taken modulo $x^5 - 1$.



Figure A.1: Adjacency graph of the cyclic IA scheme in (3.146) to (3.153). Solid lines indicate the assignments for a fixed p_{ki} given in the proof of Theorem 3.12, and dashed lines the remaining conditions that must be checked for feasibility.

 $\begin{array}{l} (3.126): x^{p_{ki}} \not\equiv x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{ij} d_{ii}^{-1} x^{p_{kj}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{ij} d_{ii}^{-1} d_{ji} d_{jj}^{-1} x^{p_{ki}} \\ \stackrel{\Rightarrow}{\Rightarrow} 0 \not\equiv \det(\boldsymbol{D}_{i,j,i,j}) \Rightarrow (\mathrm{iv}) \end{array}$

$$\begin{array}{l} (3.127): x^{p_{ji}} \not\equiv x^{p_{ii}} \\ \stackrel{(3.151)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{kk} d_{ki}^{-1} x^{p_{ik}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{kk} d_{ki}^{-1} d_{jj} d_{jk}^{-1} x^{p_{kj}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{kk} d_{ki}^{-1} d_{jj} d_{jk}^{-1} d_{ii} d_{ij}^{-1} x^{p_{j}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{kk} d_{ki}^{-1} d_{jj} d_{jk}^{-1} d_{ii} d_{ij}^{-1} x^{p_{j}} \end{array}$$

$$\begin{array}{l} (3.128): x^{p_{ki}} \neq x^{p_{ii}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ki}} \neq d_{jk} d_{ji}^{-1} x^{p_{kk}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ki}} \neq d_{jk} d_{ji}^{-1} d_{ii} d_{ik}^{-1} x^{p_{ki}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ki}} \neq d_{jk} d_{ji}^{-1} d_{ii} d_{ik}^{-1} x^{p_{ki}} \\ \stackrel{(3.129)}{\Rightarrow} x^{p_{ij}} \neq d_{ik} d_{ij}^{-1} d_{jj} d_{jk}^{-1} x^{p_{kj}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ij}} \neq d_{ik} d_{ij}^{-1} d_{jj} d_{jk}^{-1} d_{ij} d_{ij}^{-1} x^{p_{ji}} \\ \stackrel{(3.152)}{\Rightarrow} x^{p_{ij}} \neq d_{ik} d_{ij}^{-1} d_{jj} d_{jk}^{-1} \\ d_{ii} d_{ij}^{-1} d_{kj} d_{ki}^{-1} x^{p_{ij}} \\ \stackrel{(i)}{\Rightarrow} d_{ii} d_{jj} d_{kk} d_{kj} \neq d_{jk} d_{ij} d_{jk} d_{ki} d_{ij} \\ \stackrel{(i)}{\Rightarrow} d_{ii} d_{jj} d_{ik} d_{kj} \neq d_{jk} d_{ij} d_{ki} d_{kj} \\ \stackrel{(i)}{\Rightarrow} 0 \neq \det(\boldsymbol{D}_{i,j,i,j}) \Rightarrow (iv) \end{array}$$

(3.153)

Analogously, we consider the cyclically relabelled versions $i \to j \to k \to i$ of the separability conditions (3.126) to (3.139), with superscript symbol denoting the relabelled versions. Note that (3.137) contradicts (3.150) and is treated separately.

$$\begin{array}{l} (3.126) : x^{p_{kj}} \not\equiv x^{p_{ij}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ij}} \not\equiv d_{ii} d_{ij}^{-1} x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ij}} \not\equiv d_{ii} d_{ij}^{-1} d_{kj} d_{ki}^{-1} x^{p_{ij}} \\ \stackrel{(3.147)}{\Rightarrow} 0 \not\equiv \det(\boldsymbol{D}_{i,j,k}) \Rightarrow (\mathbf{v}) \\ (3.127) : x^{p_{jj}} \not\equiv x^{p_{kj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kj}} \not\equiv d_{ii} d_{ij}^{-1} x^{p_{ki}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{kj}} \not\equiv d_{ii} d_{ij}^{-1} d_{jj} d_{ji}^{-1} x^{p_{kj}} \\ \stackrel{(3.128)}{\Rightarrow} 0 \not\equiv \det(\boldsymbol{D}_{i,j,i,j}) \Rightarrow (\mathbf{iv}) \\ (3.128) : x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} x^{p_{ki}} \\ \stackrel{(3.153)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} x^{p_{ki}} \\ \stackrel{(3.153)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{ik} d_{jj}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} d_{ii} d_{jj} d_{kk} \not\equiv d_{ij} d_{jk} d_{ki} \Rightarrow (\mathbf{ix}) \end{array}$$

$$\begin{array}{l} (3.129) : d_{jk} x^{p_{jk}} \not\equiv d_{ji} x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{jk}} \not\equiv d_{ji} d_{jk}^{-1} d_{ik} d_{ii}^{-1} x^{p_{jk}} \\ \stackrel{(3.136)}{\Rightarrow} 0 \not\equiv \det(\boldsymbol{D}_{i,j,i,k}) \Rightarrow (\mathbf{v}) \\ (3.130) : d_{jj} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} x^{p_{ki}} \\ \stackrel{(3.153)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jj} d_{ji}^{-1} d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{jj}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{jj}} \not\equiv d_{jj} d_{ji}^{-1} d_{ii} d_{ij}^{-1} d_{ii} d_{ii}^{-1} x^{p_{ji}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{jj} d_{ji}^{-1} d_{ii} d_{ij}^{-1} d_{jj} d_{ji}^{-1} x^{p_{ki}} \\ \stackrel{(3.146)}{\Rightarrow} x^{p_{ji}} \not\equiv d_{jj} d_{ji}^{-1} d_{ii} d_{ij}^{-1} \\ \stackrel{(ii)}{\Rightarrow} d_{ii} d_{jj} d_{kk} d_{kk} \not\equiv d_{kk} d_{ki} d_{jk} d_{kj} \Rightarrow (\mathbf{x}) \end{array}$$

$$(3.136) : d_{ij}x^{p_{ji}} \notin d_{jj}x^{p_{kj}}$$

$$(3.132) : d_{jj}x^{p_{jj}} \notin d_{jk}x^{p_{ik}} \notin d_{jj}d_{jk}^{-1}d_{ik}d_{ij}^{-1}x^{p_{jk}}$$

$$(3.132) : d_{jj}x^{p_{jk}} \notin d_{jj}d_{jk}^{-1}d_{ik}d_{ij}^{-1}x^{p_{ki}}$$

$$(3.135) : x^{p_{jk}} \notin d_{jj}d_{jk}^{-1}d_{ik}d_{ij}^{-1}d_{kk}d_{ki}^{-1}x^{p_{jk}}$$

$$(3.137) : d_{jk}x^{p_{jk}} \notin d_{jj}x^{p_{kj}}$$

$$(3.137) : d_{jk}x^{p_{jk}} \notin d_{jj}x^{p_{ki}}$$

$$(3.133) : d_{jj}x^{p_{jj}} \notin d_{jk}d_{ki} \Rightarrow (ix)$$

$$(3.133) : d_{jj}x^{p_{jj}} \notin d_{jk}d_{ki}^{-1}x^{p_{jj}}$$

$$(3.134) : d_{jj}x^{p_{jj}} \notin d_{jk}d_{jj}^{-1}d_{ij}d_{ij}^{-1}d_{kj}x^{p_{jj}}$$

$$(3.134) : d_{ij}x^{p_{jj}} \notin d_{jk}d_{jj}^{-1}d_{ij}d_{ij}^{-1}x^{p_{jj}}$$

$$(3.134) : d_{ij}x^{p_{jj}} \notin d_{ji}d_{jj}^{-1}d_{ij}d_{ij}^{-1}x^{p_{jj}}$$

$$(3.134) : d_{ij}x^{p_{jj}} \notin d_{ji}d_{jj}^{-1}d_{ij}d_{ij}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} \notin d_{ji}d_{jj}^{-1}d_{kj}d_{kj}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} \notin d_{ji}d_{kj}d_{kj}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} \notin d_{ji}d_{jj}d_{kj}d_{kj}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} \notin d_{ji}d_{jj}d_{kj}d_{kj}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} \notin d_{ji}d_{jj}d_{kj}d_{kj}^{-1}x^{p_{jj}}$$

$$(3.135) : d_{jj}x^{p_{jj}} d_{jj$$

Analogously, we consider the cyclically relabelled versions $i \to k \to j \to i$ of the separability conditions $(3.126)^*$ to $(3.139)^*$, with superscript symbol * denoting the relabelled versions. Note that $(3.137)^*$ contradicts (3.153) and is treated separately.

$$(3.126)^* : x^{p_{ik}} \notin x^{p_{jk}}$$

$$\overset{(3.153)}{\Rightarrow} x^{p_{ik}} \notin d_{ki}d_{kk}^{-1}x^{p_{ki}}$$

$$\overset{(3.143)}{\Rightarrow} x^{p_{ik}} \notin d_{ki}d_{kk}^{-1}d_{jk}d_{ji}^{-1}x^{p_{ik}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{k,j,k,i}) \Rightarrow (\text{vii})$$

$$(3.127)^* : x^{p_{kk}} \notin x^{p_{ik}}$$

$$\overset{(3.148)}{\Rightarrow} x^{p_{kk}} \notin d_{ji}d_{jk}^{-1}x^{p_{ki}}$$

$$\overset{(3.147)}{\Rightarrow} x^{p_{kk}} \notin d_{ji}d_{jk}^{-1}d_{ik}d_{ii}^{-1}x^{p_{kk}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{k,j,k}) \Rightarrow (\text{v})$$

$$(3.128)^* : x^{p_{kk}} \notin d_{ki}d_{kk}^{-1}x^{p_{ki}}$$

$$\overset{(3.147)}{\Rightarrow} x^{p_{kk}} \notin d_{ki}d_{kk}^{-1}x^{p_{kk}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{k,j,k}) \Rightarrow (\text{v})$$

$$(3.128)^* : d_{ki}x^{p_{kk}} \notin d_{ki}d_{kk}^{-1}d_{ik}d_{ii}^{-1}x^{p_{kk}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{i,k,i,k}) \Rightarrow (\text{iii})$$

$$(3.129)^* : d_{ki}x^{p_{kk}} \notin d_{kj}d_{ki}^{-1}d_{ji}d_{jj}^{-1}x^{p_{ki}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{i,j,j,k}) \Rightarrow (\text{vi})$$

$$(3.130)^* : d_{kk}x^{p_{kk}} \notin d_{ki}d_{kk}^{-1}d_{ki}d_{ii}^{-1}x^{p_{kk}}$$

$$\Rightarrow 0 \notin \det(\mathbf{D}_{i,k,i,k}) \Rightarrow (\text{iii})$$

$$\begin{array}{l} (3.131)^* : d_{kk} x^{p_{kk}} \not\equiv d_{kj} x^{p_{kj}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{ji} d_{jj}^{-1} x^{p_{ki}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{ji} d_{jj}^{-1} d_{ik} d_{ii}^{-1} x^{p_{kk}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{ji} d_{jj}^{-1} d_{ik} d_{ii}^{-1} x^{p_{kk}} \\ \stackrel{(3.132)^*}{\Rightarrow} d_{ii} d_{jj} d_{kk} \not\equiv d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} x^{p_{kj}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} d_{ji} d_{jj}^{-1} x^{p_{ki}} \\ \stackrel{(3.148)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} d_{ji} d_{jj}^{-1} x^{p_{ki}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{ki} d_{kk}^{-1} d_{ij} d_{ii}^{-1} d_{ji} \\ d_{jj}^{-1} d_{ik} d_{ii}^{-1} d_{jk} d_{ii}^{-1} d_{ji} \\ \stackrel{(3.133)^*}{\Rightarrow} x^{p_{kk}} \not\equiv d_{ki} d_{kk}^{-1} d_{jk} d_{ji}^{-1} x^{p_{kk}} \\ \stackrel{(3.149)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj}^{-1} x^{p_{jk}} \\ \stackrel{(3.153)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj}^{-1} x^{p_{jk}} \\ \stackrel{(3.153)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj}^{-1} d_{ki} d_{kk}^{-1} x^{p_{jk}} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj}^{-1} d_{ki} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj}^{-1} d_{ki} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj} d_{ki} \\ \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj} d_{kk}^{-1} d_{jk} d_{jj} d_{ki} \\ \stackrel{(3.147)}{\Rightarrow} d_{ii} d_{jj} d_{kk} d_{kk} \not\equiv d_{kj} d_{jk} d_{jj} d_{ki} d_{kk} \Rightarrow (x) \end{array}$$

$$\begin{array}{lll} (3.135)^{*} : : d_{kk}x^{p_{kk}} \not\equiv d_{kj}x^{p_{jj}} & (3.138)^{*} : : d_{ki}x^{p_{ki}} \not\equiv d_{kj}x^{p_{ij}} \\ & \stackrel{(3.147)}{\Rightarrow} x^{p_{kk}} \not\equiv d_{kj}d_{kk}^{-1}d_{ik}d_{ij}^{-1}x^{p_{kk}} & \stackrel{(3.150)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{kj}d_{ki}^{-1}d_{jk}d_{jj}^{-1}x^{p_{jk}} \\ & \Rightarrow 0 \not\equiv \det(\boldsymbol{D}_{k,j,k,i}) \Rightarrow (\text{vii}) & \stackrel{(3.153)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{kk}x^{p_{ik}} \\ & \stackrel{(3.153)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{kk}d_{ki}^{-1}d_{jk}d_{jk}^{-1}d_{jk}d_{jk}^{-1}d_{ki}d_{kk}^{-1}x^{p_{ki}} \\ & \Rightarrow 0 \not\equiv \det(\boldsymbol{D}_{k,j,k,i}) \Rightarrow (\text{vii}) & \stackrel{(3.139)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{kj}x^{p_{jj}} \\ & \Rightarrow 0 \not\equiv \det(\boldsymbol{D}_{k,j,k,i}) \Rightarrow (\text{vii}) & \stackrel{(3.137)}{\Rightarrow} x^{p_{ki}} \not\equiv d_{kj}d_{ki}^{-1}d_{ii}d_{ij}^{-1}x^{p_{ki}} \\ & (3.137)^{*} : d_{ki}x^{p_{kj}} \not\equiv d_{kk}x^{p_{jk}} & \Rightarrow 0 \not\equiv \det(\boldsymbol{D}_{i,j,k,k}) \Rightarrow (\text{v}) \end{array}$$

The constraints (i) and (ii) are proven as follows.

$$\begin{array}{ll} (3.147): d_{ij}x^{p_{jj}} \equiv d_{ik}x^{p_{kk}} \\ (3.148): \\ \stackrel{(3.148)}{\Rightarrow} d_{ij}x^{p_{jj}} \equiv d_{ik}d_{ji}d_{jk}^{-1}x^{p_{ii}} \\ \stackrel{(3.151)}{\Rightarrow} d_{ij}x^{p_{jj}} \equiv d_{ik}d_{ji}d_{jk}^{-1}d_{kj}d_{ki}^{-1}x^{p_{jj}} \\ \stackrel{(3.151)}{\Rightarrow} d_{ij}x^{p_{jj}} \equiv d_{ik}d_{ji}d_{jk}^{-1}d_{kj}d_{ki}^{-1}x^{p_{jj}} \\ \stackrel{(3.148)}{\Rightarrow} d_{ii}x^{p_{ki}} \equiv d_{ij}d_{kk}d_{kj}^{-1}d_{jk}d_{ki}^{-1}x^{p_{ki}} \\ \stackrel{(3.148)}{\Rightarrow} d_{ii}x^{p_{ki}} \equiv d_{ij}d_{kk}d_{kj}^{-1}d_{jk}d_{ki}^{-1}x^{p_{ki}} \\ \stackrel{(3.146)}{\Rightarrow} d_{ij}d_{jk}d_{kj} \equiv d_{ik}x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} d_{ik}d_{jj}d_{jk}^{-1}x^{p_{ij}} \equiv d_{ii}x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} d_{ik}d_{jj}d_{jk}^{-1}x^{p_{ij}} \equiv d_{ii}x^{p_{ji}} \\ \stackrel{(3.146)}{\Rightarrow} d_{ij}x^{p_{kj}} \equiv d_{ji}d_{kk}d_{ki}^{-1}a^{p_{jk}} \\ \stackrel{(3.146)}{\Rightarrow} d_{jj}x^{p_{kj}} \equiv d_{ji}d_{kk}d_{ki}^{-1}d_{ij}d_{ki}^{-1}x^{p_{kj}} \\ \stackrel{(3.146)}{\Rightarrow} d_{jj}x^{p_{kj}} \equiv d_{ji}d_{kk}d_{ki}^{-1}d_{ij}d_{ki}^{-1}x^{p_{kj}} \\ \stackrel{(3.146)}{\Rightarrow} d_{jj}d_{ik}d_{kj} \equiv d_{ii}d_{jk}d_{kj} \Rightarrow (\text{ii}) \\ \end{array}$$

Since all separability conditions hold and yield the conditions as given in Theorem 3.12, the proof is complete.

B Example for Cyclic Interference Neutralization on the $2 \times 2 \times 2$ Relay-Interference Channel

Let us consider a simple example for cyclic IN on the $2 \times 2 \times 2$ relay interference channel with n = 4 dimensions. We choose the following two valid channel matrices for the UL and DL subchannels:

$$\boldsymbol{D} = \left(\begin{array}{cc} x^3 & x^3 \\ x^1 & x^0 \end{array}\right), \ \boldsymbol{E} = \left(\begin{array}{cc} x^2 & x^0 \\ x^2 & x^3 \end{array}\right).$$

By checking the conditions in Corollary 4.3, we see that they are fulfilled:

$$\begin{split} \delta_{12} + \delta_{21} + \eta_{11} + \eta_{22} &\equiv \delta_{11} + \delta_{22} + \eta_{12} + \eta_{21} \equiv 1 \pmod{4}, \\ \delta_{12} + \delta_{21} + \eta_{12} + \eta_{21} &\equiv 2 \not\equiv \delta_{11} + \delta_{22} + \eta_{11} + \eta_{22} \equiv 0 \pmod{4}, \\ \det(\boldsymbol{D}) &\equiv x^3 - x^4 \not\equiv 0 \mod(x^4 - 1), \\ \det(\boldsymbol{E}) &\equiv x^1 - x^2 \not\equiv 0 \mod(x^4 - 1). \end{split}$$

Next, we consider the interference-neutralization conditions in (4.12) and (4.13) for the parameters $\gamma_1 = 0$ and $\gamma_2 = 1$ at R_1 and R_2 :

$$\begin{split} \delta_{12} + \gamma_1 + \eta_{11} &\equiv \delta_{22} + \gamma_2 + \eta_{12} &\equiv 1 \pmod{4}, \\ \delta_{11} + \gamma_1 + \eta_{21} &\equiv \delta_{21} + \gamma_2 + \eta_{22} &\equiv 1 \pmod{4}. \end{split}$$

Furthermore, the no-signal-neutralization conditions (4.14) and (4.15) hold as well:

$$\begin{split} \delta_{11} + \gamma_1 + \eta_{11} &\equiv 1 \not\equiv \delta_{21} + \gamma_2 + \eta_{12} \equiv 2 \pmod{4}, \\ \delta_{12} + \gamma_1 + \eta_{21} &\equiv 1 \not\equiv \delta_{22} + \gamma_2 + \eta_{22} \equiv 0 \pmod{4}. \end{split}$$

By choosing parameter τ = 3, we obtain the following transmission signals for Tx1 and Tx2:

$$\begin{split} & u_1(x) = W_1^{[0]} x^0 + W_1^{[1]} x^1 + W_1^{[2]} x^2 + W_1^{[3]} x^3, \\ & u_3(x) = W_2^{[0]} x^0 + W_2^{[1]} x^1 + W_2^{[2]} x^2. \end{split}$$

The received signals at the relays R_1 and R_2 are:

$$\begin{split} r_1(x) &\equiv (W_1^{[1]} + W_2^{[1]})x^0 + (W_1^{[2]} + W_2^{[2]})x^1 + W_1^{[3]}x^2 + (W_1^{[0]} + W_2^{[0]})x^3 \mod(x^4 - 1), \\ r_2(x) &\equiv (W_1^{[3]} + W_2^{[0]})x^0 + (W_1^{[0]} + W_2^{[1]})x^1 + (W_1^{[1]} + W_2^{[2]})x^2 + W_1^{[2]}x^3 \mod(x^4 - 1). \end{split}$$

With parameter $k_2 = \tau + \delta_{22} \equiv 3 \pmod{4}$, we set $r_2^{[3]} = 0$. The resulting forwarded signals, with the given parameters $\gamma_1 = 0$ and $\gamma_2 = 1$, are:

$$\begin{split} &v_1(x) \equiv (W_1^{[1]} + W_2^{[1]})x^0 + (W_1^{[2]} + W_2^{[2]})x^1 + W_1^{[3]}x^2 + (W_1^{[0]} + W_2^{[0]})x^3 \mod(x^4 - 1), \\ &v_2(x) \equiv (-W_1^{[3]} - W_2^{[0]})x^1 + (-W_1^{[0]} - W_2^{[1]})x^2 + (-W_1^{[1]} - W_2^{[2]})x^3 \mod(x^4 - 1), \end{split}$$

so that the following signals are received at the destinations Rx_1 and Rx_2 :

$$\begin{split} t_1(x) &\equiv W_1^{[3]} x^0 + (W_1^{[0]} + W_2^{[0]} - W_1^{[3]} - W_2^{[0]}) x^1 + \\ & (W_1^{[1]} + W_2^{[1]} - W_1^{[0]} - W_2^{[1]}) x^2 + (W_1^{[2]} + W_2^{[2]} - W_1^{[1]} - W_2^{[2]}) x^3 \\ &\equiv W_1^{[3]} x^0 + (W_1^{[0]} - W_1^{[3]}) x^1 + (W_1^{[1]} - W_1^{[0]}) x^2 + (W_1^{[2]} - W_1^{[1]}) x^3 \mod(x^4 - 1), \\ t_2(x) &\equiv (W_1^{[3]} - W_1^{[3]} - W_2^{[0]}) x^0 + (W_1^{[0]} + W_2^{[0]} - W_1^{[0]} - W_2^{[1]}) x^1 + \\ & (W_1^{[1]} + W_2^{[1]} - W_1^{[1]} - W_2^{[2]}) x^2 + (W_1^{[2]} + W_2^{[2]}) x^3 \\ &\equiv -W_2^{[0]} x^0 + (W_2^{[0]} - W_2^{[1]}) x^1 + (W_2^{[1]} - W_2^{[2]}) x^2 + (W_1^{[2]} + W_2^{[2]}) x^3 \mod(x^4 - 1). \end{split}$$

At both destinations, all interfering signals are neutralized. It is obvious that the submessages $W_1^{[0]}, W_1^{[1]}, W_1^{[2]}, W_1^{[3]}$ are linear decodable at Rx₁, and that the submessages $W_2^{[0]}, W_2^{[1]}, W_2^{[2]}$ are linear decodable at Rx₂. Hence, $\frac{2n-1}{n} = \frac{7}{4}$ DoF are achieved for n = 3 in accordance to Theorem 4.2.

C Genie-Aided Upper Bounds of the $D3C(n_1, n_2, n_3)$ with $n_3 \ge n_2 \ge n_1$

The remaining upper bounds on the capacity region of the $D3C(n_1, n_2, n_3)$ are derived similar to Section 5.2.2. We only discuss the main differences in this appendix.

(i): To derive the bound $R_{12} + R_{13} + R_{23} \le n_3$, we provide W_{32} as side-information to the receiver of T_1 , and proceed similar to Section 5.2.2. That is, we prove that the enhanced T_1 can construct y_2^N from which it can decode W_{23} .

(ii): To derive the bound $R_{21} + R_{23} + R_{13} \le n_3 + n_2 - n_1$, we provide W_{31} and \hat{x}_3^N to T_2 as side-information, where \hat{x}_3^N denotes the lowermost $n_2 - n_1$ bits of x_3^N . Providing these bits is necessary since T_2 can only obtain the topmost n_1 symbols of $x_3(1)$ from $y_2(1)$ and $x_1(1)$ after decoding W_{21} (recall that given W_{31} and W_{21} , T_2 can construct $x_1(1)$). By combining the topmost n_1 bits of $x_3(1)$ and the lowermost $n_2 - n_1$ bits provided by the side-information, T_2 can construct $y_1(1)$ and hence also $x_1(2)$, since it knows W_{31} from side-information and W_{21} after decoding. Similarly, all components of y_1^N can be constructed, and W_{13} can be decode. Thus, by using Fano's inequality, we can write:

$$N(R_{21} + R_{23} + R_{13}) \leq I(W_{21}, W_{23}, W_{13}; \boldsymbol{y}_2^N, \hat{\boldsymbol{x}}_3^N, W_{12}, W_{32}, W_{31}) + N\epsilon_N \leq H(\boldsymbol{y}_2^N, \hat{\boldsymbol{x}}_3^N) - H(\boldsymbol{y}_2^N, \hat{\boldsymbol{x}}_3^N | \boldsymbol{w}) + N\epsilon_N \leq H(\boldsymbol{y}_2^N, \hat{\boldsymbol{x}}_3^N) + N\epsilon_N \leq N(\max(n_3 + n_2 - n_1, n_1 + n_2 - n_1) + \epsilon_N) = N(n_2 + n_2 - n_1 + \epsilon_N).$$

where $\epsilon_N \to 0$ as $N \to \infty$. This provides $R_{21} + R_{23} + R_{13} \le n_3 + n_2 - n_1$ after dividing by N and letting $N \to \infty$.

(iii): To derive $R_{21} + R_{23} + R_{31} \le n_3$, we provide W_{13} to T_2 and proceed similar to (i), by showing that T_2 can construct y_3^N and decode W_{31} .

(iv): To derive $R_{31} + R_{32} + R_{21} \le n_3$, we give \hat{x}_1^N and W_{12} to T_3 as side-information, where \hat{x}_1^N denotes the lowermost $n_3 - n_2$ bits of x_1^N . By proceeding similar to (ii), we can show that T_3 can construct y_2^N given this side-information, and then decode W_{21} , leading to the desired bound.

(v): To derive the bound $R_{31} + R_{32} + R_{12} \le n_3 + n_2 - n_1$, we give W_{21} and $\hat{\boldsymbol{x}}_2^N$ to T_3 , where $\hat{\boldsymbol{x}}_2^N$ denote the lowermost $n_3 - n_1$ bits of \boldsymbol{x}_2^N . Similar to (ii), T_3 is able to construct \boldsymbol{y}_1^N given this side-information, and then to decode W_{12} , leading to the desired bound.

As a result, we obtain the upper bounds on capacity region as given by (5.47) to (5.54).

List of Acronyms

2IFC	2-user interference channel
AWGN	additive white Gaussian noise
BC	broadcast channel
BD	backward decoding
BHN	backhaul network
CPCM	cyclic polynomial channel model
CSI	channel state information
D2D	device-to-device
DL	downlink
DoF	degrees-of-freedom
FF	feedforward
GDoF	generalized degrees-of-freedom
GMCM	Gaussian MIMO channel model
GPS	global positioning system
IA	interference alignment
IAC	interference alignment and cancellation
IFWD	interference forwarding
IN	interference neutralization
LDCM	linear deterministic channel model
LEaD	linear encoding and decoding
LoS	line-of-sight
MA	multiple-access
MAC	multiple-access channel
MIMO	multiple-input multiple-output
OFDM	orthogonal frequency division multiplexing
SA	signal alignment
SISO	single-input single-output
SNR	signal-to-noise-ratio
UL	uplink

List of Figures

Chapte 1.1	r 1 The Fundamental Principle of Interference Alignment	3
1.2	Notation	7
Chapte	r 2	
2.1	A Linear Deterministic MAC	13
2.2	Cyclic Polynomial Channel Model	15
Chapte	r 3	
3.1	The Cyclic Polynomial MAC	22
3.2	Orthogonal MA on the Cyclic Polynomial MAC	23
3.3	The Cyclic Polynomial BC	24
3.4	Messaging in the Cyclic Polynomial 2IFC	26
3.5	The Cyclic Polynomial 2-User X-Channel	31
3.6	Perfect Cyclic IA on the 2-User X-Channel	32
3.7	Non-Overlapping and Overlapping Frames in the X - Channel $\ldots \ldots$	40
3.8	Example of Cyclic IA with LEaD for the X-Channel	42
3.9	The Cyclic Polynomial <i>K</i> -User Interference Channel	44
3.10	The 3-User X-Network	50
3.11	Example of Cyclic Interference Alignment and Cancellation	57
Chapte	r 4	
4.1	The Principle of Interference Neutralization	62
4.2	$2 \times 2 \times 2$ Relay-Interference Channel	63
4.3	MIMO $2 \times 2 \times 2$ Relay-Interference Channel	69
4.4	The $2 \times 2 \times 2$ Two-Way Relay-Interference Channel	73
4.5	Interference-Neutralization Conditions	76
4.6	A Cascaded Two-Way Relay Network	81
4.7	UL- and DL-Phases of the Cascaded Two-way Relay Network	84
Chapte	r 5	
5.1	3-User Y-Channel	91
5.2	3-User Three-Way Channel	92
5.3	Δ -Y Channel Transformation	99
5.4	The Reciprocal Linear Deterministic 3-Way Channel	101
5.5	The Reciprocal Linear Deterministic Y-Channel	104
5.6	Δ -Y Channel Transformation	106
5.7	Auxiliary Illustration for Interference of Class (b)	107

5.8	Auxiliary Illustration for Interference of Class (c) 108	
5.9	A) (Bidirectional) Allocation of Signals over the $D3C(m, m, m)$ 110	
5.10	B) (Clock-Wise) Allocation of Signals over the $D3C(m', m', m')$ 111	
5.11	B) (Counter Clock-Wise) Allocation of Signals over the $D3C(m', m', m')$ 111	
5.12	C) (Unidirectional) Allocation of Signals over the $\mathrm{D3C}(m'',m'',m'')$ 113	
5.13	The MIMO $M_i \times M_i$ 3-Way Channel	
5.14	The Homogeneous MIMO $M_{\rm Tx} \times M_{\rm Rx}$ 3-Way Channel $\ \ldots \ \ldots \ 123$	
5.15	Sum-DoF Parameter Plane of the $M_{\rm Tx} \times M_{\rm Tx}$ MIMO 3-Way Channel . 130	
Chapte	r 6	
6.1	Node Placement for a 2-User X-Channel, 1-Dimensional Example 139	
6.2	Node Placement for a 2-User X- Channel, 2-Dimensional Example $\ . \ . \ 140$	
Appendix		
A.1	Adjacency Graph of the Cyclic IAC Scheme	

List of Theorems

Chapter 3

Lemma (Orthogonal MA on a K -user MAC)	23
Lemma (Orthogonal MA on a K-user BC)	25
Lemma (Orthogonal MA on a 2IFC with $det(D) \equiv 0 \mod(x^n - 1))$	26
Corollary (Asymptotic Orthogonal MA on a 2IFC)	27
Theorem (LEaD on a 2IFC)	29
Theorem (Perfect Cyclic IA on the 2-User X-Channel)	33
Corollary (The 2-User X-Channel with $det(D) \equiv 0 \mod(x^n - 1))$	34
Theorem (Cyclic IA & LEaD on the 2-User X-Channel)	37
Theorem (Perfect Cyclic IA on a K-User IFC)	44
Corollary (Alternative Conditions for Perfect Cyclic IA)	46
Theorem (Infeasibility of Perfect Cyclic IA on a 3-User X-Network).	51
Theorem (Cyclic IAC with FF on a 3-User X-Network)	53
Lemma (Cyclic IAC with Minimal Rx-BHN on a 3-User X-Network) .	55
Lemma (Cyclic IN with Minimal Tx-BHN on a 3-User X-Network)	56
Corollary (Cyclic IAC/IN with Min. BHN on a 3-User X-Network) $$	57
r 4	
Lemma (Cyclic IN and Lin. Decoding is Infeasible for n Submessages).	65
Theorem (Cyclic IN on the $2 \times 2 \times 2$ Relay-Interference Channel)	66
Corollary (Implicit Conditions for Cyclic IN)	67
Theorem (Cyclic IN for the 2-Way $2 \times 2 \times 2$ Relay-Interference Channel)	78
Theorem (Cyclic IA on the Open-Loop Line Network)	85
Theorem (Cyclic IA on the Closed-Loop Ring Network)	86
r 5	
Theorem (Cyclic SA on the Y-Channel)	94
Theorem (Cyclic SA on the 3-Way Channel)	99
Lemma (Outer Bounds of the Lin. Det. 3-Way Channel)	103
Theorem (Capacity Region of the Lin. Det. 3-Way Channel)	109
Theorem (IA on the Symmetric Lin. Det. 3-Way Channel)	109
Lemma (Intersection Subspace of Random Matrices)	115
Theorem (MIMO IA on the $M_i \times M_i$ 3-Way Channel)	116
Theorem (MIMO IA on the $M_{\text{Tx}} \times M_{\text{Rx}}$ 3-Way Channel)	125
	Lemma (Orthogonal MA on a <i>K</i> -user MAC) Lemma (Orthogonal MA on a <i>K</i> -user BC)

5.9 Lemma (Intersection Subspace in the CPCM) 131

Bibliography

- C. Shannon, "A mathematical theory of communication," The Bell System Technical Journal, no. 27, pp. 623–656, Jul. 1948.
- [2] D. Costello and D. Forney, "Channel coding: The road to capacity," Proceedings of the IEEE, vol. 95, no. 6, pp. 1150–1177, Jun. 2007.
- [3] V. Cadambe and S. Jafar, "Interference alignment and degrees of freedom of the K-user interference channel," *IEEE Transactions on Information Theory*, vol. 54, no. 8, pp. 3425–3441, Aug. 2008.
- [4] R. Etkin, D. Tse, and H. Wang, "Gaussian interference channel capacity to within one bit," *IEEE Transactions on Information Theory*, vol. 54, no. 12, pp. 5534–5562, Dec. 2008.
- [5] A. El Gamal and Y.-H. Kim, Network Information Theory. Cambridge University Press, 2011.
- [6] V. Annapureddy and V. Veeravalli, "Gaussian interference networks: Sum capacity in the low-interference regime and new outer bounds on the capacity region," *IEEE Transactions on Information Theory*, vol. 55, no. 7, pp. 3032–3050, Jul. 2009.
- [7] X. Shang, G. Kramer, and B. Chen, "A new outer bound and the noisyinterference sum-rate capacity for gaussian interference channels," *IEEE Transactions on Information Theory*, vol. 55, no. 2, pp. 689–699, Feb. 2009.
- [8] A. Motahari and A. Khandani, "Capacity bounds for the Gaussian interference channel," *IEEE Transactions on Information Theory*, vol. 55, no. 2, pp. 620–643, Feb. 2009.
- [9] A. Avestimehr, S. Diggavi, and D. Tse, "A deterministic approach to wireless relay networks," in *Proceedings of 45th Annual Allerton Conference on Communications, Control, and Computing (Allerton 2007)*, Monticello, IL, USA, Sep. 2007, pp. 721–728.
- [10] G. Kramer, "Outer bounds on the capacity of Gaussian interference channels," *IEEE Transactions on Information Theory*, vol. 50, no. 3, p. 581, Mar. 2004.
- [11] M. Maddah-Ali, A. Motahari, and A. Khandani, "Signaling over MIMO multibase systems: Combination of multi-access and broadcast schemes," in *Proceedings IEEE International Symposium on Information Theory (ISIT 2006)*, Seattle, WA, USA, Jul. 2006, pp. 2104–2108.

- [12] S. Jafar, Interference Alignment: A New Look at Signal Dimensions in a Communication Network. Foundations and Trends in Communications and Information Theory, 2011, vol. 7, no. 1.
- [13] A. Motahari, S. Oveis-Gharan, M. Maddah-Ali, and A. Khandani, "Real interference alignment: Exploiting the potential of single antenna systems," *IEEE Transactions on Information Theory*, vol. 60, no. 8, pp. 4799–4810, Aug. 2014.
- [14] T. Han and K. Kobayashi, "A new achievable rate region for the interference channel," *IEEE Transactions on Information Theory*, vol. IT-27, no. 1, pp. 49– 60, Jan. 1981.
- [15] B. Nazer, M. Gastpar, S. Jafar, and S. Vishwanath, "Ergodic interference alignment," *IEEE Transactions on Information Theory*, vol. 58, no. 10, pp. 6355–6371, Oct. 2012.
- [16] V. Cadambe and S. Jafar, "Degrees of freedom of wireless networks what a difference delay makes," in *Proceedings of Asilomar Conference on Signals, Systems and Computers (ACSSC 2007)*, Pacific Grove, CA, USA, Nov. 2007, pp. 133–137.
- [17] V. Cadambe, S. Jafar, and C. Wang, "Interference alignment with asymmetric complex signaling - settling the Høst-Madsen-Nosratinia conjecture," *IEEE Transaction on Information Theory*, vol. 56, no. 9, pp. 4552–4565, Sep. 2010.
- [18] S. Sridharan, A. Jafarian, S. Vishwanath, and S. Jafar, "Capacity of symmetric K-user Gaussian very strong interference channels," in *IEEE Global Communi*cations Conference (GLOBECOM 2008), New Orleans, LO, USA, Nov. 2008, pp. 1–5.
- [19] A. Dimakis, K. Ramchandran, Y. Wu, and C. Suh, "A survey on network codes for distributed storage," *Proceedings of the IEEE*, vol. 99, no. 3, pp. 476–489, Mar. 2011.
- [20] V. Cadambe, S. Jafar, and S. Shamai (Shitz), "Interference alignment on the deterministic channel and application to fully connected Gaussian interference networks," *IEEE Transactions on Information Theory*, vol. 55, no. 1, pp. 269– 274, Jan. 2009.
- [21] C. Huang, V. Cadambe, and S. Jafar, "Interference alignment and the generalized degrees of freedom of the X channel," *IEEE Transactions on Information Theory*, vol. 58, no. 8, pp. 5130–5150, Sep. 2012.
- [22] S. Jafar and S. Vishwanath, "Generalized degrees of freedom of the symmetric Gaussian K user interference channel," *IEEE Transactions on Information The*ory, vol. 56, no. 7, pp. 3297–3303, Jul. 2010.
- [23] V. Prabhakaran and P. Viswanath, "Interference channels with source/destination cooperation," in *Proceedings of Asilomar Conference on Signals, Systems*

and Computers (ACSSC 2008), Pacific Grove, CA, USA, Oct. 2008, pp. 707–710.

- [24] A. Vahid, C. Suh, and A. Avestimehr, "The two-user deterministic interference channel with rate-limited feedback," *IEEE Transactions on Information Theory*, vol. 58, no. 5, pp. 2788–2812, May 2012.
- [25] M. Gastpar and F. Kschischang, "Algebraic structure in network information theory," Banff International Research Station, Tech. Rep., Aug. 2011. [Online]. Available: http://www.birs.ca/workshops/2011/11w5074/report11w5074.pdf
- [26] R. Ahlswede, N. Cai, S. Li, and R. Yeung, "Network information flow," *IEEE Transactions on Information Theory*, vol. 46, no. 4, pp. 1204–1216, Jul. 2000.
- [27] H. Maier, J. Schmitz, and R. Mathar, "Cyclic interference alignment by propagation delay," in *Proceedings of 50th Annual Allerton Conference on Communications, Control and Computing (Allerton 2012)*, Monticello, IL, USA, Oct. 2012, pp. 1761–1768.
- [28] H. Maier and R. Mathar, "Cyclic interference neutralization on the full-duplex relay-interference channel," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2013)*, Istanbul, Turkey, Jul. 2013, pp. 2409–2413.
- [29] —, "Cyclic interference neutralization on the 2 × 2 × 2 full-duplex two-way relay-interference channel," in *Proceedings of IEEE Information Theory Work*shop (ITW 2013), Seville, Spain, Sep. 2013, pp. 644–648.
- [30] H. Maier, A. Chaaban, and R. Mathar, "Degrees of freedom of the MIMO 3way channel," in *Proceedings of IEEE International Symposium on Information Theory and Applications (ISITA 2014)*, Melbourne, VIC, Australia, Oct. 2014, pp. 638–642.
- [31] H. Maier and R. Mathar, "User-relay duality in cyclic interference alignment for cascaded two-way relay networks," in *Proceedings of IEEE International Sympo*sium on Information Theory and Applications (ISITA 2014), Melbourne, VIC, Australia, Oct. 2014, pp. 471–475.
- [32] —, "Cyclic interference alignment and cancellation in 3-user X- networks with minimal backhaul," in *Proceedings of IEEE Information Theory Workshop (ITW 2014)*, Hobart, TAS, Australia, Nov. 2014, pp. 88–92.
- [33] H. Maier, A. Chaaban, and R. Mathar, "Symmetric degrees of freedom of the MIMO 3-way channel with M_T × M_R antennas," in *Proceedings of IEEE Information Theory Workshop (ITW 2014)*, Hobart, TAS, Australia, Nov. 2014, pp. 93–97.
- [34] H. Maier, A. Chaaban, R. Mathar, and A. Sezgin, "Capacity region of the reciprocal deterministic 3-way channel via Δ-Y transformation," in *Proceedings of* 52th Allerton Conference on Communications, Control and Computing (Allerton 2014), Monticello, IL, USA, Oct. 2014, pp. 167–174.

- [35] S. Jafar and M. Fakhereddin, "Degrees of freedom for the MIMO interference channel," *IEEE Transactions on Information Theory*, vol. 53, no. 7, pp. 2637– 2642, Jul. 2007.
- [36] A. Host-Madsen and A. Nosratinia, "The multiplexing gain of wireless networks," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2005)*, Adelaide, SA, Sep. 2005, pp. 2065–2069.
- [37] V. Cadambe and S. Jafar, "Interference alignment and the degrees of freedom of wireless X networks," *IEEE Transactions on Information Theory*, vol. 55, no. 9, pp. 3893–3908, Sep. 2009.
- [38] S. Jafar and S. Shamai (Shitz), "Degrees of freedom region of the MIMO X channel," *IEEE Transactions on Information Theory*, vol. 54, no. 1, pp. 151– 170, Jan. 2008.
- [39] C. Yetis, T. Gou, S. Jafar, and A. Kayran, "Feasibility conditions for interference alignment," in *Proceedings of IEEE Global Communications Conference* (GLOBECOM 2009), Honolulu, HI, USA, Nov. 2009.
- [40] K. Gomadam, V. Cadambe, and S. Jafar, "Approaching the capacity of wireless networks through distributed interference alignment," in *Proceedings of IEEE Global Communications Conference (GLOBECOM 2008)*, New Orleans, LO, USA, Nov. 2008.
- [41] A. Avestimehr, S. Diggavi, and D. Tse, "Wireless network information flow: A deterministic approach," *IEEE Transactions on Information Theory*, vol. 57, no. 4, pp. 1872–1905, Apr. 2011.
- [42] A. Avestimehr, "Wireless network information Flow: a deterministic approach," Ph.D. dissertation, EECS Department, University of California, Berkeley, Oct. 2008.
- [43] A. El Gamal and M. Costa, "The capacity region of a class of deterministic interference channels," *IEEE Transactions on Information Theory*, vol. IT-28, no. 2, pp. 343–346, Mar. 1982.
- [44] U. Niesen and M. Maddah-Ali, "Interference alignment: From degrees of freedom to constant-gap capacity approximations," *IEEE Transactions on Information Theory*, vol. 59, no. 8, pp. 4855–4888, Aug. 2013.
- [45] T. Cover and J. Thomas, *Elements of Information Theory*, 2nd ed. Wiley, 2006.
- [46] A. Chaaban, "Relaying for Interference Managment," Ph.D. dissertation, Ruhr-Universität Bochum, 2013.
- [47] S. Mohajer, S. Diggavi, C. Fragouli, and D. Tse, "Transmission techniques for relay-interference networks," in *Proceedings of 46th Annual Allerton Conference* on Communications, Control, and Computing (Allerton 2008), Monticello, IL, USA, Sep. 2008, pp. 467–474.

- [48] —, "Capacity of deterministic Z-chain relay-interference network," in Proceedings of IEEE Information Theory Workshop (ITW 2009), Volos, Greece, Jun. 2009, pp. 331–335.
- [49] W. Peterson and D. Brown, "Cyclic codes for error detection," Proceedings of the Institute of Radio Engineers, vol. 49, no. 1, pp. 228–235, Jan. 1961.
- [50] R. Blahut, Algebraic Codes for Data Transmission. Cambridge University Press, 2003.
- [51] L. Grokop, D. Tse, and R. Yates, "Interference alignment for line-of-sight channels," *IEEE Transactions on Information Theory*, vol. 57, no. 9, pp. 5820–5839, Sep. 2011.
- [52] L. Mroueh and J.-C. Belfiore, "Feasibility of interference alignment in the timefrequency domain," *IEEE Transactions on wireless communications*, vol. 11, no. 11, pp. 3932–3941, Nov. 2012.
- [53] R. Bose, Information Theory Coding and Cryptography. McGraw-Hill, 2007.
- [54] U. Niesen, "Interference alignment in dense wireless networks," *IEEE Transac*tions on Information Theory, vol. 57, no. 5, pp. 2889–2901, May 2011.
- [55] N. Schrammar, "On deterministic models for Gaussian networks," Ph.D. dissertation, Royal Institute of Technology (KTH), 2013.
- [56] S. Hong and G. Caire, "On interference networks over finite fields," *IEEE Transactions on Information Theory*, vol. 60, no. 8, pp. 4902–4921, Aug. 2014.
- [57] C. Shannon, "Two-way communication channels," in *Proceedings of 4th Berkeley Symposium on Mathematical Statistics and Probability*, vol. 1, Berkeley, CA, USA, 1961, pp. 611–644.
- [58] A. Carleial, "Interference channels," *IEEE Transactions on Information Theory*, vol. 24, no. 1, pp. 60–70, Jan. 1978.
- [59] G. Bresler and D. Tse, "The two-user Gaussian interference channel: A deterministic view," *European Transactions on Telecommunications*, vol. 19, no. 4, pp. 333–354, Apr. 2008.
- [60] M. Mosko and I. Solis, "Ring optimal assignment of slot reservations for cyclic TDMA protocols," in *Proceedings of 20th IEEE International Conference on Computer Communications and Networks (ICCCN)*, Maui, HI, USA, Aug. 2011.
- [61] S. Gollakota, S. Perli, and D. Katabi, "Interference alignment and cancellation," Proceedings of the ACM SIGCOMM 2009 Conference on Data Communication (SIGCOMM 2009), vol. 39, no. 4, pp. 159–170, 2009.
- [62] X. Qu and C. Kang, "A closed-form solution to implement interference alignment and cancellation for Gaussian interference multiple access channel," *IEEE Transactions on Wireless Communications*, vol. 13, no. 2, pp. 710–723, Feb. 2014.

- [63] T. Gou, S. Jafar, S.-W. Jeon, and S.-Y. Chung, "Aligned interference neutralization and the degrees of freedom of the 2 × 2 × 2 interference channel," *IEEE Transactions on Information Theory*, vol. 58, no. 7, pp. 4381–4395, Jul. 2012.
- [64] V. Cadambe and S. Jafar, "Reflections on interference alignment and the degrees of freedom of the K-user interference channel," *IEEE Information Theory Society Newsletter*, vol. 59, no. 4, pp. 5–9, Dec. 2009.
- [65] T. Gou, C. Wang, and S. Jafar, "Aligned interference neutralization and the degrees of freedom of the 2 × 2 × 2 interference channel with interfering relays," in *Proceedings of 49th Annual Allerton Conference on Communications, Control* and Computing (Allerton 2011), Monticello, IL, USA, Sep. 2011.
- [66] S.-W. Jeon, C.-Y. Wang, and M. Gastpar, "Approximate ergodic capacity of a class of fading 2-user 2-hop networks," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2012)*, Cambridge, MA, USA, Jul. 2012, pp. 56–60.
- [67] I. Shomorony and A. Avestimehr, "Degrees of freedom of two-hop wireless networks: "Everyone gets the entire cake"," in *Proceedings of 50th Allerton Conference on Communications, Control and Computing (Allerton 2012)*, Monticello, IL, USA, Oct. 2012.
- [68] R. Gray, Toeplitz and Circulant Matrices: A review. Now Pub., 2006.
- [69] K. Lee, S.-H. Park, J.-S. Kim, and I. Lee, "Degrees of freedom on MIMO multilink two-way relay channels," in *Proceedings of IEEE Global Telecomunications Conference (GLOBECOM 2010)*, Miami, Florida, USA, Dec. 2010.
- [70] K. Lee, N. Lee, and I. Lee, "Achievable degrees of freedom on MIMO two-way relay interference channels," *IEEE Transactions on Wireless Communications*, vol. 12, no. 4, pp. 1472–1480, Apr. 2013.
- [71] B. Rankov and A. Wittneben, "Achievable rate regions for the two-way relay channel," in *Proceedings of IEEE International Symposium on Information The*ory (ISIT 2006), Seattle, WA, USA, Jul. 2006, pp. 1668–1672.
- [72] P. Popovski and H. Yomo, "The anti-packets can increase the achievable throughput of a wireless multi-hop network," in *Proceedings of IEEE International Conference on Communications (ICC 2006)*, Istanbul, Turkey, Jun. 2006, pp. 3885– 3890.
- [73] A. Avestimehr, A. Sezgin, and D. Tse, "Approximate capacity of the two-way relay channel: A deterministic approach," in *Proceedings of 46th Annual Aller*ton Conference on Communications, Control, and Computing (Allerton 2008), Urbana-Champaign, IL, USA, Sep. 2008, pp. 1582–1589.
- [74] A. Sezgin, M. Khajehnejad, A. Avestimehr, and B. Hassibi, "Approximate capacity region of the two-pair bidirectional Gaussian relay network," in *Proceedings*

of IEEE International Symposium on Information Theory (ISIT 2009), Seoul, Korea, Jul. 2009, pp. 2018–2022.

- [75] A. Avestimehr, A. Khajehnejad, A. Sezgin, and B. Hassibi, "Capacity region of the deterministic multi-pair bi-directional relay network," in *Proceedings of IEEE Information Theory Workshop (ITW 2009)*, Volos, Greece, Jun. 2009.
- [76] S. Zhang, S. Liew, and P. Lam, "Hot topic: Physical layer network coding," in Proceedings of 12th Annual International Conference on Mobile Computing and Networking (MobiCom2006), Los Angeles, CA, USA, Sep. 2006.
- [77] N. Lee, J.-B. Lim, and J. Chun, "Degrees of freedom of the MIMO Y channel: Signal space alignment for network coding," *IEEE Transactions on Information Theory*, vol. 56, no. 7, Jul. 2010.
- [78] A. Chaaban, K. Ochs, and A. Sezgin, "The degrees of freedom of the MIMO Y-channel," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2013)*, Istanbul, Turkey, Jul. 2013, pp. 1581–1585.
- [79] A. Chaaban and A. Sezgin, "The capacity region of the linear shift deterministic Y-channel," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2011)*, St. Petersburg, Russia, 2011, pp. 2457–2461.
- [80] C. Xu, L. Song, and Z. Han, Resource Management for Device-to-Device Underlay Communications. Springer, 2014.
- [81] A. Asadi, Q. Wang, and V. Mancuso, "A survey on device-to-device communication in cellular networks," *IEEE Communications Surveys and Tutorials*, Apr. 2014.
- [82] N. Naderializadeh and A. S. Avestimehr, "ITLinQ: A new approach for spectrum sharing in device-to-device communication systems," *IEEE Journal on Selected Areas in Communications*, vol. 32, no. 6, Jun. 2014.
- [83] L. Ong, "Capacity results for two classes of three-way channels," in *IEEE Inter*national Symposium on Communications and Information Technologies (ISCIT 2012), Gold Coast, QLD, Australia, Oct. 2012, pp. 471–476.
- [84] B. Yuan, X. Liao, F. Gao, and X. Luo, "Achievable degrees of freedom of the four-user MIMO Y channel," *IEEE Communications Letters*, vol. 18, no. 1, Jan. 2014.
- [85] A. Zewail, M. Nafie, Y. Mohasseb, and H. El Gamal, "Achievable degrees of freedom region of MIMO relay networks using detour schemes," in *Proceedings* of *IEEE International Conference on Communications (ICC)*, Sydney, Australia, Jun. 2014, pp. 1860–1865.
- [86] A. Kennelly, "The equivalence of triangles and three-pointed stars in conducting networks," *Electrical World and Engineer*, vol. 34, pp. 413–414, 1899.

- [87] A. Rosen, "A new network theorem," Journal of the Institution of Electical Engineers, vol. 62, no. 335, pp. 916–918, Nov. 1924.
- [88] C. Huang, V. Cadambe, and S. Jafar, "On the capacity and generalized degrees of freedom of the X channel," in *Proceedings of IEEE International Symposium* on Information Theory (ISIT 2009), Seoul, Korea, Jul. 2009, pp. 1929–1933.
- [89] A. Avestimehr, A. Sezgin, and D. Tse, "Capacity of the two-way relay channel within a constant gap," *European Transactions on Telecommunications*, vol. 21, no. 4, pp. 363–374, Apr. 2010.
- [90] E. Jorswieck, H. Boche, and A. Sezgin, "Delay-limited capacity and maximum throughput of spatially correlated multiple antanna systems under average and peak-power constraints," in *Proceedings of IEEE Information Theory Workshop* (*ITW 2004*), San Antonio, TX, USA, Oct. 2004, pp. 440–445.
- [91] A. Chaaban, H. Maier, and A. Sezgin, "The degrees of freedom of multi-way device-to-device communications is limited by 2," in *Proceedings of IEEE International Symposium on Information Theory (ISIT 2014)*, Honolulu, HI, USA, Jul. 2014.
- [92] O. Ayach, S. Peters, and R. Heath Jr., "The practical challenges of interference alignment," *IEEE Wireless Communications*, pp. 35–42, Jun. 2012.
- [93] C. Wang, T. Gou, and S. Jafar, "Aiming perfectly in the dark blind interference alignment through staggered antenna switching," *IEEE Transactions on Signal Processing*, vol. 59, no. 6, pp. 2734–2744, Mar. 2011.
- [94] B. Xie, Y. Li, H. Minn, and A. Nosratinia, "Adaptive interference alignment with CSI uncertainty," *IEEE Transactions on Communications*, vol. 61, no. 2, pp. 792–801, Feb. 2013.
- [95] J. García-Naya, L. Castedo, Ó. González, D. Ramírez, and I. Santamaria, "Experimental evaluation of interference alignment under imperfect channel state information," in *Proceedings of European Signal Processing Conference (EUSIPCO* 2011), Sep. 2011, pp. 1085–1089.
- [96] O. Ayach, S. Peters, and R. Heath Jr., "Real world feasibility of interference alignment using MIMO-OFDM channel measurements," in *Proceedings of Military Communications Conference (MILCOM 2009)*, Boston, MA, USA, Oct. 2009.
- [97] Ó. González, D. Ramirez, I. Santamaria, J. Garciá-Naya, and L. Castedo, "Experimental validation of interference alignment techniques using a multiuser MIMO testbed," in *International ITG Workshop on Smart Antennas (WSA 2011)*, Aachen, Germany, Feb. 2011.
- [98] J. Massey, J. Starr, S. Lee, D. Lee, A. Gerstlauer, and R. Heath, "Implementation of a real-time wireless interference alignment network," in *Proceedings of IEEE*

International Conference on Acoustics, Speech, and Signal Processing, Pacific Grove, CA, USA, Nov. 2012.

- [99] A. Loch, T. Nitsche, A. Kuehne, M. Hollick, J. Widmer, and A. Klein, "Practical interference alignment in the frequency domain for OFDM-based wirelss access networks," in *Proceedings of the International Symposium on a World of Wireless, Mobile and Multimedia Networks*, Sydney, NSW, Australia, Jun. 2014.
- [100] F. Blasco, F. Rosetto, and G. Bauch, "Time interference alignment via delay offset for long delay networks," *IEEE Transactions on Communications*, vol. 62, no. 2, pp. 590–599, Feb. 2014.
- [101] J. Gower and P. Legendre, "Metric and Euclidean properties of dissimilarity coefficients," *Journal of Classification*, vol. 3, no. 1, pp. 5–48, Mar. 1986.
- [102] M. Chitre, M. Motani, and S. Shahabudeen, "Throughput of networks with large propagation delays," *IEEE Journal of Oceanic Engineering*, vol. 37, no. 4, pp. 645–658, Oct. 2012.
- [103] M. Stojanovic, "Underwater acoustic communications: Design considerations on the physical layer," in *Proceedings of Fifth Annual Conference on Wireless on Demand Network Systems and Services (WONS 2008)*, Garmisch-Partenkirchen, Germany, Jan. 2008, pp. 1–10.
- [104] R. Mathar and M. Zivkovic, "How to position n transmitter-receiver pairs in n-1 dimensions such that each can use half of the channel with zero Interference from the others," in *Proceedings of IEEE Global Communications Conference (GLOBECOM 2009)*, Honolulu, HI, USA, Dec. 2009.
- [105] R. Mathar and G. Böcherer, "On spatial patterns of transmitter-receiver pairs that allow for interference alignment by delay," in *Proceedings of 3rd International Conference on Signal Processing and Communication Systems (ICSPCS* 2009), Omaha, NE, USA, Sep. 2009.
- [106] R. Dabora, I. Marić, and A. Goldsmith, "Relay strategies for interferenceforwarding," in *Proceedings of IEEE Information Theory Workshop (ITW 2008)*, Porto, Portugal, May 2008, pp. 46–50.

Curriculum Vitae

Henning Maier	Born on the 25th of July 1984 in Steinfurt, Germany
1991 - 1995	Johannes-Grundschule Altenberge, Germany
1995 - 2004	Städt. Gymnasium Borghorst, Steinfurt, Germany
June 2004	General qualification for university entrance (Abitur)
Oct. '04 - Nov. '09	Academic education in Computer Engineering (Technische In- formatik) at RWTH Aachen University, Germany
Oct. '06 - Jan. '08	Student assistant at the Chair for Integrated Signal Processing Systems, RWTH Aachen University, Germany
Oct. '08 - Feb. '09	Internship at Infineon Technologies AG, München, Germany
Nov. 2009	Diploma degree as engineer in Computer Engineering at RWTH Aachen University, Germany
Since Feb. 2010	Research and teaching assistant at the Chair of Theoretical Information Technology, RWTH Aachen University, Germany